SDE and BSDE on Hilbert spaces: applications to quasi-linear evolution equations and the asymptotic properties of the stochastic quasi-geostrophic equation

Rongchan Zhu

A Dissertation Submitted for the Degree of Doctor at the Department of Mathematics Bielefeld University

 $17 \ \mathrm{Feb} \ 2012$

SDE and BSDE on Hilbert spaces: applications to quasi-linear evolution equations and the asymptotic properties of the stochastic quasi-geostrophic equation

Dissertation zur Erlangung des Doktorgrades der Fakultät für Mathematik der Universität Bielefeld

> vorgelegt von Rongchan Zhu

 $17 \ \mathrm{Feb}\ 2012$

Gedruckt auf alterungsbeständigem Papier nach DIN–ISO 9706

Abstract

In this thesis the following three related problems are considered.

1. We consider the following quasi-linear parabolic system of backward partial differential equations

$$(\partial_t + L)u + f(\cdot, \cdot, u, \nabla u\sigma) = 0 \text{ on } [0, T] \times \mathbb{R}^d \qquad u_T = \phi,$$

where L is a possibly degenerate second order differential operator with merely measurable coefficients. We solve this system in the framework of generalized Dirichlet forms and employ the stochastic calculus associated to the Markov process with generator L to obtain a probabilistic representation of the solution u by solving the corresponding backward stochastic differential equation. The solution satisfies the corresponding mild equation which is equivalent to being a generalized solution of the PDE. A further main result is the generalization of the martingale representation theorem using the stochastic calculus associated to the generalized Dirichlet form given by L. The nonlinear term f satisfies a monotonicity condition with respect to u and a Lipschitz condition with respect to ∇u .

2. We consider the following quasi-linear parabolic system of backward partial differential equations on a Banach space E

$$(\partial_t + L)u + f(\cdot, \cdot, u, A^{1/2}\nabla u) = 0 \text{ on } [0, T] \times E, \qquad u_T = \phi,$$

where L is a possibly degenerate second order differential operator with merely measurable coefficients. The results in 1 can be concluded in this case.

3. We study the 2D stochastic quasi-geostrophic equation in \mathbb{T}^2 for general parameter $\alpha \in (0,1)$ and multiplicative noise. We prove it is uniquely ergodic provided the noise is non-degenerate for $\alpha > \frac{2}{3}$. In this case, the convergence to the (unique) invariant measure is exponentially fast. In the general case, we prove the existence of Markov selections.

Acknowledgements

It is a pleasure to thank many people who helped make this thesis possible.

First of all, I would like to express my sincere gratitude to my supervisors, Professor Dr. Michael Röckner and Professor Dr. Ma Zhiming . They continuously supported me in various ways with their enthusiasm, knowledge, inspiration and encouragement. Their constant encouragement and support gives me great motivation for moving forward the road of science.

I would like to thank Professors Philippe Blanchard, Yuri Kondratiev, Michael Röckner, Ludwig Streit, Barbara Gentz, Moritz Kaßmann and Gernot Akemann, who gave wonderful modularized courses and the visiting researchers of the IGK who gave short courses or talks in the IGK which opened to me the beautiful worlds of mathematics, physics and economics.

I am indebted to my sister Xiangchan Zhu for many scientific discussions and daily help. During the whole procedure of writing this thesis, I have benefited from inspiring conversations with many people. Many thanks to Prof. Dr. L. Beznea, Prof. Dr. G. Da Prato, Prof. Dr. W. Stannat, Prof. Dr. Zhenqing Chen, Prof. Dr. Shige Peng, Dr. Qingyang Guan, Dr. Wei Liu, and Dr. Shunxiang Ouyang.

Moreover, I also would like to thank my colleagues and friends in Bielefeld and Beijing, in particular Prof. Jia-An Yan, Prof. Shun-Long Luo, Prof. Fu-Zhou Gong, Prof. Zhao Dong, Prof. Xiangdong Li for their support and help. I am also very thankful to my colleagues in the IGK and the Chinese Academy of Science for their daily help in technical and scientific questions.

I owe my special thanks to Rebecca Reischuk, Stephan Merks and Sven Wiesinger for their help during my studies in Bielefeld. Lastly, but most importantly, I wish to deeply thank my parents far away in China. They supported me throughout and taught me the philosophy of hard work and persistence. This thesis is dedicated to them.

I appreciate very much the financial support from the DFG through the International Graduate College (IGK) at Bielefeld University.

Bielefeld, Feb 17, 2012

Rongchan Zhu

Contents

Pr	reface	2		\mathbf{v}	
0	Intr	oduction		1	
1	Pre	liminaries		13	
	1.1	Some basic concepts fo	r Generalized Dirichlet forms	13	
	1.2	Stochastic calculus asso	ociated with Generalized Dirichlet forms	15	
2	BSDE and generalized Dirichlet form: finite dimensional case 1				
	2.1	Preliminaries		19	
		2.1.1 Linear Equation	18	24	
		2.1.2 Solution of the	Linear Equation	25	
		2.1.3 Basic Relations	for the Linear Equation	30	
	2.2	The Non-linear Equation	on	41	
		2.2.1 The Case of Lip	schitz Conditions	42	
		2.2.2 The Case of Mo	pnotonicity Conditions	45	
	2.3	Martingale representat	ion for the processes	54	
		2.3.1 Representation	under P^x	54	
		2.3.2 Representation	under P^m	63	
	2.4	BSDE's and Generalize	ed Solutions	66	
	2.5	Further Examples			
	2.6	Appendix		75	
		2.6.1 Appendix A. Pr	coof of Theorem 2.2.8 \ldots \ldots \ldots \ldots	75	
		2.6.2 Appendix B. Pr	coof of Proposition 2.4.3	82	
3	BSI	DE and generalized D	irichlet form: Infinite dimensional case	87	
	3.1	Preliminaries		87	

		3.1.1	Linear Equations
		3.1.2	Solution of the Linear Equation
		3.1.3	Basic Relations for the Linear Equation
	3.2	The N	
		3.2.1	The Case of Lipschitz Conditions
		3.2.2	The Case of Monotonicity Conditions
	3.3	Martin	ngale representation for the processes $\ldots \ldots 121$
		3.3.1	Representation under P^x
		3.3.2	Representation under P^{μ}
	3.4	BSDE	's and Weak Solutions
	3.5	Exam	ples $\ldots \ldots 134$
	3.6	A cor	ntrol problem \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 136
	3.7	Apper	ndix
4	Sto	chastic	e quasi-geostrophic equation 149
	4.1	Notations and Preliminaries	
	4.2		w selections in the general case
	4.2 4.3	Marko	
		Marko	w selections in the general case
		Marko Ergod	w selections in the general case $\ldots \ldots \ldots$
		Marko Ergod 4.3.1	w selections in the general case $\ldots \ldots \ldots$
		Marko Ergod 4.3.1 4.3.2 4.3.3	w selections in the general case $\ldots \ldots \ldots$
	4.3	Marko Ergod 4.3.1 4.3.2 4.3.3 Expor	w selections in the general case $\ldots \ldots \ldots$
	4.34.4	Marko Ergod 4.3.1 4.3.2 4.3.3 Expor	w selections in the general case $\dots \dots \dots$
	4.34.4	Marko Ergod 4.3.1 4.3.2 4.3.3 Expor Ergod	w selections in the general case $\dots \dots \dots$
	4.34.4	Marko Ergod 4.3.1 4.3.2 4.3.3 Expor Ergod 4.5.1	w selections in the general case $\ldots \ldots \ldots$
	4.34.4	Marko Ergod 4.3.1 4.3.2 4.3.3 Expor Ergod 4.5.1 4.5.2	w selections in the general case $\dots \dots \dots$
	4.34.4	Marko Ergod 4.3.1 4.3.2 4.3.3 Expor Ergod 4.5.1 4.5.2 4.5.3	w selections in the general case $\dots \dots \dots$
	4.34.4	Marko Ergod 4.3.1 4.3.2 4.3.3 Expor Ergod 4.5.1 4.5.2 4.5.3 4.5.4	w selections in the general case $\dots \dots \dots$

Bibliography

Chapter 0

Introduction

This thesis is devoted to stochastic differential equations (SDE) and backward stochastic differential equations (BSDE) on Hilbert spaces. In the mid 1940s Itô introduced the stochastic integral and stochastic integral equations. Since then, motivated by the demand from modern applications (e.g. physics, chemistry, biology and control theory), the theory of SDE has been well developed.

Roughly speaking, the solution of a stochastic differential equation is an adapted process X satisfying

$$dX_t = b(t, X_t)dt + \sigma(t, X_t)dW_t; \quad X_0 = \xi,$$

where W is a Brownian motion. This is similar to the Cauchy problem of an ordinary differential equation. However, if we consider the terminal value problem for this stochastic equation and just take the time reversal of the solution of the SDE as a solution, the main problem lies in the adaptedness of the solution, which is essential to the definition of stochastic integral with respect to Brownian motion. This does not happen in the deterministic case. To solve this problem, Pardoux and Peng in [PP90] introduced the solution of a BSDE, which consists of a pair of adapted processes (Y, Z) satisfying

$$-dY_t = f(t, Y_t, Z_t)dt - Z_t dW_t; \quad Y_T = \xi_t$$

where ξ is the terminal condition. Since this type of equation appears in numerous problems in finance, the subject has become increasingly important and popular.

The existence and uniqueness of the solution of the BSDE with Lipschitz coefficients has been obtained by Pardoux and Peng in [PP90]. Later on, there have been a series of papers (c.f. [Pa99], [BDHPS03], [FT02], [BC08] and the references therein) extending their results for more general coefficients and more general state spaces. Important results concerning the link between those BSDEs and PDEs are also stated in Pardoux and Peng ([PP92]) (see below). The first aim of this thesis is to generalize their results in the framework of generalized Dirichlet forms.

BSDE and generalized Dirichlet forms: finite dimensional case

In Chapter 2 we consider the following quasi-linear parabolic system of backward partial differential equations

$$(\partial_t + L)u + f(\cdot, \cdot, u, \nabla u\sigma) = 0 \text{ on } [0, T] \times \mathbb{R}^d \qquad u_T = \phi, \tag{1.1}$$

where L is a second order linear differential operator and f is monotone in u and Lipschitz in ∇u and σ is the diffusion coefficient for the process associated with L. If L has sufficiently regular coefficients there is a well-known theory to obtain a probabilistic representation of the solutions to (1.1), using corresponding backward stochastic differential equations (BSDE) and also to solve BSDE with the help of (1.1), originally due to E. Pardoux and S. Peng ([PP92]). The main aim of this chapter is to implement this approach for a very general class of linear operators L, which are possibly degenerate, have merely measurable cofficients and are in general non-symmetric. Solving (1.1) for such general L is the first main task of this chapter (see Theorem 2.2.8). The second main contribution is to prove the martingale representation theorem (Theorem 2.3.8) for the underlying reference diffusions generated by such general operators L.

If f and the coefficients of the second-order differential operator L are sufficiently smooth, the PDE has a classical solution u. Consider $Y_t^{s,x} := u(t, X_t^{s,x}), Z_t^{s,x} :=$ $\nabla u \sigma(t, X_t^{s,x})$ where $X_t^{s,x}, s \leq t \leq T$, is the diffusion process with infinitesimal generator L which starts from x at time s and σ is the diffusion coefficient of X. Then, using Itô's formula one checks that $(Y_t^{s,x}, Z_t^{s,x})_{s \leq t \leq T}$ solves the BSDEs

$$Y_t^{s,x} = \phi(X_T^{s,x}) + \int_t^T f(r, X_r^{s,x}, Y_r^{s,x}, Z_r^{s,x}) dr - \int_t^T Z_r^{s,x} dB_r.$$
 (1.2)

Conversely, by standard methods one can prove that (1.2) has a unique solution $(Y_t^{s,x}, Z_t^{s,x})_{s \le t \le T}$ and then $u(s, x) := Y_s^{s,x}$ is a solution to PDE (1.1). If f and the coefficients of L are Lipschitz continuous then a series of papers (e.g. [BPS05], [Pa99] and the reference therein) prove that the above relation between PDE (1.1) and BSDE (1.2) remains true, if one considers viscosity solutions to PDE (1.1). In both these approaches, since the coefficients are Lipschitz continuous, the Markov process X with infinitesimal operator L is a diffusion process which satisfies an SDE and so one may use its associated stochastic calculus.

In [BPS05] Bally, Pardoux and Stoica consider a semi-elliptic symmetric secondorder differential operator L (which is written in divergence form) with measurable coefficients. They prove that the above system of PDE has a unique solution u in some functional space. Then using the theory of symmetric Dirichlet forms and its associated stochastic calculus, they prove that the solution $Y^{s.x}$ of the BSDE yields a precised version of the solution u so that, moreover, one has $Y_t^{s,x} = u(t, X_{t-s}), P^x$ a.s. In [S09], the analytic part of [BPS05] has been generalized to a non-symmetric case with L satisfying the weak sector condition. Here the weak sector condition means

$$((1-L)u, v) \le K((1-L)u, u)^{1/2}((1-L)v, v)^{1/2}, \text{ for } u, v \in \mathcal{D}(L),$$

for some constant K > 0, i.e. the non-symmetric part of the operator L can be dominated by the symmetric part. In [L01], A.Lejay considers the generator $L = \frac{1}{2} \sum_{i,j=1}^{d} \frac{\partial}{\partial x_i} (a_{ij} \frac{\partial}{\partial x_j}) + \sum_{i=1}^{d} b_i(x) \frac{\partial}{\partial x_i}$ for bounded a, b. In [ZR11], T.S. Zhang and Q.K.Ran (see also [Z]) consider L of a more general form, but $a = (a_{ij})$ is required to be uniformly elliptic and $b \in L^p$ for p > d. Anyway, since L satisfies the weak sector condition in these cases, it generates a sectorial (i.e. a small perturbation of a symmetric) Dirichlet form, so the theory of Dirichlet forms from [MR92] can be applied in [L01], [Z], [ZR11].

In [St2] Stannat extends the known framework of Dirichlet forms to the class of generalized Dirichlet forms. By this we can analyze differential operators where the second order part may be degenerate and at the same time the first order part may be unbounded satisfying no global L^p -condition for $p \ge d$. The motivation for the first chapter is to extend the results in [BPS05] to the case, where L generates a generalized Dirichlet form so that we can allow the coefficients of L to be more general.

In Chapter 2, we consider PDE (1.1) for a non-symmetric second order differential operator L, which is associated to the bilinear form

$$\mathcal{E}(u,v) := \sum_{i,j=1}^{d} \int a_{ij}(x) \frac{\partial u}{\partial x^{i}}(x) \frac{\partial v}{\partial x^{j}}(x) m(dx) + \int c(x)u(x)v(x)m(dx) + \sum_{i=1}^{d} \int \sum_{j=1}^{d} a_{ij}(x)(b_{j}(x) + \hat{b}_{j}(x)) \frac{\partial u}{\partial x^{i}}v(x)m(dx) \ \forall u,v \in C_{0}^{\infty}(\mathbb{R}^{d}).$$

$$(1.3)$$

where $C_0^{\infty}(\mathbb{R}^d)$ denotes the space of infinitely differentiable functions with compact support. We stress that (a_{ij}) is not necessarily assumed to be (locally) strictly positive definite, but may be degenerate in general. When $b \equiv 0$, the bilinear form \mathcal{E} satisfies the weak sector condition. For the perturbation term given by b we need $b\sigma \in L^2(\mathbb{R}^d; \mathbb{R}^d, m)$, where $\sigma\sigma^* = a$ and σ^* is the transpose of the matrix of σ . That implies that we do not have the weak sector condition for the bilinear form. We use the theory of generalized Dirichlet forms and its associated stochastic calculus (cf [St1, St2, Tr1, Tr2]) to generalize the results in [BPS05]. Here m is a finite measure or Lebesgue measure on \mathbb{R}^d . If D is a bounded open domain, we choose mas $1_D(x)dx$. Then in certain cases the solution of PDE (1.1) satisfies the Neumann boundary condition. If we replace $C_0^{\infty}(\mathbb{R}^d)$ by $C_0^{\infty}(D)$, the solution of PDE (1.1) satisfies the Dirichlet boundary condition.

In the analytic part of Chapter 2, we do not need \mathcal{E} to be a generalized Dirichlet form. We start from a semigroup (P_t) satisfying conditions (A1)-(A4), specified in Section 2.1 below. Such a semigroup can, however, be constructed from a generalized Dirichlet form. It can also be constructed by other methods (see e.g. [DR02]). Under conditions (A1)-(A4), the coefficients of L may be quite singular and only very broad assumptions on a and b are needed (see the examples in Sections 2.3 and 2.4).

Chapter 2 is organized as follows. In Sections 2.1 and 2.2, we use functional analytical methods to solve PDE (1.1) (see Theorems 2.2.8 and 2.2.11) in the sense of Definition 2.1.5, i.e. there are sequences $\{u^n\}$ which are strong solutions with data (ϕ^n, f^n) such that

$$||u^n - u||_T \to 0, ||\phi^n - \phi||_2 \to 0, \lim_{n \to \infty} f^n = f \text{ in } L^1([0, T]; L^2).$$

Here $\|\cdot\|_T := (\sup_{t \leq T} \|\cdot\|_2^2 + \int_0^T \mathcal{E}_{c_2+1}^{a,\hat{b}}(\cdot)dt)^{1/2}$, where $\mathcal{E}^{a,\hat{b}}$ is the summand in the left hand side of (1.3) with $b \equiv 0$. The above definition for the solution is equivalent to that of the following mild equation in L^2 -sense

$$u(t,x) = P_{T-t}\phi(x) + \int_t^T P_{s-t}f(s,\cdot,u_s,D_\sigma u_s)(x)ds,$$

(see Proposition 2.1.9). If we use the definition of weak solution to define our solution as in [BPS05], uniqueness of the solution cannot be obtained since only $|b\sigma| \in L^2(\mathbb{R}^d; m)$. Furthermore, the function f in PDE (1.1) need not to be Lipschitz continuous with respect to the third variable; monotonicity suffices. And μ which appears in the monotonicity conditions (see condition (H2) in Section 2.2.2 below) can depend on t. f is, however, assumed to be Lipschitz continuous with respect to the last variable. We emphasize that the first order term of L cannot be incorporated into f without the condition that b is bounded. Hence we are forced to take it as part of L and hence have to consider a diffusion process X in (1.2) which is generated by an operator L associated with a (in general non-sectorial) generalized Dirichlet form. We also emphasize that under our conditions, PDE (1.1) cannot be tackled

by standard monotonicity methods (see e.g. [Ba10]) because of the lack of a suitable Gelfand triple $V \subset H \subset V^*$ with V being a reflexive Banach space.

In Section 2.3, we extend the stochastic calculus of generalized Dirichlet forms in order to generalize the martingale representation theorem. In order to treat BSDE, we show in Theorem 2.3.8 that there exists a set of null capacity \mathcal{N} outside of which the following representation result holds : for every bounded \mathcal{F}_{∞} -measurable random variable ξ , there exists a predictable process $(\phi_1, ..., \phi_d) : [0, \infty) \times \Omega \to \mathbb{R}^d$, such that for each probability measure ν , supported by $\mathbb{R}^d \setminus \mathcal{N}$, one has

$$\xi = E^{\nu}(\xi|\mathcal{F}_0) + \sum_{i=0}^d \int_0^\infty \phi_s^i dM_s^{(i)} \qquad P^{\nu} - a.s..$$

where $M^i, i = 1, ..., d$ are the coordinate martingales associated with the process X. As a result, one can choose the exceptional set \mathcal{N} such that if the process X starts from a point of \mathcal{N}^c , it remains always in this set. As a consequence we deduce the existence of solutions for the BSDE using the existence for PDE (1.1) in the usual way, however, only under P^m , because of our general coefficients of L (c.f. Theorem 2.3.12).

In Section 2.4, we employ the martingale representation to deduce existence and uniqueness for the solutions of BSDE (1.2). As a consequence, in Theorem 2.4.7, the existence and uniqueness of solutions for PDE (1.1), not covered by our analytic results in Section 2.2, is obtained by $u(s, x) = Y_s^s$, where Y_t^s is the solution of the BSDE. Moreover we have, $Y_t^s = u(t, X_{t-s})$, P^x -a.s., $x \in \mathbb{R}^d \setminus \mathcal{N}$. Further examples are given in Section 2.5.

BSDE and generalized Dirichlet form: infinite dimensional case

In Chapter 3, we consider the following quasi-linear parabolic system of backward partial differential equations on a (real) Banach space E

$$(\partial_t + L)u + f(\cdot, \cdot, u, A^{1/2}\nabla u) = 0 \text{ on } [0, T] \times E, \qquad u_T = \phi, \tag{1.4}$$

where L is a second order differential operator with measurable coefficients, ∇u is the H-gradient of u and $(H, \langle \cdot, \cdot \rangle_H)$ is a separable real Hilbert space such that $H \subset E$ densely and continuously. A is a symmetric, positive-definite and bounded operator on H. This equation is also called *nonlinear Kolmogorov equation* on an infinite dimensional space. In fact, in this chapter we study systems of PDE of type (1.4), i.e. u takes values in \mathbb{R}^l for some fixed $l \in \mathbb{N}$. For simplicity, in this introductory section we explain our results in the case l = 1. Various concepts of solution are known for (linear and) nonlinear parabolic equations in infinite dimensions. In Chapter 3 we will consider solutions in the sense of Definition 3.1.4, i.e. there is a sequence $\{u^n\}$ of strong solutions with data (ϕ^n, f^n) such that

$$||u^n - u||_T \to 0, ||\phi^n - \phi||_2 \to 0 \text{ and } \lim_{n \to \infty} f^n = f \text{ in } L^1([0, T]; L^2).$$

We will prove the above definition for solution is equivalent to being a solution of the following mild equation in L^2 sense

$$u(t,x) = P_{T-t}\phi(x) + \int_{t}^{T} P_{s-t}f(s,\cdot,u_s,A^{1/2}\nabla u_s)(x)ds,$$
(1.5)

(see Proposition 3.1.7). This formula is meaningful provided u is even only once differentiable with respect to x. Thus, the solutions we consider are in a sense intermediate between classical and viscosity solutions.

The notion of viscosity solution, developed by many authors, in particular M. Crandall and P. L. Lions and their collaborators, is not discussed here. Generally speaking, the class of equations that can be treated by this method (c.f. [L88,L89,L92]) is much more general than those considered in this paper: it includes fully nonlinear operators. However, none of these results are applicable to our situation because the coefficients of the operator L are only measurable in our case.

In [FT02], mild solutions of the above PDE (1.4) have been considered, and a probabilistic technique, based on backward stochastic differential equations, has been used to prove the existence and uniqueness for the mild solution. Furthermore, their results has been extended in [BC08] and [M11]. All these results need some regular conditions for the coefficients of L and f to make sure that the process Xhas regular dependence on parameters, which are not required for our results. In Chapter 3, we will prove the existence and uniqueness of a solution u of (1.4) for a general non-symmetric operator L by methods from functional analysis (Theorem 3.2.8). In fact Chapter 3 is an extension of Chapter 2 to the infinite dimensional case. Though Chapter 2 serves as guideline, serious obstacles appear at various places if E is infinite dimensional, which we overcome in this work.

The connection between backward stochastic equations and nonlinear partial differential equations was proved for the finite dimensional case e.g. in [BPS05], [PP92] (see also the references therein). A further motivation of Chapter 3 is to give a probabilistic interpretation for the solutions of the above PDE's, i.e. in this infinite dimensional case.

If E is a Hilbert space, which equals to H, f and the coefficients of the secondorder differential operator L are sufficiently regular, then PDE (1.4) has a classical solution and one may construct the pair of processes $Y_s^{t,x} := u(s, X_s^{t,x}), Z_s^{t,x} :=$ $A^{1/2}\nabla u(s, X_s^{t,x})$ where $X_s^{t,x}, t \leq s \leq T$, is the diffusion process with infinitesimal operator L which starts from x at time t and A is the diffusion coefficient for X. Then, using Itô's formula one checks that $(Y_s^{t,x}, Z_s^{t,x})_{t\leq s\leq T}$ solves the BSDE

$$Y_s^{t,x} = \phi(X_T^{t,x}) + \int_s^T f(r, X_r^{t,x}, Y_r^{t,x}, Z_r^{t,x}) dr - \int_s^T \langle Z_r^{t,x}, dW_r \rangle_H,$$
(1.6)

where W_r is a cylindrical Wiener process in H. Conversely, for regular coefficients by standard methods one can prove that (1.6) has a unique solution $(Y_t^{s,x}, Z_t^{s,x})_{s \leq t \leq T}$ and then $u(s, x) := Y_s^{s,x}$ is a solution to PDE (1.4). If f and the coefficients of L are Lipschitz continuous then in [FT02] the authors prove that the probabilistic interpretation above remains true, if one considers mild solutions to PDE (1.4). There are many papers that study forward-backward systems in infinite dimension (cf [FT02], [FH07] and the references therein). In these approaches, since the coefficients are Lipschitz continuous, the Markov process X with infinitesimal operator L is a diffusion process which satisfies an SDE and so one can use its associated stochastic calculus to conclude the results.

In Chapter 3, we consider PDE (1.4) for a non-symmetric second order differential operator L in infinite dimensions, which is associated to the bilinear form

$$\mathcal{E}(u,v) = \int \langle A(z)\nabla u(z), \nabla v(z) \rangle_H d\mu(z) + \int \langle A(z)b(z), \nabla u(z) \rangle_H v(z)d\mu(z), u, v \in \mathcal{F}C_b^{\infty},$$

where $\mathcal{F}C_b^{\infty}$ will be defined in Section 3.1. Here we only need $|A^{1/2}b|_H \in L^2(E;\mu)$. That is to say, in general the above bilinear form \mathcal{E} does not satisfy any weak sector condition. We use the theory of generalized Dirichlet forms and the associated stochastic calculus(cf. [St1, St2, Tr1, Tr2]) to generalize the results in [BPS05].

In the analytic part of Chapter 3, we don't need \mathcal{E} to be a generalized Dirichlet form. We start from a semigroup (P_t) satisfying conditions (A1)-(A3), specified in Section 3.1 below. Such a semigroup can e.g. be constructed from a generalized Dirichlet form. It can also be constructed by other methods (see e.g. [DR02]). Under conditions (A1)-(A3), the coefficients of L may be quite singular and only very broad assumptions on A and b are needed.

Chapter 3 is organized as follows. In Sections 3.1 and 3.2, we use functional analytical methods to solve PDE (1.4) (see Theorem 3.2.8) in the sense of Definition 3.1.4 or equivalently in the sense of (1.5). Here the function f need not to be Lipschitz continuous with respect to y; monotonicity suffices. And μ which appears in the monotonicity conditions (see condition (H2) in Section 3.2.2 below) can depend on t. f is, however, assumed to be Lipschitz continuous with respect to the last variable. We emphasize that the first order term with coefficient Ab of L cannot be incorporated into f unless it is bounded. Hence we are forced to take it as a part of L and hence we have to consider a diffusion process X which is generated by an operator L which is the generator of a (in general non-sectorial) generalized Dirichlet form. We also emphasize that under our conditions PDE (1.4) cannot be tackled by standard monotonicity methods (see e.g. [Ba10]) because of lack of a suitable Gelfand triple $\mathcal{V} \subset \mathcal{H} \subset \mathcal{V}^*$ with \mathcal{V} being a reflexive Banach space.

In Section 3.3, we assume that \mathcal{E} is a generalized Dirichlet form and is associated with a strong Markov process $X = (\Omega, \mathcal{F}_{\infty}, \mathcal{F}_t, X_t, P^x)$. Such a process can be constructed if \mathcal{E} is quasi-regular. We extend the stochastic calculus for the Markov process in order to generalize the martingale representation theorem. More precisely, in order to treat BSDE's, in Theorem 3.3.3 we show that there is a set \mathcal{N} of null capacity outside of which the following representation theorem holds : for every bounded \mathcal{F}_{∞} -measurable random variable ξ , there exists a predictable process ϕ : $[0, \infty) \times \Omega \to H$, such that for each probability measure ν , supported by $E \setminus \mathcal{N}$, one has

$$\xi = E^{\nu}(\xi|\mathcal{F}_0) + \sum_{i=0}^{\infty} \int_0^{\infty} \phi_s^i dM_s^i \qquad P^{\nu} - a.e.,$$

where $M^i, i \in \mathbb{N}$ are the coordinate martingales associated with the process X. In fact, one may choose the exceptional set \mathcal{N} such that if the process X starts from a point of \mathcal{N}^c , it remains always in \mathcal{N}^c . As a consequence we deduce the existence of solutions for the BSDE using the existence of solutions for PDE (1.4) in the usual way, however, only under P^{μ} , because of our very general coefficients of L (c.f. Theorem 3.3.7).

In Section 3.4, we employ the above results to deduce existence and uniqueness for the solutions of the BSDE under P^x for $x \in \mathcal{N}^c$. As a consequence, in Theorem 3.4.4 one finds a version of the solution to PDE (1.4) which satisfies the mild equation pointwise, i.e. for the solution Y^s of the BSDE, we have $Y_t^s = u(t, X_{t-s}), P^x$ -a.s. In particular, Y_t^t is P^x -a.s. equal to u(t, x).

In Section 3.5, we give some examples of the operator L satisfying our general conditions (A1)-(A5). In Section 3.6, we consider an application of our results to a control problem. An admissible control $\theta(t, \omega)$ is a progressively measurable process with respect to the filtration $(\mathcal{F}_t)_{t\geq 0}$ and takes values in some metric space K. Given a measurable function $c : [0,T] \times E \times K \to H$ and a admissible control θ , we define $N_t^{\theta} = \int_0^t c_s(X_s, \theta_s) dM_s, \Gamma_t^{\theta} = \exp(N_t^{\theta} - \frac{1}{2} \langle N^{\theta} \rangle_t)$, and $P^{\theta,x} = \Gamma^{\theta} P^x$. The aim is to choose a control process θ , within a set of admissible controls, to minimize a cost

functional of the form:

$$J^{\theta}(x) = E^{\theta,x}[\phi(X_T) + \int_0^T h(s, X_s, \theta_s)ds],$$

where ϕ and h are measurable functions and $E^{\theta,x}$ means taking expectation under $P^{\theta,x}$. There is a vast literature on such control problems in infinite dimensions if X is a solution of an SDE on a Hilbert space (c.f. [FT02] [G96] and the reference therein). In our case, the process X is generated by a linear operator L with merely measurable coefficients as above and X does not need to satisfy an SDE. As the coefficients of L are very general, X doesnot have regular dependence on parameters, which is essential in [FT02]. Moreover, we also do not need that ϕ and h are Gâteaux differentiable with respect to x. By the results in Sections 3 and 5, we directly provide a mild solution of the Hamilton-Jacobi-Bellman equation.

Ergodicity of the stochastic quasi-geostrophic equation

Up to the early 1960s, most works on SDE has been confined to ordinary differential equation. Later on, a large number of models were found that could be described by partial differential equations with random parameters, such as the coefficients or the forcing term. As a result, the study of SDE in infinite dimensional space has begun to attract a lot of attention of many researchers. In this thesis, we are concerned with the long time behavior of the stochastic quasi-geostrophic equation, which is an interesting SDE in infinite dimensional space.

In Chapter 4, we study the long time behavior of the stochastic partial differential equation by proving the uniqueness of invariant measures and strong asymptotic stability, i.e. the law of the process converges to the invariant measure in total variation norm. In order to have uniqueness of the invariant measure, the Markov process should satisfy some irreducibility property, together with some regularity. Here we prove the strong Feller property and the irreducibility of the associated Markov process. Then the classical results in the ergodic theory of Markov processes, as developed by Doob, Khas'minskii and others, can be applied to obtain the uniqueness of invariant measures as well as the strong asymptotic stability (see e.g. [DZ96]).

Consider the following 2D stochastic quasi-geostrophic equation in the periodic domain $\mathbb{T}^2 = \mathbb{R}^2/(2\pi\mathbb{Z})^2$:

$$\frac{\partial\theta(t,\xi)}{\partial t} = -u(t,\xi) \cdot \nabla\theta(t,\xi) - \kappa(-\triangle)^{\alpha}\theta(t,\xi) + (G(\theta)\eta)(t,\xi), \qquad (1.7)$$

with initial condition

$$\theta(0,\xi) = \theta_0(\xi), \tag{1.2}$$

where $\theta(t,\xi)$ is a real-valued function of $\xi \in \mathbb{T}^2$ and $t \ge 0$, $0 < \alpha < 1, \kappa > 0$ are real numbers. u is determined by θ through a stream function ψ via the following relations:

$$u = (u_1, u_2) = (-R_2\theta, R_1\theta).$$
(1.9)

Here R_j is the *j*-th periodic Riesz transform and $\eta(t,\xi)$ is a Gaussian random field, white noise in time, subject to the restrictions imposed below. The case $\alpha = \frac{1}{2}$ is called the critical case, the case $\alpha > \frac{1}{2}$ sub-critical and the case $\alpha < \frac{1}{2}$ super-critical.

This equation is an important model in geophysical fluid dynamics. The case $\alpha = 1/2$ exhibits similar features (singularities) as the 3D Navier-Stokes equations and can therefore serve as a model case for the latter. In the deterministic case this equation has been intensively investigated because of both its mathematical importance and its background in geophysical fluid dynamics, (see for instance [CV06], [Re95], [CW99], [Ju03], [Ju04], [KNV07] and the references therein). In the deterministic case, the global existence of weak solutions has been obtained in [Re95] and one most remarkable result by [CV06] proves the existence of a classical solution for $\alpha = 1/2$ and the other by [KNV07] proves solutions for $\alpha = 1/2$ with periodic C^{∞} data remain C^{∞} for all the time.

In Chapter 4 we study the 2D stochastic quasi-geostrophic equation in \mathbb{T}^2 for general parameter $\alpha \in (0, 1)$ and multiplicative noise. First using an abstract result for obtaining Markov selections from [GRZ09], we prove the existence of an a.s. Markov family for general parameter $\alpha \in (0, 1)$ (see Theorem 4.2.5).

Then we prove the ergodicity of the solution in the subcritical case, provided that the noise is non-degenerate and regular (see Theorem 4.3.10). The proof follows from employing the weak-strong uniqueness principle in [FR08] (Theorem 4.3.4) and as usual first establishing the strong Feller property (Theorem 4.3.3). Though one would expect to get ergodicity for $\alpha > \frac{1}{2}$, surprisingly it turns out that one needs $\alpha > \frac{2}{3}$. As the dynamics exists only in the martingale sense and standard tools of stochastic analysis are not available, the computations are made for an approximating cutoff dynamics, which is equal to the original dynamics on a small random time interval. As the noise is non-degenerate, we can use the Bismut-Elworthy-Li formula to prove the strong Feller property. Since in our case $\alpha < 1$, it is more difficult to use the H^{α} -norm to control the nonlinear term even though the equation is on \mathbb{T}^2 . To prove the weak-strong uniqueness principle we need some regularity for the trajectories of the noise. Therefore, we need conditions on G so that it is enough regularizing. However, in order to apply the Bismut-Elworthy-Li formula, we also need G^{-1} to be regularizing enough. As a result, $\alpha > 2/3$ is required (see Remark 4.3.2 below). It seems difficult to use the Kolmogorov equation method

as in [DD03], [DO06] or a coupling approach as in [O08] in our situation (see Remark 4.3.2 below).

In order to prove the exponential convergence, we need to show decay of the solution's L^p -norm for suitable p. To prove it, we have to improve the crucial positivity lemma from [Re95] (see Lemma 4.4.1 below).

Chapter 4 is organized as follows. For the general case the existence of Markov selections is obtained in Section 4.2. In Section 4.3, we prove the ergodicity of the solution for $\alpha > 2/3$ provided the noise is non-degenerate. The exponential convergence to the (unique) invariant measure is shown in Section 4.4 (Theorem 4.4.5). We also consider the ergodicity of the equation driven by the mildly degenerate noise following the idea of [EH01] in Section 4.5 (Theorem 4.5.17).

Chapter 1

Preliminaries

In this chapter, we collect some results about the generalized Dirichlet form and the associated stochastic calculus for the following chapters. We omit all proofs and refer the reader to [St2, Tr1, Tr2] for details. In the first part, we recall the definitions of a generalized Dirichlet form and a quasi-regular generalized Dirichlet form. In the second part, we collect some useful results about the stochastic calculus associated with the generalized Dirichlet form, such as the Fukushima decomposition.

1.1 Some basic concepts for Generalized Dirichlet forms

Let us recall the definition of a generalized Dirichlet form from [St2]. Let E be a Hausdorff topological space and assume that its Borel σ -algebra $\mathcal{B}(E)$ is generated by the set C(E) of all continuous functions on E. Let m be a σ -finite measure on $(E, \mathcal{B}(E))$ such that $\mathcal{H} := L^2(E, m)$ is a separable (real) Hilbert space. Let $(\mathcal{A}, \mathcal{V})$ be a coercive closed form on \mathcal{H} in the sense of [MR92], i.e. \mathcal{V} is a dense linear subspace of $\mathcal{H}, \mathcal{A} : \mathcal{V} \times \mathcal{V} \to \mathbb{R}$ is a positive definite bilinear map, \mathcal{V} is a Hilbert space with inner product $\widetilde{\mathcal{A}}_1(u, v) := \frac{1}{2}(\mathcal{A}(u, v) + \mathcal{A}(v, u)) + (u, v)_{\mathcal{H}}$, and \mathcal{A} satisfies the weak sector condition

$$|\mathcal{A}_1(u,v)| \le K\mathcal{A}_1(u,u)^{1/2}\mathcal{A}_1(v,v)^{1/2},$$

 $u, v \in \mathcal{V}$, with sector constant K. We will always denote the corresponding norm by $\|\cdot\|_{\mathcal{V}}$. Identifying \mathcal{H} with its dual \mathcal{H}' we obtain that $\mathcal{V} \to \mathcal{H} \cong \mathcal{H}' \to \mathcal{V}'$ densely and continuously.

Let $(\Lambda, D(\Lambda, \mathcal{H}))$ be a linear operator on \mathcal{H} satisfying the following assumptions: (i) $(\Lambda, D(\Lambda, \mathcal{H}))$ generates a C_0 -semigroup of contractions $(U_t)_{t\geq 0}$ on \mathcal{H} . (ii) \mathcal{V} is Λ -admissible, i.e. $(U_t)_{t\geq}$ can be restricted to a C_0 -semigroup on \mathcal{V} .

Let (Λ, \mathcal{F}) with corresponding norm $\|\cdot\|_{\mathcal{F}}$ be the closure of $\Lambda : D(\Lambda, \mathcal{H}) \cap \mathcal{V} \to \mathcal{V}'$ as an operator from \mathcal{V} to \mathcal{V}' and $(\hat{\Lambda}, \hat{\mathcal{F}})$ be its dual operator.

Let

$$\mathcal{E}(u,v) = \begin{cases} \mathcal{A}(u,v) - \langle \Lambda u, v \rangle & \text{if } u \in \mathcal{F}, v \in \mathcal{V} \\ \mathcal{A}(u,v) - \langle \hat{\Lambda} v, u \rangle & \text{if } u \in \mathcal{V}, v \in \hat{\mathcal{F}}, \end{cases}$$

where $\langle \cdot, \cdot \rangle$ denotes the dualization between \mathcal{V}' and \mathcal{V} and $\langle \cdot, \cdot \rangle$ coincides with the inner product $(\cdot, \cdot)_H$ in H when restricted to $H \times \mathcal{V}$. Define $\mathcal{E}_{\alpha}(u, v) := \mathcal{E}(u, v) + \alpha(u, v)_{\mathcal{H}}$ for $\alpha > 0$. We call \mathcal{E} the bilinear form associated with $(\mathcal{A}, \mathcal{V})$ and $(\Lambda, D(\Lambda, \mathcal{H}))$.

Definition 1.1 The bilinear form \mathcal{E} associated with $(\mathcal{A}, \mathcal{V})$ and $(\Lambda, D(\Lambda, \mathcal{H}))$ is called *a generalized Dirichlet form*, if

$$u \in \mathcal{F} \Rightarrow u^+ \land 1 \in \mathcal{V} \text{ and } \mathcal{E}(u, u - u^+ \land 1) \ge 0.$$

We also recall the definition of semi-Dirichlet form from [MOR95]. For the closed coercive form $(\mathcal{A}, \mathcal{V})$ is called a *semi-Dirichlet form* if $u \in \mathcal{V}$, $u^+ \wedge 1 \in \mathcal{V}$ and $\mathcal{A}(u + u^+ \wedge 1, u - u^+ \wedge 1) \geq 0$.

Suppose the adjoint semigroup $(\hat{U}_t)_{t\geq 0}$ of $(U_t)_{t\geq 0}$ can also be restricted to a C_0 semigroup on \mathcal{V} . Let $(\hat{\Lambda}, D(\hat{\Lambda}, \mathcal{H}))$ denote the generator of $(\hat{U}_t)_{t\geq 0}$ on $\mathcal{H}, \hat{\mathcal{A}}(u, v) :=$ $\mathcal{A}(v, u), u, v \in \mathcal{V}$ and let the coform $\hat{\mathcal{E}}$ be defined as the bilinear form associated
with $(\hat{\mathcal{A}}, \mathcal{V})$ and $(\hat{\Lambda}, D(\hat{\Lambda}, \mathcal{H}))$.

In [St2, Section I.3], they construct the resolvent $(G_{\alpha})_{\alpha>0}$ such that for all $\alpha > 0$, $\mathcal{E}(G_{\alpha}f, v) = (f, v)_{\mathcal{H}}, \forall f \in \mathcal{H}, v \in \mathcal{V}$. The resolvent $(G_{\alpha})_{\alpha>0}$ is called the *resolvent* associated with \mathcal{E} . Let $(\hat{G}_{\alpha})_{\alpha>0}$ be the adjoint of $(G_{\alpha})_{\alpha>0}$ in \mathcal{H} . $(\hat{G}_{\alpha})_{\alpha>0}$ is called the *coresolvent* associated with \mathcal{E} . By [St2, Proposition 3.6] $(G_{\alpha})_{\alpha>0}$ is a strongly continuous contraction resolvent on \mathcal{H} .

For the generalized Dirichlet form, we also have the concept of the quasi-regular generalized Dirichlet form. By this we can construct a strong Markov process associated with it. This will be used in the probabilistic part of this chapter (see Section 2.3). We recall the definition of the quasi-regular generalized Dirichlet form here. For this reason we introduce some useful notations.

An element of u of \mathcal{H} is called 1-excessive (resp. 1-coexcessive) if $\beta G_{\beta+1}u \leq u$ (resp. $\beta \hat{G}_{\beta+1}u \leq u$) for all $\beta \geq 0$. Let \mathcal{P} (resp. $\hat{\mathcal{P}}$) denote the 1-excessive (resp. 1-coexcessive) elements of \mathcal{V} .

Definition 1.2 (i) An increasing sequence of closed subset $(F_k)_{k\geq 1}$ is called an \mathcal{E} -nest, if for every function $u \in \mathcal{P} \cap \mathcal{F}$, $u_{F_k^c} \to 0$ in \mathcal{H} and weakly in \mathcal{V} , where $u_{F_k^c} := e_{u \cdot 1_{F_c^c}}$ is the 1-reduced function defined in [St2, Definition III.1.8].

(ii) A subset $N \subset E$ is called \mathcal{E} -exceptional if there is an \mathcal{E} -nest $(F_k)_{k\geq 1}$ such that $N \subset \bigcap_{k\geq 1} E \setminus F_k$.

(iii) A property of points in E holds \mathcal{E} -quasi-everywhere($\mathcal{E} - q.e.$) if the property holds outside some \mathcal{E} -exceptional set.

(iv) A function f defined up to some \mathcal{E} -exceptional set $N \subset E$ is called \mathcal{E} -quasicontinuous (\mathcal{E} -q.c.) if there exists an \mathcal{E} -nest $(F_k)_{k\geq 1}$, such that $\bigcup_{k\geq 1}F_k \subset E \setminus N$ and $f|_{F_k}$ is continuous for all k.

Definition 1.3 The generalized Dirichlet form \mathcal{E} is called *quasi-regular* if:

(i) There exists an \mathcal{E} -nest consisting of compact sets.

(ii) There exists a dense subset of \mathcal{F} whose elements have \mathcal{E} -quasi-continuous m-versions.

(iii) There exist $u_n \in \mathcal{F}, n \in \mathbb{N}$, having \mathcal{E} -quasi-continuous m-versions $\tilde{u}_n, n \in \mathbb{N}$, and an \mathcal{E} -exceptional set $N \subset E$ such that $\{\tilde{u}_n | n \in \mathbb{N}\}$ separates the points of $E \setminus N$.

1.2 Stochastic calculus associated with Generalized Dirichlet forms

In this section we assume that an *m*-tight special standard process ([MR92, IV Definition 1.13]) $X = (\Omega, \mathcal{F}_{\infty}, \mathcal{F}_t, X_t, P^x)$ is properly associated in the resolvent sense with the quasi-regular generalized Dirichlet form \mathcal{E} , i.e. $R_{\alpha}f := E^x \int_0^{\infty} e^{-\alpha t} f(X_t) dt$ is an \mathcal{E} -quasi-continuous *m*-version of $G_{\alpha}f$, where $G_{\alpha}, \alpha > 0$ is the resolvent of \mathcal{E} and $f \in \mathcal{B}_b(\mathbb{R}^d) \cap L^2(\mathbb{R}^d; m)$. The coform $\hat{\mathcal{E}}$ introduced in Section 1.1 is a generalized Dirichlet form with the associated resolvent $(\hat{G}_{\alpha})_{\alpha>0}$ and there exists an *m*-tight special standard process properly associated in the resolvent sense with $\hat{\mathcal{E}}$. In this section we will obtain the results under this assumption.

Now we recall [Tr2, Theorem 1.9], which give a description of the \mathcal{E} -exceptional set and will be used for the proof of the martingale representation theorem. $\hat{P}_{\hat{G}_1b\mathcal{H}^+}$ denotes the set of all 1-coexcessive elements in \mathcal{V} which are dominated by elements of $\hat{G}_1b\mathcal{H}^+$, where $\hat{G}_1b\mathcal{H}^+ := {\hat{G}_1h|h \in b\mathcal{H}^+}$. $\tilde{\mathcal{P}}_{\mathcal{F}}$ denotes the set of all the \mathcal{E} -q.e. *m*-versions of 1-excessive elements in \mathcal{V} which are dominated by elements of \mathcal{F} .

By [Tr2, Theorem 1.4], we obtain for $\hat{u} \in \hat{P}_{\hat{G}_1 b \mathcal{H}^+}$, there exists a unique σ -finite and positive measure $\mu_{\hat{u}}$ on $(E, \mathcal{B}(E))$ charging no \mathcal{E} -exceptional set such that for all $\tilde{f} \in \widetilde{\mathcal{P}}_{\mathcal{F}} - \widetilde{\mathcal{P}}_{\mathcal{F}}$,

$$\int \widetilde{f} d\mu_{\hat{u}} = \lim_{\alpha \to \infty} \mathcal{E}_1(f, \alpha \hat{G}_{\alpha+1} \hat{u})$$

Define

$$\hat{S}_{00} := \{ \mu_{\hat{u}} | \hat{u} \in \hat{P}_{\hat{G}_1 b \mathcal{H}^+} \text{ and } \mu_{\hat{u}}(E) < \infty \}.$$

Then we have the following results from [Tr2, Theorem 1.9].

Theorem 1.4 For $B \in \mathcal{B}(E)$, B is \mathcal{E} -exceptional if and only if $\nu(B) = 0, \forall \nu \in \hat{S}_{00}$.

Definition 1.5 A positive measure μ on $(E, \mathcal{B}(E))$ is said to be of finite 1-order co-energy integral if there exists $\hat{U}_1 \mu \in \mathcal{V}$, such that

$$\int_E \widetilde{G_1 h} d\mu = \mathcal{E}_1(G_1 h, \hat{U}_1 \mu),$$

for all $h \in \mathcal{H}$ and for all \mathcal{E} -q.c. *m*-versions $\widetilde{G_1h}$ of G_1h . The measures of finite 1-order co-energy integral are denoted by \hat{S}_0 .

By [Tr2, Section 1.3], $\hat{S}_{00} \subset \hat{S}_0$.

Now we introduce the spaces which will be relevant for our further investigations.

Definition 1.6 A family $(A_t)_{t\geq 0}$ of extended real valued functions on Ω is called an *additive functional* of X if:

(i) $A_t(\cdot)$ is \mathcal{F}_t -measurable for all $t \ge 0$.

(ii) There exists a defining set $\Lambda \in \mathcal{F}_{\infty}$ and an \mathcal{E} -exceptional set $N \subset E$, such that $P^{z}[\Lambda] = 1$ for all $z \in E \setminus N, \theta_{t}(\Lambda) \subset \Lambda$ for all t > 0 and for each $\omega \in \Lambda, t \mapsto A_{t}(\omega)$ is right continuous on $[0, \infty)$ and has left limits on $(0, \zeta(\omega)), A_{0}(\omega) = 0, |A_{t}(\omega)| < \infty$ for $t < \zeta(\omega), A_{t}(\omega) = A_{\zeta}(\omega)$ for $t \geq \zeta(\omega)$ and $A_{t+s}(\omega) = A_{t}(\omega) + A_{s}(\theta_{t}\omega)$ for $s, t \geq 0$.

Define

$$\mathcal{M} := \{ M | M \text{ is a finite additive functional, } E^{z}[M_{t}^{2}] < \infty, E^{z}[M_{t}] = 0$$
for $\mathcal{E} - q.e.z \in E$ and all $t \geq 0 \}.$

 $M \in \mathcal{M}$ is called a *martingale additive functional*(MAF). Furthermore, define

$$\dot{\mathcal{M}} = \{ M \in \mathcal{M} | e(M) < \infty \}.$$

Here $e(M) = \frac{1}{2} \lim_{\alpha \to \infty} \alpha^2 E^m [\int_0^\infty e^{-\alpha t} M_t^2 dt]$. The elements of $\dot{\mathcal{M}}$ are called martingale additive functional's (MAF) of finite energy. Let $M \in \mathcal{M}$. There exists an \mathcal{E} -exceptional set N, such that $(M_t, \mathcal{F}_t, P_z)_{t\geq 0}$ is a square integrable martingale for all $z \in E \setminus N$. Moreover, there exists a unique (up to equivalence) positive continuous additive functional $\langle M \rangle$, called the sharp bracket of M, such that $(M_t^2 - \langle M \rangle_t, \mathcal{F}_t, P_z)_{t\geq 0}$ is a martingale for all $z \in E \setminus N$. By [Tr1, Theorem 2.10] $\dot{\mathcal{M}}$ is a real Hilbert space with inner product e. It now follows that one half of the total mass of the Revuz measure $\mu_{\langle M \rangle}$ associated to the sharp bracket of $M \in \mathcal{M}$ is equal

to the energy of M, i.e.

$$e(M) = \frac{1}{2} \int d\mu_{\langle M \rangle}$$

For $M, L \in \dot{\mathcal{M}}$ let

$$\langle M, L \rangle_t = \frac{1}{2} (\langle M + L \rangle_t - \langle M \rangle_t - \langle L \rangle_t).$$

Then the finite signed measure $\mu_{\langle M,L\rangle}$ defined by $\mu_{\langle M,L\rangle} = \frac{1}{2}(\mu_{\langle M+L\rangle} - \mu_{\langle M\rangle} - \mu_{\langle L\rangle})$ is the Revuz measure related to $\langle M,L\rangle$. We also define

$$e(M,L) = \frac{1}{2} \lim_{\alpha \to \infty} \alpha^2 E^m \left[\int_0^\infty e^{-\alpha t} M_t L_t dt \right].$$

Define the following space:

$$\mathcal{N}_c := \{N | N \text{ is a finite continuous additive functional, } e(N) = 0, E^z[|N_t|] < \infty$$
for $\mathcal{E} - q.e.z \in E$ and all $t \ge 0\}.$

Now we recall the well-known Fukushima decomposition in the framework of generalized Dirichlet forms.

Theorem 1.7 ([Tr1, Theorem 4.5]) If \hat{G}_{α} is sub-Markovian and strongly continuous on \mathcal{V} , then for $u \in \mathcal{F}$, there exists a unique $M^{[u]} \in \dot{\mathcal{M}}$ and a unique $N^{[u]} \in \mathcal{N}_c$ such that

$$u(X) - u(X_0) = M^{[u]} + N^{[u]}.$$

Furthermore, by [Tr2, Lemma 2.12], we obtain that for $f \in \mathcal{B}_b(\mathbb{R}^d)$ and $M \in \dot{\mathcal{M}}$, there exists a unique element denoted by $f \cdot M \in \dot{\mathcal{M}}$ such that for all $L \in \dot{\mathcal{M}}$

$$\frac{1}{2}\int f d\mu_{\langle M,L\rangle} = e(f\cdot M,L).$$

Chapter 2

BSDE and generalized Dirichlet form: finite dimensional case

In this chapter we establish that the relation between PDE (1.1) and BSDE (1.2) mentioned in introduction holds under the condition that the operator L is associated with a generalized Dirichlet form. In Section 2.1 we give some basic assumptions on the operator L and prove some basic relations for linear equation. In Section 2.2, we use analytic methods to solve PDE (1.1). In Section 2.3, we prove the martingale representation theorem for the process associated with the operator L. By this we obtain the existence and uniqueness of solution of BSDE (1.2) in Section 2.4. The relation between PDE and BSDE is also established in this section. Further extensions and examples are given in Section 2.5. The main results of this chapter have already been submitted for publication, see [Zhu a].

2.1 Preliminaries

Let $\sigma : \mathbb{R}^d \mapsto \mathbb{R}^d \otimes \mathbb{R}^k$ be a measurable map. Then there exists a measurable map $\tau : \mathbb{R}^d \mapsto \mathbb{R}^k \otimes \mathbb{R}^d$ such that

$$\sigma\tau = \tau^*\sigma^*, \qquad \tau\sigma = \sigma^*\tau^*, \qquad \sigma\tau\sigma = \sigma,$$

where σ^* is the transpose of the matrix of σ (see e.g. [BPS05, Lemma A.1]). Then $a := \sigma \sigma^* = (a_{ij})_{1 \le i,j \le d}$ takes values in the space of symmetric non-negative definite matrices. Let also $b : \mathbb{R}^d \to \mathbb{R}^d$ be measurable. Assume that the basic measure m(dx) for the generalized Dirichlet form, to be defined below, is a finite measure or Lebesgue measure on \mathbb{R}^d .

Denote the Euclidean norm and the scalar product in \mathbb{R}^d by $|\cdot|, \langle \cdot, \cdot \rangle$ respec-

tively, while on the space of matrices $\mathbb{R}^d \otimes \mathbb{R}^k$ we use the trace scalar product and its associated norm, i.e., for $z = (z_{ij}) \in \mathbb{R}^d \otimes \mathbb{R}^k$, $\langle z_1, z_2 \rangle = \operatorname{trace}(z_1 z_2^*), |z| =$ $(\sum_{i=1}^d \sum_{j=1}^k z_{ij}^2)^{1/2}$. Let L^2 , $L^2(\mathbb{R}^d; \mathbb{R}^k)$ denote $L^2(\mathbb{R}^d, m)$, $L^2(\mathbb{R}^d, m; \mathbb{R}^k)$ respectively. (\cdot, \cdot) denotes the L^2 -inner product. For $1 \leq p \leq \infty$, $\|\cdot\|_p$ denotes the usual norm in $L^p(\mathbb{R}^d; m)$. If W is a function space, we use bW to denote the bounded function in W.

Furthermore, let a_{ij} , $\sum_{j=1}^{d} a_{ij}b_j$, $\sum_{j=1}^{d} a_{ij}\hat{b}_j \in L^1_{\text{loc}}(\mathbb{R}^d, m)$ and $c \in L^1_{\text{loc}}(\mathbb{R}^d, \mathbb{R}^+; m)$. We introduce the bilinear form

$$\mathcal{E}(u,v) := \sum_{i,j=1}^{d} \int a_{ij}(x) \frac{\partial u}{\partial x^{i}}(x) \frac{\partial v}{\partial x^{j}}(x) m(dx) + \int c(x)u(x)v(x) + \sum_{i=1}^{d} \int \sum_{j=1}^{d} a_{ij}(x)(b_{j}(x) + \hat{b}_{j}(x)) \frac{\partial u}{\partial x^{i}}v(x)m(dx) \ \forall u, v \in C_{0}^{\infty}(\mathbb{R}^{d}).$$

Consider the following conditions:

(A1) The bilinear form

$$\mathcal{E}^{a}(u,v) = \sum_{i,j=1}^{d} \int a_{ij}(x) \frac{\partial u}{\partial x^{i}}(x) \frac{\partial v}{\partial x^{j}}(x) m(dx) \ \forall u, v \in C_{0}^{\infty}(\mathbb{R}^{d}),$$

is closable on $L^2(\mathbb{R}^d, m)$.

Define $\mathcal{E}_1^a(\cdot, \cdot) := \mathcal{E}^a(\cdot, \cdot) + (\cdot, \cdot)$. The closure of $C_0^{\infty}(\mathbb{R}^d)$ with respect to \mathcal{E}_1^a is denoted by F^a . Then (\mathcal{E}^a, F^a) is a well-defined symmetric Dirichlet form on $L^2(\mathbb{R}^d, m)$.

For the bilinear form

$$\mathcal{E}^{a,\hat{b}}(u,v) := \sum_{i,j=1}^{d} \int a_{ij}(x) \frac{\partial u}{\partial x^{i}}(x) \frac{\partial v}{\partial x^{j}}(x) m(dx) + \int c(x) u(x) v(x) + \sum_{i=1}^{d} \int \sum_{j=1}^{d} a_{ij}(x) \hat{b}_{j}(x) \frac{\partial u}{\partial x^{i}} v(x) m(dx),$$

we consider the following conditions:

(A2) There exists a constant $c_2 \ge 0$ such that $\mathcal{E}_{c_2}^{a,\hat{b}}(\cdot,\cdot) := \mathcal{E}^{a,\hat{b}}(\cdot,\cdot) + c_2(\cdot,\cdot)$ is a semi-Dirichlet form (see Section 1.1) with domain $F := \overline{C_0^{\infty}(\mathbb{R}^d)}^{\tilde{\mathcal{E}}_{c_2^{+1}}^{a,\hat{b}}}$, and there exist constants $c_1, c_3 > 0$ such that for $u \in C_0^{\infty}(\mathbb{R}^d)$

(2.1.1)
$$c_1 \mathcal{E}^a(u, u) \le \mathcal{E}^{a, \bar{b}}_{c_2}(u, u),$$

and

(2.1.2)
$$\int cu^2 dm \le c_3 \mathcal{E}_{c_2+1}^{a,\hat{b}}(u,u).$$

where $\tilde{\mathcal{E}}^{a,\hat{b}}(u,v) := \mathcal{E}^{a,\hat{b}}(u,v) + \mathcal{E}^{a,\hat{b}}(v,u).$

By (2.1.1) we have that $F \subset F^a$ and that for $u \in F$ (2.1.1) and (2.1.2) are satisfied.

(A3) $|b\sigma| \in L^2(\mathbb{R}^d; m)$ and there exists $\alpha \ge 0$ such that

(2.1.3)
$$\int \langle b\sigma, (\nabla u^2)\sigma \rangle dm \ge -\alpha \|u\|_2^2, \quad u \in C_0^\infty(\mathbb{R}^d).$$

(A4) There exists a positivity preserving C_0 -semigroup P_t on $L^1(\mathbb{R}^d; m)$ such that for any $t \in [0, T], \exists C_T > 0$

$$\|P_t f\|_{\infty} \le C_T \|f\|_{\infty}.$$

Then for $0 \leq t \leq T$, P_t extends to a semigroup on $L^p(\mathbb{R}^d; m)$ for all $p \in [1, \infty)$ by the Riesz-Thorin Interpolation Theorem (denoted by P_t for simplicity) which is strongly continuous on $L^p(\mathbb{R}^d; m)$. We denote its L^2 -generator by $(L, \mathcal{D}(L))$ and assume that $b\mathcal{D}(L) \subset bF$ and for any $u \in bF$ there exists uniformly bounded $u_n \in \mathcal{D}(L)$ such that $\tilde{\mathcal{E}}_{c_2+1}^{a,\hat{b}}(u_n - u) \to 0$ and that it is associated with the bilinear form in the sense that $\mathcal{E}(u, v) = -(Lu, v)$ for $u, v \in b\mathcal{D}(L)$.

We emphasize that in contrast to previous work P_t in (A4) is no longer analytic on $L^2(\mathbb{R}^d; m)$. By (A4) there exist constants M_0, c_0 such that

(2.1.4)
$$||P_t f||_2 \le M_0 e^{c_0 t} ||f||_2, \ \forall f \in L^2(\mathbb{R}^d; m).$$

To obtain a semigroup P_t satisfying the above conditions, we can use generalized Dirichlet forms (Definition 1.1).

Remark 2.1.1 (i) Some general criteria imposing conditions on a in order that \mathcal{E}^{a} to be closable are e.g. given in [FOT94, Section 3.1] and [MR92, Chap II, Section 2].

(ii) There are examples considered in [MR92, Chap. II, Subsection 2d] satisfying

(A2). Assume the Sobolev inequality

$$||u||_q \le C(\mathcal{E}^a(u, u) + ||u||_2^2)^{1/2}, \ \forall u \in C_0^\infty(\mathbb{R}^d),$$

is satisfied, where $\frac{1}{q} + \frac{1}{d} = \frac{1}{2}$ and $\|\cdot\|_q$ denotes the usual norm in L^q . If $|\hat{b}\sigma| \in L^d(\mathbb{R}^d; m) + L^{\infty}(\mathbb{R}^d; m)$ and $c \in L^{d/2}(\mathbb{R}^d; m) + L^{\infty}(\mathbb{R}^d; m)$, then (A2) is satisfied (see [MOR95]). In [ZR11] they consider the bilinear form $Q(u, v) = \mathcal{E}^{a,\hat{b}}(u, v) + \int \langle d_1(x), \nabla v(x) \rangle u(x) dm$, where $d_1 \in L^q(\mathbb{R}^d), q > d$. In their case, the result for the existence of the solution of the nonlinear PDE can be obtained by [PR07, Theorem 4.2.4] since the nonlinear part is Lipschitz in u and ∇u . In our case, we have more general conditions on b and f, so that we cannot find a suitable Gelfand triple $V \subset H \subset V^*$ with V being a reflexive Banach space and use monotonicity methods as in [PR07].

(iii) We can construct a semigroup P_t satisfying (A4) by the theory of generalized Dirichlet forms. More precisely, suppose there exists a constant $\hat{c} \geq 0$ such that $\mathcal{E}_{\hat{c}}(\cdot, \cdot) := \mathcal{E}(\cdot, \cdot) + \hat{c}(\cdot, \cdot)$ is a generalized Dirichlet form with domain $\mathcal{F} \times \mathcal{V}$ in one of the following three senses:

(a)
$$(E, \mathcal{B}(E), m) = (\mathbb{R}^d, \mathcal{B}(\mathbb{R}^d), m),$$

 $(\mathcal{E}_{c_2}^{a,\hat{b}}, F) = (\mathcal{A}, \mathcal{V}),$
 $-\langle \Lambda u, v \rangle - (\hat{c} - c_2)(u, v) = \sum_i^d \int \sum_{j=1}^d a_{ij}(x) b_j(x) \frac{\partial u}{\partial x^i} v(x) m(dx) \text{ for } u, v \in C_0^{\infty}(\mathbb{R}^d);$
(b) $(E, \mathcal{B}(E), m) = (\mathbb{R}^d, \mathcal{B}(\mathbb{R}^d), m),$
 $\mathcal{A} \equiv 0 \text{ and } \mathcal{V} = L^2(\mathbb{R}^d, m),$
 $-\langle \Lambda u, v \rangle = \mathcal{E}_{\hat{c}}(u, v) \text{ for } u, v \in C_0^{\infty}(\mathbb{R}^d) \text{ and } C_0^{\infty}(\mathbb{R}^d) \subset \mathcal{D}(L);$
(c) $\mathcal{E}_{\hat{c}} = \mathcal{A}, \Lambda \equiv 0$ (In this case $(\mathcal{E}_{\hat{c}}, \mathcal{V})$ is a sectorial Dirichlet form in the sense of [MR92]).

Then there exists a sub-Markovian C_0 -semigroup of contractions $P_t^{\hat{c}}$ associated with the generalized Dirichlet form $\mathcal{E}_{\hat{c}}$. Define $P_t := e^{\hat{c}t}P_t^{\hat{c}}$. If it is a C_0 -semigroup on L^1 then it satisfies (A4). Then we have

$$\mathcal{D}(L) \subset \mathcal{F} \subset F.$$

(iv) The semigroup can be also constructed by other methods. (see e.g. [DR02], [BDR09]).

(v) By (A3) we have that \mathcal{E} is positivity preserving i.e.

$$\mathcal{E}(u, u^+) \ge 0 \ \forall u \in \mathcal{D}(L),$$

which can be obtained by the same arguments as [St2, Proposition 4.4].

2.1. Preliminaries

(vi) The condition that for any $u \in bF$ there exists uniformly bounded $u_n \in \mathcal{D}(L)$ such that $\tilde{\mathcal{E}}_{c_2+1}^{a,\hat{b}}(u_n - u) \to 0$ is satisfied if $C_0^{\infty}(\mathbb{R}^d) \subset \mathcal{D}(L)$. It can also be satisfied in the case of (iii) by the theory of generalized Dirichlet form.

(vii) All the conditions are satisfied by the bilinear form considered in [DR02], [L01], [St1, Section 1 (a)] and the following example which is considered in [St2].

Example 2.1.2 Let $b_i \in L^2(\mathbb{R}^d; dx), 1 \leq i \leq d$. Consider the bilinear form

$$\mathcal{E}(u,v) := \sum_{i,j=1}^{d} \int_{\mathbb{R}^d} \frac{\partial u}{\partial x_i} \frac{\partial v}{\partial x_j} dx - \sum_{i=1}^{d} \int b_i \frac{\partial u}{\partial x_i} v dx; u, v \in C_0^{\infty}(\mathbb{R}^d)$$

Assume there exist constants $c, L \ge 0$ such that

$$\begin{split} \int \langle b, \nabla u \rangle dx &\leq 2c \|u\|_1 \text{ for all } u \in C_0^\infty(\mathbb{R}^d), u \geq 0, \\ -\sum_{i,j=1}^d \int b_i \frac{\partial u}{\partial x_j} dx h_i h_j \leq L \|u\|_1 |h|^2, \\ \text{ for all } u \in C_0^\infty(\mathbb{R}^d), u \geq 0, h \in \mathbb{R}^d, \end{split}$$

(or equivalently, b is quasi-monotone, i.e.

$$\langle b(x) - b(y), x - y \rangle \le L |x - y|^2, \forall x, y \in \mathbb{R}^d,)$$

and for some continuous, monotone increasing function $f:[0,\infty) \to [1,\infty)$ with $\int_0^\infty \frac{dr}{f(r)} = \infty$ we have that

$$|b(x)| \le f(|x|), x \in \mathbb{R}^d.$$

Then in [St2, Subsection II.2] it is proved that there exists a generalized Dirichlet form in $L^2(\mathbb{R}^d)$ extending \mathcal{E}_c . We denote the semigroup associated with \mathcal{E}_c by P_t^c . If we define $P_t := e^{ct} P_t^c$, then it is the semigroup associated with \mathcal{E} . By the computation in [St2, Subsection II.2], P_t is sub-Markovian. So it satisfies the conditions (A1)-(A4).

Further examples are presented in Section 2.3 (see Examples 2.3.2 and 2.3.3) and Sections 2.4, 2.5.

Then we use the same notations $\hat{F}, \mathcal{C}_T, \|\cdot\|_T$ associated with $\mathcal{E}^{a,\hat{b}}$ as in [BPS05]: $\mathcal{C}_T = C^1([0,T]; L^2) \cap L^2([0,T]; F)$, which turns out to be the appropriate space of test functions, i.e.

$$\mathcal{C}_T = \{ \varphi : [0,T] \times \mathbb{R}^d \to \mathbb{R} | \varphi_t \in F \text{ for almost each } t, \int_0^T \mathcal{E}^{a,\hat{b}}(\varphi_t) dt < \infty, \\ t \to \varphi_t \text{ is differentiable in } L^2 \text{ and } t \to \partial_t \varphi_t \text{ is } L^2 - \text{ continuous on } [0,T] \}.$$

Here and below we set $\mathcal{E}^{a,\hat{b}}(u)$ for $\mathcal{E}^{a,\hat{b}}(u,u)$. We also set $\mathcal{C}_{[a,b]} = C^1([a,b];L^2) \cap L^2([a,b];F)$. For $\varphi \in \mathcal{C}_T$, we define

$$\|\varphi\|_T := (\sup_{t \le T} \|\varphi_t\|_2^2 + \int_0^T \mathcal{E}_{c_2}^{a,\hat{b}}(\varphi_t) dt)^{1/2}.$$

 \hat{F} is the completion of \mathcal{C}_T with respect to $\|\cdot\|_T$. By [BPS05], $\hat{F} = C([0,T]; L^2) \cap L^2(0,T;F)$. Define the space \hat{F}^a w.r.t. \mathcal{E}_1^a analogous to \hat{F} . Then we have $\hat{F} \subset \hat{F}^a$. We also introduce the following space

$$W^{1,2}([0,T];L^2(\mathbb{R}^d)) = \{ u \in L^2([0,T];L^2); \partial_t u \in L^2([0,T];L^2) \},\$$

where $\partial_t u$ is the derivative of u in the weak sense (see e.g. [Ba10]).

2.1.1 Linear Equations

Consider the linear equation

(2.1.5)
$$\begin{aligned} (\partial_t + L)u + f &= 0, \quad 0 \le t \le T, \\ u_T(x) &= \phi(x), \quad x \in \mathbb{R}^d, \end{aligned}$$

where $f \in L^1([0, T]; L^2), \phi \in L^2$.

As in [BPS05] we set $D_{\sigma}\varphi := (\nabla \varphi)\sigma$ for any $\varphi \in C_0^{\infty}(\mathbb{R}^d)$, define $V_0 = \{D_{\sigma}\varphi : \varphi \in C_0^{\infty}(\mathbb{R}^d)\}$, and let V be the closure of V_0 in $L^2(\mathbb{R}^d;\mathbb{R}^k)$. Then we have the following results:

Proposition 2.1.3 Assume (A1)-(A3) hold. Then:

(i) For every $u \in F^a$ there is a unique element of V, which we denote by $D_{\sigma}u$, such that

$$\mathcal{E}^{a}(u) = \int \langle D_{\sigma}u(x), D_{\sigma}u(x) \rangle m(dx).$$

(ii) Furthermore, if $u \in \hat{F}^a$, then there exists a measurable function $\phi : [0, T] \times \mathbb{R}^d \mapsto \mathbb{R}^d$ such that $|\phi\sigma| \in L^2((0, T) \times \mathbb{R}^d)$ and $D_{\sigma}u_t = \phi_t \sigma$ for almost all $t \in [0, T]$. (iii)Let $u^n, u \in \hat{F}^a$ be such that $u^n \to u$ in $L^2((0, T) \times \mathbb{R}^d)$ and $(D_{\sigma}u^n)_n$ is Cauchy in $L^2([0,T] \times \mathbb{R}^d; \mathbb{R}^k)$. Then $D_{\sigma}u^n \to D_{\sigma}u$ in $L^2((0,T) \times \mathbb{R}^d; \mathbb{R}^k)$, i.e. D_{σ} is closed as an operator from \hat{F}^a into $L^2((0,T) \times \mathbb{R}^d)$.

Proof See [BPS05, Proposition 2.3].

For $u \in F, v \in bF$, we define

$$\mathcal{E}(u,v) := \mathcal{E}^{a,\hat{b}}(u,v) + \int \langle b\sigma, D_{\sigma}u \rangle vm(dx).$$

Notation We denote by $\tilde{\nabla}u$ the set of all measurable functions $\phi : \mathbb{R}^d \to \mathbb{R}^d$ such that $\phi\sigma = D_{\sigma}u$ as elements of $L^2(\mathbb{R}^d, \mathbb{R}^k)$.

2.1.2 Solution of the Linear Equation

We recall the following standard notions.

Definition 2.1.4 (strong solutions) A function $u \in \hat{F} \cap L^1((0,T); \mathcal{D}(L))$ is called a strong solution of equation (2.1.5) with data ϕ, f , if $t \mapsto u_t = u(t, \cdot)$ is L^2 differentiable on $[0,T], \partial_t u_t \in L^1((0,T); L^2)$ and the equalities in (2.1.5) hold *m*-a.e.

Definition 2.1.5 (generalized solutions) A function $u \in \hat{F}$ is called a generalized solution of equation (2.1.5), if there are sequences $\{u^n\}$ which are strong solutions with data (ϕ^n, f^n) such that

$$||u^n - u||_T \to 0, ||\phi^n - \phi||_2 \to 0, \lim_{n \to \infty} f^n = f \text{ in } L^1([0, T]; L^2).$$

Proposition 2.1.6 Assume (A3)-(A4) hold.

(i) Let $f \in C^1([0,T]; L^p)$ for $p \in [1,\infty)$. Then

$$w_t := \int_t^T P_{s-t} f_s ds \in C^1([0,T]; L^p),$$

and

$$\partial_t w_t = -P_{T-t} f_T + \int_t^T P_{s-t} \partial_s f_s ds$$

(ii) Assume that $\phi \in \mathcal{D}(L)$, $f \in C^1([0,T]; L^2)$ and for each $t \in [0,T]$, $f_t \in \mathcal{D}(L)$. Define

$$u_t := P_{T-t}\phi + \int_t^T P_{s-t}f_s ds.$$

Then u is a strong solution of (2.1.5) and, moreover, $u \in C^1([0,T]; L^2)$.

Proof By the same arguments as in the proof of [BPS05, Proposition 2.6] the results follow. \Box

Remark 2.1.7 Compared to [BPS05, Proposition 2.6], in (ii) we add the assumption $\phi \in \mathcal{D}(L)$ and $f_t \in \mathcal{D}(L)$, $t \in [0, T]$, as we cannot deduce $P_t \phi \in \mathcal{D}(L)$ for $\phi \in L^2$, since (P_t) might not be analytic.

Proposition 2.1.8 Suppose (A4) holds. If u is a strong solution for (2.1.5), it is a mild solution for (2.1.5) i.e.

$$u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds.$$

Proof For fixed $t, \varphi \in \mathcal{D}(\hat{L})$

$$(u_T, \hat{P}_{T-t}\varphi) - (u_t, \varphi) = \int_t^T (-Lu_s - f_s, \hat{P}_{s-t}\varphi)ds + \int_t^T (u_s, \hat{L}\hat{P}_{s-t}\varphi)ds,$$

where \hat{L}, \hat{P}_t denote the adjoints on $L^2(\mathbb{R}^d, m)$ of L, P_t respectively. As u is a strong solution, we can deduce that

$$(u_t, \varphi) = (P_{T-t}\phi + \int_t^T P_{s-t}f_s ds, \varphi).$$

Since $\mathcal{D}(\hat{L})$ is dense in L^2 , the result follows.

Proposition 2.1.9 Suppose that conditions (A1)-(A4) hold, $f \in L^1([0,T]; L^2)$ and $\phi \in L^2$. Then equation (2.1.5) has a unique generalized solution $u \in \hat{F}$ and

(2.1.6)
$$u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds$$

The solution satisfies the three relations:

(2.1.7)
$$\|u_t\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(u_s)ds \le 2\int_t^T (f_s, u_s)ds + \|\phi\|_2^2 + 2\alpha\int_t^T \|u_s\|_2^2ds, \qquad 0 \le t \le T.$$

(2.1.8)
$$\|u\|_T^2 \le M_T(\|\phi\|_2^2 + (\int_0^T \|f_t\|_2 dt)^2).$$

$$(2.1.9) \int_{t_0}^T ((u_t, \partial_t \varphi_t) + \mathcal{E}^{a, \hat{b}}(u_t, \varphi_t) + \int \langle b\sigma, D_\sigma u_t \rangle \varphi_t dm) dt = \int_{t_0}^T (f_t, \varphi_t) dt + (\phi, \varphi_T) - (u_{t_0}, \varphi_{t_0}),$$

for any $\varphi \in b\mathcal{C}_T, t_0 \in [0, T]$. M_T is a constant depending on T. (2.1.9) can be extended easily for $\varphi \in bW^{1,2}([0, T]; L^2) \cap L^2([0, T]; F)$.

Moreover, if $u \in \hat{F}$ is bounded and satisfies (2.1.9) for any $\varphi \in b\mathcal{C}_T$ with bounded f, ϕ , then u is a generalized solution given by (2.1.6).

Proof [Existence] Define u by (2.1.6). First assume that ϕ , f are bounded and satisfy the conditions of Proposition 2.1.6 (ii). Then, since u is bounded and by Proposition 2.1.6 we know that u is a strong solution of (2.1.5), hence it obviously satisfies (2.1.9). Furthermore, $u \in C^1([0,T]; L^2)$. Hence, actually $u \in bC_T$ and consequently, for $t \in [0,T]$

$$\int_{t}^{T} ((u_s, \partial_t u_s) + \mathcal{E}^{a, \hat{b}}(u_s, u_s) + \int \langle b\sigma, D_{\sigma} u_s \rangle u_s dm) ds = \int_{t}^{T} (f_s, u_s) ds + (\phi, u_T) - (u_t, u_t).$$

By (2.1.3) we have $\int \langle b\sigma, D_{\sigma}u_s \rangle u_s dm \geq -\alpha \|u_s\|_2^2$. Hence

$$\|u_t\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(u_s)ds \le 2\int_t^T (f_s, u_s)ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|u_s\|_2^2 ds, \qquad 0 \le t \le T.$$

As

$$\begin{split} \int_{t}^{T} (f_{s}, u_{s}) ds &= \int_{t}^{T} ((f_{s}, P_{T-s}\phi) + (f_{s}, \int_{s}^{T} P_{r-s}f_{r}dr)) ds \\ &\leq \int_{t}^{T} \|f_{s}\|_{2} \|P_{T-s}\phi\|_{2} ds + \int_{t}^{T} \|f_{s}\|_{2} \|\int_{s}^{T} P_{r-s}f_{r}dr\|_{2} ds \\ &\leq M_{0}e^{T-t} (\|\phi\|_{2} \int_{t}^{T} \|f_{s}\|_{2} ds + \int_{t}^{T} (\|f_{s}\|_{2} \int_{s}^{T} \|f_{r}\|_{2} dr) ds), \end{split}$$

and

$$\int_{t}^{T} \|u_{s}\|_{2}^{2} ds \leq M_{T-t}(\|\phi\|_{2}^{2} + (\int_{0}^{T} \|f_{t}\|_{2} dt)^{2}),$$

we get

$$\|u_t\|_2^2 + \int_t^T \mathcal{E}^{a,\hat{b}}(u_s) ds \le M_{T-t}(\|\phi\|_2^2 + (\int_0^T \|f_t\|_2 dt)^2).$$

Hence, it follows that

(2.1.10)
$$\|u\|_T^2 \le M_T(\|\phi\|_2^2 + (\int_0^T \|f_t\|_2 dt)^2).$$

Here M_{T-t} can change from line to line and is independent of f, ϕ . Now we will prove the result for general data ϕ and f. Let $(f^n)_{n \in \mathbb{N}}$ be a sequence of bounded function in $C^1([0,T]; L^2)$ such that $f_t \in \mathcal{D}(L)$ for a.e. $t \in [0,T]$ and $\int_0^T ||f_t^n - f_t||_2 dt \to 0$ (This sequence can be obtained since $\{\alpha_t g(x); \alpha_t \in C_0^\infty[0,T], g \in b\mathcal{D}(L)\}$ is dense in $L^1([0,T]; L^2)$). Take bounded functions $(\phi^n)_{n \in \mathbb{N}} \subset \mathcal{D}(L)$ such that $\phi^n \to \phi$ in L^2 . Let u^n denote the solution given by (2.1.6) with $f = f^n, \phi = \phi^n$.

By linearity, $u^n - u^m$ is associated with $(\phi^n - \phi^m, f^n - f^m)$. Since by (2.1.10)

$$||u^{n} - u^{m}||_{T}^{2} \le M_{T}(||\phi^{n} - \phi^{m}||_{2}^{2} + (\int_{0}^{T} ||f_{t}^{n} - f_{t}^{m}||_{2}dt)^{2}),$$

we deduce that $(u^n)_{n \in \mathbb{N}}$ is a Cauchy sequence in \hat{F} . Hence $u = \lim_{n \to \infty} u^n$ in $\|\cdot\|_T$ is the generalized solution of (2.1.5) and we have

$$u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds$$

Next we prove (2.1.7)(2.1.8) (2.1.9) for u. We have (2.1.9) for u^n with f^n, ϕ^n and $\varphi \in b\mathcal{C}_T$, i.e.

$$\int_0^T ((u_t^n, \partial_t \varphi_t) + \mathcal{E}^{a, \hat{b}}(u_t^n, \varphi_t) + \int \langle b\sigma, D_\sigma u_t^n \rangle \varphi_t dm) dt = \int_0^T (f_t^n, \varphi_t) dt + (\phi^n, \varphi_T) - (u_0^n, \varphi_0).$$

Since we have

$$\begin{split} |\int_{0}^{T} \mathcal{E}^{a,\hat{b}}(u_{t}^{n}-u_{t},\varphi_{t})dt| \leq & K(\int_{0}^{T} \mathcal{E}^{a,\hat{b}}_{c_{2}+1}(u_{t}^{n}-u_{t})dt)^{\frac{1}{2}} (\int_{0}^{T} \mathcal{E}^{a,\hat{b}}_{c_{2}+1}(\varphi_{t})dt)^{\frac{1}{2}} \\ &+ \int_{0}^{T} (c_{2}+1)(u_{t}^{n}-u_{t},\varphi_{t})dt \\ &\to 0, \text{ as } n \to \infty, \end{split}$$

and

$$\begin{split} |\int_0^T \int \langle b\sigma, D_\sigma(u_t^n - u_t) \rangle \varphi_t dm dt | &\leq \|\varphi\|_{\infty} (\int_0^T \int |b\sigma|^2 dm dt)^{\frac{1}{2}} (\int_0^T \int |D_\sigma(u_t^n - u_t)|^2 dm dt)^{\frac{1}{2}} \\ &= \|\varphi\|_{\infty} (\int_0^T \int |b\sigma|^2 dm dt)^{\frac{1}{2}} (\int_0^T \mathcal{E}^a(u_t^n - u_t) dt)^{\frac{1}{2}} \\ &\to 0, \text{ as } n \to \infty, \end{split}$$

we deduce that

$$\int_0^T ((u_t, \partial_t \varphi_t) + \mathcal{E}^{a, \hat{b}}(u_t, \varphi_t) + \int \langle b\sigma, D_\sigma u_t \rangle \varphi_t dm) dt = \int_0^T (f_t, \varphi_t) dt + (\phi, \varphi_T) - (u_0, \varphi_0),$$

for any $\varphi \in b\mathcal{C}_T$.

The relations (2.1.7), (2.1.8) hold for the approximating functions:

$$\|u_t^n\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(u_s^n) ds \le 2\int_t^T (f_s^n, u_s^n) ds + \|\phi^n\|_2^2 + 2\alpha \int_t^T \|u_s^n\|_2^2 ds, \qquad 0 \le t \le T,$$

and

$$||u^n||_T^2 \le M_T(||\phi^n||_2^2 + (\int_0^T ||f_t^n||_2 dt)^2).$$

Since $||u_t^n||_T \to ||u_t||_T$, $n \to \infty$, we conclude

$$\lim_{n \to \infty} \int_0^T \mathcal{E}^{a,\hat{b}}(u_t^n) dt = \int_0^T \mathcal{E}^{a,\hat{b}}(u_t) dt$$

It is easy to see that $\lim_{n\to\infty} \int_t^T (f_s^n, u_s^n) ds = \int_t^T (f_s, u_s) ds$. Then by passing to the limit in the above relations, (2.1.7) and (2.1.8) hold for u.

[Uniqueness] Let $v \in \hat{F}$ be another generalized solution of (2.1.5) and $(v^n)_{n \in \mathbb{N}}$, $(\tilde{\phi}^n)_{n \in \mathbb{N}}, (\tilde{f}^n)_{n \in \mathbb{N}}$ be the corresponding approximating sequences in the definition of generalized solutions. By Proposition 2.1.8

$$\sup_{t \in [0,T]} \|u_t^n - v_t^n\|_2^2 \le M_T(\|\phi^n - \tilde{\phi}^n\|_2^2 + (\int_0^T \|f_t^n - \tilde{f}_t^n\|_2 dt)^2).$$

Letting $n \to \infty$, this implies u = v.

For the last result we note that
$$\forall t_0 \in [0, T], \varphi \in b\mathcal{C}_T$$

(2.1.11)
 $\int_{t_0}^T ((u_t, \partial_t \varphi_t) + \mathcal{E}^{a, \hat{b}}(u_t, \varphi_t) + \int \langle b\sigma, D_\sigma u_t \rangle \varphi_t dm) dt = \int_{t_0}^T (f_t, \varphi_t) dt + (\phi, \varphi_T) - (u_{t_0}, \varphi_{t_0}).$

For $t \geq \frac{1}{n}$, define

$$u_t^n := n \int_0^{\frac{1}{n}} u_{t-s} ds, \qquad f_t^n := n \int_0^{\frac{1}{n}} f_{t-s} ds, \qquad \phi^n := n \int_0^{\frac{1}{n}} u_{T-s} ds.$$

Let us check that each u^n also fulfills (2.1.11) with f^n, ϕ^n . We set $\varphi_r^s := \varphi_{r+s}$ for $0 \leq s+r \leq T$. Then for fixed $t_0 \in (0,T]$, and $n \geq \frac{1}{t_0}$,

$$\int_{t_0}^T ((u_t^n, \partial_t \varphi_t) + \mathcal{E}^{a, \hat{b}}(u_t^n, \varphi_t) + \int \langle b\sigma, D_\sigma u_t^n \rangle \varphi_t dm) dt$$
$$= n \int_0^{\frac{1}{n}} \int_{t_0}^T (u_{t-s}, \partial_t \varphi_t) + \mathcal{E}^{a, \hat{b}}(u_{t-s}, \varphi_t) + \int \langle b\sigma, D_\sigma u_{t-s} \rangle \varphi_t dm dt ds$$

$$=n\int_{0}^{\frac{1}{n}}\int_{t_{0}-s}^{T-s}(u_{t},\partial_{t}\varphi_{t}^{s}) + \mathcal{E}^{a,\hat{b}}(u_{t},\varphi_{t}^{s}) + \int \langle b\sigma, D_{\sigma}u_{t}\rangle\varphi_{t}^{s}dmdtds$$

$$=n\int_{0}^{\frac{1}{n}}[\int_{t_{0}-s}^{T-s}(f_{t},\varphi_{t}^{s})dt + (u_{T},\varphi_{T-s}^{s}) - (u_{t_{0}-s},\varphi_{t_{0}-s}^{s})]ds$$

$$=n\int_{0}^{\frac{1}{n}}[\int_{t_{0}}^{T}(f_{t-s},\varphi_{t})dt + (u_{T-s},\varphi_{T}) - (u_{t_{0}-s},\varphi_{t_{0}})]ds$$

$$=\int_{t_{0}}^{T}(f_{t}^{n},\varphi_{t})dt + (\phi^{n},\varphi_{T}) - (u_{t_{0}}^{n},\varphi_{t_{0}}).$$

For the mild solution v associated with f, ϕ , the above relation also holds with v^n replacing u^n . Hence we have

$$\int_{t_0}^T (((u-v)_t^n, \partial_t \varphi_t) + \mathcal{E}^{a,\hat{b}}((u-v)_t^n, \varphi_t) + \int \langle b\sigma, D_\sigma(u-v)_t^n \rangle \varphi_t dm) dt$$
$$= -((u-v)_{t_0}^n, \varphi_{t_0}).$$

Since $(u-v)_t^n \in b\mathcal{C}_{[\frac{1}{n},T]}$ the above equation holds with $(u-v)_t^n$ as a test function, i.e. for $n \geq \frac{1}{t_0}$

$$\int_{t_0}^T (((u-v)_t^n, \partial_t (u-v)_t^n) + \mathcal{E}^{a,\hat{b}}((u-v)_t^n, (u-v)_t^n) + \int \langle b\sigma, D_\sigma (u-v)_t^n \rangle (u-v)_t^n dm) dt$$

= - ((u-v)_{t_0}^n, (u-v)_{t_0}^n).

So we have

$$\|(u-v)_{t_0}^n\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}((u-v)_t^n, (u-v)_t^n)dt \le 2\alpha \int_{t_0}^T \|(u-v)_t^n\|_2^2 dt.$$

By Gronwall's lemma it follows that

$$\|(u-v)_{t_0}^n\|_2^2 = 0.$$

Letting $n \to \infty$, we have $||u_{t_0} - v_{t_0}||_2 = 0$. Then letting $t_0 \to 0$, we have $||u_0 - v_0|| = 0$. Then $u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds$ is a generalized solution for (2.1.5).

2.1.3 Basic Relations for the Linear Equation

In this section we assume that (A1)-(A4) hold.

Lemma 2.1.10 If u is a bounded generalized solution of equation (2.1.5) with some function $\phi \ge 0$, $\phi \in L^2 \cap L^\infty$, then u^+ satisfies the following relation for

2.1. Preliminaries

 $0 \le t_1 < t_2 \le T$

$$||u_{t_1}^+||_2^2 \le 2 \int_{t_1}^{t_2} (f_s, u_s^+) ds + ||u_{t_2}^+||_2^2.$$

Proof Choose the approximation sequence u^n for u as in the existence proof of Proposition 2.1.9. Denote its related data by f^n, ϕ^n .

We have the following equations:

$$\lim_{n \to \infty} \sup_{t \in [0,T]} \|u_t^n - u_t\|_2 = 0, \qquad \lim_{n \to \infty} \int_0^T \mathcal{E}^{a,\hat{b}}(u_t^n - u_t)dt = 0,$$
$$\lim_{n \to \infty} \int_0^T \|f_t^n - f_t\|_2 dt = 0, \qquad \lim_{n \to \infty} \|\phi^n - \phi\|_2 = 0.$$

Suppose that the following holds

(2.1.12)
$$\|(u_{t_1}^n)^+\|_2^2 \le 2 \int_{t_1}^{t_2} (f_s^n, (u_s^n)^+) ds + \|(u_{t_2}^n)^+\|_2^2,$$

where $0 \le t_1 \le t_2 \le T$. Since $||u^n||_2$ are uniformly bounded, we obtain

$$\lim_{n \to \infty} \int_{t_1}^{t_2} (f_s^n, (u_s^n)^+) ds = \int_{t_1}^{t_2} (f_s, u_s^+) ds.$$

By passing n to the limit in equation (2.1.12) we get for $0 \le t_1 \le t_2 \le T$,

$$||u_{t_1}^+||_2^2 \le 2 \int_{t_1}^{t_2} (f_s, u_s^+) ds + ||u_{t_2}^+||_2^2.$$

Therefore, the problem is reduced to the case that u belongs to $b\mathcal{C}_T$; in the remainder we assume $u \in b\mathcal{C}_T$. (2.1.9), written with $u^+ \in bW^{1,2}([0,T];L^2) \cap L^2([0,T];F)$ as test function, takes the form

(2.1.13)
$$\int_{t_1}^{t_2} (u_t, \partial_t(u_t^+)) dt + \int_{t_1}^{t_2} \mathcal{E}^{a,\hat{b}}(u_t, u_t^+) dt + \int_{t_1}^{t_2} \int \langle b\sigma, D_\sigma u_t \rangle u_t^+ dm dt \\ = \int_{t_1}^{t_2} (f_t, u_t^+) dt + (u_{t_2}, u_{t_2}^+)) - (u_{t_1}, u_{t_1}^+)).$$

By [Ba10, Theorem 1.19] we obtain

$$\int_{t_1}^{t_2} (u_t, \partial_t(u_t^+)) dt = \frac{1}{2} (\|u_{t_2}^+\|_2^2 - \|u_{t_1}^+\|_2^2).$$

Then

(2.1.14)
$$\|u_{t_1}^+\|_2^2 + 2\int_{t_1}^{t_2} \mathcal{E}^{a,\hat{b}}(u_t, u_t^+)dt + 2\int_{t_1}^{t_2} \int \langle b\sigma, D_{\sigma}u_t \rangle u_t^+ dm dt$$
$$= 2\int_{t_1}^{t_2} (f_t, u_t^+)dt + \|u_{t_2}^+\|_2^2.$$

Next we prove for $u \in bF$

$$(2.1.15)\qquad\qquad\qquad \mathcal{E}(u,u^+)\geq 0.$$

By Remark 2.1.1 (v), we have the above relation for $u \in \mathcal{D}(L)$. For $u \in bF$, by (A4) we can choose a uniformly bounded sequence $\{u_n\} \subset \mathcal{D}(L)$ such that $\mathcal{E}_{c_2+1}^{a,\hat{b}}(u_n-u) \to 0$. Then we have

$$\begin{split} &|\int \langle b\sigma, D_{\sigma}u \rangle u^{+}dm - \int \langle b\sigma, D_{\sigma}u_{n} \rangle u_{n}^{+}dm| \\ &\leq |\int \langle b\sigma, D_{\sigma}u_{n} - D_{\sigma}u \rangle u_{n}^{+}dm| + |\int \langle b\sigma, D_{\sigma}u \rangle (u_{n}^{+} - u^{+})dm| \\ &\leq M(\int |D_{\sigma}u_{n} - D_{\sigma}u|^{2}dm)^{\frac{1}{2}} + |\int \langle b\sigma, D_{\sigma}u \rangle (u_{n}^{+} - u^{+})dm| \\ &\rightarrow 0. \end{split}$$

By (A2) and [MOR95] $\mathcal{E}^{a,\hat{b}}(u^+) \leq 4K^2 \mathcal{E}^{a,\hat{b}}(u)$, $\sup_n \mathcal{E}^{a,\hat{b}}(u_n^+) \leq 4K^2 \sup_n \mathcal{E}^{a,\hat{b}}(u_n) < \infty$, we also have

$$\begin{split} |\mathcal{E}^{a,b}(u_n,(u_n)^+) - \mathcal{E}^{a,b}(u,u^+)| \\ \leq |\mathcal{E}^{a,\hat{b}}_{c_2+1}(u_n - u,(u_n)^+) + \mathcal{E}^{a,\hat{b}}_{c_2+1}(u,(u_n)^+ - u^+)| \\ + (c_2 + 1)|(u_n - u,(u_n)^+)| + (c_2 + 1)|(u,(u_n)^+ - u^+)| \\ \leq K(\mathcal{E}^{a,\hat{b}}_{c_2+1}(u_n - u))^{\frac{1}{2}}(\mathcal{E}^{a,\hat{b}}_{c_2+1}((u_n)^+))^{\frac{1}{2}} + |\mathcal{E}^{a,\hat{b}}_{c_2+1}(u,(u_n)^+ - u^+)| \\ + (c_2 + 1)(||(u_n)^+||_2||u_n - u||_2 + (c_2 + 1)||(u_n)^+ - u^+||_2||u||_2) \\ \rightarrow 0. \end{split}$$

As a result, we obtain (2.1.15) for $u \in bF$. Then the assertion follows.

To extend the class of solutions we are working with to allow $f \in L^1(dt \times dm)$, we need the following proposition. It is a modified version of the above lemma.

Lemma 2.1.11 Let $u \in b\hat{F}$ and $f \in L^1(dt \times dm)$ satisfying the weak relation (2.1.9) with test functions in $b\mathcal{C}_T$ and some function $\phi \geq 0, \phi \in L^2 \cap L^\infty$. Then u^+

satisfies the following relation with $0 \le t_1 < t_2 \le T$

$$||u_{t_1}^+||_2^2 \le 2 \int_{t_1}^{t_2} (f_s, u_s^+) ds + ||u_{t_2}^+||_2^2.$$

Proof First note that we can prove analogously to the proof of Lemma 2.1.10 that for each $u \in bC_T$ satisfying the weak relation (2.1.9) with data (ϕ, f) over the interval $[t_1, t_2]$, where $\varepsilon \leq t_1 \leq t_2 \leq T$ for $\varepsilon > 0$, the following holds

$$||u_{t_1}^+||_2^2 \le 2 \int_{t_1}^{t_2} (f_t, u_t^+) dt + ||u_{t_2}^+||_2^2.$$

For $u \in \hat{F}$ we take approximating functions u^n and (ϕ^n, f^n) as in the last part of the proof of Proposition 2.1.9. Then we have that u^n satisfies the weak relation (2.1.9) for the data ϕ^n, f^n with test functions in $b\mathcal{C}_T$ over the interval $[\varepsilon, t_2]$ and $\frac{1}{n} \leq \varepsilon \leq t_2 \leq T$. Note

$$\lim_{n \to \infty} \int_{\varepsilon}^{T} \|f_t^n - f_t\|_1 dt = 0.$$

Then we obtain

$$\|(u_{t_1}^n)^+\|_2^2 \le 2\int_{t_1}^{t_2} (f_t^n, (u_t^n)^+)dt + \|(u_{t_2}^n)^+\|_2^2,$$

where $\varepsilon \leq t_1 \leq t_2 \leq T$ for $\varepsilon > 0$. The convergence of all terms, which does not depend on f, follows by the same arguments as the proof of Lemma 2.1.10. Since u is bounded, it is easy to see that u^n is uniformly bounded. Then we have

$$\lim_{n \to \infty} \left| \int_{t_1}^{t_2} (f_s^n, (u_s^n)^+) ds - \int_{t_1}^{t_2} (f_s, u_s^+) ds \right|$$

$$\leq M \lim_{n \to \infty} \int_{t_1}^{t_2} \|f_s^n - f_s\|_1 ds + \lim_{n \to \infty} \int_{t_1}^{t_2} (f_s, (u_s^n)^+ - u_s^+) ds$$

=0.

Finally, we obtain

$$\|u_{t_1}^+\|_2^2 \le 2\int_{t_1}^{t_2} (f_t, u_t^+) dt + \|u_{t_2}^+\|_2^2,$$

where $\varepsilon \leq t_1 \leq t_2 \leq T$ for $\varepsilon > 0$. Letting $\varepsilon \to 0$, the assertion follows.

The next proposition is a modification of [BPS05, Proposition 2.9]. It represents a version of the maximum principle.

Proposition 2.1.12 Let $u \in b\hat{F}$ and $f \in L^1(dt \times dm), f \ge 0$, satisfying the weak

relation (2.1.9) with test functions in bC_T and some function $\phi \ge 0$, $\phi \in L^2 \cap L^{\infty}$. Then $u \ge 0$ and it is represented by the following relation:

$$u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds.$$

Here we use P_t is a C_0 -semigroup on $L^1(\mathbb{R}^d; m)$ to make $P_{s-t}f_s$ meaningful.

Proof Let $(f^n)_{n \in \mathbb{N}}$ be a sequence of bounded functions such that

$$0 \le f^n \le f^{n+1} \le f, \qquad \lim_{n \to \infty} f^n = f$$

Since f^n is bounded, we have $f^n \in L^1([0,T]; L^2)$. Define

$$u_t^n := P_{T-t}\phi + \int_t^T P_{s-t}f_s^n ds.$$

Then by Proposition 2.1.9, $u^n \in \hat{F}$ is a unique generalized solution for the data (ϕ, f^n) . Clearly $0 \leq u^n \leq u^{n+1}$ for $n \in \mathbb{N}$. Define $y := u^n - u$ and $\tilde{f} := f^n - f$. Then $\tilde{f} \leq 0$ and y satisfies the weak relation (2.1.9) for the data $(0, \tilde{f})$. Therefore by Lemma 2.1.11, we have for $t_1 \in [0, T]$

$$||y_{t_1}^+||_2^2 \le 2 \int_{t_1}^T (\tilde{f}_s, y_s^+) ds \le 0.$$

We conclude that $||y_{t_1}^+||_2^2 = 0$. Therefore, $u \ge u^n \ge 0$ for $n \in \mathbb{N}$. Set $v := \lim_{n \to \infty} u^n$. By (2.1.7) we have

$$\|u_t^n\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(u_s^n) ds \le 2\int_t^T (f_s^n, u_s^n) ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|u_s^n\|_2^2 ds$$

which implies that

$$\|u_t^n\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(u_s^n) ds \le 2M \int_t^T \int |f_s^n| dm ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|u_s^n\|_2^2 ds.$$

By Gronwall's lemma, we have $\sup_n \sup_{t\in[0,T]} \|u_t^n\|_2^2 \le \text{const.}$ We obtain $\lim_{n\to\infty} \|u_t^n - v_t\|_2^2 = 0$ and

$$\lim_{n \to \infty} \left| \int_t^T \int (f_s^n u_s^n - f_s v_s) dm ds \right| = 0.$$

By [MR92, Lemma 2.12] we obtain

$$\int_t^T \mathcal{E}_{c_2+1}^{a,\hat{b}}(v_s) ds \le \int_t^T \liminf_{n \to \infty} \mathcal{E}_{c_2+1}^{a,\hat{b}}(u_s^n) ds \le \liminf_{n \to \infty} \int_t^T \mathcal{E}_{c_2+1}^{a,\hat{b}}(u_s^n) ds.$$

Finally, for $t \in [0, T]$ we get

$$\begin{aligned} \|v_t\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(v_s)ds &\leq \lim_{n \to \infty} \|u_t^n\|_2^2 + \liminf_{n \to \infty} \int_t^T \mathcal{E}^{a,\hat{b}}(u_s^n)ds \\ &\leq \lim_{n \to \infty} (2\int_t^T (f_s^n, u_s^n)ds + \|\phi\|_2^2) + \lim_{n \to \infty} 2\alpha \int_t^T \|u_s^n\|_2^2 ds \\ &= 2\int_t^T (f_s, v_s)ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|v_s\|_2^2 ds. \end{aligned}$$

Since the right hand side of this inequality is finite and $t \mapsto v_t$ is L^2 -continuous, it follows that $v \in \hat{F}$.

Now we show that v satisfies the weak relation (2.1.9) for the data (ϕ, f) . As $\varphi^n(t) := \|u_t^n - v_t\|_2$ is continuous and decreasing, we conclude by Dini's theorem

$$\lim_{n \to \infty} \sup_{t \in [0,T]} \|u_t^n - v_t\|_2 = 0,$$

and therefore

$$\lim_{n \to \infty} \int_0^T \|u_t^n - v_t\|_2^2 = 0.$$

Furthermore, there exists $K \in \mathbb{R}_+$ and a subsequence $(n_k)_{k \in \mathbb{N}}$ such that

$$\left|\int_{0}^{T} \mathcal{E}_{c_{2}+1}^{a,\hat{b}}(u_{s}^{n_{k}})ds\right| \leq K \qquad \forall k \in \mathbb{N}.$$

in particular

$$\int_0^T \int |D_{\sigma} u_s^{n_k}|^2 dm ds \le \frac{K}{c_1} \qquad \forall k \in \mathbb{N}.$$

We obtain

$$\lim_{k \to \infty} \int_0^T \mathcal{E}^{a,\hat{b}}(u_s^{n_k},\varphi_s) ds = \int_0^T \mathcal{E}^{a,\hat{b}}(v_s,\varphi_s) ds,$$

and

$$\lim_{k \to \infty} \int_0^T \int \langle b\sigma, D_\sigma u_s^{n_k} \rangle \varphi_s dm ds = \int_0^T \int \langle b\sigma, D_\sigma v_s \rangle \varphi_s dm ds,$$

which implies (2.1.9) for v associated to (ϕ, f) . Clearly u - v satisfies (2.1.9) with data (0,0) for $\varphi \in b\mathcal{C}_T$. By Proposition 2.1.9 we have u - v = 0. Since

$$v_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds,$$

the assertion follows.

Corollary 2.1.13 Let $u \in b\hat{F}$ and $f \in L^1(dt \times dm)$ satisfy the weak relation (2.1.9)

with test functions in bC_T and some function $\phi \in L^2 \cap L^\infty$. Let $g \in L^1(dt \times dm)$ be a bounded function such that $f \leq g$. Then u is represented by the following relation:

$$u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds.$$

Proof Define $f^n := (f \lor (-n)) \land g, n \in \mathbb{N}$. Then $(f^n)_{n \in \mathbb{N}}$ is a sequence of bounded functions such that $f^n \downarrow f$ and $f^n \leq g$ then by the same arguments as the proof of Proposition 2.1.12, the assertion follows.

The following proposition is a modification of [BPS05, Proposition 2.10]. It is essential for the analytic treatment of the non-linear equation (1.1) which is done in the next section.

Proposition 2.1.14 Let $u = (u^1, ..., u^l)$ be a vector valued function where each component is a generalized solution of the linear equation (2.1.5) associated to certain data f^i, ϕ^i , which are bounded and satisfy the conditions in Proposition 2.1.6 (ii) for i = 1, ..., l. Denote by ϕ, f the vectors $\phi = (\phi^1, ..., \phi^l), f = (f^1, ..., f^l)$ and by $D_{\sigma}u$ the matrix whose rows consist of the row vectors $D_{\sigma}u^i$. Then the following relations hold *m*-almost everywhere

$$(2.1.16) \quad |u_t|^2 + 2\int_t^T P_{s-t}(|D_{\sigma}u_s|^2 + \frac{1}{2}c|u_s|^2)ds = P_{T-t}|\phi|^2 + 2\int_t^T P_{s-t}\langle u_s, f_s\rangle ds.$$

$$(2.1.17) |u_t| \le P_{T-t}|\phi| + \int_t^T P_{s-t}\langle \hat{u}_s, f_s \rangle ds.$$

Here we write $\hat{x} = x/|x|$, for $x \in \mathbb{R}^l$, $x \neq 0$ and $\hat{x} = 0$, if x = 0.

Proof By Proposition 2.1.6 (ii) we have $u \in bC_T$.

First we assume l = 1. If we can check that u^2 satisfies (2.1.9) with data $(2uf - 2|D_{\sigma}u|^2 - cu^2, \phi^2)$ for $\varphi \in bC_T$, then (2.1.16) will follow by Corollary 2.1.13. We have the following relations:

$$\int_{0}^{T} (u_{t}^{2}, \partial_{t}\varphi_{t})dt = 2 \int_{0}^{T} (u_{t}, \partial_{t}(u_{t}\varphi_{t}))dt + (u_{0}^{2}, \varphi_{0}) - (u_{T}^{2}, \varphi_{T}),$$
$$\mathcal{E}^{a,\hat{b}}(u_{t}^{2}, \varphi_{t}) = 2\mathcal{E}^{a,\hat{b}}(u_{t}, u_{t}\varphi_{t}) - (2|D_{\sigma}u_{t}|^{2} + cu_{t}^{2}, \varphi_{t}),$$

and

$$\int \langle b\sigma, D_{\sigma}(u_t^2) \rangle \varphi_t dm = 2 \int \langle b\sigma, D_{\sigma}u_t \rangle u_t \varphi_t dm$$

2.1. Preliminaries

For the second relation we use (2.1.2). Since u is a generalized solution of (2.1.5), we obtain

$$\int_0^T (u_t, \partial_t (u_t \varphi_t)) dt - (u_T, u_T \varphi_T) + (u_0, u_0 \varphi_0) - \int_0^T (f_t, u_t \varphi_t) dt$$
$$= -\int_0^T \mathcal{E}^{a, \hat{b}}(u_t, u_t \varphi_t) dt - \int_0^T \int \langle b\sigma, D_\sigma u_t \rangle u_t \varphi_t dm dt.$$

By the above relations we have

$$\begin{aligned} &(2.1.18)\\ &\int_0^T (u_t^2, \partial_t \varphi_t) dt + (u_0^2, \varphi_0) - (u_T^2, \varphi_T) + \int_0^T (\mathcal{E}^{a, \hat{b}}(u_t^2, \varphi_t) + \int \langle b\sigma, D_\sigma(u_t^2) \rangle \varphi_t dm) dt \\ &= 2 \int_0^T (f_t u_t, \varphi_t) dt - \int_0^T (2|D_\sigma u_t|^2 + c|u_t|^2, \varphi_t) dt. \end{aligned}$$

Hence, by Corollary 2.1.13 (2.1.16) holds in the case l = 1. To deduce this relation in the case l > 1, it suffices to add the relations corresponding to the components $|u_t^i|^2, i = 1, ..., l$. For (2.1.17), define for $\varepsilon > 0$, $h_{\varepsilon}(t) := \sqrt{t + \varepsilon} - \sqrt{\varepsilon}$ for $t \ge 0$. Then by integration by parts we have

$$\mathcal{E}^{a,\hat{b}}(h_{\varepsilon}(|u|^{2}),\varphi) = \mathcal{E}^{a,\hat{b}}(|u|^{2},h_{\varepsilon}'(|u|^{2})\varphi) - (h_{\varepsilon}''(|u|^{2})|D_{\sigma}(|u|^{2})|^{2},\varphi) + (c(h_{\varepsilon}(|u|^{2}) - |u|^{2}h_{\varepsilon}'(|u|^{2})),\varphi),$$

and

$$\int_0^T (h_{\varepsilon}(|u_t|^2), \partial_t \varphi_t) dt = \int_0^T (|u_t|^2, \partial_t(\varphi_t h_{\varepsilon}'(|u_t|^2))) dt - (|u_T|^2, \varphi_T h_{\varepsilon}'(|u_T|^2)) + (|u_0|^2, \varphi_0 h_{\varepsilon}'(|u_0|^2)) + (h_{\varepsilon}(|u_T|^2), \varphi_T) - (h_{\varepsilon}(|u_0|^2), \varphi_0).$$

If we choose $\varphi h'_{\varepsilon}(|u|^2)$ as test function in (2.1.18), we obtain

$$\begin{split} &\int_{0}^{T} (|u_{t}|^{2}, \partial_{t}(\varphi_{t}h_{\varepsilon}'(|u_{t}|^{2})))dt + (|u_{0}|^{2}, \varphi_{0}h_{\varepsilon}'(|u_{0}|^{2})) - (|u_{T}|^{2}, \varphi_{T}h_{\varepsilon}'(|u_{T}|^{2})) \\ &+ \int_{0}^{T} (\mathcal{E}^{a,\hat{b}}(|u_{t}|^{2}, \varphi_{t}h_{\varepsilon}'(|u_{t}|^{2})) + \int \langle b\sigma, D_{\sigma}(|u_{t}|^{2}) \rangle \varphi_{t}h_{\varepsilon}'(|u_{t}|^{2})dm)dt \\ &= 2\int_{0}^{T} (\langle f_{t}, u_{t} \rangle, \varphi_{t}h_{\varepsilon}'(|u_{t}|^{2}))dt - \int_{0}^{T} (2|D_{\sigma}u_{t}|^{2} + c|u_{t}|^{2}, \varphi_{t}h_{\varepsilon}'(|u_{t}|^{2}))dt. \end{split}$$

By the above relations we have

$$\begin{split} &\int_{0}^{T} (h_{\varepsilon}(|u_{t}|^{2}), \partial_{t}\varphi_{t})dt - (h_{\varepsilon}(|u_{T}|^{2}), \varphi_{T}) + (h_{\varepsilon}(|u_{0}|^{2}), \varphi_{0}) \\ &+ \int_{0}^{T} (\mathcal{E}^{a,\hat{b}}(h_{\varepsilon}(|u_{t}|^{2}), \varphi_{t}) + \int \langle b\sigma, D_{\sigma}(h_{\varepsilon}(|u_{t}|^{2})) \rangle \varphi_{t}dm)dt \\ &= \int_{0}^{T} - (h_{\varepsilon}''(|u_{t}|^{2})|D_{\sigma}(|u_{t}|^{2})|^{2}, \varphi_{t}) + (c(h_{\varepsilon}(|u|^{2}) - |u|^{2}h_{\varepsilon}'(|u|^{2})), \varphi)dt \\ &+ 2\int_{0}^{T} (\langle f_{t}, u_{t} \rangle h_{\varepsilon}'(|u_{t}|^{2}), \varphi_{t})dt - \int_{0}^{T} (h_{\varepsilon}'(|u_{t}|^{2})(2|D_{\sigma}u_{t}|^{2} + c|u_{t}|^{2}, \varphi_{t})dt. \end{split}$$

As

$$|D_{\sigma}(|u|^2)|^2 = 4\langle u, D_{\sigma}u(D_{\sigma}u)^*u \rangle_{2}$$

we deduce

$$\begin{split} & 2\langle f, u \rangle h_{\varepsilon}'(|u|^{2}) - 2h_{\varepsilon}'(|u|^{2})|D_{\sigma}u|^{2} - h_{\varepsilon}''(|u|^{2})|D_{\sigma}(|u|^{2})|^{2} \\ &= \frac{\langle f, u \rangle - |D_{\sigma}u|^{2}}{(|u|^{2} + \varepsilon)^{\frac{1}{2}}} + \frac{|u|^{2}\langle \hat{u}, D_{\sigma}u(D_{\sigma}u)^{*}\hat{u} \rangle}{(|u|^{2} + \varepsilon)^{\frac{3}{2}}} \\ &= \frac{\langle f, u \rangle}{(|u|^{2} + \varepsilon)^{\frac{1}{2}}} - \frac{\varepsilon |D_{\sigma}u|^{2} + |u|^{2}(|D_{\sigma}u|^{2} - \langle \hat{u}, D_{\sigma}u(D_{\sigma}u)^{*}\hat{u} \rangle)}{(|u|^{2} + \varepsilon)^{\frac{3}{2}}} \\ &\leq \frac{\langle f, u \rangle}{(|u|^{2} + \varepsilon)^{\frac{1}{2}}}. \end{split}$$

By Proposition 2.1.14 and since $c(h_{\varepsilon}(|u_s|^2) - 2|u_s|^2 h'_{\varepsilon}(|u_s|^2)) \leq 0$, we deduce

$$h_{\varepsilon}(|u_t|^2) \le P_{T-t}h_{\varepsilon}(|\phi|^2) + \int_t^T P_{s-t}\frac{\langle f_s, u_s \rangle}{(|u_s|^2 + \varepsilon)^{\frac{1}{2}}} ds.$$

Letting $\varepsilon \to 0$ the results follow.

The next corollary is a version of the above proposition for general data, where we use P_t is a C_0 -semigroup on L^1 .

Corollary 2.1.15 Let $u = (u^1, ..., u^l)$ be a vector-valued function, where each component is a generalized solution of the linear equation (2.1.5) associated to certain data $f^i \in L^1([0,T]; L^2), \phi^i \in L^2$ for i = 1, ..., l. Denote by ϕ, f the vectors $\phi = (\phi^1, ..., \phi^l), f = (f^1, ..., f^l)$ and by $D_{\sigma}u$ the matrix whose rows consist of the row vectors $D_{\sigma}u^i$. Then the following relations hold *m*-almost everywhere

$$(2.1.19) \quad |u_t|^2 + 2\int_t^T P_{s-t}(|D_{\sigma}u|^2 + \frac{1}{2}c|u_s|^2)ds = P_{T-t}|\phi|^2 + 2\int_t^T P_{s-t}\langle u_s, f_s\rangle ds.$$

$$(2.1.20) |u_t| \le P_{T-t} |\phi| + \int_t^T P_{s-t} \langle \hat{u}_s, f_s \rangle ds.$$

Proof Analogously to the proof of Proposition 2.1.14 it is enough to verify (2.1.19) for l = 1. For $\phi \in L^2$, $f \in L^1([0,T], L^2)$, take ϕ_n , f_n as in the proof of Proposition 2.1.9, then we have

(a). $u_{n,t} := P_{T-t}\phi_n + \int_t^T P_{s-t}f_{n,s}ds$ is a generalized solution,

(b).
$$\lim_{n \to \infty} \int_t^T ||f_{n,s} - f_s||_2 ds = 0$$
,

(c).
$$\lim_{n \to \infty} \|\phi_n - \phi\|_2 = 0$$
,

(d).
$$\lim_{n \to \infty} ||u_n - u||_T = 0.$$

By Proposition 2.1.14 we have

$$(2.1.21) |u_{n,t}|^2 + 2\int_t^T P_{s-t}(|D_{\sigma}u_{n,s}|^2 + \frac{1}{2}c|u_{n,s}|^2)ds = P_{T-t}|\phi_n|^2 + 2\int_t^T P_{s-t}\langle u_{n,s}, f_{n,s}\rangle ds.$$

By (b) and (d) we obtain

$$\begin{split} &\|\int_{t}^{T} P_{s-t}((u_{n,s}, f_{n,s}) - (u_{s}, f_{s}))ds\|_{1} \\ \leq &C \int_{t}^{T} (\|u_{n,s}\|_{2} \|f_{n,s} - f_{s}\|_{2} + \|f_{s}\|_{2} \|u_{n,s} - u_{s}\|_{2})ds \\ \leq &C(\sup_{s \in [0,T]} \|u_{n,s}\|_{2} \int_{t}^{T} \|f_{n,s} - f_{s}\|_{2}ds + \sup_{s \in [0,T]} \|u_{n,s} - u_{s}\| \int_{t}^{T} \|f_{s}\|_{2}ds) \\ \rightarrow &0, \text{ as } n \to \infty. \end{split}$$

Here we used that P_t is a C_0 -semigroup on $L^1(\mathbb{R}^d; m)$. By (d) we conclude that

$$\begin{split} &\int_{t}^{T} \||D_{\sigma}u_{n,s}|^{2} - |D_{\sigma}u_{s}|^{2}\|_{1}ds \\ \leq &((\int_{t}^{T} \|D_{\sigma}u_{n,s}\|_{2}^{2}ds)^{\frac{1}{2}} + (\int_{t}^{T} \|D_{\sigma}u_{s}\|_{2}^{2}ds)^{\frac{1}{2}})(\int_{t}^{T} \|D_{\sigma}u_{n,s} - D_{\sigma}u_{s}\|_{2}^{2}ds)^{\frac{1}{2}} \\ = &((\int_{t}^{T} \mathcal{E}^{a}(u_{n,s})ds)^{\frac{1}{2}} + (\int_{t}^{T} \mathcal{E}^{a}(u_{s})ds)^{\frac{1}{2}})(\int_{t}^{T} \mathcal{E}^{a}(u_{n,s} - u_{s})ds)^{\frac{1}{2}} \\ \to 0, \text{ as } n \to \infty, \end{split}$$

and that

$$\begin{split} &\int_{t}^{T} \||cu_{n,s}|^{2} - |cu_{s}|^{2}\|_{1}ds \\ \leq &((\int_{t}^{T} \|c^{1/2}u_{n,s}\|_{2}^{2}ds)^{\frac{1}{2}} + (\int_{t}^{T} \|c^{1/2}u_{s}\|_{2}^{2}ds)^{\frac{1}{2}})(\int_{t}^{T} \|c^{1/2}u_{n,s} - c^{1/2}u_{s}\|_{2}^{2}ds)^{\frac{1}{2}} \\ \leq &M((\int_{t}^{T} \mathcal{E}_{c_{2}+1}^{a,\hat{b}}(u_{n,s})ds)^{\frac{1}{2}} + (\int_{t}^{T} \mathcal{E}_{c_{2}+1}^{a,\hat{b}}(u_{s})ds)^{\frac{1}{2}})(\int_{t}^{T} \mathcal{E}_{c_{2}+1}^{a,\hat{b}}(u_{n,s} - u_{s})ds)^{\frac{1}{2}} \\ \rightarrow 0, \text{ as } n \rightarrow \infty, \end{split}$$

where we used (2.1.1) and (2.1.2) in the second inequality. Thus, we obtain

$$\lim_{n \to \infty} \int_t^T P_{s-t}(|D_\sigma u_{n,s}|^2) ds = \int_t^T P_{s-t}|D_\sigma u_s|^2 ds,$$

and

$$\lim_{n \to \infty} \int_t^T P_{s-t}(|cu_{n,s}|^2) ds = \int_t^T P_{s-t} |cu_s|^2 ds.$$

Passing to the limit $n \to \infty$ in equation (2.1.21) (2.1.19) follows. (2.1.20) also follows by using the same method.

Lemma 2.1.16 If $f, g \in L^1([0,T]; L^2)$ and $\phi \in L^2$, then:

(2.1.22)
$$\int_{t}^{T} P_{s-t}(f_{s}P_{T-s}\phi)ds \leq \frac{1}{2}P_{T-t}\phi^{2} + \int_{t}^{T} \int_{s}^{T} P_{s-t}(f_{s}P_{r-s}f_{r})drds, \ m-a.e.$$

Proof Define

$$h_t := P_{T-t}\phi, \qquad v_t := \int_t^T P_{s-t}f_s ds.$$

By (2.1.19) we deduce

$$h_t^2 + 2\int_t^T P_{s-t}(|D_{\sigma}h_s|^2 + \frac{1}{2}c|h_s|^2)ds = P_{T-t}\phi^2,$$
$$v_t^2 + 2\int_t^T P_{s-t}(|D_{\sigma}v_s|^2 + \frac{1}{2}c|v_s|^2)ds = 2\int_t^T P_{s-t}(f_s\int_s^T P_{r-s}f_rdr)ds,$$

and

$$h_t v_t + 2\int_t^T P_{s-t}(\langle D_\sigma h_s, D_\sigma v_s \rangle + \frac{1}{2}ch_s v_s)ds = \int_t^T P_{s-t}(f_s P_{T-s}\phi)ds$$

So, we obtain

$$\begin{split} \int_{t}^{T} P_{s-t}(f_{s}P_{T-s}\phi)ds =& h_{t}v_{t} + 2\int_{t}^{T} P_{s-t}(\langle D_{\sigma}h_{s}, D_{\sigma}v_{s}\rangle + \frac{1}{2}ch_{s}v_{s})ds \\ \leq & \frac{1}{2}(h_{t}^{2} + v_{t}^{2}) + \int_{t}^{T} P_{s-t}(|D_{\sigma}h_{s}|^{2} + |D_{\sigma}v_{s}|^{2} + \frac{1}{2}c|v_{s}|^{2} + \frac{1}{2}c|h_{s}|^{2})ds \\ = & \frac{1}{2}P_{T-t}\phi^{2} + \int_{t}^{T} \int_{s}^{T} P_{s-t}(f_{s}P_{r-s}f_{r})drds. \end{split}$$

2.2 The Non-linear Equation

In this section, we solve the non-linear equation (2.2.1). In the case of non-linear equations, we are going to consider systems of equations, with the unknown functions and their first-order derivatives mixed in the non-linear term of the equation. The non-linear term is a given measurable function $f : [0,T] \times \mathbb{R}^d \times \mathbb{R}^l \times \mathbb{R}^l \otimes \mathbb{R}^k \to \mathbb{R}^l$, $l \in \mathbb{N}$. We are going to treat the following system of equations.

(2.2.1)
$$(\partial_t + L)u + f(\cdot, \cdot, u, D_\sigma u) = 0 \qquad u_T = \phi.$$

Here $\phi \in L^2(\mathbb{R}^d, dm; \mathbb{R}^l)$.

Definition 2.2.1 (Generalized solutions of the nonlinear equation) A generalized solution of equation (2.2.1) is a system $u = (u^1, u^2, ..., u^l)$ of l elements in \hat{F} with the property that $f^i(\cdot, \cdot, u, D_{\sigma}u)$ belongs to $L^1([0, T]; L^2)$ and there are sequences $\{u_n\}$ which are strong solutions of (2.2.1) with data (ϕ_n, f_n) such that

$$||u_n - u||_T \to 0, ||\phi_n - \phi||_2 \to 0, \text{ and } \lim_{n \to \infty} f_n(\cdot, \cdot, u_n, D_\sigma u_n) = f(\cdot, \cdot, u, D_\sigma u) \text{ in } L^1([0, T]; L^2).$$

Definition 2.2.2 (*Mild equation*) A mild solution of equation (2.2.1) is a system $u = (u^1, u^2, ..., u^l)$ of l elements in \hat{F} , which has the property that each function $f^i(\cdot, \cdot, u, D_{\sigma}u)$ belongs to $L^1([0, T]; L^2(m))$ and such that for every $i \in \{1, ..., l\}$,

(2.2.2)
$$u^{i}(t,x) = P_{T-t}\phi^{i}(x) + \int_{t}^{T} P_{s-t}f^{i}(s,\cdot,u_{s},D_{\sigma}u_{s})(x)ds, m-a.e.$$

Lemma 2.2.3 u is a generalized solution of the nonlinear equation (2.2.1) if and only if it solves the mild equation (2.2.2).

Proof The assertion follows by Proposition 2.1.9.

We will use the following notations:

$$\begin{split} \|\phi\|_{2}^{2} &= \sum_{i=1}^{l} \|\phi^{i}\|_{2}^{2}, \phi \in L^{2}(\mathbb{R}^{d}; \mathbb{R}^{l}), \\ \mathcal{E}(u, v) &= \sum_{i=1}^{l} \mathcal{E}(u^{i}, v^{i}), \mathcal{E}^{a}(u, v) = \sum_{i=1}^{l} \mathcal{E}^{a}(u^{i}, v^{i}), u, v \in F^{l}, \\ &\qquad \mathcal{E}^{a, \hat{b}}(u, v) = \sum_{i=1}^{l} \mathcal{E}^{a, \hat{b}}(u^{i}, v^{i}), u, v \in F^{l}, \\ &\qquad \|u\|_{T}^{2} := \sup_{t \leq T} \|u_{t}\|_{2}^{2} + \int_{0}^{T} \mathcal{E}^{a, \hat{b}}_{c_{2}+1}(u_{t}) dt, u \in \hat{F}^{l}. \end{split}$$

2.2.1 The Case of Lipschitz Conditions

In this subsection we consider a measurable function $f: [0,T] \times \mathbb{R}^d \times \mathbb{R}^l \otimes \mathbb{R}^k \to \mathbb{R}^l$ such that

(2.2.3)
$$|f(t, x, y, z) - f(t, x, y', z')| \le C(|y - y'| + |z - z'|),$$

with t, x, y, y', z, z' arbitrary and C is a constant independent of t, x. Set $f^0(t, x) := f(t, x, 0, 0)$.

Proposition 2.2.4 Suppose that the conditions (A1)-(A4) hold and that f satisfies condition (2.2.3), $f^0 \in L^2([0,T] \times \mathbb{R}^d, dt \times dm; \mathbb{R}^l)$ and $\phi \in L^2(\mathbb{R}^d; \mathbb{R}^l)$. Then the equation (2.2.1) admits a unique generalized solution $u \in \hat{F}^l$ and it satisfies the following estimate

$$\|u\|_T^2 \le e^{T(1+2C+\frac{C^2}{c_1}+2\alpha+c_2)} (\|\phi\|_2^2 + \|f^0\|_{L^2([0,T]\times\mathbb{R}^d)}^2).$$

Proof If $u \in \hat{F}^l$, then by relation (2.2.3) we have

$$|f(\cdot, \cdot, u, D_{\sigma}u)| \le |f(\cdot, \cdot, u, D_{\sigma}u) - f(\cdot, \cdot, 0, 0)| + |f(\cdot, \cdot, 0, 0)| \le C(|u| + |D_{\sigma}u|) + |f^0|.$$

As $f^0 \in L^2([0,T] \times \mathbb{R}^d, dt \times dm; \mathbb{R}^l)$ and $|D_{\sigma}u|$ is an element of $L^2([0,T] \times \mathbb{R}^d)$, we get $f(\cdot, \cdot, u, D_{\sigma}u) \in L^2([0,T] \times \mathbb{R}^d; \mathbb{R}^l)$.

Now we define the operator $A: \hat{F}^l \to \hat{F}^l$ by

$$(Au)_{t}^{i} := P_{T-t}\phi^{i} + \int_{t}^{T} P_{s-t}f^{i}(s, \cdot, u_{s}, D_{\sigma}u_{s})ds, \qquad i = 1, ..., l.$$

Then Proposition 2.1.9 implies that $Au \in \hat{F}^l$. In the following we write $f_{u,s}^i := f^i(s, \cdot, u_s, D_\sigma u_s)$. Since $(Au)_t^i - (Av)_t^i = \int_t^T P_{s-t}(f_{u,s}^i - f_{v,s}^i)ds$ is the mild solution with data $(f_u^i - f_v^i, 0)$, by the same arguments as in the proof of Proposition 2.1.9 we have

$$\begin{split} \|\int_{t}^{T} P_{s-t}(f_{u,s}^{i} - f_{v,s}^{i}) ds\|_{[t,T]}^{2} &\leq M_{T}(\int_{t}^{T} \|f_{u,s} - f_{v,s}\|_{2} ds)^{2} \\ &\leq M_{T}(T-t) \int_{t}^{T} \|f_{u,s} - f_{v,s}\|_{2}^{2} ds \\ &\leq M_{T}(T-t) \int_{t}^{T} (\|u_{s} - v_{s}\|_{2}^{2} + \|D_{\sigma}u_{s} - D_{\sigma}v_{s}\|_{2}^{2}) ds \\ &\leq M_{T}(T-t) \|u-v\|_{[t,T]}^{2}, \end{split}$$

where M_T can change from line to line. Here $||u||_{[T_a,T_b]} := (\sup_{t \in [T_a,T_b]} ||u_t||_2^2 + \int_{T_a}^{T_b} \mathcal{E}_{c_2+1}^{a,\hat{b}}(u_t) dt)^{\frac{1}{2}}$, where $0 \leq T_a \leq T_b \leq T$. Fix T_1 sufficiently small such that $\gamma := M_T(T - T_1) < 1$. Then we have :

$$||Au - Av||_{[T_1,T]}^2 \le \gamma ||u - v||_{[T_1,T]}^2.$$

Then there exists a unique $u_1 \in \hat{F}_{[T_1,T]}$ such that $Au_1 = u_1$ where $\hat{F}_{[T_a,T_b]} := C([T_a, T_b]; L^2) \cap L^2((T_a, T_b); F)$ for $T_a \in [0, T]$ and $T_b \in [T_a, T]$.

We define the operator $A^1: \hat{F}^l \to \hat{F}^l$ by

$$(A^{1}u)_{t}^{i} := P_{T_{1}-t}u_{1,T_{1}}^{i} + \int_{t}^{T_{1}} P_{s-t}f^{i}(s,\cdot,u_{s},D_{\sigma}u_{s})ds, \qquad i = 1,...,l.$$

Then by the same method as above, we get

$$||A^{1}u - A^{1}v||_{[t,T_{1}]}^{2} \le M_{T}(T_{1} - t)||u - v||_{[t,T_{1}]}^{2}$$

Now we choose $T_2 < T_1$ such that $M_T(T_1 - t) < 1$. We obtain that there exists a unique $u_2 \in \hat{F}_{[T_2,T_1]}$ such that $A^1u_2 = u_2$. If we define $u := u_1 \mathbf{1}_{[T_1,T]} + u_2 \mathbf{1}_{[T_2,T_1]}$, then for $T_2 \leq t \leq T_1$

$$P_{T-t}\phi^i + \int_t^T P_{s-t}f^i(s,\cdot,u_s,D_\sigma u_s)ds$$

$$= P_{T-t}\phi^{i} + \int_{t}^{T_{1}} P_{s-t}f^{i}(s, \cdot, u_{s}, D_{\sigma}u_{s})ds + \int_{T_{1}}^{T} P_{s-t}f^{i}(s, \cdot, u_{1,s}, D_{\sigma}u_{1,s})ds$$

$$= P_{T-t}\phi^{i} + \int_{t}^{T_{1}} P_{s-t}f^{i}(s, \cdot, u_{s}, D_{\sigma}u_{s})ds + P_{T_{1}-t}(u_{1,T_{1}}^{i} - P_{T-T_{1}}\phi^{i})$$

$$= P_{T-t}\phi^{i} + \int_{t}^{T_{1}} P_{s-t}f^{i}(s, \cdot, u_{s}, D_{\sigma}u_{s})ds + P_{T_{1}-t}u_{1,T_{1}}^{i} - P_{T-t}\phi^{i}$$

$$= u_{2,t}^{i}.$$

If $t > T_1$,

$$(Au)_{t}^{i} = P_{T-t}\phi^{i} + \int_{t}^{T} P_{s-t}f^{i}(s, \cdot, u_{1,s}, D_{\sigma}u_{1,s})ds$$
$$= u_{1,t}^{i},$$

Therefore, we can construct a solution over the interval $[T_2, T]$. Clearly there exists $n \in \mathbb{N}$ such that $T < n(T - T_1)$. Hence, the construction is done after n steps.

In order to obtain the estimate in the statement, we write

$$\begin{split} &|\int_{t}^{T} (f_{u,s}, u_{s})ds| \\ &\leq \int_{t}^{T} \|f_{s}^{0}\|_{2} \|u_{s}\|_{2}ds + C \int_{t}^{T} \|u_{s}\|_{2}^{2}ds + C \int_{t}^{T} \|D_{\sigma}u_{s}\|_{2} \|u_{s}\|_{2}ds \\ &\leq \frac{1}{2} \int_{t}^{T} \|f_{s}^{0}\|_{2}^{2}ds + (\frac{1}{2} + C + \frac{1}{2c_{1}}C^{2}) \int_{t}^{T} \|u_{s}\|_{2}^{2}ds + \frac{c_{1}}{2} \int_{t}^{T} \mathcal{E}^{a}(u_{s})ds. \end{split}$$

By relation (2.1.7) of Proposition 2.1.9 it follows that

$$\begin{split} \|u_t\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(u_s)ds &\leq 2\int_t^T (f_{u,s}, u_s)ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|u_s\|_2^2 ds \\ &\leq \|\phi\|_2^2 + \int_t^T \|f_s^0\|_2^2 ds + (1 + 2C + \frac{C^2}{c_1} + 2\alpha + c_2) \int_t^T \|u_s\|_2^2 ds \\ &+ \int_t^T \mathcal{E}^{a,\hat{b}}(u_s)ds. \end{split}$$

Now by Gronwall's lemma the desired estimate follows.

[Uniqueness] Let u_1 and u_2 be two solutions of equation (2.2.1). By using (2.1.7) for the difference $u_1 - u_2$ we get

$$\begin{aligned} \|u_{1,t} - u_{2,t}\|_{2}^{2} + 2\int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{1,s} - u_{2,s})ds \\ \leq & 2\int_{t}^{T} (f(s, \cdot, u_{1,s}, D_{\sigma}u_{1,s}) - f(s, \cdot, u_{2,s}, D_{\sigma}u_{2,s}), u_{1,s} - u_{2,s})ds + 2\alpha \int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds \end{aligned}$$

$$\leq 2 \int_{t}^{T} C(|D_{\sigma}u_{1,s} - D_{\sigma}u_{2,s}|, |u_{1,s} - u_{2,s}|) ds + (2\alpha + C) \int_{t}^{T} ||u_{1,s} - u_{2,s}||_{2}^{2} ds$$
$$\leq (\frac{C^{2}}{c_{1}} + c_{2} + 2\alpha + C) \int_{t}^{T} ||u_{1,s} - u_{2,s}||_{2}^{2} ds + \int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{1,s} - u_{2,s}) ds.$$

By Gronwall's lemma it follows that

$$||u_{1,t} - u_{2,t}||_2^2 = 0,$$

hence $u_1 = u_2$.

2.2.2 The Case of Monotonicity Conditions

Let $f: [0,T] \times \mathbb{R}^d \times \mathbb{R}^l \times \mathbb{R}^l \otimes \mathbb{R}^k \to \mathbb{R}^l$ be a measurable function and $\phi \in L^2(\mathbb{R}^d, m; \mathbb{R}^l)$ be the final condition of (2.2.1). We impose the following conditions:

(H1) (Lipschitz condition in z) There exists a fixed constant C > 0 such that for t, x, y, z, z' arbitrary

$$|f(t, x, y, z) - f(t, x, y, z')| \le C|z - z'|.$$

(H2) (Monotonicity condition in y) For x, y, y', z arbitrary, there exists a function $\mu_t \in L^1([0,T];\mathbb{R})$ such that

$$\langle y - y', f(t, x, y, z) - f(t, x, y', z) \rangle \le \mu_t |y - y'|^2.$$

We set $\alpha_t := \int_0^t \mu_s ds$.

(H3) (Continuity condition in y) For t, x and z fixed, the map

$$y \mapsto f(t, x, y, z)$$

is continuous.

We need the following notations:

$$f^{0}(t,x) := f(t,x,0,0), \qquad f'(t,x,y) := f(t,x,y,0) - f(t,x,0,0),$$
$$f^{',r}(t,x) := \sup_{|y| \le r} |f'(t,x,y)|.$$

(H4) For each $r > 0, f'^{,r} \in L^1([0,T]; L^2)$.

(H5) $\|\phi\|_{\infty} < \infty, \|f^0\|_{\infty} < \infty, |\phi| \in L^2, |f^0| \in L^2([0,T];L^2).$

If $m(\mathbb{R}^d) < \infty$ the last two conditions in (H5) are ensured by the boundedness of ϕ and f^0 . The conditions (H1), (H4), and (H5) imply that if $u \in b\hat{F}$, then

 $|f(u, D_{\sigma}u)| \in L^1([0, T]; L^2)$. It seems impossible to apply general monotonicity methods to the map $\mathcal{V} \ni u \mapsto f(t, \cdot, u(\cdot), D_{\sigma}u) \in \mathcal{V}'$ because of lack of a suitable reflexive Banach space \mathcal{V} such that $\mathcal{V} \subset \mathcal{H} \subset \mathcal{V}'$. Therefore, also here we proceed developing a hands-on approach to prove existence and uniqueness of solutions for equation (2.2.1) as done in [BPS05].

Lemma 2.2.5 In (H2) without loss of generality we can assume that $\mu_t \equiv 0$.

Proof Let us make the change $u_t^* = \exp(\alpha_t)u_t$ and

$$\phi^* = \exp(\alpha_T)\phi, \qquad f_t^*(y, z) = \exp(\alpha_t)f_t(\exp(-\alpha_t)y, \exp(-\alpha_t)z) - \mu_t y,$$

for the data. Next we will prove that u is a generalized solution associated to the data (ϕ, f) if and only if u^* is a solution associated to the data (ϕ^*, f^*) . Hence we can write

$$u_t^i = P_{T-t}\phi^i(x) + \int_t^T P_{s-t}f^i(s, \cdot, u_s, D_\sigma u_s)(x)ds,$$

equivalently as

$$\begin{split} u_t^{i,*} &= \exp(\alpha_t) P_{T-t} \phi^i(x) + \exp(\alpha_t) \int_t^T P_{s-t} f^i(s, \cdot, u_s, D_\sigma u_s)(x) ds \\ &= \exp(\alpha_T) P_{T-t} \phi^i(x) + (\exp(\alpha_t) - \exp(\alpha_T)) P_{T-t} \phi^i(x) \\ &+ \int_t^T (\exp(\alpha_t) - \exp(\alpha_s)) P_{s-t} f^i(s, \cdot, u_s, D_\sigma u_s)(x) ds \\ &+ \int_t^T \exp(\alpha_s) P_{s-t} f^i(s, \cdot, u_s, D_\sigma u_s)(x) ds \\ &= P_{T-t} \phi^{i,*}(x) + \int_t^T P_{s-t}(\exp(\alpha_s) f^i(s, \cdot, u_s, D_\sigma u_s)(x)) ds \\ &- \int_t^T \mu_s \exp(\alpha_s) P_{T-t} \phi^i(x) ds - \int_t^T \int_t^l \mu_s \exp(\alpha_s) P_{l-t} f^i(l, \cdot, u_l, D_\sigma u_l) ds dl \\ &= P_{T-t} \phi^{i,*}(x) + \int_t^T P_{s-t}(\exp(\alpha_s) f^i(s, \cdot, \exp(\alpha_s) u_s^*, \exp(\alpha_s) D_\sigma u_s^*)(x)) ds \\ &- \int_t^T P_{s-t}(\mu_s \exp(\alpha_s) P_{T-s} \phi^i(x) + \int_s^T \mu_s \exp(\alpha_s) P_{l-s} f^i(l, \cdot, u_l, D_\sigma u_l) dl ds \\ &= P_{T-t} \phi^{i,*}(x) + \int_t^T P_{s-t} f^{i,*}(s, \cdot, u_s^*, D_\sigma u_s^*)(x) ds. \end{split}$$

Next we prove f^* satisfies (H1)-(H5). It is obvious that (H1), (H3)-(H5) are satisfied.

Let us prove that f^* satisfies (H2) with $\mu_t \equiv 0.$ We have

$$\begin{aligned} \langle y - y', f^*(t, x, y, z) - f^*(t, x, y', z) \rangle \\ = \langle y - y', \mu_t y' - \mu_t y \rangle \\ + (\exp(\alpha_t))^2 \langle \exp(-\alpha_t) y - \exp(-\alpha_t) y', f(t, x, \exp(-\alpha_t) y, \exp(-\alpha_t) z) \rangle \\ - (\exp(\alpha_t))^2 \langle \exp(-\alpha_t) y - \exp(-\alpha_t) y', f(t, x, \exp(-\alpha_t) y', \exp(-\alpha_t) z) \rangle \\ \leq - |y - y'|^2 \mu_t + \mu_t (\exp(\alpha_t))^2 |\exp(-\alpha_t) y - \exp(-\alpha_t) y'|^2 \\ = 0. \end{aligned}$$

Thus, by making the transformation $f \to f^*$, we can assume that $\mu_t \equiv 0$.

Lemma 2.2.6 Suppose that conditions (A1)-(A4), (H1) and the following weaker form of condition (H2) (with $\mu_t \equiv 0$) hold:

(H2')
$$\langle y, f'(t, x, y) \rangle \leq 0$$
 for all t, x, y .

If u is a generalized solution of (2.2.1), then there exists a constant K depending on C, μ_t, T, α such that

(2.2.4)
$$\|u\|_T^2 \le K(\|\phi\|_2^2 + \int_0^T \|f_t^0\|_2^2 dt)$$

Proof Since u is a solution of (2.2.1), by Proposition 2.1.9 we have

$$\|u_t\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(u_s)ds \le 2\int_t^T (f_s, u_s)ds + \|u_T\|_2^2 + 2\alpha\int_t^T \|u_s\|_2^2ds.$$

Conditions (H1) and (H2') yield

$$\langle f_s(u_s, D_{\sigma}u_s), u_s \rangle = \langle f_s(u_s, D_{\sigma}u_s) - f_s(u_s, 0) + f'_s(u_s) + f^0_s, u_s \rangle \\ \leq |f_s(u_s, D_{\sigma}u_s) - f_s(u_s, 0)||u_s| + \langle f'_s(u_s), u_s \rangle + |f^0_s||u_s| \\ \leq (C|D_{\sigma}u_s| + |f^0_s|)|u_s|.$$

Hence, it follows that

$$\begin{split} \|u_t\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(u_s)ds &\leq 2\int_t^T \int (C|D_{\sigma}u_s| + |f_s^0|)|u_s|dmds + \|u_T\|_2^2 + 2\alpha \int_t^T \|u_s\|_2^2ds \\ &\leq \int_t^T \mathcal{E}^{a,\hat{b}}(u_s)ds + (\frac{C^2}{c_1} + 1 + 2\alpha + c_2)\int_t^T \|u_s\|_2^2ds + \int_t^T \|f_s^0\|_2^2ds \\ &+ \|u_T\|_2^2. \end{split}$$

Then Gronwall's lemma yields

$$\|u\|_T^2 \le K(\|\phi\|_2^2 + \int_0^T \|f_t^0\|_2^2 dt).$$

By a modification of the arguments in [BPS05, Lemma 3.3] we obtain the following estimates.

Lemma 2.2.7 Suppose that conditions (A1)-(A4), (H1) and (H2') hold. If u is a generalized solution of (2.2.1), there exists a constant K, which depends on C, μ and T, such that

(2.2.5)
$$||u||_{\infty} \le K(||\phi||_{\infty} + ||f^0||_{\infty}).$$

Proof By Corollary 2.1.15 we have

$$(2.2.6) \qquad |u_t|^2 + 2\int_t^T P_{s-t}(|D_{\sigma}u|^2) \le P_{T-t}|\phi|^2 + 2\int_t^T P_{s-t}\langle u_s, f_s(u_s, D_{\sigma}u_s)\rangle ds.$$

Following the same arguments as the proof of Lemma 2.2.6 we deduce

$$\langle f_s(u_s, D_\sigma u_s), u_s \rangle \le (C|D_\sigma u_s| + |f_s^0|)|u_s|.$$

By Corollary 2.1.15 (2.1.20) we obtain

$$|u_s| \le P_{T-s}|\phi| + \int_s^T P_{r-s}(C|D_{\sigma}u_r| + |f_r^0|)dr.$$

Then we have

$$\int_{t}^{T} P_{s-t} \langle f_{s}(u_{s}, D_{\sigma}u_{s}), u_{s} \rangle ds$$

$$\leq \int_{t}^{T} P_{s-t}[(P_{T-s}|\phi| + \int_{s}^{T} P_{r-s}(C|D_{\sigma}u_{r}| + |f_{r}^{0}|)dr)(C|D_{\sigma}u_{s}| + |f_{s}^{0}|)]ds$$

So, by (2.2.6) and Lemma 2.1.16 we obtain

$$|u_t|^2 + 2\int_t^T P_{s-t}(|D_{\sigma}u_s|^2)ds$$

$$\leq P_{T-t}|\phi|^2 + 2(\int_t^T P_{s-t}[(P_{T-s}|\phi| + \int_s^T P_{r-s}(C|D_{\sigma}u_r| + |f_r^0|)dr)(C|D_{\sigma}u_s| + |f_s^0|)]ds)$$

$$\leq 3P_{T-t}|\phi|^2 + 2C^2 \int_t^T \int_s^T P_{s-t}(|D_{\sigma}u_s|P_{r-s}|D_{\sigma}u_r|)drds + 2\int_t^T \int_s^T P_{s-t}(|f_s^0|P_{r-s}|f_r^0|)drds \\ + 2\int_t^T \int_s^T P_{s-t}[P_{r-s}(C|D_{\sigma}u_r| + |f_r^0|)(C|D_{\sigma}u_s| + |f_s^0|)]drds.$$

Since

$$\begin{split} &\int_{t}^{T} \int_{s}^{T} P_{s-t} [P_{r-s}(C|D_{\sigma}u_{r}| + |f_{r}^{0}|)(C|D_{\sigma}u_{s}| + |f_{s}^{0}|)] dr ds \\ \leq &\frac{1}{2} \int_{t}^{T} \int_{s}^{T} [P_{s-t}(C|D_{\sigma}u_{s}| + |f_{s}^{0}|)^{2}] + P_{s-t} [(P_{r-s}(C|D_{\sigma}u_{r}| + |f_{r}^{0}|))^{2}] dr ds \\ \leq &\int_{t}^{T} \int_{s}^{T} C^{2} P_{s-t} |D_{\sigma}u_{s}|^{2} + P_{s-t} |f_{s}^{0}|^{2} + \frac{1}{2} P_{r-t}(C|D_{\sigma}u_{r}| + |f_{r}^{0}|)^{2} dr ds \\ \leq &2 C^{2}(T-t) \int_{t}^{T} P_{s-t} |D_{\sigma}u_{s}|^{2} ds + 2(T-t) \int_{t}^{T} P_{s-t} |f_{s}^{0}|^{2} ds, \end{split}$$

and by Schwartz's inequality one has

$$\begin{split} &\int_{t}^{T} \int_{s}^{T} P_{s-t}(|D_{\sigma}u_{s}|P_{r-s}|D_{\sigma}u_{r}|) dr ds \\ &\leq \int_{t}^{T} \int_{s}^{T} \frac{1}{2} (P_{s-t}|D_{\sigma}u_{s}|^{2}) dr ds + \int_{t}^{T} \int_{s}^{T} \frac{1}{2} (P_{r-t}|D_{\sigma}u_{r}|^{2}) dr ds \\ &\leq (T-t) \int_{t}^{T} P_{s-t}|D_{\sigma}u_{s}|^{2} ds, \end{split}$$

we conclude

$$|u_t|^2 + 2\int_t^T P_{s-t}(|D_{\sigma}u_s|^2)ds$$

$$\leq 3P_{T-t}|\phi|^2 + 6C^2(T-t)\int_t^T P_{s-t}|D_{\sigma}u_s|^2ds + 6(T-t)\int_t^T P_{s-t}|f_s^0|^2ds.$$

So we can deduce by iteration the estimate over the interval [0, T] and obtain

$$|u_t|^2 \leq \sup_{t \in [0,T]} \sup_{x \in \mathbb{R}^d} \tilde{K}(P_{T-t}|\phi|^2 + (T-t) \int_t^T P_{s-t}|f_s^0|^2 ds)$$

$$\leq \sup_{t \in [0,T]} \tilde{K}(\|\phi^2\|_{\infty} + T^2\|f^0\|_{\infty}^2)$$

$$\leq K^2(\|\phi\|_{\infty}^2 + \|f^0\|_{\infty}^2),$$

which implies (2.2.5).

By the same methods as in [BPS05, Theorem 3.2], we obtain the following results. As the method is similar as in the proof of [BPS05, Theorem 3.2], we will give the

proof in the Appendix A.

Theorem 2.2.8 Suppose that m(dx) is a finite measure and that conditions (A1)-(A4), (H1)-(H5) hold. Then there exists a unique generalized solution of equation (2.2.1) and it satisfies the following estimates for some K_1 and K_2 independent of u, ϕ, f

$$\|u\|_{T}^{2} \leq K_{1}(\|\phi\|_{2}^{2} + \int_{0}^{T} \|f_{t}^{0}\|_{2}^{2} dt)$$
$$\|u\|_{\infty} \leq K_{2}(\|\phi\|_{\infty} + \|f^{0}\|_{\infty}).$$

The following lemma is essential to the case that m(dx) = dx.

Lemma 2.2.9 Assume conditions (A1)-(A4),(H1)-(H5) hold. If $u \in \hat{F}$ is bounded and for $\varphi \in bC_T$ satisfies

$$\int_0^T \mathcal{E}(u_t, \varphi_t) + (u_t, \partial_t \varphi_t) dt = \int_0^T (f_t(u_t, D_\sigma u_t), \varphi_t) dt + (u_T, \varphi_T) - (u_0, \varphi_0) dt$$

Then we have

$$\|u_t\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(u_s)ds \le 2\int_t^T (f_s(u_s, D_\sigma u_s), u_s)ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|u_s\|_2^2 ds, \quad 0 \le t \le T.$$

Proof Define $u_t^h = \frac{1}{2h} \int_{t-h}^{t+h} u_s ds$. Choose $\phi_t = u_t^h$, then we have for $t_0 \in [0,T]$

$$\int_{t_0}^T \mathcal{E}(u_t, u_t^h) + \frac{1}{2h}(u_t, u_{t+h}) - \frac{1}{2h}(u_t, u_{t-h})dt = \int_{t_0}^T (f_t(u_t, D_\sigma u_t), u_t^h)dt + (u_T, u_T^h) - (u_{t_0}, u_{t_0}^h)dt$$

That is to say,

(2.2.7)
$$\frac{1}{2h} \int_{T-h}^{T} (u_t, u_{t+h}) dt - \frac{1}{2h} \int_{t_0-h}^{t_0} (u_t, u_{t+h}) dt + \int_{t_0}^{T} \mathcal{E}(u_t, u_t^h) dt \\ = \int_{t_0}^{T} (f_t(u_t, D_\sigma u_t), u_t^h) dt + (u_T, u_T^h) - (u_{t_0}, u_{t_0}^h).$$

Letting $h \to 0$ in (2.2.7), the assertion follows .

For the case m(dx) = dx, we will use a weight function of the form $\pi(x) = \exp[-\rho\theta(x)]$, with $\theta \in C^1(\mathbb{R}^d)$ being a fixed function such that $0 \leq \theta(x) \leq |x|$, and $\theta(x) = |x|$ if $|x| \geq 1$, and $\rho \in \mathbb{R}^+$. If one chooses $\rho > 0$, then clearly one has $m(\mathbb{R}^d) < \infty$. We denote the generalized Dirichlet form, function spaces and the generator associated with $\rho > 0$ by \mathcal{E}^{ρ} , \hat{F}^{ρ} , \mathcal{C}^{ρ}_T , L_{ρ} respectively. In the case $\rho = 0$, we drop ρ in the notation, i.e. $\mathcal{E} = \mathcal{E}^0$. And for the case $\rho = 0$, we need the following

condition.

(A2') (Sobolev inequality) For $\rho = 0, \sigma$ is a bounded measurable field in \mathbb{R}^d and

$$\|u\|_q \le C\mathcal{E}^a(u,u)^{1/2}, \ \forall u \in C_0^\infty(\mathbb{R}^d),$$

where $\frac{1}{q} + \frac{1}{d} = \frac{1}{2}$ and $\|\cdot\|_q$ denotes the usual norm in L^q . And $|\hat{b}\sigma| \in L^d(\mathbb{R}^d; dx) + L^{\infty}(\mathbb{R}^d; dx), c \in L^{d/2}(\mathbb{R}^d; dx) + L^{\infty}(\mathbb{R}^d; dx).$

If (A2'), (A3) are satisfied, for $u, v \in bF$, we have

$$\mathcal{E}^{\rho}(u,v) = \int \langle D_{\sigma}u, D_{\sigma}v \rangle dm + \int cuv dm + \int \langle (b\sigma + \hat{b}\sigma, D_{\sigma}u \rangle v dm.$$

If $\rho = 0$, we additionally have

$$\mathcal{E}^{a,b}(u,u) \le C\mathcal{E}_1^a(u,u),$$

and that $F = F^a$. We consider the following condition, which is a technical condition for our proof:

(H6). $\mathcal{E}^{a}(u) < \infty, u \in L^{2} \Rightarrow u \in F.$

The Sobolev inequality and (H6) are satisfied if a is uniformly elliptic. By [S09, Lemma 4.20] we have:

Lemma 2.2.10 Assume conditions (A2'), (A3) and (H6) hold. Let $\rho > 0$. Then it holds

$$\mathcal{E}^{\rho}(u,\varphi) = \mathcal{E}(u,\varphi\exp(-\theta\rho)) + (M_{\rho}u,\varphi)_{\rho},$$

for $u \in F_{\rho}, \varphi \in bF_{\rho}$, where $M_{\rho}u = \rho \langle D_{\sigma}\theta, D_{\sigma}u \rangle$.

Theorem 2.2.11 Suppose that m(dx) = dx and that the conditions (A1),(A2'),(A3), (A4) (H1)-(H5), (H6) hold. Then there exists a unique generalized solution of equation (2.2.1) and it satisfies the following estimates with constants K_1 and K_2 independent of u, ϕ, f

$$||u||_T^2 \le K_1(||\phi||_2^2 + \int_0^T ||f_t^0||_2^2 dt).$$
$$||u||_{\infty} \le K_2(||\phi||_{\infty} + ||f^0||_{\infty}).$$

Proof Set for $\rho > 0$

$$f^{\rho}(t,x,y,z) := f(t,x,y,z) + \rho \sum_{l=1}^{k} \sum_{i=1}^{d} \sigma_l^i(x) \partial_i \theta(x) z_l(x),$$

and consider

(2.2.8)
$$(\partial_t + L_\rho)u + f^\rho(u, D_\sigma u) = 0, \qquad u_T = \phi.$$

The associated weak equation has the form $\forall \varphi \in b \mathcal{C}_T^{\rho}$ (2.2.9)

$$\int_0^T \mathcal{E}^{\rho}(u_t,\varphi_t) + (u_t,\partial_t\varphi_t)_{\rho} dt = \int_0^T (f_t^{\rho}(u_t,D_{\sigma}u_t),\varphi_t)_{\rho} dt + (u_T,\varphi_T)_{\rho} - (u_0,\varphi_0)_{\rho}.$$

As f^{ρ} satisfies conditions (H1)-(H5), we have a generalized solution u^{ρ} of (2.2.8).

Fix $\rho > 0$ and take $f_n \in C_0^{\infty}(\mathbb{R}^d)$ such that $f_n(x) = 1$ for $x \in B_n(0)$, $f_n(x) = 0$ for $x \in B_{2n}^c(0)$, $\partial_{x_i} f_n(x)$ are uniformly bounded and $\partial_{x_i} f_n(x) \to 0$ as $n \to \infty$. If $\varphi \in b\mathcal{C}_T$, then $\varphi f_n \exp(\theta \rho) \in b\mathcal{C}_T^{\rho}$. As

$$\int_0^T \mathcal{E}^{\rho}(u_t^{\rho}, \varphi_t f_n \exp(\theta \rho)) + (u_t^{\rho}, \partial_t \varphi_t f_n) dt$$
$$= \int_0^T (f_t^{\rho}(u_t^{\rho}, D_{\sigma} u_t^{\rho}), f_n \varphi_t) dt + (u_T^{\rho}, f_n \varphi_T) - (u_0^{\rho}, f_n \varphi_0)$$

,

by Lemma 2.2.10 we have

$$(2.2.10)$$

$$\int_0^T \mathcal{E}(u_t^{\rho}, \varphi_t f_n) + (u_t^{\rho}, \partial_t \varphi_t f_n) dt = \int_0^T (f_t(u_t^{\rho}, D_\sigma u_t^{\rho}), f_n \varphi_t) dt + (u_T^{\rho}, f_n \varphi_T) - (u_0^{\rho}, f_n \varphi_0).$$

If $u \in \hat{F}_{\tilde{\rho}}$ satisfies (2.2.10) for fixed $\tilde{\rho}$ with test function $\varphi \in b\mathcal{C}_T$, then u satisfies (2.2.9) for $\rho \geq \tilde{\rho}$, with test functions φ where $\varphi \in b\mathcal{C}_T^{\rho}$.

Now fix $\rho_1 > 0$. Then there exists a solution u^{ρ_1} of (2.2.8) associated to ρ_1 . We conclude that u^{ρ_1} satisfies the weak equation (2.2.9) for all $\rho > \rho_1$ with $\varphi \in bC_T^{\rho}$. Then by Lemma 2.2.9 and the same arguments as the uniqueness proof of Theorem 2.2.8 we have $u^{\rho_1} = u^{\rho}$ for all $\rho > \rho_1$.

Finally, we deduce that a solution $u^{\tilde{\rho}}$ of (2.2.8) associated to $\tilde{\rho}$ is a solution of (2.2.8) for all $\rho > 0$. Then by Theorem 2.2.8, we have

$$\|u^{\tilde{\rho}}\|_{T,\rho}^{2} \leq K_{1}(\|\phi\|_{2,\rho}^{2} + \int_{0}^{T} \|f_{t}^{0}\|_{2,\rho}^{2} dt).$$

Letting $\rho \to 0$, by Fatou's Lemma, we obtain

$$\begin{split} \|u^{\tilde{\rho}}\|_{T}^{2} &= \lim_{\rho \to 0} \|u^{\tilde{\rho}}\|_{T,\rho}^{2} \\ &\leq \lim_{\rho \to 0} K_{1}(\|\phi\|_{2,\rho}^{2} + \int_{0}^{T} \|f_{t}^{0}\|_{2,\rho}^{2} dt) \\ &= K_{1}(\|\phi\|_{2}^{2} + \int_{0}^{T} \|f_{t}^{0}\|_{2}^{2} dt), \end{split}$$

and

$$||u^{\tilde{\rho}}||_{\infty} \le K_2(||\phi||_{\infty} + ||f^0||_{\infty}).$$

By (H6), we have $u^{\tilde{\rho}} \in L^2((0,T), F)$. For $u^{\tilde{\rho}} \in \hat{F}^{\rho}$ for $\rho > 0$, we obtain for any $h_n \to 0$,

$$||u_{t+h_n}^{\tilde{\rho}} - u_t^{\tilde{\rho}}||_{2,\rho} \to 0.$$

Then there exists a subsequence such that $u_{t+h_{n_k}}^{\tilde{\rho}} \to u_t^{\tilde{\rho}}$ for m_{ρ} -almost every x. Hence, $u_{t+h_{n_k}}^{\tilde{\rho}} \to u_t^{\tilde{\rho}}$ for dx-almost every x. Then by the same arguments as the proof of Lemma 2.2.9, we have

$$\begin{split} |\|u_{t}^{\tilde{\rho}}\|_{2,\rho}^{2} - \|u_{t+h}^{\tilde{\rho}}\|_{2,\rho}^{2}| \leq & 2[|\int_{t}^{t+h} (u_{s}^{\tilde{\rho}}, f_{s}^{\rho})_{\rho} ds| + |\int_{t}^{t+h} \mathcal{E}^{a,\hat{b}}(u_{s}^{\tilde{\rho}}) ds| + \alpha \int_{t}^{t+h} \|u_{s}^{\tilde{\rho}}\|_{2,\rho}^{2} ds] \\ \leq & M \int_{t}^{t+h} \|f_{s}\|_{2,\rho} ds + M \int_{t}^{t+h} \mathcal{E}^{a,\hat{b}}_{c_{2}+1}(u_{s}^{\tilde{\rho}}) ds. \end{split}$$

Letting $\rho \to 0$, we get

$$|||u_t^{\tilde{\rho}}||_2^2 - ||u_{t+h}^{\tilde{\rho}}||_2^2| \le M \int_t^{t+h} ||f_s||_2 ds + M \int_t^{t+h} \mathcal{E}_{c_2+1}^{a,\hat{b}}(u_s^{\tilde{\rho}}) ds$$

Hence we have $u_{t+h_{n_k}}^{\tilde{\rho}} \to u_t^{\tilde{\rho}}$ in $L^2(\mathbb{R}^d, dx)$. Since this reason holds for every sequence $h_n \to 0$, we have $u^{\tilde{\rho}} \in \mathcal{C}([0,T], L^2)$, hence $u^{\tilde{\rho}} \in \hat{F}$. By the above arguments, we deduce that

$$\int_0^T \mathcal{E}(u_t^{\tilde{\rho}},\varphi_t f_n) + (u_t^{\tilde{\rho}},\partial_t \varphi_t f_n) dt = \int_0^T (f_t(u_t^{\tilde{\rho}},D_\sigma u_t^{\tilde{\rho}}),f_n \varphi_t) dt + (u_T^{\tilde{\rho}},f_n \varphi_T) - (u_0^{\tilde{\rho}},f_n \varphi_0).$$

Letting $n \to \infty$, we conclude that

$$\int_0^T \mathcal{E}(u_t^{\tilde{\rho}},\varphi_t) + (u_t^{\tilde{\rho}},\partial_t\varphi_t)dt = \int_0^T (f_t(u_t^{\tilde{\rho}},D_\sigma u_t^{\tilde{\rho}}),\varphi_t)dt + (u_T^{\tilde{\rho}},\varphi_T) - (u_0^{\tilde{\rho}},\varphi_0)dt$$

Since $f_t(u_t^{\tilde{\rho}}, D_{\sigma}u_t^{\tilde{\rho}}) \in L^1([0, T]; L^2)$, we can choose $(f^n)_{n \in \mathbb{N}} \subset C_0^{\infty}([0, T] \times \mathbb{R}^d)$ such that $\int_0^T \|f_t^n - f_t(u_t^{\tilde{\rho}}, D_{\sigma}u_t^{\tilde{\rho}})\|_2 dt \to 0$. Let v^n be the generalized solution associated

with (f^n, ϕ) . Then v^n is bounded. For

$$v_t := P_{T-t}\phi(x) + \int_t^T P_{s-t}f(s, \cdot, u_s^{\tilde{\rho}}, D_{\sigma}u_s^{\tilde{\rho}})(x)ds,$$

we have $||v^n - v||_T \to 0$. On the other hand, by Lemma 2.2.9 we have

$$\begin{aligned} &\|u_t^{\tilde{\rho}} - v_t^n\|_2^2 + 2\int_t^T \mathcal{E}^{a,\hat{b}}(u_s^{\tilde{\rho}} - v_s^n)ds \\ &\leq 2\int_t^T (f_s(u_s^{\tilde{\rho}}, D_\sigma u_s^{\tilde{\rho}}) - f_s^n, u_s^{\tilde{\rho}} - v_s^n)ds + 2\alpha\int_t^T \|u_s^{\tilde{\rho}} - v_s^n\|_2^2ds \\ &\leq 2M\int_t^T \|f_s(u_s^{\tilde{\rho}}, D_\sigma u_s^{\tilde{\rho}}) - f_s^n\|_2ds + 2\alpha\int_t^T \|u_s^{\tilde{\rho}} - v_s^n\|_2^2ds. \end{aligned}$$

By Gronwall's lemma we obtain $||v^n - u^{\tilde{\rho}}||_T \to 0$, as $n \to \infty$. Therefore, we have $u_t^{\tilde{\rho}} = v_t$. That is to say $u^{\tilde{\rho}}$ is a mild solution of (2.2.1).

2.3 Martingale representation for the processes

The Brownian motion has the *martingale representation property*: any martingale with respect to the filtration generated by the Brownian motion can be expressed as an Itô integral against the Brownian motion. In classic case, this property is essential to the existence of the solution of a BSDE.

The martingale representation property of a family of martingales has been studied in a huge of literature. Several general results have been obtained, for example, Jacod and Yor [JY77] have discovered the equivalence between the martingale representation property and the extremal property of martingale measures. However, when applied to specific situation, further work and hard estimates are often required. Recently, a lot of work (see e.g. [BPS05], [QY10], [Zh]) extend the martingale representation property to Markov processes associated with Dirichlet forms. In this section, we extend the martingale representation theorem under the framework of generalized Dirichlet forms.

2.3.1 Representation under P^x

In order to obtain the results for the probabilistic part, we need \mathcal{E} to be a quasiregular generalized Dirichlet form (Definition 1.3) in the sense of Remark 2.1.1 (iii) with $c_2, \hat{c} \equiv 0$ and $c \equiv 0$. The Markov process $X = (\Omega, \mathcal{F}_{\infty}, \mathcal{F}_t, X_t, P^x)$ with shift operator $(\theta_t)_{t\geq 0}$ is properly associated in the resolvent sense with \mathcal{E} , i.e. $R_{\alpha}f :=$ $E^x \int_0^\infty e^{-\alpha t} f(X_t) dt$ is an \mathcal{E} -quasi-continuous *m*-version of $G_{\alpha}f$, where $G_{\alpha}, \alpha > 0$ is the resolvent of \mathcal{E} and $f \in \mathcal{B}_b(\mathbb{R}^d) \cap L^2(\mathbb{R}^d; m)$. The coform $\hat{\mathcal{E}}$ introduced in Section 1.1 is a generalized Dirichlet form with the associated resolvent $(\hat{G}_{\alpha})_{\alpha>0}$ and there exists an *m*-tight special standard process properly associated in the resolvent sense with $\hat{\mathcal{E}}$. We always assume that $(\mathcal{F}_t)_{t\geq 0}$ is the (universally completed) natural filtration of X_t . From now on, we obtain all the results under the above assumption.

For the concepts related to additive functionals that we used in this section, we refer to Section 1.2. We consider the following conditions:

(A5) X is a continuous conservative Hunt process in the state space $\mathbb{R}^d \cup \{\partial\}$. \hat{G}_{α} is strongly continuous on \mathcal{V} and $\hat{\mathcal{E}}$ is quasi-regular. $C_0^{\infty}(\mathbb{R}^d) \subset \mathcal{F}$ and for $u \in \mathcal{F}$, there exists a sequence $\{u_n\} \subset C_0^{\infty}(\mathbb{R}^d)$ such that $\mathcal{E}(u_n - u, u_n - u) \to 0, n \to \infty$. $F_k := \{x \in \mathbb{R}^d, |x| \leq k\}$ is an \mathcal{E} -nest (Definition 1.2).

Remark 2.3.1 The last two conditions in (A5) are satisfied if $C_0^{\infty}(\mathbb{R}^d)$ is dense in \mathcal{F} . It is easy to verify the condition (A5), if \mathcal{E} satisfies the weak sector condition. The following two examples satisfy (A5) and they don't satisfy the weak sector condition.

Example 2.3.2 Consider $b = (b^i) : \mathbb{R}^d \to \mathbb{R}^d$ be a Borel-measurable vector field. Let us define

$$Lu = \Delta u + \langle b, \nabla u \rangle, \qquad \forall u \in C_b^{\infty}(\mathbb{R}^d).$$

Assume that

$$\lim_{|x|\to\infty} \langle b(x), x \rangle = -\infty,$$

and that there exist $C_1, C_2, m \in [0, \infty)$ such that

$$|b(x)| \le C_1 + C_2 |x|^m \qquad x \in \mathbb{R}^d.$$

Then by [BR95, Theorem 5.3], there exists a probability measure μ on \mathbb{R}^d such that

$$\int_{\mathbb{R}^d} Lud\mu = 0 \qquad \forall u \in C_b^\infty(\mathbb{R}^d)$$

and

$$b \in L^2(\mu).$$

By [BR95, Theorem 3.1] we have $d\mu$ is absolutely continuous w.r.t. dx and the density admits a representation φ^2 , where $\varphi \in H^{1,2}(\mathbb{R}^d, dx)$. The closure of

$$\mathcal{E}^{0}(u,v) = \frac{1}{2} \int \langle \nabla u, \nabla v \rangle d\mu; \ u, v \in C_{0}^{\infty}(\mathbb{R}^{d}),$$

on $L^2(\mathbb{R}^d, \mu)$ is a Dirichlet form. Denote $b^0 := 2\nabla \varphi/\varphi$ and $\beta := b - b^0$. Then we have $\beta \in L^2(\mathbb{R}^d; \mathbb{R}^d, \mu)$. Then by [St1, Proposition 1.10 and Proposition 2.4] $(L, C_0^{\infty}(\mathbb{R}^d))$ is L^1 -unique. Then by the proof of [St1, Proposition 2.4] for $u \in b\mathcal{F}$ there exists a sequence $\{u_n\} \subset C_0^{\infty}(\mathbb{R}^d)$ such that $\mathcal{E}(u_n - u, u_n - u) \to 0, n \to \infty$.

Consider the bilinear form

$$\mathcal{E}(u,v) = \frac{1}{2} \int \langle \nabla u, \nabla v \rangle d\mu - \int \langle \frac{1}{2}\beta, \nabla u \rangle v d\mu \ u, v \in C_0^{\infty}(\mathbb{R}^d).$$

Then by the computation in [Tr2, Section 4d] we have that conditions (A1)-(A5) hold for the bilinear form \mathcal{E} .

Example 2.3.3 Consider $d \ge 2$, $A = (a^{ij})$ a Borel-measurable mapping on \mathbb{R}^d with values in the non-negative symmetric matrices on \mathbb{R}^d , and let $b = (b^i) : \mathbb{R}^d \to \mathbb{R}^d$ be a Borel-measurable vector field. Consider the operator

$$L_{A,b}\psi = \partial_i(a^{ij}\partial_j\psi) + b^i\partial_i\psi, \qquad \forall \psi \in C_0^\infty(\mathbb{R}^d),$$

where we use the standard summation rule for repeated indices. By $H^{1,p}(\mathbb{R}^d, dx)$ we denote the standard Sobolev space of functions on \mathbb{R}^d whose first order derivatives are in $L^p(\mathbb{R}^d, dx)$. Assume that for p > d

 $(C1)a^{ij} \in H^{1,p}_{\text{loc}}(\mathbb{R}^d, dx), (a^{ij}) \text{ is uniformly strictly elliptic in } \mathbb{R}^d.$ $(C2)b^i \in L^p_{\text{loc}}(\mathbb{R}^d, dx).$

Here by $H^{1,p}_{loc}(\mathbb{R}^d, dx)$ we denote the class of all functions f on \mathbb{R}^d such that $f\chi \in H^{1,p}(\mathbb{R}^d, dx)$ for all $\chi \in C^{\infty}_0(\mathbb{R}^d)$. And $L^p_{loc}(\mathbb{R}^d, dx)$ denotes the class of all functions f on \mathbb{R}^d such that $f\chi \in L^p(\mathbb{R}^d)$ for all $\chi \in C^{\infty}_0(\mathbb{R}^d)$. Assume that there exists $V \in C^2(\mathbb{R}^d)$ ("Lyapunov function") such that

$$\lim_{|x|\to\infty} V(x) = +\infty, \qquad \lim_{|x|\to\infty} L_{A,b}V(x) = -\infty.$$

Examples of V can be found in [BRS00] and the reference therein.

Then by [BRS00, Theorem 2.2] there exists a probability measure μ on \mathbb{R}^d such that

$$\int_{\mathbb{R}^d} L_{A,b} \psi d\mu = 0 \qquad \forall \psi \in C_0^\infty(\mathbb{R}^d).$$

Then by [BRS00, Theorem 2.1] we have $d\mu$ is absolutely continuous w.r.t. dxand that the density admits a representation φ^2 , where $\varphi^2 \in H^{1,p}_{loc}(\mathbb{R}^d, dx)$. The closure of

$$\mathcal{E}^{0}(u,v) = \frac{1}{2} \int \langle \nabla ua, \nabla v \rangle d\mu; \ u, v \in C_{0}^{\infty}(\mathbb{R}^{d}),$$

on $L^2(\mathbb{R}^d, \mu)$ is a Dirichlet form.

If in addition, there is a positive Borel function θ on $[0, \infty)$ such that $\lim_{t\to\infty} \theta(t) = +\infty$ and

$$L_{A,b}V(x) \le c_1 - c_2\theta(|bA^{-\frac{1}{2}}|)|bA^{-\frac{1}{2}}|^2$$

outside some ball, then by [BKR06, Theorem 2.6] $b \in L^2(\mathbb{R}^d; \mathbb{R}^d, \mu)$. Set $b^0 = (b_1^0, ..., b_d^0)$, where $b_i^0 := 2 \sum_{j=1}^d a^{ij} \partial_j \varphi / \varphi, i = 1, ..., d$ and $\beta := b - b^0$. By [BKR06, Theorem 2.6] $\beta \in L^2(\mathbb{R}^d; \mathbb{R}^d, \mu)$. Then by [St1, Proposition 1.10 and Proposition 2.4] $(L, C_0^\infty(\mathbb{R}^d))$ is L^1 -unique. Then by the proof of [St1, Proposition 2.4] for $u \in b\mathcal{F}$ there exists a sequence $\{u_n\} \subset C_0^\infty(\mathbb{R}^d)$ such that $\mathcal{E}(u_n - u, u_n - u) \to 0, n \to \infty$.

Consider the bilinear form

$$\mathcal{E}(u,v) = \frac{1}{2} \int \langle \nabla ua, \nabla v \rangle d\mu - \int \langle \frac{1}{2}\beta, \nabla u \rangle v d\mu \ u, v \in C_0^{\infty}(\mathbb{R}^d).$$

Then by the computation in [Tr2, Section 4d] we have that conditions (A1)-(A5) hold for the bilinear form \mathcal{E} .

For an initial distribution $\mu \in \mathcal{P}(\mathbb{R}^d)$, here $\mathcal{P}(\mathbb{R}^d)$ denotes all the probabilities on \mathbb{R}^d , we prove the *Fukushima reprensentation property* mentioned in [QY10] holds for X, i.e. there is an algebra $K(\mathbb{R}^d) \subset \mathcal{B}_b(\mathbb{R}^d)$ which generates the Borel σ -algebra $\mathcal{B}(\mathbb{R}^d)$ and is invariant under R_α for $\alpha > 0$, and there are finitely many continuous martingales $M^1, ..., M^d$ over $(\Omega, \mathcal{F}^\mu, \mathcal{F}^\mu_t, P^\mu)$ such that for any potential $u = R_\alpha f$, where $\alpha > 0$ and $f \in K(\mathbb{R}^d)$, the martingale part $M^{[u]}$ of the semimartingale $u(X_t) - u(X_0)$ has the martingale representation in terms of $(M^1, ..., M^d)$, that is, there are predictable processes $F_1, ..., F_d$ on $(\Omega, \mathcal{F}^\mu, \mathcal{F}^\mu_t)$ such that

$$M_t^{[u]} = \sum_{j=1}^d \int_0^t F_s^j dM_s^j \qquad P^{\mu} - a.s..$$

Let us first calculate the energy measure related to $\langle M^{[u]} \rangle, u \in C_0^{\infty}(\mathbb{R}^d)$. By [Tr2, (23)], for bounded $g \in L^1(\mathbb{R}^d, m)$, we have

$$\int \hat{G}_{\gamma}gd\mu_{\langle M^{[u]}\rangle}$$

$$= \lim_{\alpha \to \infty} \alpha (U^{\alpha+\gamma}_{\langle M^{[u]}\rangle} 1, \hat{G}_{\gamma}g)$$

$$= \lim_{\alpha \to \infty} \lim_{t \to \infty} E_{\hat{G}_{\gamma}g \cdot m}(\alpha e^{-(\gamma+\alpha)t} \langle M^{[u]}\rangle_t) + \lim_{\alpha \to \infty} E_{\hat{G}_{\gamma}g \cdot m}(\int_0^\infty \langle M^{[u]}\rangle_t \alpha(\gamma+\alpha)e^{-(\gamma+\alpha)t}dt)$$

$$= \lim_{\alpha \to \infty} \lim_{t \to \infty} \alpha \langle \mu_{\langle M^{[u]}\rangle}, e^{-(\gamma+\alpha)t} \int_0^t \hat{P}_s \hat{G}_{\gamma}gds \rangle$$

$$+ \lim_{\alpha \to \infty} \alpha(\gamma + \alpha) \left(\int_{0}^{\infty} e^{-(\gamma + \alpha)t} E_{\hat{G}_{\gamma}g \cdot m}((u(X_{t}) - u(X_{0}) - N_{t}^{[u]})^{2}) dt \right)$$

$$= \lim_{\alpha \to \infty} \alpha(\gamma + \alpha) \left(\int_{0}^{\infty} e^{-(\gamma + \alpha)t} E_{\hat{G}_{\gamma}g \cdot m}((u(X_{t}) - u(X_{0}))^{2}) dt \right)$$

$$= \lim_{\alpha \to \infty} 2\alpha(u - \alpha G_{\alpha}u, u\hat{G}_{\gamma}g) - \alpha(u^{2}, \hat{G}_{\gamma}g - \alpha \hat{G}_{\alpha}\hat{G}_{\gamma}g)$$

$$= 2(-Lu, u\hat{G}_{\gamma}g) - (-Lu^{2}, \hat{G}_{\gamma}g)$$

$$= 2\mathcal{E}(u, u\hat{G}_{\gamma}g) - \mathcal{E}(u^{2}, \hat{G}_{\gamma}g) + 2\int \langle b\sigma, D_{\sigma}u \rangle u\hat{G}_{\gamma}gm(dx) - \int \langle b\sigma, D_{\sigma}u^{2} \rangle \hat{G}_{\gamma}gm(dx)$$

$$= 2\mathcal{E}^{a}(u, u\hat{G}_{\gamma}g) - \mathcal{E}^{a}(u^{2}, \hat{G}_{\gamma}g)$$

$$= 2\int \langle D_{\sigma}u, D_{\sigma}(u\hat{G}_{\gamma}g) \rangle dm - \int \langle D_{\sigma}u^{2}, D_{\sigma}(\hat{G}_{\gamma}g) \rangle dm$$

$$= 2\int \langle D_{\sigma}u, D_{\sigma}u \rangle \hat{G}_{\gamma}gdm.$$

Thus, by [Tr2, Theorem 2.5] we obtain

$$\mu_{\langle M^{[u]}\rangle} = 2 \langle D_{\sigma} u, D_{\sigma} u \rangle \cdot dm.$$

So, for $u, v \in C_0^{\infty}(\mathbb{R}^d)$, for q.e. x under P^x ,

(2.3.1)
$$\langle M^{[u]}, M^{[v]} \rangle_t = 2 \int_0^t \langle D_\sigma u, D_\sigma v \rangle(X_s) ds.$$

Then by (A5) and [Tr1, Theorem 4.4], we deduce (2.3.1) for every $u, v \in \mathcal{F}$.

By (A5) $F_k = \{x \in \mathbb{R}^d, |x| \leq k\}, k \in \mathbb{N} \text{ is an } \mathcal{E}\text{-nest.}$ By [Tr2, Theorem 3.6], for $u_i(x) = x_i$, we have the Fukushima decomposition for $A^{[u_i]} := u_i(X) - u_i(X_0)$, and let $M^{(i)} \in \dot{\mathcal{M}}_{loc,(F_k)_{k\in\mathbb{N}}}$ be the associated local martingale additive functional. Here $\dot{\mathcal{M}}_{loc,(F_k)_{k\in\mathbb{N}}}$ means that there exists $(M^k)_{k\in\mathbb{N}} \subset \dot{\mathcal{M}}$ such that for any k

$$M_t = M_t^k \quad \forall t \le \sigma_{F_k^c},$$

where $\sigma_{F_k^c} = \inf\{t > 0 | X_t \in F_k^c\}.$

We define the stochastic integral $f \cdot M^{(i)} \in \dot{\mathcal{M}}$ for $f \in L^2(\mathbb{R}^d; \mu_{\langle M^{(i)} \rangle})$ as in [FOT94, p243], and for $L \in \dot{\mathcal{M}}$ we have

$$\langle f \cdot M^{(i)}, L \rangle = f \cdot \langle M^{(i)}, L \rangle,$$

where $f \cdot \langle M^{(i)}, L \rangle_t = \int_0^t f(X_s) d\langle M^{(i)}, L \rangle_s$.

Theorem 2.3.4 Suppose (A5) holds. Let $u \in C_0^1(\mathbb{R}^d)$, where $C_0^1(\mathbb{R}^d)$ denotes

the continuous function with compact support and continuous first order derivative. Then for q.e. x under P^x ,

$$M^{[u]} = \sum_{i=1}^{d} u_{x_i} \cdot M^{(i)}$$

Proof By [Tr2, Theorem 3.6] we obtain that for q.e. x under P^x , for $t \ge 0$

$$\langle M^{[u]} - \sum_{i=1}^{d} u_{x_i} \cdot M^{(i)} \rangle_t = \sum_{i,j=1}^{n} \int_0^t u_{x_i}(X_s) u_{x_j}(X_s) d\langle M^{(i)}, M^{(j)} \rangle_s - 2 \sum_{i,j=1}^{n} \int_0^t u_{x_i}(X_s) u_{x_j}(X_s) d\langle M^{(i)}, M^{(j)} \rangle_s + \sum_{i,j=1}^{n} \int_0^t u_{x_i}(X_s) u_{x_j}(X_s) d\langle M^{(i)}, M^{(j)} \rangle_s = 0.$$

Then the assertion follows.

Then by [Tr2, Lemma 2.4, Lemma 1.18], we have for q.e. x under P^x ,

(2.3.2)
$$\langle M^{(i)}, M^{(j)} \rangle_t = 2 \int_0^t \sum_{l=1}^k \sigma_l^i(X_s) \sigma_l^j(X_s) ds.$$

Lemma 2.3.5 Suppose (A5) holds. Let C_1 be a uniformly dense subset of $C_0(\mathbb{R}^d)$. Here $C_0(\mathbb{R}^d)$ denotes the continuous function with compact support. Then the family $\{f \cdot M^{[u]} : f \in C_1, u \in C_0^{\infty}(\mathbb{R}^d)\}$ of stochastic integrals is dense in $(\dot{\mathcal{M}}, e)$.

Proof Suppose that an MAF $M \in \dot{\mathcal{M}}$ is *e*-orthogonal to the above family, namely, $\int_X f d\mu_{\langle M, M^{[u]} \rangle} = 0, \forall f \in \mathcal{C}_1, u \in C_0^{\infty}(\mathbb{R}^d)$. This identity extends to all $u \in \mathcal{F}$ by [Tr1, (13)] and (A5). Hence,

$$\langle M, M^{[u]} \rangle = 0 \qquad \forall u \in \mathcal{F}.$$

In particular, this holds for $u = G_{\alpha}g, \alpha > 0, \forall g \in C_0(\mathbb{R}^d)$. By [FOT94, Theorem A.3.20] we deduce that M = 0.

Theorem 2.3.6 Suppose (A5) holds. Then the space $\dot{\mathcal{M}}$ can be represented by stochastic integrals based on $M^{(i)} = M^{[x_i]}, 1 \leq i \leq d$:

(2.3.3)
$$\dot{\mathcal{M}} = \{ \sum_{i=1}^{d} f_i \cdot M^{(i)} : \sum_{i,j=1}^{d} \sum_{l=1}^{k} \int_{\mathbb{R}^d} (f_i f_j \sigma_i^l \sigma_j^l)(x) m(dx) < \infty \},$$

and

$$e(\sum_{i=1}^{d} f_{i} \cdot M^{(i)}) = \sum_{i,j=1}^{d} \sum_{l=1}^{k} \int_{\mathbb{R}^{d}} (f_{i}f_{j}\sigma_{i}^{l}\sigma_{j}^{l})(x)m(dx).$$

Proof The space of the right hand side of (2.3.3) is dense in (\mathcal{M}, e) , since it contains the set

$$\{f \cdot M^{[u]} = \sum_{i=1}^{u} (fu_{x_i}) \cdot M^{(i)}; f \in C_0(\mathbb{R}^d), u \in C_0^1(\mathbb{R}^d)\},\$$

which is dense in $(\dot{\mathcal{M}}, e)$ by Lemma 2.3.5. Hence, it is enough to show that the right hand side of (2.3.3) is closed in $(\dot{\mathcal{M}}, e)$.

Suppose that $\lim_{n\to\infty} e(M_n - M) = 0$, where

$$M_n = \sum_{i=1}^d f_i^{(n)} \cdot M^{(i)}, \sum_{i,j=1}^d \int_{\mathbb{R}^d} a_{ij} f_i^{(n)} f_j^{(n)} dm < \infty, M \in \dot{\mathcal{M}}.$$

Set $f^n := (f_1^n, ..., f_d^n)$. Since

$$e(M_n - M_m) = \int_{\mathbb{R}^d} |(f^n - f^m)\sigma|^2 dm,$$

we deduce that $f^n \sigma$ converges in $L^2(\mathbb{R}^d, \mathbb{R}^k; m)$ to some function $h \in L^2(\mathbb{R}^d, \mathbb{R}^k; m)$ Let $f = h\tau$, where τ is the matrix that we have introduced at the beginning of Section 2.1 and $M' = \sum_{i=1}^d f_i \cdot M^{(i)}$, then

$$e(M_n - M') = \int_D |(f^n - f)\sigma|^2 dm$$
$$= \int_D |f^n \sigma - h|^2 dm.$$

which converges to zero as $n \to \infty$. Therefore, we have M = M' and

$$e(M) = \sum_{i,j=1}^{d} \sum_{l=1}^{k} \int_{\mathbb{R}^d} (f_i f_j \sigma_i^l \sigma_j^l)(x) m(dx) < \infty.$$

As a consequence, X satisfies *Fukushima representation theorem* mentioned before. To prove main results in this section we need the following lemma which is proved by Meyer in [M67] (see also [QY10, Lemma 2.2]).

Lemma 2.3.7 Let $K(\mathbb{R}^d) \subset \mathcal{B}_b(\mathbb{R}^d)$ be an algebra, which generates the Borel σ -algebra $\mathcal{B}(\mathbb{R}^d)$, and \mathcal{C}_0 be all $\xi = \xi_1 \cdots \xi_n$, $n \in \mathbb{N}$, $\xi_j = \int_0^\infty e^{-\alpha_j t} f_j(X_t) dt$, where

 $\alpha_j \in \mathbb{Q}^+, f_j \in K(\mathbb{R}^d), j = 1, ..., n$. Then the completion of the σ -algebra generated by \mathcal{C}_0 is \mathcal{F}_{∞} .

Moreover, by an analogous method to the proof of [QY10, Theorem 3.1] we have the martingale representation theorem for X.

Theorem 2.3.8 Suppose (A5) holds. Then there exists some \mathcal{E} -exceptional set \mathcal{N} such that the following representation result holds: For every bounded \mathcal{F}_{∞} -measurable random variable ξ , there exists an predictable process $(\phi_1, ..., \phi_d)$: $[0, \infty) \times \Omega \to \mathbb{R}^d$, such that for each probability measure ν , supported by $\mathbb{R}^d \setminus \mathcal{N}$, one has

$$\xi = E^{\nu}(\xi|\mathcal{F}_0) + \sum_{i=0}^d \int_0^\infty \phi_s^i dM_s^{(i)} \qquad P^{\nu} - a.s.,$$

and

$$E^{\nu} \int_0^\infty |\phi_s \sigma(X_s)|^2 ds \le \frac{1}{2} E^{\nu} \xi^2.$$

If another predictable process $\phi' = (\phi'_1, ..., \phi'_d)$ satisfies the same relations under a certain measure P^{ν} , then one has $\phi'_t \sigma(X_t) = \phi_t \sigma(X_t), dt \times dP^{\nu} - a.s.$

Proof Suppose that \mathcal{N} is some fixed exceptional set. By \mathcal{K} we denote the class of bounded random variables for which the statement holds outside this set. First we prove that if $(\xi_n) \subset \mathcal{K}$ is a uniformly bounded increasing sequence and $\xi = \lim_{n \to \infty} \xi_n$ then $\xi \in \mathcal{K}$.

Indeed, since ξ and ξ_0 are bounded, $E^x |\xi_n - \xi|^2 \to 0$. Denoting by ϕ^n the process which represents ξ^n , we obtain

$$E^{x} \int_{0}^{\infty} |(\phi_{s}^{p} - \phi_{s}^{n})\sigma(X_{s})|^{2} ds \le E^{x}(\xi_{p} - \xi_{n})^{2} \to 0, \text{ as } n, p \to \infty.$$

Now we want to pass to the limit with ϕ^n pointwise, so that the limit be predictable. In order to obtain a sequence of representable variables that converges rapidly enough under all measures $P^x, x \in \mathcal{N}^c$, we are going to construct them as follows. For each l = 0, 1, ... set $n_l(x) = \inf\{n | E^x(\xi - \xi_n)^2 < \frac{1}{2^l}\}, \bar{\xi}_l = \xi_{n_l(x)}$. The process which represents ξ^l is simply obtained by the formula $\bar{\phi}^l = \phi^{n_l(X_0)}$. With this sequence we may pass to the limit and define $\psi_s = \limsup_{l\to\infty} \bar{\phi}^l_s \sigma(X_s)$ (where limsup is taken on each coordinate) and $\phi_s = \psi_s \tau(X_s)$ where τ is the matrix that we have introduced in the beginning of Section 2.1. Then we obtain

$$E^x \int_0^\infty |(\bar{\phi}_s^l - \phi_s)\sigma(X_s)|^2 ds \to 0, \text{ as } l \to \infty.$$

By this we obtain $\xi \in \mathcal{K}$.

Let $K(\mathbb{R}^d) \subset \mathcal{B}_b(\mathbb{R}^d)$ be a countable set which is closed under multiplication, generates the Borel σ -algebra $\mathcal{B}(\mathbb{R}^d)$ and $R_{\alpha}(K(\mathbb{R}^d)) \subset K(\mathbb{R}^d)$ for each $\alpha \in \mathbb{Q}^+$. Such $K(\mathbb{R}^d)$ can be constructed as follows. Choose $N_0 \subset \mathcal{B}_b(\mathbb{R}^d)$ to be a countable set which generates the Borel σ -algebra $\mathcal{B}(\mathbb{R}^d)$. For $l \geq 1$ define $N_{l+1} =$ $\{g_1...g_k, R_{\alpha}fg_1...g_k, f, g_i \in N_l, k \in \mathbb{N} \cup \{0\}, \alpha \in Q^+\}$ and $K(\mathbb{R}^d) := \bigcup_{l=0}^{\infty} N_l$.

Let \mathcal{C}_0 be all $\xi = \xi_1 \cdots \xi_n$, $n \in \mathbb{N}$, $\xi_j = \int_0^\infty e^{-\alpha_j t} f_j(X_t) dt$, where $\alpha_j \in \mathbb{Q}^+$, $f_j \in K(\mathbb{R}^d), j = 1, ..., n$. By Lemma 2.3.7 the completion of the σ -algebra generated by \mathcal{C}_0 is \mathcal{F}_∞ . By the first part of our proof a monotone class argument reduces the proof to the representation of a random variable in \mathcal{C}_0 .

Let $\xi \in C_0$. By Markov property of the process X, we obtain the following result (see e.g. [QY10, Theorem 3.1])

$$N_t = E^x(\xi|\mathcal{F}_t) = \sum_m Z_t^m,$$

where the sum is a finite one, and for each $m, Z^m = Z_t$ has the following form

$$Z_t = V_t u(X_t),$$

(the superscript *m* will be dropped if no confusion may arise), where $V_t = \prod_{i=1}^{k'} \int_0^t e^{-\beta_i s} g_i(X_s) ds$ and $u(x) = R_{\beta_1 + \ldots + \beta_k}(h_1(R_{\beta_2 + \ldots + \beta_k}h_2 \dots (R_{\beta_k}h_k) \dots))$ for $\beta_i \in \mathbb{Q}^+, g_i, h_i \in K(\mathbb{R}^d)$. We have $u \in K(\mathbb{R}^d)$. Hence, by the Fukushima decomposition and the Fukushima representation we obtain

(2.3.4)
$$u(X_t) - u(X_0) = M_t^{[u]} + A_t^{[u]} = \sum_{j=1}^d \int_0^t G_s^j dM_s^{(j)} + A_t^{[u]} P^x - a.s.$$

for some predictable processes G^{j} . Then by Itô's formula, we obtain

$$Z_t = Z_0 + \int_0^t u(X_s) dV_s + \int_0^t V_s dA_t^{[u]} + \sum_{j=1}^d \int_0^t V_s \cdot G_s^j dM_s^{(j)} \quad P^x - a.s..$$

Hence the martingale part of Z_t is $\sum_{j=1}^d \int_0^t V_s \cdot G_s^j dM_s^{(j)}$. We deduce that

$$N_t = \sum_{i=1}^d \int_0^t \sum_m V_s^m \cdot G_s^{m,i} dM_s^{(i)} \qquad P^x - a.s..$$

As a result, the representation holds for $\xi \in C_0$. As (2.3.4) holds for every x outside a set of zero capacity. Then we take the exceptional set \mathcal{N} in the assertion to be the union of all these exceptional sets corresponding to $u \in K(\mathbb{R}^d)$. One may represent separately the positive and the negative parts and then we have the following corollary.

Corollary 2.3.9 Suppose (A5) holds. Let \mathcal{N} be the set obtained in the preceding theorem. Then for any \mathcal{F}_{∞} -measurable nonnegative random variable $\xi \geq 0$ there exists a predictable process $\phi = (\phi_1, ..., \phi_d) : [0, \infty) \times \Omega \to \mathbb{R}^d$ such that the following holds

$$\xi = E^{x}(\xi|\mathcal{F}_{0}) + \sum_{i=0}^{d} \int_{0}^{\infty} \phi_{s}^{i} dM_{s}^{(i)} \qquad P^{x} - a.s.,$$

and

$$E^x \int_0^\infty |\phi_s \sigma(X_s)|^2 ds \le \frac{1}{2} E^x \xi^2,$$

for each point $x \in \mathcal{N}^c$ such that $E^x \xi < \infty$.

If another predictable process $\phi' = (\phi'_1, ..., \phi'_d)$ satisfies the same relations under a certain measure P^x , then one has $\phi'_t \sigma(X_t) = \phi_t \sigma(X_t), dt \times dP^x - a.s.$

2.3.2 Representation under P^m

In the following, we use the notation $\int_0^t \psi(s, X_s) dM_s := \sum_{i=1}^d \int_0^t \psi_i(s, X_s) dM_s^{(i)}$.

Lemma 2.3.10 Suppose (A1)-(A5) hold. If $u \in \mathcal{D}(L)$ and $\psi \in \tilde{\nabla} u$, then

$$u(X_t) - u(X_0) = \int_0^t \psi(X_s) dM_s + \int_0^t Lu(X_s) ds \qquad P^m - a.s..$$

Proof The assertion follows by the Fukushima decomposition, (2.3.1), (2.3.2) and Theorem 2.3.6.

The aim of the rest of this section is to extend this representation to time dependent function u(t, x).

Lemma 2.3.11 Suppose (A1)-(A5) hold. Let $u : [0,T] \times \mathbb{R}^d \to \mathbb{R}$ be such that

- (i) $\forall s, u_s \in \mathcal{D}(L)$ and $s \to Lu_s$ is continuous in L^2 .
- (ii) $u \in C^1([0,T]; L^2)$.

Then clearly $u \in C_T$. Moreover, for any $\psi \in \tilde{\nabla} u$ and any s, t > 0 such that s + t < T, the following relation holds P^m -a.s.

$$u(s+t, X_t) - u(s, X_0) = \int_0^t \psi(s+r, X_r) dM_r + \int_0^t (\partial_s + L) u(s+r, X_r) dr$$

Proof We prove the above relation with s = 0, the general case being similar. Let

 $0 = t_0 < t_1 < \dots < t_p = t$ be a partition of the interval [0, t] and write

$$u(t, X_t) - u(0, X_0) = \sum_{n=0}^{p-1} (u(t_{n+1}, X_{t_{n+1}}) - u(t_n, X_{t_n})).$$

Then, on account of the preceding lemma, each term of the sum is expressed as

$$u(t_{n+1}, X_{t_{n+1}}) - u(t_n, X_{t_n})$$

= $u(t_{n+1}, X_{t_{n+1}}) - u(t_{n+1}, X_{t_n}) + u(t_{n+1}, X_{t_n}) - u(t_n, X_{t_n})$
= $\int_{t_n}^{t_{n+1}} \psi^{n+1}(X_s) dM_s + \int_{t_n}^{t_{n+1}} Lu_{t_{n+1}}(X_s) ds + \int_{t_n}^{t_{n+1}} \partial_s u_s(X_{t_n}) ds,$

where $\psi^{n+1} = (\psi_1^{n+1}, ..., \psi_d^{n+1}) \in \tilde{\nabla} u_{t_{n+1}}$ and the last integral is obtained by using the Leibnitz-Newton formula for the L^2 -valued function $s \to u_s$. Below we estimate in L^2 the differences between each term in the last expression and the similar terms corresponding to the formula we have to prove. Here we use $mP_t \leq m$ i.e. $\int P_t f dm \leq \int f dm$ for $f \in \mathcal{B}^+$. This holds since \hat{P}_t is sub-Markovian. Then we have

$$\begin{split} & E^{m}(\int_{t_{n}}^{t_{n+1}}\psi^{n+1}(X_{s}).dM_{s} - \int_{t_{n}}^{t_{n+1}}\psi(s,X_{s}).dM_{s})^{2} \\ &= & E^{m}\int_{t_{n}}^{t_{n+1}}|(\psi^{n+1}(X_{s}) - \psi(s,X_{s}))\sigma(X_{s})|^{2}ds \\ &\leq & \int_{t_{n}}^{t_{n+1}}\mathcal{E}^{a}(u_{t_{n+1}} - u_{s})ds. \end{split}$$

Since $s \to Lu_s$ is continuous in L^2 , it follows that $s \to u_s$ is continuous w.r.t. \mathcal{E}_1^a -norm. Hence the difference appearing in the last integral $\mathcal{E}^a(u_{t_{n+1}} - u_s)$ is uniformly small, provided the partition is fine enough. From this one deduces that

$$\sum_{n=0}^{p-1} \int_{t_n}^{t_{n+1}} \psi^{n+1}(X_s) dM_s \to \int_0^t \psi(s+r, X_r) dM_r.$$

The next difference is estimated by using Minkowski's inequality

$$(E^{m}(\sum_{n=0}^{p-1}\int_{t_{n}}^{t_{n+1}}(Lu_{t_{n+1}}-Lu_{s})(X_{s})ds)^{2})^{1/2}$$

$$\leq \sum_{n=0}^{p-1}\int_{t_{n}}^{t_{n+1}}(E^{m}(Lu_{t_{n+1}}-Lu_{s})^{2}(X_{s}))^{1/2}ds$$

$$\leq \sum_{n=0}^{p-1}\int_{t_{n}}^{t_{n+1}}\|Lu_{t_{n+1}}-Lu_{s}\|_{2}ds,$$

so that it is similarly expressed as in integral of a uniformly small quantity.

For the last difference we write

$$(E^{m}(\sum_{n=0}^{p-1}\int_{t_{n}}^{t_{n+1}}(\partial_{s}u_{s}(X_{t_{n}}) - \partial_{s}u_{s}(X_{s}))ds)^{2})^{1/2}$$

$$\leq \sum_{n=0}^{p-1}\int_{t_{n}}^{t_{n+1}}(E^{m}(\partial_{s}u_{s}(X_{t_{n}}) - \partial_{s}u_{s}(X_{s}))^{2})^{1/2}ds$$

$$= \sum_{n=0}^{p-1}\int_{t_{n}}^{t_{n+1}}(E^{m}(\partial_{s}u_{s}(X_{t_{n}})^{2} + P_{s-t_{n}}(\partial_{s}u_{s})^{2}(X_{t_{n}}) - 2\partial_{s}u_{s}(X_{t_{n}})(P_{s-t_{n}}\partial_{s}u)(X_{t_{n}})))^{1/2}ds$$

$$= \sum_{n=0}^{p-1}\int_{t_{n}}^{t_{n+1}}(E^{m}((\partial_{s}u_{s}(X_{t_{n}}) - (P_{s-t_{n}}\partial_{s}u_{s})(X_{t_{n}}))^{2} + (P_{s-t_{n}}(\partial_{s}u_{s})^{2}(X_{t_{n}}) - ((P_{s-t_{n}}\partial_{s}u_{s})(X_{t_{n}}))^{2}))^{1/2}ds$$

$$\leq \sum_{n=0}^{p-1}(\int_{t_{n}}^{t_{n+1}}\int(\partial_{s}u_{s} - P_{s-t_{n}}\partial_{s}u_{s})^{2} + P_{s-t_{n}}(\partial_{s}u_{s})^{2} - (P_{s-t_{n}}\partial_{s}u_{s})^{2}dm))^{1/2}ds.$$

From the hypotheses it follows that this will tend also to zero if the partition is fine enough. Hence the assertions follow. $\hfill \Box$

Theorem 2.3.12 Suppose (A1)-(A5) hold. Let $f \in L^1([0,T]; L^2)$ and $\phi \in L^2(\mathbb{R}^d)$ and define

$$u_t := P_{T-t}\phi + \int_t^T P_{s-t}f_s ds.$$

Then for each $\psi \in \tilde{\nabla} u$ and for each $s \in [0, T]$, the following relation holds P^m -a.s.

$$u(s+t, X_t) - u(s, X_0) = \int_0^t \psi(s+r, X_r) dM_r - \int_0^t f(s+r, X_r) dr$$

In particular, if u is a generalized solution of PDE (2.2.1), for each $t \in [s, T]$ the following BSDE holds P^{m} -a.s.

$$u(t, X_{t-s}) = \phi(X_{T-s}) + \int_{t}^{T} f(r, X_{r-s}, u(r, X_{r-s}), D_{\sigma}u(r, X_{r-s})) dr - \int_{t-s}^{T-s} \psi(s+r, X_{r}) dM_{r}$$

Proof Assume first that ϕ and f satisfy the conditions in Proposition 2.1.6 (ii). Then we have u satisfies the conditions in Lemma 2.3.11. Then by Lemma 2.3.11, the assertion follows. For the general case we choose u^n associated (f^n, ϕ^n) as in Proposition 2.1.9. Then we obtain that if $n \to \infty$, $||u^n - u||_T \to 0$. For u^n we have

(2.3.5)
$$u^n(s+t, X_t) - u^n(s, X_0) = \int_0^t \psi^n(s+r, X_r) dM_r - \int_0^t f^n(s+r, X_r) dr.$$

As

$$E^{m} |\int_{0}^{t} (\psi^{n}(s+r, X_{r}) - \psi^{p}(s+r, X_{r}).dM_{r}|^{2}$$

$$\leq E^{m} \int_{0}^{t} |(\psi^{n}(s+r, X_{r}) - \psi^{p}(s+r, X_{r}))\sigma(X_{r})|^{2} dr$$

$$\leq \int_{0}^{t} \mathcal{E}^{a}(u_{s+r}^{n} - u_{s+r}^{p}) dr,$$

letting $n \to \infty$ in (2.3.5) we obtain the assertion.

2.4 BSDE's and Generalized Solutions

The set \mathcal{N} obtained in Theorem 2.3.8 will be fixed throughout this section. By Theorem 2.3.8 we can solve BSDE's under all measures P^x , $x \in \mathcal{N}^c$, at the same time. We will treat systems of l equations, $l \in \mathbb{N}$, associated to \mathbb{R}^l -valued functions $f: [0,T] \times \Omega \times \mathbb{R}^l \times \mathbb{R}^l \otimes \mathbb{R}^k \mapsto \mathbb{R}^l$. These functions are assumed to depend on the past in general and it turns out that a good theory is developed assuming that they are predictable. This means that we consider the map $(s,\omega) \mapsto f(s,\omega,\cdot,\cdot)$ as a predictable process with respect to the canonical filtration of our process (\mathcal{F}_t) .

Lemma 2.4.1 Suppose (A5) holds. Let ξ be an \mathcal{F}_T -measurable random variable and $f : [0, T] \times \Omega \mapsto \mathbb{R}$ an $(\mathcal{F}_t)_{t \geq 0}$ -predictable process. Let A be the set of all points $x \in \mathcal{N}^c$ for which the following integrability condition holds

$$E^{x}(|\xi| + \int_{0}^{T} |f(s,\omega)|ds)^{2} < \infty.$$

Then there exists a pair $(Y_t, Z_t)_{0 \le t \le T}$ of predictable processes $Y : [0, T) \times \Omega \mapsto \mathbb{R}, Z : [0, T) \times \Omega \mapsto \mathbb{R}^d$, such that under all measures $P^x, x \in A$, they have the following properties:

- (i) Y is continuous,
- (ii) Z satisfy the integrability condition

$$\int_0^T |Z_t \sigma(X_t)|^2 dt < \infty, \qquad P^x - a.s.,$$

(iii) The local martingale $\int_0^t Z_s dM_s$, obtained by integrating Z against the coordi-

nate martingales, is a uniformly integrable martingale,

(iv) they satisfy the equation

$$Y_t = \xi + \int_t^T f(s,\omega)ds - \int_t^T Z_s dM_s, \qquad P^x - a.s., 0 \le t \le T.$$

If another pair (Y'_t, Z'_t) of predictable processes satisfies the above conditions (i),(ii),(iii),(iv), under a certain measure P^{ν} with the initial distribution ν supported by A, then one has $Y_{\cdot} = Y'_{\cdot}, P^{\nu} - a.s.$ and $Z_t \sigma(X_t) = Z'_t \sigma(X_t), dt \times P^{\nu} - a.s.$

Proof The representation of the positive and negative parts of the random variable $\xi + \int_0^T f_s ds$ give us the predictable process Z such that

$$\xi + \int_0^T f_s ds = E^{X_0}(\xi + \int_0^T f_s ds) + \int_0^T Z_s dM_s.$$

Then we get the process Y by the formula

$$Y_t = E^{X_0}(\xi + \int_0^T f_s ds) + \int_0^t Z_s dM_s - \int_0^t f_s ds.$$

Definition 2.4.2 Let ξ be an \mathbb{R}^l -valued, \mathcal{F}_T -measurable, random variable and $f : [0,T] \times \Omega \times \mathbb{R}^l \times \mathbb{R}^l \otimes \mathbb{R}^k \mapsto \mathbb{R}^l$ a measurable \mathbb{R}^l -valued function such that $(s,\omega) \mapsto f(s,\omega,\cdot,\cdot)$ as a process is predictable. Let p > 1 and ν be a probability measure supported by \mathcal{N}^c such that $E^{\nu}|\xi|^p < \infty$. We say that a pair $(Y_t, Z_t)_{0 \le t \le T}$ of predictable processes $Y : [0,T) \times \Omega \mapsto \mathbb{R}^l$, $Z : [0,T) \times \Omega \mapsto \mathbb{R}^l \otimes \mathbb{R}^d$ is a solution of the BSDE (2.4.1) in $L^p(P^{\nu})$ with data (ξ, f) provided that Y is continuous under P^{ν} and satisfies both the integrability conditions

$$\int_0^T |f(t, \cdot, Y_t, Z_t \sigma(X_t))| dt < \infty, \qquad P^\nu - a.s.,$$
$$E^\nu (\int_0^T |Z_t \sigma(X_t)|^2 dt)^{p/2} < \infty,$$

and the following equation, with $0 \le t \le T$,

(2.4.1)
$$Y_t = \xi + \int_t^T f(s, \omega, Y_s, Z_s \sigma(X_s)) ds - \int_t^T Z_s dM_s, \qquad P^{\nu} - a.s..$$

Let $f: [0,T] \times \Omega \times \mathbb{R}^l \times \mathbb{R}^l \otimes \mathbb{R}^k \mapsto \mathbb{R}^l$ be a measurable \mathbb{R}^l -valued function such that $(s, \omega) \mapsto f(s, \omega, \cdot, \cdot)$ is predictable and satisfies the following conditions:

(\Omega1) (Lipschitz condition in z) There exists a constant C > 0 such that for all t, ω, y, z, z'

$$|f(t,\omega,y,z) - f(t,\omega,y,z')| \le C|z-z'|.$$

($\Omega 2$) (Monotonicity condition in y) There exists a function $\mu_t \in L^1([0,T],\mathbb{R})$ such that for all ω, y, y', z ,

$$\langle y-y', f(t,\omega,y,z) - f(t,\omega,y',z) \rangle \leq \mu_t |y-y'|^2,$$

and $\alpha_t := \int_0^t \mu_s ds < \infty$.

(Ω 3) (Continuity condition in y) For t, ω and z fixed, the map

$$y \mapsto f(t, \omega, y, z),$$

is continuous.

We need the following notation

$$f^{0}(t,\omega) := f(t,\omega,0,0), \qquad f'(t,\omega,y) := f(t,\omega,y,0) - f(t,\omega,0,0),$$
$$f^{',r}(t,\omega) := \sup_{|y| \le r} |f'(t,\omega,y)|.$$

Let ξ be an \mathbb{R}^l -valued, \mathcal{F}_T -measurable, random variable and, for each p > 0 denote by A_p the set of all points $x \in \mathcal{N}^c$ for which the following integrability conditions hold,

(2.4.2)
$$E^x \int_0^T f_t^{\prime,r} dt < \infty, \qquad \forall r \ge 0,$$
$$E^x (|\xi|^p + (\int_0^T |f^0(s,\omega)| ds)^p) < \infty.$$

Denote by A_{∞} the set of points $x \in \mathcal{N}^c$ for which (2.4.2) holds and with the property that $|\xi|, |f^0| \in L^{\infty}(P^x)$.

Proposition 2.4.3 Under the conditions (A5), $(\Omega 1), (\Omega 2), (\Omega 3)$, there exists a pair $(Y_t, Z_t)_{0 \le t \le T}$ of predictable processes $Y : [0, T) \times \Omega \mapsto \mathbb{R}^l, Z : [0, T) \times \Omega \mapsto \mathbb{R}^l \otimes \mathbb{R}^d$ that forms a solution of the BSDE (2.4.1) in $L^p(P^x)$ with data (ξ, f) for each point $x \in A_p$. Moreover, the following estimate holds with some constant K that depends only on C, μ and T,

$$E^{x}(\sup_{t\in[0,T]}|Y_{t}|^{p}+(\int_{0}^{T}|Z_{t}\sigma(X_{t})|^{2}dt)^{p/2}) \leq KE^{x}(|\xi|^{p}+(\int_{0}^{T}|f^{0}(s,\omega)|ds)^{p}), \qquad x\in A_{p}.$$

If $x \in A_{\infty}$, then $\sup_{t \in [0,T]} |Y_t| \in L^{\infty}(P^x)$.

If (Y'_t, Z'_t) is another solution in $L^p(P^x)$, for some point $x \in A_p$, then one has $Y_t = Y'_t$ and $Z_t \sigma(X_t) = Z'_t \sigma(X_t), dt \times P^x - a.s.$

The proof is based on more or less standard methods. Therefore, we include it not here, but in the appendix below.

We shall now look at the connection between the solutions of BSDE's introduced in this section and PDE's studied in Section 2.2. In order to do this we have to consider BSDE's over time intervals like [s, T], with $0 \le s \le T$. Since the present approach is based on the theory of Markov processes, which is a time homogeneous theory, we have to discuss solutions over the interval [s, T], while the process and the coordinate martingales are indexed by a parameter in the interval [0, T - s].

Let us give a formal definition for the natural notion of solution over a time interval [s, T]. Let ξ be an \mathcal{F}_{T-s} -measurable, \mathbb{R}^{l} -valued, random variable and f: $[s, T] \times \Omega \times \mathbb{R}^{l} \times \mathbb{R}^{l} \otimes \mathbb{R}^{k} \to \mathbb{R}^{l}$ an \mathbb{R}^{l} -valued, measurable map such that $(f(s + l, \omega, \cdot, \cdot))_{l \in [0, T-s]}$ is predictable with respect to $(\mathcal{F}_{l})_{l \in [0, T-s]}$. Let ν be a probability measure supported by \mathcal{N}^{c} such that $E^{\nu}|\xi|^{p} < \infty$. We say a pair $(Y_{t}, Z_{t})_{s \leq t \leq T}$ of processes $Y : [s, T] \times \Omega \to \mathbb{R}^{l}, Z : [s, T] \times \Omega \to \mathbb{R}^{l} \otimes \mathbb{R}^{d}$ is a solution in $L^{p}(P^{\nu})$ of the BSDE (2.4.3) over the interval [s, T] with data (ξ, f) , provided that they have the property that reindexed as $(Y_{s+l}, Z_{s+l})_{l \in [0, T-s]}$ these processes are $(\mathcal{F}_{l})_{l \in [0, T-s]}$ predictable, Y is continuous and together they satisfy the integrability conditions

$$\int_{s}^{T} |f(t,\cdot,Y_{t},Z_{t}\sigma(X_{t-s}))|dt < \infty, \qquad P^{\nu} - a.s..$$
$$E^{\nu} (\int_{s}^{T} |Z_{t}\sigma(X_{t-s})|^{2} dt)^{p/2} < \infty.$$

and the following equation under P^{ν} ,

(2.4.3)
$$Y_t = \xi + \int_t^T f(r, Y_r, Z_r \sigma(X_{r-s})) dr - \int_{t-s}^{T-s} Z_{s+l} dM_l, \qquad s \le t \le T.$$

The next result gives a probabilistic interpretation of Theorem 2.2.8. Let us assume that $f: [0,T] \times \mathbb{R}^d \times \mathbb{R}^l \times \mathbb{R}^l \otimes \mathbb{R}^k \to \mathbb{R}^l$ is the measurable function appearing in the basic equation (2.2.1). Let $\phi: \mathbb{R}^d \to \mathbb{R}^l$ be measurable and for each p > 1, denote by A_p the set of points $(s,x) \in [0,T) \times \mathcal{N}^c$ with the following properties

(2.4.4)
$$E^x \int_s^T f'^{,r}(t, X_{t-s}) dt < \infty, \qquad \forall r \ge 0.$$

$$E^{x}(|\phi|^{p}(X_{T-s}) + (\int_{s}^{T} |f^{0}(t, X_{t-s})|ds)^{p}) < \infty.$$

Set $A = \bigcup_{p>1} A_p$, $A_{p,s} = \{x \in \mathcal{N}^c, (s, x) \in A_p\}$, and $A_s = \bigcup_{p>1} A_{p,s}$, $s \in [0, T)$. By the same arguments as in [BPS05, Theorem 5.4], we have the following results. In particular, we can reconstruct solutions to PDE (2.2.1) using Proposition 2.4.3.

Theorem 2.4.4 Assume that (A5) holds and f satisfies conditions (H1),(H2),(H3). Then there exist nearly Borel measurable functions $(u, \psi), u : A \to \mathbb{R}^l, \psi : A \to \mathbb{R}^l \otimes \mathbb{R}^d$, such that, for each $s \in [0, T)$ and each $x \in A_{p,s}$, the pair $(u(t, X_{t-s}), \psi(t, X_{t-s}))_{s \leq t \leq T}$ solves the BSDE (2.4.3) in $L^p(P^x)$ with data $(\phi(X_{T-s}), f(t, X_{t-s}, y, z))$ over the interval [s, T].

In particular, the functions u, ψ satisfy the following estimates, for $(s, x) \in A_p$,

$$E^{x}(\sup_{t\in[s,T]}|u(t,X_{t-s})|^{p} + (\int_{s}^{T}|\psi\sigma(t,X_{t-s})|^{2}dt)^{p/2}) \leq KE^{x}(|\phi(X_{T-s})|^{p} + (\int_{s}^{T}|f^{0}(t,X_{t-s})|dt)^{p}).$$

Moreover, suppose (A1)-(A4) hold, and the conditions in Theorem 2.2.11 hold when m(dx) = dx. If f and ϕ satisfy the conditions (H4) and (H5) then the complement of $A_{2,s}$ is *m*-negligible (i.e. $m(A_{2,s}^c) = 0$) for each $s \in [0, T)$, the class of $u1_{A_2}$ is an element of \hat{F}^l which is a generalized solution of PDE (2.2.1), $\psi\sigma$ represents a version of $D_{\sigma}u$ and the following relations hold for each $(s, x) \in A$ and $1 \leq i \leq l$,

(2.4.5)
$$u^{i}(s,x) = E^{x}(\phi^{i}(X_{T-s})) + \int_{s}^{T} E^{x} f^{i}(t, X_{t-s}, u(t, X_{t-s}), D_{\sigma}u(t, X_{t-s})) dt.$$

Proof We will assume that ϕ and f^0 are bounded; the general case is then obtained by approximation. Then the sets $A_p, p > 0$, are all equal. We construct the functions (u, ψ) on A as follows. For $s \in [0, T)$, denote by $(Y_t^s, Z_t^s)_{s \le t \le T}$ the solution in Proposition 2.4.3, of the BSDE (2.4.3) over the interval [s, T], in $L^2(P^x), x \in A_s$ with data $(\phi(X_{T-s}(\omega)), f(t, X_{t-s}(\omega), y, z))$. Since $X_r \in A_{s+r}, P^x$ -a.s., by the uniqueness part of that proposition one deduces that

$$Y_t^{s+r} \circ \theta_r = Y_t^s, \qquad t \in [s+r,T), P^x - a.s.,$$
$$(Z_t^{s+r} \sigma(X_{t-s-r})) \circ \theta_r = Z_t^s \sigma(X_{t-s}), \qquad dt \times P^x - a.s$$

for each fixed $r \in [0, T - s)$ and all measures $P^x, x \in A_s$. In particular, if we define

$$u(s,x) := E^x(Y^s_s),$$

we will have, for any $x \in A_s$,

$$u(t, X_{t-s}) = E^{X_{t-s}}(Y_t^t) = E^x(Y_t^t \circ \theta_{t-s} | \mathcal{F}_{t-s}) = Y_t^s \qquad P^x - a.s..$$

Set $W_l(s,\omega) := Z_{l+s}^s \sigma(X_l)(\omega)$, for $(s,\omega) \in [0,T) \times \Omega$ and $l \in [0,T-s)$. One has $W_l(r+s,\theta_r(\omega)) = W_{l+r}(s,\omega), dl \times P^x - a.s.$ In terms of the time-space Markov process \hat{X} (see e.g. [BPS05, Section 4.2]), we have $W_l(\hat{\theta}_r(s,\omega)) = W_{l+r}(s,\omega)$. Therefore, $t \to U_{j,t}^i(s,\omega) = \int_0^{t\wedge T} W_{j,l}^i(s,\omega) dl$, with $1 \leq i \leq l$ and $1 \leq j \leq k$, represents an additive functional for the time-space process \hat{X} . By [Sh88, Theorem 66.2] we deduce that there exists a nearly Borel measurable function $\tilde{\psi}_j^i : [0,T) \times \mathbb{R}^d \to R$, such that $\tilde{\psi}_j^i(t, X_{t-s}(\omega)) = W_{j,t-s}^i(s,\omega), dt \times P^x$ -a.s. for each $x \in A_s$. Define

$$\psi := \tilde{\psi} \tau$$

Then we have $Z_t^s \sigma(X_{t-s}) = \psi \sigma(t, X_{t-s}) dt \times P^x - a.s., \forall x \in A_s.$ Now we have

$$u^{i}(s,x) = E^{x}(\phi^{i}(X_{T-s})) + \int_{s}^{T} E^{x} f^{i}(t, X_{t-s}, u(t, X_{t-s}), \psi\sigma(t, X_{t-s})) dt,$$

since ϕ and f^0 are bounded. In particular, we have that $t \to u(t, X_{t-s})$ is continuous P^x -a.s. for each $x \in A_s$, because u may be written as the difference of two \hat{X} -excessive functions with regular potential part (cf. [BG68]). This implies $u(\cdot, X_{\cdot-s}) = Y^s$. $u(\cdot, X_{\cdot-s}), \psi(\cdot, X_{\cdot-s})$ solves the BSDE (2.4.3) in $L^p(P^x)$ over the time interval [s, T]. By Theorem 2.3.12, we have that u is a generalized solution of (2.2.1) and that $\psi\sigma$ represents a version of $D_{\sigma}u$.

Remark 2.4.5 In the above theorem, we need the analytic results, i.e. the existence of a generalized solution of nonlinear equation (2.2.1), to obtain the above results. In the following example, we drop the conditions (A1)-(A4), in particular, we don't need $|b\sigma| \in L^2(\mathbb{R}^d; m)$ and use the results that the existence of the solution of BSDE (2.4.3) to obtain the existence of a generalized solution of nonlinear equation (2.2.1), which is not covered by our analytic results in Section 2.2.

Example 2.4.6 Consider $d \geq 2$, $A = (a^{ij})$ a Borel-measurable mapping on \mathbb{R}^d with values in the non-negative symmetric matrices on \mathbb{R}^d , and let $b = (b^i) : \mathbb{R}^d \to \mathbb{R}^d$ be a Borel-measurable vector field. Consider the operator

$$L_{A,b}\psi = a^{ij}\partial_i\partial_j\psi + b^i\partial_i\psi, \qquad \forall \psi \in C_0^\infty(\mathbb{R}^d),$$

where we use the standard summation rule for repeated indices. By $H^{1,p}(\mathbb{R}^d, dx)$ we denote the standard Sobolev space of functions on \mathbb{R}^d whose first order derivatives are in $L^p(\mathbb{R}^d, dx)$. Assume that for p > d

$$(C1)a^{ij} = a^{ji} \in H^{1,p}_{\text{loc}}(\mathbb{R}^d, dx), 1 \le i, j \le d.$$

$$(C2)b^i \in L^p_{\text{loc}}(\mathbb{R}^d, dx).$$

(C3) for all V relatively compact in \mathbb{R}^d there exist $\nu_V > 0$ such that

$$u_V^{-1}|h|^2 \leq \langle ha,h \rangle \leq \nu_V |h|^2 \text{ for all } h \in \mathbb{R}^d, x \in V$$

Here by $H^{1,p}_{loc}(\mathbb{R}^d, dx)$ we denote the class of all functions f on \mathbb{R}^d such that $f\chi \in H^{1,p}(\mathbb{R}^d, dx)$ for all $\chi \in C^{\infty}_0(\mathbb{R}^d)$. And $L^p_{loc}(\mathbb{R}^d, dx)$ denotes the class of all functions f on \mathbb{R}^d such that $f\chi \in L^p(\mathbb{R}^d)$ for all $\chi \in C^{\infty}_0(\mathbb{R}^d)$. Assume that there exists $V \in C^2(\mathbb{R}^d)$ ("Lyapunov function") such that

$$\lim_{|x|\to\infty} V(x) = +\infty, \qquad \lim_{|x|\to\infty} L_{A,b}V(x) = -\infty.$$

Examples of V can be found in [BRS00] and the reference therein.

Then by [BRS00, Theorem 2.2] there exists a probability measure μ on \mathbb{R}^d such that

$$\int_{\mathbb{R}^d} L_{A,b} \psi d\mu = 0 \qquad \forall \psi \in C_0^\infty(\mathbb{R}^d).$$

Then by [BRS00, Theorem 2.1] we have $d\mu$ is absolutely continuous w.r.t. dxand that the density admits a representation φ^2 , where $\varphi^2 \in H^{1,p}_{loc}(\mathbb{R}^d, dx)$. The closure of

$$\mathcal{E}^{0}(u,v) = \frac{1}{2} \int \langle \nabla ua, \nabla v \rangle d\mu; \ u, v \in C_{0}^{\infty}(\mathbb{R}^{d}),$$

on $L^2(\mathbb{R}^d, \mu)$ is a Dirichlet form.

Set $b^0 = (b_1^0, ..., b_d^0)$, where $b_i^0 := \sum_{j=1}^d (\partial_j a_{ij} + 2a^{ij}\partial_j \varphi/\varphi)$, i = 1, ..., d, and $\beta := b - b^0$. Then, $\beta \in L^2_{\text{loc}}(\mathbb{R}^d; \mathbb{R}^d, \mu)$. By [St1, Proposition 1.10 and Proposition 2.4] $(L, C_0^\infty(\mathbb{R}^d))$ is L^1 -unique. By the proof of [St1, Proposition 2.4] we conclude that for $u \in b\mathcal{F}$ there exists a sequence $\{u_n\} \subset C_0^\infty(\mathbb{R}^d)$ such that $\mathcal{E}(u_n - u, u_n - u) \to 0, n \to \infty$.

Consider the bilinear form

$$\mathcal{E}(u,v) = \frac{1}{2} \int \langle \nabla ua, \nabla v \rangle d\mu - \int \langle \frac{1}{2}\beta, \nabla u \rangle v d\mu \ u, v \in C_0^{\infty}(\mathbb{R}^d).$$

Then by the computation in [Tr2, Section 4d] we have that conditions (A5) hold for the bilinear form \mathcal{E} . Then we can use the first part of Theorem 2.4.4 to obtain the following results.

Theorem 2.4.7 Consider the bilinear form obtained in Example 2.4.6. If f satis-

fies conditions (H1),(H2),(H3). Then there exist nearly Borel measurable functions $(u, \psi), u : A \to \mathbb{R}^l, \psi : A \to \mathbb{R}^l \otimes \mathbb{R}^d$, such that, for each $s \in [0, T)$ and each $x \in A_{p,s}$, the pair $(u(t, X_{t-s}), \psi(t, X_{t-s}))_{s \le t \le T}$ solves the BSDE (2.4.3) in $L^p(P^x)$ with data $(\phi(X_{T-s}), f(t, X_{t-s}, y, z))$ over the interval [s, T].

In particular, the functions u, ψ satisfy the following estimates, for $(s, x) \in A_p$,

$$E^{x}(\sup_{t\in[s,T]}|u(t,X_{t-s})|^{p} + (\int_{s}^{T}|\psi\sigma(t,X_{t-s})|^{2}dt)^{p/2}) \leq KE^{x}(|\phi(X_{T-s})|^{p} + (\int_{s}^{T}|f^{0}(t,X_{t-s})|dt)^{p}).$$

Moreover, suppose f and ϕ satisfy the conditions (H4) and (H5) then the complement of $A_{2,s}$ is μ -negligible (i.e. $\mu(A_{2,s}^c) = 0$) for each $s \in [0, T)$, the class of $u_{1_{A_2}}$ is an element of \hat{F}^l which is the unique generalized solution of (2.2.1), $\psi\sigma$ represents a version of $D_{\sigma}u$ and the following relations hold for each $(s, x) \in A$ and $1 \leq i \leq l$,

$$u^{i}(s,x) = E^{x}(\phi^{i}(X_{T-s})) + \int_{s}^{T} E^{x} f^{i}(t, X_{t-s}, u(t, X_{t-s}), D_{\sigma}u(t, X_{t-s})) dt.$$

Proof By [St1, Lemma 3.1] we have that for $u \in D(L_{A,b})$, $u \in D(\mathcal{E}^0)$ and $\mathcal{E}^0(u, u) \leq -\int Luud\mu$. Hence the first part of proof in Proposition 2.1.9 hold in this case i.e. the mild solution is equivalent to the generalized solution and (2.1.7), (2.1.8) hold. Hence, the uniqueness of the solution (2.2.1) follows by the same arguments as the uniqueness proof of Theorem 2.2.8. Moreover, the results in Theorem 2.3.12 hold. By the same arguments as in the proof of Theorem 2.4.4 the assertion follows. \Box

2.5 Further Examples

The following two examples discuss the case where PDE satisfies some boundary conditions.

Example 2.5.1 . Let $D \subset \mathbb{R}^d$ be a bounded domain. We choose $m(dx) = 1_D(x)dx$. If \mathcal{E} is a sectorial Dirichlet form, it is associated to a reflecting diffusion X in the state space \overline{D} . Then by Theorem 2.2.8 there exists a solution to the non-linear parabolic equation

$$\begin{aligned} (\partial_t + L)u + f(t, x, u, D_\sigma u) &= 0, & 0 \le t \le T, \\ u_T(x) &= \phi(x), & x \in \mathbb{R}^d, \\ \frac{\partial u(t, \cdot)}{\partial \nu}|_{\partial D} &= 0, & t > 0, \end{aligned}$$

where $\frac{\partial}{\partial \nu}$ denotes the normal derivative. Then Theorem 2.4.4 provides a probabilistic

interpretation for this equation.

Example 2.5.2 . Let $D \subset \mathbb{R}^d$ be a bounded domain satisfying the cone condition. We choose $m(dx) = 1_D(x)dx$ and replace $C_0^{\infty}(\mathbb{R}^d)$ by $C_0^{\infty}(D)$. Then the results in Theorem 2.2.8 apply and there exists a solution $u_1 \in F = H_0^1(D)$ to the following non-linear parabolic equation:

$$(\partial_t + L)u + f(t, x, u, D_\sigma u) = 0, \qquad 0 \le t \le T,$$
$$u_T(x) = \phi(x), \qquad x \in \mathbb{R}^d.$$

Assume \mathcal{E} satisfies the weak sector condition. Let X^0 denote the diffusion associated with \mathcal{E}^R , where \mathcal{E}^R denotes the Dirichlet form which has the same form as \mathcal{E} with the reference measure m(dx) replaced by dx. Then define

$$X_t := \begin{cases} X_t^0, & \text{if } t < \tau, \\ \Delta & \text{otherwise,} \end{cases}$$

where $\tau = \inf\{t \ge 0, X_t^0 \in D^c \cup \Delta\}$. Assume (A5) holds for X^0 . We use Theorem 2.4.4 for X^0 with the data $(\phi(X_{T-s}^0)1_{\{T-s<\tau\}}, 1_{[0,\tau+s]}(r)f(r, X_{r-s}^0, Y_r, Z_r\sigma(X_{r-s}^0)))$. Then there exist nearly Borel measurable functions $(u, \psi), u : A \to \mathbb{R}^l, \psi : A \to \mathbb{R}^l \otimes \mathbb{R}^d$, such that, for each $s \in [0, T)$ and each $x \in A_{p,s}$, the pair $(u(t, X_{t-s}^0), \psi(t, X_{t-s}^0))_{s \le t \le T}$ solves the BSDE

$$Y_t = \phi(X_{T-s}^0) \mathbf{1}_{\{T-s<\tau\}} + \int_{t\wedge(\tau+s)}^{T\wedge(\tau+s)} f(r, X_{r-s}^0, Y_r, Z_r \sigma(X_{r-s}^0)) dr - \int_{t-s}^{T-s} Z_{s+l} dM_l, \qquad s \le t \le T$$

Then by [Pa99, Proposition 2.6] we have

$$Y_t = 0, Z_t = 0$$
 when $t \in [\tau + s, T],$

and the pair $(u(t, X_{t-s}), \psi(t, X_{t-s}))_{s \le t \le T}$ solves the BSDE

$$Y_t = \phi(X_{T-s}) + \int_t^T f(r, X_{r-s}, Y_r, Z_r \sigma(X_{r-s})) dr - \int_{t-s}^{T-s} Z_{s+l} dM_l, \quad s \le t \le T.$$

In particular, the functions u, ψ satisfy the following estimates, for $(s, x) \in A_p$,

$$E^{x}(\sup_{t\in[s,T]}|u(t,X_{t-s})|^{p} + (\int_{s}^{T}|\psi\sigma(t,X_{t-s})|^{2}dt)^{p/2}) \leq KE^{x}(|\phi(X_{T-s})|^{p} + (\int_{s}^{T}|f^{0}(t,X_{t-s})|dt)^{p})$$

The class of $u 1_{A_2}$ is an element in \hat{F}^l which is an *m*-version of $u_1, \psi \sigma$ represents a

version of $D_{\sigma}u$ and the following relations hold for each $(s, x) \in A$ and $1 \le i \le l$, (2.5.1)

$$u^{i}(s,x) = E^{x}(\phi^{i}(X_{T-s})) + \int_{s}^{T} E^{x} f^{i}(t, X_{t-s}, u(t, X_{t-s}), \psi(t, X_{t-s})\sigma(X_{t-s})) dt.$$

2.6 Appendix

2.6.1 Appendix A. Proof of Theorem 2.2.8

[Uniqueness]

Let u_1 and u_2 be two solutions of equation (2.2.1). By using (2.1.7) for the difference $u_1 - u_2$ we get

$$\begin{split} \|u_{1,t} - u_{2,t}\|_{2}^{2} &+ 2\int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{1,s} - u_{2,s})ds \\ &\leq 2\int_{t}^{T} (f(s,\cdot,u_{1,s},D_{\sigma}u_{1,s}) - f(s,\cdot,u_{2,s},D_{\sigma}u_{2,s}), u_{1,s} - u_{2,s})ds + 2\alpha\int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds \\ &\leq 2\int_{t}^{T} C(|D_{\sigma}u_{1,s} - D_{\sigma}u_{2,s}|, |u_{1,s} - u_{2,s}|)ds + 2\alpha\int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds \\ &\leq (\frac{C^{2}}{c_{1}} + c_{2} + 2\alpha)\int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds + \int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{1,s} - u_{2,s})ds. \end{split}$$

By Gronwall's lemma it follows that

$$||u_{1,t} - u_{2,t}||_2^2 = 0,$$

hence $u_1 = u_2$.

[Existence] The existence will be proved in four steps.

Step 1: Suppose there exists $r \in \mathbb{R}$ such that

$$r \ge 1 + K(\|\phi\|_{\infty} + \|f^0\|_{\infty} + \|f'^{,1}\|_{\infty}),$$

where K is the constant appearing in Lemma 2.2.7 (2.2.5), and f is uniformly bounded on the set

$$A_r = [0, T] \times \mathbb{R}^d \times \{ |y| \le r \} \times \mathbb{R}^l \otimes \mathbb{R}^k.$$

Define

$$M := \sup\{|f(t, x, y, z)| : (t, x, y, z) \in A_r\} < \infty.$$

Next we regularize f with respect to the variable y by convolution

$$f_n(t, x, y, z) = n^l \int_{\mathbb{R}^l} f(t, x, y', z) \varphi(n(y - y')) dy',$$

where φ is a smooth nonnegative function with support contained in the ball $\{|y| \leq 1\}$ such that $\int \varphi = 1$. Then $f = \lim_{n \to \infty} f_n$ and for each n, $\partial_{y_i} f_n$ are uniformly bounded on A_{r-1} . Set

$$h_n(t, x, y, z) := f_n(t, x, \frac{r-1}{|y| \lor (r-1)}y, z).$$

Then each h_n satisfies the Lipschitz condition with respect to both y and z. Thus by Proposition 2.2.4 each h_n determines a solution $u_n \in \hat{F}^l$ of (2.2.1) with data (ϕ, h_n) . By the same arguments as in [S09, Theorem 4.19], we have that h_n satisfies conditions (H1) and (H2') with the same constants $(C > 0 \text{ and } \mu = 0)$. As m is a finite measure and $f'^{,1} \in L^{\infty}([0, T] \times \mathbb{R}^d)$, we have $f'^{,1} \in L^2([0, T]; L^2)$. Since

$$|h_n(t, x, 0, 0) = |f_n(t, x, 0, 0)|$$

$$\leq n^l \int_{R^l} |f(t, x, y') - f^0(t, x) + f^0(t, x)| |\varphi(n(-y'))| dy'$$

$$\leq |f^0(t, x)| + f'^{,1}(t, x),$$

one deduces from Lemma 2.2.7 that $||u_n||_{\infty} \leq r-1$ and $||u_n||_T \leq K_T$. Since $h_n = f_n$ on A_{r-1} , it follows that u_n satisfies (2.2.1) with data (ϕ, f_n) .

Now for b > 0, set

$$d_{n,b}(t,x) := \sup_{|y| \le r-1, |z| \le b} |f(t,x,y,z) - f_n(t,x,y,z)|.$$

Obviously one has $|d_{n,b}| \leq 2M$. Moreover, on account of the *y*-continuity and of the uniform *z*-continuity, one sees that for fixed t, x, b, the family of functions

$$\{f(t, x, \cdot, z) ||z| \le b\},\$$

is equicontinuous and then compact in $C(\{|y| \leq r-1\})$. Since the convolution operators approach the identity uniformly on such a compact set, we get

$$\lim_{n \to \infty} d_{n,b}(t, x) = 0,$$

which implies $\lim_{n\to\infty} d_{n,b}(t,x) = 0$ in $L^2(dt \times m)$ because of our assumption that

 $m(\mathbb{R}^d) < \infty$. Moreover, for $u \in \hat{F}^l, |u| \le r - 1$

$$|f(u, D_{\sigma}u) - f_n(u, D_{\sigma}u)| \leq 1_{\{|D_{\sigma}u| \leq b\}} d_{n,b} + 2M 1_{\{|D_{\sigma}u| > b\}}$$
$$\leq d_{n,b} + \frac{2M}{b} |D_{\sigma}u|.$$

Next we will show that $(u_n)_{n\in\mathbb{N}}$ is a $\|\cdot\|_T$ -Cauchy sequence. By (2.1.7) for the difference $u_l - u_n$, we have

$$\begin{split} \|u_{l,t} - u_{n,t}\|_{2}^{2} &+ 2\int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{l,s} - u_{n,s})ds \\ \leq & 2\int_{t}^{T} (f_{t}(s, \cdot, u_{l,s}, D_{\sigma}u_{l,s}) - f_{n}(s, \cdot, u_{n,s}, D_{\sigma}u_{n,s}), u_{l,s} - u_{n,s})ds + 2\alpha \int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds \\ \leq & 2\int_{t}^{T} (|f_{t}(s, \cdot, u_{l,s}, D_{\sigma}u_{l,s}) - f(s, \cdot, u_{l,s}, D_{\sigma}u_{l,s})|, |u_{l,s} - u_{n,s}|)ds \\ &+ 2\int_{t}^{T} (|f_{n}(s, \cdot, u_{n,s}, D_{\sigma}u_{n,s}) - f(s, \cdot, u_{n,s}, D_{\sigma}u_{n,s})|, |u_{l,s} - u_{n,s}|)ds \\ &+ 2\int_{t}^{T} (|f(s, \cdot, u_{l,s}, D_{\sigma}u_{l,s}) - f(s, \cdot, u_{l,s}, D_{\sigma}u_{n,s})|, |u_{l,s} - u_{n,s}|)ds \\ &+ 2\int_{t}^{T} (f(s, \cdot, u_{l,s}, D_{\sigma}u_{l,s}) - f(s, \cdot, u_{l,s}, D_{\sigma}u_{n,s}), u_{l,s} - u_{n,s}|)ds \\ &+ 2\int_{t}^{T} (f(s, \cdot, u_{l,s}, D_{\sigma}u_{l,s}) - f(s, \cdot, u_{l,s}, D_{\sigma}u_{n,s}), u_{l,s} - u_{n,s}|)ds \\ &+ 2\int_{t}^{T} (|f_{n}(s, \cdot, u_{l,s}, D_{\sigma}u_{l,s}) - f(s, \cdot, u_{l,s}, D_{\sigma}u_{n,s})|, |u_{l,s} - u_{n,s}|)ds \\ &+ 2\int_{t}^{T} C(|D_{\sigma}u_{l,s} - D_{\sigma}u_{n,s}|, |u_{l,s} - u_{n,s}|)ds + 2\alpha\int_{t}^{T} ||u_{l,s} - u_{n,s}||^{2}ds \\ &\leq 2\int_{t}^{T} C(|D_{\sigma}u_{l,s} - D_{\sigma}u_{n,s}|, |u_{l,s} - u_{n,s}|)ds + 2\int_{t}^{T} \frac{2M}{b}(|D_{\sigma}u_{l,s}| + |D_{\sigma}u_{n,s}|, |u_{l,s} - u_{n,s}|)ds \\ &+ 2\int_{t}^{T} C(|D_{\sigma}u_{l,s} - D_{\sigma}u_{n,s}|, |u_{l,s} - u_{n,s}|)ds + 2\alpha\int_{t}^{T} ||u_{l,s} - u_{n,s}||^{2}ds \\ &\leq \int_{t}^{T} ||d_{l,b}(s, \cdot)||^{2}_{2}ds + \int_{t}^{T} ||d_{n,b}(s, \cdot)|^{2}_{2}ds + \frac{1}{b^{2}}\int_{t}^{T} (|D_{\sigma}u_{l,s}||^{2}_{2} + |D_{\sigma}u_{n,s}||^{2}_{2})ds \\ &+ (1 + 4M^{2} + \frac{C^{2}}{c_{1}} + 2\alpha + c_{2})\int_{t}^{T} ||u_{l,s} - u_{n,s}||^{2}_{2}ds + \int_{t}^{T} \mathcal{E}^{a,b}(u_{l,s} - u_{n,s})ds. \end{aligned}$$

Since $||u_n||_T \leq K_T$, we have

$$\int_0^T \|D_{\sigma} u_{l,s}\|_2^2 ds \le \frac{K_T}{c_1},$$

where the K_T is independent of l and b. Thus, for b, l, n large enough, for arbitrary $\varepsilon > 0$ we get

$$\|u_{l,t} - u_{n,t}\|_{2}^{2} + \int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{l,s} - u_{n,s}) ds \leq \varepsilon + \tilde{K} \int_{t}^{T} \|u_{l,t} - u_{n,t}\|_{2}^{2} ds,$$

where \tilde{K} depends on C, M, μ, α . It is easy to see that Gronwall's lemma implies that $(u_n)_{n \in \mathbb{N}}$ is a Cauchy-sequence in \hat{F} . Define $u := \lim_{n \to \infty} u_n$ and take a subsequence $(n_k)_{k \in \mathbb{N}}$ such that $u_{n_k} \to u$ a.e. We have

$$f(\cdot, \cdot, u_{n_k}, D_\sigma u) \to f(\cdot, \cdot, u, D_\sigma u)$$
 in $L^2(dt \times m)$.

Since $||u_{n_k} - u||_T \to 0$, we obtain

$$||D_{\sigma}u - D_{\sigma}u_{n_k}||_{L^2(dt \times m)} \to 0.$$

Then by (H1), it follows that

$$\lim_{k \to \infty} \|f(\cdot, \cdot, u_{n_k}, D_{\sigma} u) - f(\cdot, \cdot, u_{n_k}, D_{\sigma} u_{n_k})\|_{L^2(dt \times m)}$$
$$\leq \lim_{k \to \infty} C \|D_{\sigma} u - D_{\sigma} u_{n_k}\|_{L^2(dt \times m)}$$
$$= 0.$$

We also have

$$\|f(\cdot, \cdot, u_{n_k}, D_{\sigma} u_{n_k}) - f_{n_k}(\cdot, \cdot, u_{n_k}, D_{\sigma} u_{n_k})\|_{L^2(dt \times m)}$$

$$\leq \|d_{n_k, b}\|_{L^2(dt \times m)} + \frac{2M}{b} \|D_{\sigma} u_{n_k}\|_{L^2(dt \times m)}.$$

Letting $k \to \infty$ and then $b \to \infty$ the above equality converges to zero. Finally, we conclude

$$\lim_{k \to \infty} \|f_{n_k}(u_{n_k}, D_{\sigma} u_{n_k}) - f(u, D_{\sigma} u)\|_{L^2(dt \times m)}$$

$$\leq \lim_{k \to \infty} \|f_{n_k}(u_{n_k}, D_{\sigma} u_{n_k}) - f(u_{n_k}, D_{\sigma} u_{n_k})\|_{L^2(dt \times m)}$$

$$+ \lim_{k \to \infty} \|f(u_{n_k}, D_{\sigma} u_{n_k}) - f(u_{n_k}, D_{\sigma} u)\|_{L^2(dt \times m)}$$

$$+ \lim_{k \to \infty} \|f(u_{n_k}, D_{\sigma} u) - f(u, D_{\sigma} u)\|_{L^2(dt \times m)}$$

$$= 0.$$

By passing to the limit in the mild equation associated to u_{n_k} with data (ϕ, f_{n_k}) , it follows that u is the solution associated to (ϕ, f) .

Step 2: In this step we will prove the assertion under the assumption that there

exists some constant r such that $f'^{,r}$ is uniformly bounded and

$$r \ge 1 + K(\|\phi\|_{\infty} + \|f^0\|_{\infty} + \|f'^{,1}\|_{\infty}),$$

where K is the constant appearing in Lemma 2.2.7 (2.2.5). Define

$$f_n(t, x, y, z) := f(t, x, y, \frac{n}{|z| \vee n} z).$$

 $f_n \leq Cn + ||f',r||_{\infty} + ||f^0||_{\infty}$ on A_r . Each of the functions f_n satisfies the same conditions as f and by Step 1, there exists a solution u_n associated to the data (ϕ, f_n) . One has $||u_n||_{\infty} \leq r - 1$, $||u_n||_T \leq K_T$. Conditions (H1) and (H2) yield

$$|(f_l(u_l, D_{\sigma}u_l) - f_n(u_n, D_{\sigma}u_n), u_l - u_n)| \le C(|D_{\sigma}u_l - D_{\sigma}u_n|, |u_l - u_n|) + |(f_l(u_n, D_{\sigma}u_n) - f_n(u_n, D_{\sigma}u_n), u_l - u_n)|.$$

Since $f_n(t, x, y, z) \mathbf{1}_{|z| \le n} = f(t, x, y, z) \mathbf{1}_{|z| \le n}$, and for $n \le l$, $|f_l - f_n| \mathbf{1}_{|z| \ge n} \le 2C|z| \mathbf{1}_{|z| \ge n}$, we have

$$|(f_l(u_n, D_{\sigma}u_n) - f_n(u_n, D_{\sigma}u_n), u_l - u_n)| \le |(2C|D_{\sigma}u_n|1_{\{|D_{\sigma}u_n| \ge n\}}, |u_l - u_n|)|$$

Then,

$$\begin{split} \|u_{l,t} - u_{n,t}\|_{2}^{2} &+ 2\int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{l,s} - u_{n,s})ds \\ \leq & 2\int_{t}^{T} (f_{l}(u_{l,s}, D_{\sigma}u_{l,s}) - f_{n}(u_{n,s}, D_{\sigma}u_{n,s}), u_{l,s} - u_{n,s})ds + 2\alpha \int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds \\ \leq & 2\int_{t}^{T} C(|D_{\sigma}u_{l} - D_{\sigma}u_{n}|, |u_{l} - u_{n}|)ds + 2\int_{t}^{T} |(2C|D_{\sigma}u_{n}|1_{\{|D_{\sigma}u_{n}| \geq n\}}, |u_{l} - u_{n}|)|ds \\ &+ 2\alpha \int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds \\ \leq & (\frac{C^{2}}{c_{1}} + 2\alpha + c_{2})\int_{t}^{T} \|u_{l} - u_{n}\|_{2}^{2}ds + \int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{l} - u_{n})ds \\ &+ 8C(r-1)\int_{t}^{T} \int |D_{\sigma}u_{n}|1_{\{|D_{\sigma}u_{n}| \geq n\}}dmds \\ \leq & (\frac{C^{2}}{c_{1}} + 2\alpha + c_{2})\int_{t}^{T} \|u_{l} - u_{n}\|_{2}^{2}ds + \int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{l} - u_{n})ds \\ &+ 8C(r-1)(\int_{t}^{T} \|1_{\{|D_{\sigma}u_{n}| \geq n\}}\|_{2}^{2}ds)^{\frac{1}{2}}(\int_{t}^{T} \|D_{\sigma}u_{n}\|_{2}^{2}ds)^{\frac{1}{2}}. \end{split}$$

As $||u_n||_T^2 \leq K_T$, we have $\int_0^T ||D_\sigma u_n||_2^2 ds \leq \frac{K_T}{c_1}$. Hence,

$$n^{2} \int_{t}^{T} \|1_{\{|D_{\sigma}u_{n}| \ge n\}}\|_{2}^{2} ds \le \int_{t}^{T} \|D_{\sigma}u_{n}1_{\{|D_{\sigma}u_{n}| \ge n\}}\|_{2}^{2} ds \le \frac{K_{T}}{c_{1}}.$$

Therefore, for n big enough

$$\|u_{l,t} - u_{n,t}\|_{2}^{2} + \int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{l,s} - u_{n,s}) ds \le \left(\frac{C^{2}}{c_{1}} + 2\alpha + c_{2}\right) \int_{t}^{T} \|u_{l} - u_{n}\|_{2}^{2} ds + \varepsilon.$$

By Gronwalls' lemma it follows that $(u_n)_{n\in\mathbb{N}}$ is a Cauchy sequence in \hat{F}^l . Hence, $u := \lim_{n\to\infty} u_n$ is well defined. We can find a subsequence such that $(u_{n_k}, D_{\sigma}u_{n_k}) \to (u, D_{\sigma}u)$ a.e. and conclude

$$|f_{n_k}(u_{n_k}, D_{\sigma}u_{n_k}) - f(u, D_{\sigma}u)| \le C |\frac{n_k}{|D_{\sigma}u_{n_k}| \vee n_k} D_{\sigma}u_{n_k} - D_{\sigma}u| + |f(u_{n_k}, D_{\sigma}u) - f(u, D_{\sigma}u)| \to 0.$$

Since

$$\begin{aligned} &|f_{n_k}(u_{n_k}, D_{\sigma} u_{n_k}) - f(u, D_{\sigma} u)| \\ \leq &|f(u, 0) - f(u, D_{\sigma} u)| + |f_{n_k}(u_{n_k}, D_{\sigma} u_{n_k}) - f_{n_k}(u_{n_k}, 0)| + |f_{n_k}(u_{n_k}, 0) - f^0| + |f^0 - f(u, 0)| \\ \leq &C(|D_{\sigma} u| + |D_{\sigma} u_{n_k}|) + 2f'^{,r}, \end{aligned}$$

we have

$$f_{n_k}(u_{n_k}, D_{\sigma}u_{n_k}) \to f(u, D_{\sigma}u) \text{ in } L^1([0, T], L^2).$$

We conclude that u is a solution of (2.2.1) associated to the data (ϕ, f) .

Step 3: Now we only suppose that $f'^{,1}$ is bounded. Hence, we can choose a constant r such that

$$r \ge 1 + K(\|\phi\|_{\infty} + \|f^0\|_{\infty} + \|f'^{,1}\|_{\infty}),$$

where K is the constant appearing in Lemma 2.2.7 (2.2.5). Let us define

$$f_n := \frac{n}{f'^{,r} \vee n} (f - f^0) + f^0.$$

Easily we see that the f_n have the same properties as f. Since $f_n(t, x, y, z) = f(t, x, y, z)$ for $f'^{,r} \leq n$, we have

$$\lim_{n \to \infty} f_n = f.$$

We introduce the following notation:

$$f'_n(t,x) := \sup_{|y| \le r} |f'_n(t,x,y)|$$
, and $f'_n(t,x,y) := f_n(t,x,y,0) - f^0(t,x)$.

By the same arguments as in [S09, Theorem 4.19] we have

$$|f_n',r| \le n \land |f',r|.$$

Hence, by Step 2 we obtain that there exists a solution u_n associated to the data (ϕ, f^n) such that $||u_n||_{\infty} \leq r-1, ||u_n||_T \leq M$, where M is a constant. For $n \leq l$, we obtain

$$|f_l - f_n| \le (C|z| + |f'|) |\frac{l}{f'^{,r} \lor l} - \frac{n}{f'^{,r} \lor n}| \le (C|z| + |f'|) \mathbb{1}_{\{f'^{,r} > n\}}.$$

Hence

$$\int_{t}^{T} |(f_{l}(u_{n}, D_{\sigma}u_{n}) - f_{n}(u_{n}, D_{\sigma}u_{n}), u_{l} - u_{n})|ds \leq 2(r-1) \int_{t}^{T} \int_{\{f', r > n\}} (C|D_{\sigma}u_{n}| + f'^{,r}) dm ds.$$

We obtain as in the preceding steps:

$$\begin{split} \|u_{l,t} - u_{n,t}\|_{2}^{2} &+ 2\int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{l,s} - u_{n,s})ds \\ &\leq 2\int_{t}^{T} (f_{l}(u_{l,s}, D_{\sigma}u_{l,s}) - f_{n}(u_{n,s}, D_{\sigma}u_{n,s}), u_{l,s} - u_{n,s})ds + 2\alpha \int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds \\ &\leq 2\int_{t}^{T} C(|D_{\sigma}u_{l} - D_{\sigma}u_{n}|, |u_{l} - u_{n}|)ds + 2\int_{t}^{T} |(f_{l}(u_{n}, D_{\sigma}u_{n}) - f_{n}(u_{n}, D_{\sigma}u_{n}), u_{l} - u_{n})|ds \\ &+ 2\alpha \int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds \\ &\leq (\frac{C^{2}}{c_{1}} + 2\alpha + c_{2}) \int_{t}^{T} \|u_{l} - u_{n}\|_{2}^{2}ds + \int_{t}^{T} \mathcal{E}^{a,\hat{b}}(u_{l} - u_{n})ds \\ &+ 4(r-1) \int_{t}^{T} \int_{\{f',r>n\}} (C|D_{\sigma}u_{n}| + f'^{,r})dmds. \end{split}$$

 As

$$\lim_{n \to \infty} \int_t^T \int_{\{f', r > n\}} f'^{,r} dm ds = 0,$$

and

æ

$$\int_{t}^{T} \int_{\{f',r>n\}} |D_{\sigma}u_{n}| dm dt \leq ||1_{\{f',r>n\}}||_{L^{2}(dt\times m)} ||D_{\sigma}u_{n}||_{L^{2}(dt\times m)} \to 0,$$

we have as above that $(u_n)_{n\in\mathbb{N}}$ is a Cauchy sequence in \hat{F}^l . Hence, $u := \lim_{n\to\infty} u_n$

exists in \hat{F}^l . We can find a subsequence such that $(u_{n_k}, D_{\sigma}u_{n_k}) \to (u, D_{\sigma}u)$ a.e. and we have that

$$\begin{aligned} &|f_{n_k}(u_{n_k}, D_{\sigma} u_{n_k}) - f(u, D_{\sigma} u)| \\ \leq & 1_{\{f', r \leq n_k\}} |f(u, D_{\sigma} u) - f(u_{n_k}, D_{\sigma} u_{n_k})| + 1_{\{f', r > n_k\}} [|f(u, D_{\sigma} u) - f^0| + |f(u, D_{\sigma} u)| \\ &- f(u_{n_k}, D_{\sigma} u_{n_k})|] \\ \leq & |f(u, D_{\sigma} u) - f(u_{n_k}, D_{\sigma} u_{n_k})| + 1_{\{f', r > n_k\}} |f(u, D_{\sigma} u) - f^0| \\ \leq & |f(u_{n_k}, D_{\sigma} u) - f(u_{n_k}, D_{\sigma} u_{n_k})| + |f(u_{n_k}, D_{\sigma} u) - f(u, D_{\sigma} u)| + 1_{\{f', r > n_k\}} |f(u, D_{\sigma} u) - f^0| \end{aligned}$$

As in the above proof we have

$$f_{n_k}(u_{n_k}, D_\sigma u_{n_k}) \to f(u, D_\sigma u),$$

in $L^1([0,T], L^2)$. We conclude that u is a solution of (2.2.1) associated to the data (ϕ, f) .

Step 4: Now we prove the theorem without additional conditions. Define

$$f_n := \frac{n}{f'^{,1} \vee n} (f - f^0) + f^0.$$

Since $f_n(t, x, y, z) = f(t, x, y, z)$ for $f'^{,1} \le n$, we have

$$\lim_{n \to \infty} f_n = f_n$$

Introduce the following notation:

$$f_n^{\prime,1}(t,x) := \sup_{|y| \le 1} |f_n^\prime(t,x,y)|$$
 and $f_n^\prime(t,x,y) := f_n(t,x,y,0) - f^0(t,x).$

As in Step 3 we have

$$|f_n'^{,1}| \le n \land |f'^{,1}|.$$

Since f'_n is uniformly bounded, we can apply Step 3. Then we get a solution u_n for the data (ϕ, f_n) . The convergence of u_n can be shown analogously to Step 3.

2.6.2 Appendix B. Proof of Proposition 2.4.3

Let $M_x^p(\mathbb{R}^l)$ denote the set of (equivalence classes of)predictable processes $\{\phi_t\}_{t\in[0,T]}$ with values in \mathbb{R}^l such that

$$\|\phi\|_{M^p_x} := (E^x[(\int_0^T |\phi_r|^2 dr)^{p/2}])^{1/p} < \infty.$$

 $M^p_{\sigma,x}(\mathbb{R}^l \otimes \mathbb{R}^d)$ denotes the set of (equivalence classes of) predictable processes $\{\phi_t\}_{t \in [0,T]}$ with values in $\mathbb{R}^l \otimes \mathbb{R}^d$ such that

$$\|\phi\|_{M^p_{\sigma,x}} := (E^x[(\int_0^T |\phi_r \sigma(X_r)|^2 dr)^{p/2}])^{1/p} < \infty$$

Fix $x \in A_p$.

We note that (Y, Z) solves the BSDE (2.4.1) with data (ξ, f) iff

$$(\bar{Y}_t, \bar{Z}_t) := (e^{\alpha_t} Y_t, e^{\alpha_t} Z_t),$$

solve the BSDE (2.4.1) with data $(e^{\alpha_T}\xi, f')$, where

$$f'(t, y, z) := e^{\alpha_t} f(t, e^{-\alpha_t} y, e^{-\alpha_t} z) - \mu_t y.$$

Therefore, we may replace $(\Omega 2)$ by

$$\langle y - y', f(t, \omega, y, z) - f(t, \omega, y', z) \rangle \le 0$$
, for all t, x, y, y', z

Step 1 Assume that f is Lipschitz continuous with respect to both y and z. Define a mapping Φ from $B_x^2 := M_x^2(\mathbb{R}^l) \times M_{\sigma,x}^2(\mathbb{R}^l \otimes \mathbb{R}^d)$ into itself as follows. Given $(U, V) \in B_x^2$, we can set $\Phi(U, V) := (Y, Z)$, where (Y, Z) is the solution of the BSDE (2.4.1) associated with data $(\xi, f(U, V\sigma(X)))$ given by Lemma 2.4.1. Then by Itô's formula and BDG inequality we get

$$E^x[\sup_{t\in[0,T]}|Y_t|^2]<\infty.$$

Let $(U, V), (U', V') \in B_x^2, (Y, Z) = \Phi(U, V), (Y', Z') = \Phi(U', V'), (\overline{U}, \overline{V}) = (U - U', V - V'), (\overline{Y}, \overline{Z}) = (Y - Y', Z - Z')$. It follows from Itô's formula that for each $\gamma \in \mathbb{R}$,

$$e^{\gamma t} E^{x} |\bar{Y}_{t}|^{2} + E^{x} \int_{t}^{T} e^{\gamma s} (\gamma |\bar{Y}_{s}|^{2} + |\bar{Z}_{s}\sigma(X_{s})|^{2}) ds$$

$$\leq 2K E^{x} \int_{t}^{T} e^{\gamma s} |\bar{Y}_{s}| (|\bar{U}_{s}| + |\bar{V}_{s}\sigma(X_{s})|) ds$$

$$\leq 4K^{2} E^{x} \int_{t}^{T} e^{\gamma s} |\bar{Y}_{s}|^{2} + \frac{1}{2} E^{x} \int_{t}^{T} e^{\gamma s} (|\bar{U}_{s}|^{2} + |\bar{V}_{s}\sigma(X_{s})|^{2}) ds,$$

where K is the Lipschitz constant of f. We choose $\gamma = 1 + 4K^2$. Then

$$E^{x} \int_{0}^{T} e^{\gamma s} (|\bar{Y}_{s}|^{2} + |\bar{Z}_{s}\sigma(X_{s})|^{2}) ds \leq \frac{1}{2} E^{x} \int_{0}^{T} e^{\gamma s} (|\bar{U}_{s}|^{2} + |\bar{V}_{s}\sigma(X_{s})|^{2}) ds,$$

from which it follows that Φ is a strict contraction on B_x^2 equipped with the norm:

$$|||(Y,Z)|||_{\gamma}^{x} = (E^{x} \int_{0}^{T} e^{\gamma t} (|Y_{t}|^{2} + |Z_{t}\sigma(X_{t})|^{2}) dt)^{1/2}.$$

Define a sequence (Y^n, Z^n) by $(Y^{n+1}, Z^{n+1}) := \Phi(Y^n, Z^n)$. We have for $\gamma = 1 + 4K^2$

$$E^{x} \int_{0}^{1} e^{\gamma s} (|Y_{s}^{n} - Y_{s}^{n+1}|^{2} + |(Z_{s}^{n} - Z_{s}^{n+1})\sigma(X_{s})|^{2}) ds$$

$$\leq \frac{1}{2^{n}} E^{x} \int_{0}^{T} e^{\gamma s} (|Y_{s}^{0} - Y_{s}^{1}|^{2} + |(Z_{s}^{0} - Z_{s}^{1})\sigma(X_{s})|^{2}) ds.$$

Then we have the a.s. pointwise convergence of $(Y_s^n, Z_s^n \sigma(X_s))$ under each measure P^x , $x \in A^2$. Denote the limit by $(Y_s, Z_s \sigma(X_s))$. Then this is the fixed point of Φ under the norm $|||(Y, Z)|||_{\gamma}^x$. So we have (Y_s, Z_s) is the solution of BSDE (2.4.1).

Step 2 We assume that f, ξ are bounded.

We need the following proposition.

Proposition B.1 Assume condition (A5). Given $V \in \bigcap_x M^2_{\sigma,x}(\mathbb{R}^l \otimes \mathbb{R}^d)$, there exists a unique pair of predictable processes $(Y_t, Z_t) \in M^2_x \times M^2_{\sigma,x}(\mathbb{R}^l \otimes \mathbb{R}^d), \forall x \in \mathcal{N}^c$ satisfying under all P^x , $x \in \mathcal{N}^c$

$$Y_t = \xi + \int_t^T f(s, Y_s, V_s) ds - \int_t^T Z_s dM_s, \qquad 0 \le t \le T.$$

Using Proposition B.1, we can construct a mapping Φ from B_x^2 into itself as follows. For any $(U, V) \in B_x^2$, $(Y, Z) = \Phi(U, V)$ is the solution of the BSDE

$$Y_t = \xi + \int_t^T f(s, Y_s, V_s) ds - \int_t^T Z_s dM_s, \qquad 0 \le t \le T.$$

Then as in **Step 1**, we have

$$\begin{split} e^{\gamma t} E^x |\bar{Y}_t|^2 + E^x \int_t^T e^{\gamma s} (\gamma |\bar{Y}_s|^2 + |\bar{Z}_s \sigma(X_s)|^2) ds \\ = & 2E^x \int_t^T e^{\gamma s} \langle \bar{Y}_s, f(Y_s, V_s \sigma(X_s)) - f(Y'_s, V'_s \sigma(X_s)) \rangle ds \\ \leq & 2KE^x \int_t^T e^{\gamma s} |\bar{Y}_s| \times |\bar{V}_s \sigma(X_s)| ds \\ \leq & E^x \int_t^T e^{\gamma s} (2K^2 |\bar{Y}_s|^2 + \frac{1}{2} |\bar{V}_s \sigma(X_s)|^2) ds. \end{split}$$

Then by the same arguments as in **Step 1**, we obtain the assertion of Proposition 2.4.3 if f, ξ are bounded.

Proof of Proposition B.1 We write f(s, y) for $f(s, y, V_s)$.

By C we denote the constant satisfying $|\xi|^2 + \sup_t |f(t,0)|^2 \leq C$ a.s.. Define

$$f^n(t,y) := (\rho_n * f(t,\cdot))(y),$$

where $\rho_n : \mathbb{R}^l \to \mathbb{R}^+$ is a sequence of smooth functions with compact support satisfying $\int \rho_n(z) dz = 1$, which approximate the Dirac measure at 0. Then each f^n is locally Lipschitz in y, uniformly with respect to s and ω .

Define for each $m \in \mathbb{N}$,

$$f^{n,m}(t,y) := f^n(t, \frac{\inf(m, |y|)}{|y|}y).$$

Then $f^{n,m}$ is globally Lipschitz and bounded, uniformly w.r.t. (t, ω) . As in **Step 1**, we have a unique pair $(Y_t^{n,m}, Z_t^{n,m}) \in M_x^2 \times M_{\sigma,x}^2(\mathbb{R}^l \otimes \mathbb{R}^d)$ such that

$$Y_t^{n,m} = \xi + \int_t^T f^{n,m}(s, Y_s^{n,m}) ds - \int_t^T Z_s^{n,m} dM_s, \qquad 0 \le t \le T$$

By Itô's formula we have

$$|Y^{n,m}_t|^2 \le e^T C, \qquad 0 \le t \le T.$$

Consequently, for $m^2 > e^T C$, $(Y_t^{n,m}, Z_t^{n,m})$ does not depend on m. Therefore, we denote it by (Y_t^n, Z_t^n) . Then by the same arguments as [BDHPS03, Proposition 3.2] we have

$$E^{x}(\sup_{0 \le t \le T} |Y_{t}^{k} - Y_{t}^{l}|^{2}) + E^{x}(\int_{0}^{T} |(Z_{t}^{k} - Z_{t}^{l})\sigma(X_{t})|^{2}dt) \le KE^{x}[\int_{0}^{T} |f^{k}(t, Y_{t}^{k}) - f^{l}(t, Y_{t}^{k})|^{2}dt].$$

By a similar argument as in the proof of Theorem 2.2.8, we obtain for fixed ω ,

$$\sup_{k>l} \int_0^T |f^k(t, Y_t^k) - f^l(t, Y_t^k)|^2 dt \to 0, \text{ as } l \to \infty.$$

Then we have

$$\sup_{k>l} E^x \int_0^T |f^k(t, Y^k_t) - f^l(t, Y^k_t)|^2 dt \le E^x \sup_{k>l} \int_0^T |f^k(t, Y^k_t) - f^l(t, Y^k_t)|^2 dt \to 0, \qquad l \to \infty$$

and we can obtain a sequence of representable variables that converges rapidly

enough under all measures $P^x, x \in \mathcal{N}^c$. For each $l = 0, 1, \dots$ set

$$n_{l}(x) = \inf\{n > n_{l-1}(x); \sup_{k \ge n} E^{x} \left[\int_{0}^{T} |f^{k}(t, Y_{t}^{k}) - f^{n}(t, Y_{t}^{k})|^{2} dt\right] < \frac{1}{2^{l}}\},$$

$$\bar{Y}^{l} = Y^{n_{l}(X_{0})}, \bar{Z}^{l} = Z^{n_{l}(X_{0})}.$$

With this sequence one may pass to the limit and define $Z'_s = \limsup_{l\to\infty} \overline{Z}^l_s \sigma(X_s)$ and $Z_s = Z'_s \tau(X_s)$. Then we obtain the claimed results.

So far we have proved the assertion when ξ , f are bounded. Then by the same arguments as in [BDHPS03, Theorem 4.2], one proves the general case.

Chapter 3

BSDE and generalized Dirichlet form: Infinite dimensional case

In this chapter we extend results in the previous chapter to infinite dimensional case. In Section 3.1 we give some basic assumptions on the operator L and prove some basic relations for linear equation. In Section 3.2, we use analytic methods to solve PDE (1.4). In Section 3.3, we prove the martingale representation theorem for the process associated with the operator L. By this we obtain the existence and uniqueness of the solution of BSDE (1.6) in Section 3.4. The relation between PDE and BSDE is also established in this section. Examples are given in Section 3.5. In Section 3.6, we use our results to a control problem for an application. The main results of this chapter have already been submitted for publication, see [Zhu b].

3.1 Preliminaries

Let E be a separable real Banach space and $(H, \langle \cdot, \cdot \rangle_H)$ a separable real Hilbert space such that $H \subset E$ densely and continuously. Identifying H with its topological dual H' we obtain that $E' \subset H \subset E$ densely and continuously and $_{E'}\langle \cdot, \cdot \rangle_E = \langle \cdot, \cdot \rangle_H$ on $E' \times H$. Define the linear space of finitely based smooth functions on E by

$$\mathcal{F}C_b^{\infty} := \{ f(l_1, ..., l_m) | m \in \mathbb{N}, f \in C_b^{\infty}(\mathbb{R}^m), l_1, ..., l_m \in E' \}.$$

Here $C_b^{\infty}(\mathbb{R}^m)$ denotes the set of all infinitely differentiable (real-valued) functions with all partial derivatives bounded. For $u \in \mathcal{F}C_b^{\infty}$ and $k \in E$ let

$$\frac{\partial u}{\partial k}(z) := \frac{d}{ds}u(z+sk)|_{s=0}, z \in E$$

be the Gâteaux derivative of u in direction k. It follows that for $u = f(l_1, ..., l_m) \in \mathcal{F}C_b^{\infty}$ and $k \in H$ we have that

$$\frac{\partial u}{\partial k}(z) = \sum_{i=1}^{m} \frac{\partial f}{\partial x_i}(l_1(z), ..., l_m(z)) \langle l_i, k \rangle_H, z \in E.$$

Consequently, $k \mapsto \frac{\partial u}{\partial k}(z)$ is continuous on H and we can define $\nabla u(z) \in H$ by

$$\langle \nabla u(z), k \rangle_H = \frac{\partial u}{\partial k}(z)$$

Let μ be a finite positive measure on $(E, \mathcal{B}(E))$. By $L_{sym}(H)$ we denote the linear space of all symmetric and bounded linear operators on H equipped with usual operator norm $\|\cdot\|_{L^{\infty}(H)}$. Let $A: E \mapsto L_{sym}(H)$ be measurable such that $\langle A(z)h, h \rangle_{H} \geq 0$ for all $z \in E, h \in H$ and let $b: E \to H$ be $\mathcal{B}(E)/\mathcal{B}(H)$ -measurable. Suppose the Pseudo inverse A^{-1} of A is measurable.

We denote by $|\cdot|_H$ the *H*-norm and set $||u(z)||_2^2 := \int |u(z)|^2 d\mu(z)$ for $u \in L^2(E,\mu)$. We also denote $(u,v)_{L^2(E,\mu)}$ by (u,v) for $u,v \in L^2(E,\mu)$. For $p \ge 1$, let $L^p(\mu)$, $L^p(\mu; H)$ denote $L^p(E,\mu)$, $L^p(E,\mu; H)$ respectively. If *W* is a function space, we will use *bW* to denote set of all the bounded functions in *W*.

Furthermore, we introduce the bilinear form

$$(3.1.1)$$

$$\mathcal{E}(u,v) := \int \langle A(z)\nabla u(z), \nabla v(z) \rangle_H d\mu(z) + \int \langle A(z)b(z), \nabla u(z) \rangle_H v(z)d\mu(z), u, v \in \mathcal{F}C_b^{\infty}.$$

We introduce the following conditions,

(A1) $\langle A(\cdot)k,k\rangle \in L^1(\mu)$ for $k \in H$ and the bilinear form

$$\mathcal{E}^{A}(u,v) = \int \langle A(z)\nabla u(z), \nabla v(z) \rangle_{H} d\mu(z); u, v \in \mathcal{F}C_{b}^{\infty}.$$

is closable on $L^2(E;\mu)$.

The closure of $\mathcal{F}C_b^{\infty}$ with respect to $\mathcal{E}_1^A := \mathcal{E}^A + \langle \cdot, \cdot \rangle_H$ is denoted by F. Then (\mathcal{E}^A, F) is a well-defined symmetric Dirichlet form on $L^2(E, \mu)$. Set $\mathcal{E}_1^A(u) := \mathcal{E}_1^A(u, u), u \in F$.

(A2) Let $A^{1/2}b \in L^2(E; H, \mu)$, i.e. $\int |A^{1/2}b|_H^2 d\mu < \infty$, and there exists $\alpha \ge 0$ such that

(3.1.2)
$$\int \langle Ab, \nabla u^2 \rangle_H d\mu \ge -\alpha \|u\|_2^2, \qquad u \in \mathcal{F}C_b^{\infty}, \forall u \ge 0.$$

Obviously, \mathcal{E} from (3.1.1) immediately extends to all $u \in F, v \in bF$.

(A3) There exists a positivity preserving C_0 -semigroup P_t on $L^2(E; \mu)$ such that for any $t \in [0, T], \exists C_T > 0$ such that

$$\|P_t f\|_{\infty} \le C_T \|f\|_{\infty}$$

and such that its L^2 -generator $(L, \mathcal{D}(L))$ has the following properties: $b\mathcal{D}(L) \subset bF$ and for any $u \in bF$ there exists uniformly bounded $u_n \in \mathcal{D}(L)$ such that $\mathcal{E}_1^A(u_n - u) \to 0$ as $n \to \infty$ and that it is associated with the bilinear form \mathcal{E} in (3.1.1) in the sense that $\mathcal{E}(u, v) = -(Lu, v)$ for $u, v \in b\mathcal{D}(L)$.

To obtain a semigroup P_t satisfying the above conditions, we can use generalized Dirichlet forms as introduced in Section 1.1.

Remark 3.1.1 (i) Some general criteria imposing conditions on A and μ in order that \mathcal{E}^A be closable are e.g. given in [MR92, Chap II, Section 2] and [AR90].

(ii)In our case, due to our general conditions on b and f, we can't find a suitable Gelfand triple $V \subset H \subset V^*$ with V being a reflexive Banach space to apply the monotonicity method as in [Ba10] or [PR07].

(iii) We can construct a semigroup P_t satisfying (A3) by the theory of generalized Dirichlet forms. More precisely, if there exists a constant $\hat{c} \geq 0$ such that $\mathcal{E}_{\hat{c}}(\cdot, \cdot) :=$ $\mathcal{E}(\cdot, \cdot) + \hat{c}(\cdot, \cdot)$ is a generalized Dirichlet form on a Hausdorff space E_1 with domain $\mathcal{F} \times \mathcal{V}$ in one of the following three senses:

 $\begin{aligned} (\mathbf{a})(E_1, \mathcal{B}(E_1), m) &= (E, \mathcal{B}(E), \mu), \\ (\mathcal{A}, \mathcal{V}) &= (\mathcal{E}^A, F), \\ -\langle \Lambda u, v \rangle - \hat{c}(u, v) &= \int \langle A(z)b(z), \nabla u(z) \rangle_H v(z) d\mu(z) \text{ for } u, v \in \mathcal{F}C_b^{\infty}; \\ (\mathbf{b})(E_1, \mathcal{B}(E_1), m) &= (E, \mathcal{B}(E), \mu), \\ \mathcal{A} &\equiv 0 \text{ and } \mathcal{V} = L^2(E, \mu), \end{aligned}$

 $-\langle \Lambda u, v \rangle = \mathcal{E}_{\hat{c}}(u, v)$ for $u, v \in D$, where $D \subset \mathcal{F}C_b^{\infty}$ densely w.r.t. \mathcal{E}_1^A -norm and $D \subset \mathcal{D}(L)$;

(c) $\mathcal{E}_{\hat{c}} = \mathcal{A}, \Lambda \equiv 0$ (In this case $(\mathcal{E}_{\hat{c}}, \mathcal{V})$ is a sectorial Dirichlet form in the sense of [MR92]);

then there exists a sub-Markovian C_0 -semigroup of contraction $P_t^{\hat{c}}$ associated with the generalized Dirichlet form $\mathcal{E}_{\hat{c}}$. Then $P_t := e^{\hat{c}t}P_t^{\hat{c}}$ satisfies (A3) and we have

$$\mathcal{D}(L) \subset \mathcal{F} \subset F.$$

(iv)The semigroup can also be constructed by other methods. (see e.g. [DR02],

[BDR09]).

90

(v) By (A3) we have that \mathcal{E} is positivity preserving, i.e.

$$\mathcal{E}(u, u^+) \ge 0 \ \forall u \in \mathcal{D}(L),$$

which can be obtained by the same arguments as in [St2, I Proposition 4.4]. By (3.1.2) and (A3), we have for $u \in b\mathcal{D}(L), u \geq 0$

$$\int Lud\mu = -\mathcal{E}(u,1) = -\int \langle Ab, \nabla u \rangle_H d\mu = -\int \langle Ab, \nabla (u+\varepsilon) \rangle_H d\mu \le -\alpha \int (u+\varepsilon) d\mu.$$

Letting $\varepsilon \to 0$, we have $\int Lud\mu \leq -\alpha \int ud\mu$. $(P_t)_{t \in [0,T]}$ is a C_0 -semigroup on $L^1(E;\mu)$.

(vi) All the conditions are satisfied by the bilinear form considered in [St1, Section 4] and the operator in [D04, Chapter II,III,IV] (see Section 3.5 below).

(vii) The notion of quasi-regularity for generalized Dirichlet forms analogously to [MR92] has been introduced in [St2]. By this and a technical assumption an associated m-tight special standard process can be constructed. We will use stochastic calculus associated with this process to conclude our probabilistic results (see Section 3.3 below).

Let us recall the notations $\hat{F}, \mathcal{C}_T, \|\cdot\|_T$ associated with \mathcal{E}^A from [BPS05]: $\mathcal{C}_T := C^1((0,T); L^2) \cap L^2(0,T;F)$, which turns out to be the appropriate space of test functions, i.e.

$$\mathcal{C}_T = \{ \varphi : [0,T] \times E \to \mathbb{R} | \varphi_t \in F \text{ for almost each } t, \int_0^T \mathcal{E}^A(\varphi_t,\varphi_t) dt < \infty, \\ t \to \varphi_t \text{ is differentiable in } L^2 \text{ and } t \to \partial_t \varphi_t \text{ is } L^2 - \text{ continuous on } [0,T] \}$$

We also set $\mathcal{C}_{[a,b]} := C^1([a,b];L^2) \cap L^2([a,b];F)$. For $\varphi \in \mathcal{C}_T$, we define

$$\|\varphi\|_T := (\sup_{t \le T} \|\varphi_t\|_2^2 + \int_0^T \mathcal{E}^A(\varphi_t) dt)^{1/2}.$$

 \hat{F} is the completion of \mathcal{C}_T with respect to $\|\cdot\|_T$. By [BPS05], $\hat{F} = C([0,T]; L^2) \cap L^2(0,T;F)$, and for every $u \in \hat{F}$ there exists a sequence $u^n \in \mathcal{F}C_b^{\infty,T}, n \in \mathbb{N}$, such that $\int_0^T \mathcal{E}_1^A(u_t - u_t^n) dt \to 0$, where

$$\mathcal{F}C_b^{\infty,T} := \{ f(t, l_1, ..., l_m) | m \in \mathbb{N}, f \in C_b^{\infty}([0, T] \times \mathbb{R}^m), l_1, ..., l_m \in E' \}.$$

We also introduce the following space

$$W^{1,2}([0,T];L^2(E)) = \{ u \in L^2([0,T];L^2); \partial_t u \in L^2([0,T];L^2) \},\$$

where $\partial_t u$ is the derivative of u in the weak sense (see e.g. [Ba10]).

3.1.1 Linear Equations

We consider the linear equation

(3.1.3)
$$\begin{aligned} (\partial_t + L)u + f &= 0, \qquad 0 \le t \le T \\ u_T(x) &= \phi(x), \qquad x \in E \end{aligned}$$

where $f \in L^1([0,T]; L^2(E,\mu)), \phi \in L^2(E,\mu).$

By [BPS05] we set $D_{A^{1/2}}\varphi := A^{1/2}\nabla\varphi$ for any $\varphi \in \mathcal{F}C_b^{\infty}$, define $V_0 = \{D_{A^{1/2}}\varphi : \varphi \in \mathcal{F}C_b^{\infty}\}$, and let V be the closure of V_0 in $L^2(E; H, \mu)$. Then we have the following results.

Proposition 3.1.2 Assume (A1) holds.

(i) For every $u \in F$ there is a unique element of V, which we denote by $D_{A^{1/2}} u$ such that

$$\mathcal{E}^{A}(u,\varphi) = \int \langle D_{A^{1/2}}u(x), D_{A^{1/2}}\varphi(x) \rangle_{H}\mu(dx), \qquad \forall \varphi \in \mathcal{F}C_{b}^{\infty}$$

One has $A^{1/2}A^{-1/2}D_{A^{1/2}}u(x) = D_{A^{1/2}}u(x)$. Moreover, the above formula extends to $u, v \in F$,

$$\mathcal{E}^{A}(u,v) = \int \langle D_{A^{1/2}}u(x), D_{A^{1/2}}v(x) \rangle_{H}\mu(dx)$$

(ii) Furthermore, if $u \in \hat{F}$, there exists a measurable function $\phi : [0, T] \times E \mapsto H$ such that $|A^{1/2}\phi|_H \in L^2([0, T] \times E)$ and $D_{A^{1/2}}u_t = A^{1/2}\phi_t$ for almost all $t \in [0, T]$.

(iii)Let $u^n, u \in \hat{F}$ be such that $u^n \to u$ in $L^2((0,T) \times E)$ and $(D_{A^{1/2}}u^n)_n$ is a Cauchy-sequence in $L^2([0,T] \times E; H)$. Then $D_{A^{1/2}}u^n \to D_{A^{1/2}}u$ in $L^2((0,T) \times E; H)$, i.e. $D_{A^{1/2}}$ is closable as an operator from \hat{F} into $L^2((0,T) \times E; H)$.

Proof (i) Since \mathcal{E}^A is closable on $L^2(E; \mu)$, the assertion follows.

(ii) If $u \in \hat{F}$, we have $u^n \in \mathcal{F}C_b^{\infty,T}$, $n \in \mathbb{N}$, such that $\int_0^T \mathcal{E}_1^A(u^n - u)dt \to 0$. Hence, we define $\varphi := \lim_{n \to \infty} D_{A^{1/2}}u^n$ in $L^2((0,T) \times E; H)$ and $\phi := A^{-1/2}\varphi$. Since $A^{1/2}A^{-1/2}A^{1/2} = A^{1/2}$ and $||A^{-1/2}A^{1/2}|| \leq 1$, $D_{A^{1/2}}u^n = A^{1/2}A^{-1/2}A^{1/2}\nabla u^n \to A^{1/2}A^{-1/2}\varphi = A^{1/2}\phi$ in $L^2([0,T] \times E; H)$. Passing to a subsequence we may find a set $\Lambda \subset [0,T]$ such that $[0,T] \setminus \Lambda$ is negligible and for every $t \in \Lambda$, $\mathcal{E}_1^A(u_t^n - u_t) \to 0$ and $\|D_{A^{1/2}}u_t^n - A^{1/2}\phi_t\|_{L^2(E;H)} \to 0$. Then we have

$$\mathcal{E}^{A}(u_{t},\varphi) = \int \langle A^{1/2}\phi_{t}(x), D_{A^{1/2}}\varphi(x) \rangle_{H}\mu(dx).$$

(iii) Let $v = \lim_{n} D_{A^{1/2}} u^n$. Passing to a subsequence we assume for almost every $t \in [0, T]$, $\|v_t - D_{A^{1/2}} u^n_t\|_{L^2(E;H)} \to 0$. We take $\varphi \in \mathcal{D}(L^A)$ where L^A is the generator associated to (\mathcal{E}^A, F) . Then

$$\int \langle v_t, D_{A^{1/2}}\varphi \rangle_H d\mu = \lim_{n \to \infty} \int \langle D_{A^{1/2}}u_t^n, D_{A^{1/2}}\varphi \rangle_H d\mu = \lim_n \mathcal{E}^A(u_t^n, \varphi) = -\lim_n (u_t^n, L^A\varphi)$$
$$= -(u_t, L^A\varphi) = \mathcal{E}^A(u_t, \varphi) = \int \langle D_{A^{1/2}}u_t, D_{A^{1/2}}\varphi \rangle_H d\mu.$$

It follows that $v_t = D_{A^{1/2}} u_t$.

For $u \in F, v \in bF$ we denote

$$\mathcal{E}(u,v) := \int \langle D_{A^{1/2}}u(x), D_{A^{1/2}}v(x) \rangle_H \mu(dx) + \int \langle A^{1/2}b, D_{A^{1/2}}u \rangle_H v \mu(dx) + \int \langle A^{1/2}b,$$

Notation By $\tilde{\nabla}u$ we denote the set of all measurable functions $\phi: E \to H$, such that $A^{1/2}\phi = D_{A^{1/2}}u$ as elements of $L^2(\mu; H)$.

3.1.2 Solution of the Linear Equation

Definition 3.1.3 [strong solution] A function $u \in \hat{F} \cap L^1((0,T); \mathcal{D}(L))$ is called a strong solution of equation (3.1.3) with data (ϕ, f) , if $t \mapsto u_t = u(t, \cdot)$ is L^2 differentiable on $[0,T], \partial_t u_t \in L^1((0,T); L^2)$ and the equalities in (3.1.3) hold in $L^2(\mu)$.

Definition 3.1.4 [generalized solution] A function $u \in \hat{F}$ is called a generalized solution of equation (3.1.3), if there exists a sequence of $\{u^n\}$ consisting of strong solutions with data (ϕ^n, f^n) such that

$$||u^n - u||_T \to 0, ||\phi^n - \phi||_2 \to 0, \lim_{n \to \infty} f^n = f \text{ in } L^1([0, T]; L^2(\mu)).$$

By (A3) and Remark 3.1.1 (v), for $0 \le t \le T$, P_t , as C_0 -semigroup on $L^1(E;\mu)$, can be restricted to a semigroup on $L^p(E;\mu)$ for all $p \in [1,\infty)$ by the Riesz-Thorin Interpolation Theorem and the restricted semigroup (denoted again by P_t for simplicity) is strongly continuous on $L^p(E;\mu)$.

Proposition 3.1.5 Assume that (A1)-(A3) hold.

3.1. Preliminaries

(i) Let $f \in C^1([0,T]; L^p)$ for $p \in [1,\infty)$. Then

$$w_t := \int_t^T P_{s-t} f_s ds \in C^1([0,T]; L^p),$$

and

$$\partial_t w_t(x) = -P_{T-t} f_T(x) + \int_t^T P_{s-t} \partial_s f_s(x) ds.$$

(ii) Assume that $\phi \in \mathcal{D}(L)$, $f \in C^1([0,T]; L^2)$ and for each $t \in [0,T]$, $f_t \in \mathcal{D}(L)$. Define

$$u_t := P_{T-t}\phi + \int_t^T P_{s-t}f_s ds.$$

Then u is a strong solution of (3.1.5) and, moreover, $u \in C^1([0,T]; L^2)$.

Proof See the proof of [BPS05, Proposition 2.6].

Proposition 3.1.6 Suppose that conditions (A1)-(A3) hold. If u is a strong solution for (3.1.3), it is a mild solution for (3.1.3) i.e.

$$u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds.$$

Proof For fixed $t, \varphi \in \mathcal{D}(\hat{L})$

$$(u_T, \hat{P}_{T-t}\varphi) - (u_t, \varphi) = \int_t^T (-Lu_s - f_s, \hat{P}_{s-t}\varphi)ds + \int_t^T (u_s, \hat{L}\hat{P}_{s-t}\varphi)ds$$

where \hat{L}, \hat{P}_t denote the adjoints on $L^2(E, \mu)$ of L and P_t respectively. As u is a strong solution, we deduce that

$$(u_t,\varphi) = (P_{T-t}\phi + \int_t^T P_{s-t}f_s ds,\varphi).$$

Since $\mathcal{D}(\hat{L})$ is dense in L^2 , we have the result.

Proposition 3.1.7 Suppose that conditions (A1)-(A3) hold, $f \in L^1([0,T]; L^2)$ and $\phi \in L^2$. Then the equation (3.1.3) has a unique generalized solution $u \in \hat{F}$

(3.1.4)
$$u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds.$$

The solution satisfies the three relations:

(3.1.5)
$$\|u_t\|_2^2 + 2\int_t^T \mathcal{E}^A(u_s)ds \le 2\int_t^T (f_s, u_s)ds + \|\phi\|_2^2 + 2\alpha\int_t^T \|u_s\|_2^2ds, \qquad 0 \le t \le T,$$

(3.1.6)
$$\|u\|_T^2 \le M_T(\|\phi\|_2^2 + (\int_0^T \|f_t\|_2 dt)^2),$$

$$\int_0^{T} ((u_t, \partial_t \varphi_t) + \mathcal{E}^A(u_t, \varphi_t) + \int \langle A^{1/2}b, D_{A^{1/2}}u_t \rangle_H \varphi_t d\mu) dt = \int_0^T (f_t, \varphi_t) dt + (\phi, \varphi_T) - (u_0, \varphi_0),$$

for any $\varphi \in b\mathcal{C}_T$, where M_T is a constant depending on T. (3.1.7) can be extended easily to $\varphi \in bW^{1,2}([0,T]; L^2) \cap L^2([0,T]; F)$.

Moreover, if $u \in \hat{F}$ is bounded and satisfies (3.1.7) for any $\varphi \in b\mathcal{C}_T$ with bounded (f, ϕ) , then u is a weak solution given by (3.1.4).

Proof Define u by (3.1.4). First assume that ϕ , f are bounded and satisfy the conditions of Proposition 3.1.5 (ii). Then, since u is bounded and by Proposition 3.1.5 we know that u is a strong solution of (3.1.3), hence it obviously satisfies (3.1.7). Furthermore, $u \in C^1([0,T]; L^2)$. Hence, actually $u \in b\mathcal{C}_T$ and consequently, for $t_0 \in [0,T]$

$$\int_{t_0}^T ((u_t, \partial_t u_t) + \mathcal{E}^A(u_t, u_t) + \int \langle A^{1/2}b, D_{A^{1/2}}u_t \rangle_H u_t d\mu) dt = \int_{t_0}^T (f_t, u_t) dt + (\phi, u_T) - (u_{t_0}, u_{t_0}) dt +$$

By (3.1.2) we have $\int \langle A^{1/2}b, D_{A^{1/2}}u_t \rangle_H u_t d\mu \ge -\alpha \|u_t\|_2^2$ then we have (3.1.8) $\|u_t\|_2^2 + 2 \int_t^T \mathcal{E}^A(u_s) ds \le 2 \int_t^T (f_s, u_s) ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|u_s\|_2^2 ds, \qquad 0 \le t \le T.$

As

$$\begin{split} \int_{t}^{T} (f_{s}, u_{s}) ds &= \int_{t}^{T} ((f_{s}, P_{T-s}\phi) + (f_{s}, \int_{s}^{T} P_{r-s}f_{r}dr)) ds \\ &\leq \int_{t}^{T} \|f_{s}\|_{2} \|P_{T-s}\phi\|_{2} ds + \int_{t}^{T} \|f_{s}\|_{2} \|\int_{s}^{T} P_{r-s}f_{r}dr\|_{2} ds \\ &\leq M_{0}e^{T-t} (\|\phi\|_{2} \int_{t}^{T} \|f_{s}\|_{2} ds + \int_{t}^{T} (\|f_{s}\|_{2} \int_{s}^{T} \|f_{r}\|_{2} dr) ds), \end{split}$$

and

$$\int_{t}^{T} \|u_{s}\|_{2}^{2} ds \leq M_{T-t}(\|\phi\|_{2}^{2} + (\int_{0}^{T} \|f_{t}\|_{2} dt)^{2}),$$

we obtain

$$||u_t||_2^2 + \int_t^T \mathcal{E}^A(u_s) ds \le M_{T-t}(||\phi||_2^2 + (\int_0^T ||f_t||_2 dt)^2).$$

Hence, it follows that

(3.1.9)
$$\|u\|_T^2 \le M_T(\|\phi\|_2^2 + (\int_0^T \|f_t\|_2 dt)^2).$$

Here the constant M_{T-t} may change from line to line, but it is independent of f, ϕ . Next we will prove the result for general data ϕ and f. Let $(f^n)_{n \in \mathbb{N}}$ be a sequence of functions in $bC^1([0,T]; L^2(\mu))$ such that $f_t \in \mathcal{D}(L)$ for a.e. $t \in [0,T]$ and $\int_0^T \|f_t^n - f_t\|_2 dt \to 0$. (Such a sequence exists, since $\{\alpha_t g(x); \alpha_t \in C_0^\infty[0,T], g \in b\mathcal{D}(L)\}$ is dense in $L^1([0,T]; L^2)$). Take functions $(\phi^n)_{n \in \mathbb{N}} \subset b\mathcal{D}(L)$ such that $\phi^n \to \phi$ in L^2 . Let u^n denote the solution given by (3.1.4) with $f = f^n, \phi = \phi^n$.

By linearity, $u^n - u^m$ is associated with $(\phi^n - \phi^m, f^n - f^m)$. Since (3.1.9) implies that

$$||u^{n} - u^{m}||_{T}^{2} \le M_{T}(||\phi^{n} - \phi^{m}||_{2}^{2} + (\int_{0}^{T} ||f_{t}^{n} - f_{t}^{m}||_{2}dt)^{2}),$$

we deduce that $(u^n)_{n\in\mathbb{N}}$ is a Cauchy sequence in \hat{F} . Then $u = \lim_{n\to\infty} u^n$ in $\|\cdot\|_T$ is a generalized solution of (3.1.3) and we have

$$u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds.$$

Next we prove (3.1.5) (3.1.6) (3.1.7) for u. For $\varphi \in b\mathcal{C}_T$, we have (3.1.10) $\int_0^T ((u_t^n, \partial_t \varphi_t) + \mathcal{E}^A(u_t^n, \varphi_t) + \int \langle A^{1/2}b, D_{A^{1/2}}u_t^n \rangle_H \varphi_t d\mu) dt = \int_0^T (f_t^n, \varphi_t) dt + (\phi^n, \varphi_T) - (u_0^n, \varphi_0).$

Since we have

$$\left|\int_{0}^{T} \mathcal{E}^{A}(u_{t}^{n}-u_{t},\varphi_{t})dt\right| \leq \left(\int_{0}^{T} \mathcal{E}^{A}(u_{t}^{n}-u_{t})dt\right)^{\frac{1}{2}} \left(\int_{0}^{T} \mathcal{E}^{A}(\varphi_{t})dt\right)^{\frac{1}{2}} \to 0, \text{ as } n \to \infty,$$

and

$$\begin{split} &|\int_{0}^{T} \int \langle A^{1/2}b, D_{A^{1/2}}(u_{t}^{n}-u_{t})\rangle_{H}\varphi_{t}d\mu dt| \\ \leq &\|\varphi\|_{\infty} (\int_{0}^{T} \int |A^{1/2}b|_{H}^{2}d\mu dt)^{\frac{1}{2}} (\int_{0}^{T} \int |D_{A^{1/2}}(u_{t}^{n}-u_{t})|_{H}^{2}d\mu dt)^{\frac{1}{2}} \\ &= &\|\varphi\|_{\infty} (\int_{0}^{T} \int |A^{1/2}b|_{H}^{2}d\mu dt)^{\frac{1}{2}} (\int_{0}^{T} \mathcal{E}^{A}(u_{t}^{n}-u_{t})dt)^{\frac{1}{2}} \\ \to &0, \text{ as } n \to \infty, \end{split}$$

we deduce that

$$\int_0^T ((u_t, \partial_t \varphi_t) + \mathcal{E}^A(u_t, \varphi_t) + \int \langle A^{1/2}b, D_{A^{1/2}}u_t \rangle_H \varphi_t d\mu) dt = \int_0^T (f_t, \varphi_t) dt + (\phi, \varphi_T) - (u_0, \varphi_0),$$

for any $\varphi \in b\mathcal{C}_T$

for any $\varphi \in \partial \mathcal{L}_T$.

The relations (3.1.5) (3.1.6) hold for the approximating functions:

$$\|u_t^n\|_2^2 + 2\int_t^T \mathcal{E}^A(u_s^n) ds \le 2\int_t^T (f_s^n, u_s^n) ds + \|\phi^n\|_2^2 + 2\alpha \int_t^T \|u_s^n\|_2^2 ds, \qquad 0 \le t \le T.$$
$$\|u^n\|_T^2 \le M_T(\|\phi^n\|_2^2 + (\int_0^T \|f_t^n\|_2 dt)^2).$$

Since $||u_t^n||_T \to ||u_t||_T$, $n \to \infty$, we conclude

$$\lim_{n \to \infty} \int_0^T \mathcal{E}^A(u_t^n) dt = \int_0^T \mathcal{E}^A(u_t) dt$$

It is easy to see that $\lim_{n\to\infty}\int_t^T (f_s^n, u_s^n) ds = \int_t^T (f_s, u_s) ds$. Then by passing to the limit, $n \to \infty$ in the above relations, we get (3.1.5) and (3.1.6) for u.

[Uniqueness] Let $v \in \hat{F}$ be another generalized solution of (3.1.3) and let $(v^n)_{n \in \mathbb{N}}$, $(\tilde{\phi}^n)_{n\in\mathbb{N}}, (\tilde{f}^n)_{n\in\mathbb{N}}$ be the corresponding approximating sequences in the definition of the generalized solution. By Proposition 3.1.6

$$\sup_{t \in [0,T]} \|u_t^n - v_t^n\|_2^2 \le M_T(\|\phi^n - \tilde{\phi}^n\|_2^2 + (\int_0^T \|f_t^n - \tilde{f}_t^n\|_2 dt)^2).$$

Letting $n \to \infty$, this implies u = v.

For the last result we have $\forall t_0 \in [0, T], \varphi \in b\mathcal{C}_T$ (3.1.11) $\int_{t_0}^T ((u_t, \partial_t \varphi_t) + \mathcal{E}^A(u_t, \varphi_t) + \int \langle A^{1/2}b, D_{A^{1/2}}u_t \rangle \varphi_t dm) dt = \int_{t_0}^T (f_t, \varphi_t) dt + (\phi, \varphi_T) - (u_{t_0}, \varphi_{t_0}).$

For $t \geq \frac{1}{n}$, define

$$u_t^n := n \int_0^{\frac{1}{n}} u_{t-s} ds, \qquad f_t^n := n \int_0^{\frac{1}{n}} f_{t-s} ds, \qquad \phi^n := n \int_0^{\frac{1}{n}} u_{T-s} ds.$$

By a similar argument as the proof of Proposition 2.1.9, u^n also fulfills (3.1.11) with f^n, ϕ^n . For the mild solution v associated with f, ϕ , the above relation also holds with v^n replacing u^n . Hence we have

$$\int_{t_0}^T (((u-v)_t^n, \partial_t \varphi_t) + \mathcal{E}^A((u-v)_t^n, \varphi_t) + \int \langle A^{1/2}b, D_{A^{1/2}}(u-v)_t^n \rangle \varphi_t dm) dt = -((u-v)_{t_0}^n, \varphi_{t_0})$$

Since $(u-v)_t^n \in b\mathcal{C}_{[\frac{1}{n},T]}$, the above equation holds with $(u-v)_t^n$ as a test function. So we have

$$\|(u-v)_{t_0}^n\|_2^2 + 2\int_{t_0}^T \mathcal{E}^A((u-v)_t^n, (u-v)_t^n)dt \le 2\alpha \int_{t_0}^T \|(u-v)_t^n\|_2^2 dt.$$

By Gronwall's lemma it follows that

$$||(u-v)_{t_0}^n||_2^2 = 0.$$

Letting $n \to \infty$, we have $||u_{t_0} - v_{t_0}||_2 = 0$. Then letting $t_0 \to 0$, we have $||u_0 - v_0|| = 0$. Therefore, $u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds$ is a generalized solution for (3.1.3).

3.1.3 Basic Relations for the Linear Equation

In this section we assume that (A1)-(A3) hold.

Lemma 3.1.8 If u is a bounded generalized solution of equation (3.1.3) with some function $\phi \ge 0, \phi \in L^2 \cap L^\infty$, then u^+ satisfies the following relation with $0 \le t_1 < t_2 \le T$

$$||u_{t_1}^+||_2^2 \le 2 \int_{t_1}^{t_2} (f_s, u_s^+) ds + ||u_{t_2}^+||_2^2.$$

Proof Choose the approximation sequence u^n for u as in the existence proof of Proposition 3.1.7. Denote its related data by f^n, ϕ^n .

We have the following equations:

$$\lim_{n \to \infty} \sup_{t \in [0,T]} \|u_t^n - u_t\|_2 = 0, \qquad \lim_{n \to \infty} \int_0^T \mathcal{E}^A (u_t^n - u_t) dt = 0$$
$$\lim_{n \to \infty} \int_0^T \|f_t^n - f_t\|_2 dt = 0, \qquad \lim_{n \to \infty} \|\phi^n - \phi\|_2 = 0.$$

Suppose that the following holds

(3.1.12)
$$\|(u_{t_1}^n)^+\|_2^2 \le 2 \int_{t_1}^{t_2} (f_s^n, (u_s^n)^+) ds + \|(u_{t_2}^n)^+\|_2^2,$$

where $0 \le t_1 \le t_2 \le T$. Since $||u^n||_2$ are uniformly bounded, we have

$$\lim_{n \to \infty} \int_{t_1}^{t_2} (f_s^n, (u_s^n)^+) ds = \int_{t_1}^{t_2} (f_s, u_s^+) ds.$$

By letting $n \to \infty$ in equation (3.1.12) we get for $0 \le t_1 \le t_2 \le T$,

$$\|u_{t_1}^+\|_2^2 \le 2\int_{t_1}^{t_2} (f_s, u_s^+) ds + \|u_{t_2}^+\|_2^2.$$

Therefore, the problem is reduced to the case where u belongs to $b\mathcal{C}_T$; in the remainder we assume $u \in b\mathcal{C}_T$. (3.1.7), written with $u^+ \in bW^{1,2}([0,T];L^2) \cap L^2([0,T];F)$ as test functions, takes the form

(3.1.13)
$$\int_{t_1}^{t_2} (u_t, \partial_t(u_t^+)) dt + \int_{t_1}^{t_2} \mathcal{E}^A(u_t, u_t^+) dt + \int_{t_1}^{t_2} \int \langle A^{1/2}b, D_{A^{1/2}}u_t \rangle u_t^+ d\mu dt$$
$$= \int_{t_1}^{t_2} (f_t, u_t^+) dt + (u_{t_2}, u_{t_2}^+)) - (u_{t_1}, u_{t_1}^+)).$$

By [Ba10, Theorem 1.19] we obtain

$$\int_{t_1}^{t_2} (u_t, \partial_t(u_t^+)) dt = \frac{1}{2} (\|u_{t_2}^+\|_2^2 - \|u_{t_1}^+\|_2^2)$$

Then

(3.1.14)
$$\|u_{t_1}^+\|_2^2 + 2\int_{t_1}^{t_2} \mathcal{E}^A(u_t, u_t^+)dt + 2\int_{t_1}^{t_2} \int \langle A^{1/2}b, D_{A^{1/2}}u_t \rangle_H u_t^+ d\mu dt$$
$$= 2\int_{t_1}^{t_2} (f_t, u_t^+)dt + \|u_{t_2}^+\|_2^2.$$

Next we prove for $u \in bF$

$$(3.1.15)\qquad\qquad\qquad \mathcal{E}(u,u^+)\ge 0.$$

We have the above relation for $u \in \mathcal{D}(L)$. For $u \in bF$, by (A3) we choose a uniformly bounded sequence $\{u_n\} \subset \mathcal{D}(L)$ such that $\mathcal{E}_1^A(u_n - u) \to 0, n \to \infty$. Then we have

$$\begin{split} &|\int \langle A^{1/2}b, D_{A^{1/2}}u \rangle_{H} u^{+} d\mu - \int \langle A^{1/2}b, D_{A^{1/2}}u_{n} \rangle_{H} u_{n}^{+} d\mu |\\ &\leq |\int \langle A^{1/2}b, D_{A^{1/2}}u_{n} - D_{A^{1/2}}u \rangle_{H} u_{n}^{+} d\mu | + |\int \langle A^{1/2}b, D_{A^{1/2}}u \rangle_{H} (u_{n}^{+} - u^{+}) d\mu | \end{split}$$

$$\leq M(\int |D_{A^{1/2}}u_n - D_{A^{1/2}}u|_H^2 d\mu)^{\frac{1}{2}} + |\int \langle A^{1/2}b, D_{A^{1/2}}u \rangle_H (u_n^+ - u^+) d\mu|$$

 $\to 0, \text{ as } n \to \infty.$

Since $\mathcal{E}^{A}(u^{+}) \leq \mathcal{E}^{A}(u)$, $\sup_{n} \mathcal{E}^{A}(u_{n}^{+}) \leq \sup_{n} \mathcal{E}^{A}(u_{n}) < \infty$, we also have

$$\begin{aligned} |\mathcal{E}^{A}(u_{n},(u_{n})^{+}) - \mathcal{E}^{A}(u,u^{+})| \\ \leq |\mathcal{E}^{A}_{1}(u_{n}-u,(u_{n})^{+}) + \mathcal{E}^{A}_{1}(u,(u_{n})^{+}-u^{+})| \\ + |(u_{n}-u,(u_{n})^{+})| + |(u,(u_{n})^{+}-u^{+})| \\ \leq (\mathcal{E}^{A}_{1}(u_{n}-u))^{\frac{1}{2}}(\mathcal{E}^{0}_{1}((u_{n})^{+}))^{\frac{1}{2}} + |\mathcal{E}^{A}_{1}(u,(u_{n})^{+}-u^{+})| \\ + (||(u_{n})^{+}||_{2}||u_{n}-u||_{2} + ||(u_{n})^{+}-u^{+}||_{2}||u||_{2}) \\ \rightarrow 0, \text{ as } n \rightarrow \infty. \end{aligned}$$

As a result we obtain (3.1.15) for bounded $u \in F$. So we have

$$\|u_{t_1}^+\|_2^2 \le 2\int_{t_1}^{t_2} (f_t, u_t^+) dt + \|u_{t_2}^+\|_2^2.$$

To extend the class of solutions we are working with, to allow f to belong to $L^1(dt \times d\mu)$, we need the following proposition. It is a modified version of the above lemma.

Lemma 3.1.9 Let $u \in \hat{F}$ be bounded and $f \in L^1(dt \times d\mu)$, be such that the weak relation (3.1.7) is satisfied with test functions in $b\mathcal{C}_T$ and some function $\phi \geq 0$, $\phi \in L^2 \cap L^\infty$. Then u^+ satisfies the following relation for $0 \leq t_1 < t_2 \leq T$

$$||u_{t_1}^+||_2^2 \le 2 \int_{t_1}^{t_2} (f_s, u_s^+) ds + ||u_{t_2}^+||_2^2.$$

Proof First note that we prove analogously to the proof of Lemma 3.1.8 that for each $u \in bC_T$ satisfying the weak relation (3.1.7) with data (ϕ, f) over the interval $[t_1, t_2]$, where $\varepsilon \leq t_1 \leq t_2 \leq T$ for $\varepsilon > 0$, the following holds:

$$||u_{t_1}^+||_2^2 \le 2 \int_{t_1}^{t_2} (f_t, u_t^+) dt + ||u_{t_2}^+||_2^2.$$

For $u \in \hat{F}$ we take approximating functions u^n with data (ϕ^n, f^n) as in the last proof of Proposition 3.1.7. Then u^n satisfies the weak relation (3.1.7) for the data ϕ^n, f^n with test functions in $b\mathcal{C}_T$ over the interval $[\varepsilon, t_2]$ and $\frac{1}{n} \leq \varepsilon \leq t_2 \leq T$. Note that

$$\lim_{n \to \infty} \int_{\varepsilon}^{T} \|f_t^n - f_t\|_1 dt = 0.$$

We have

$$\|(u_{t_1}^n)^+\|_2^2 \le 2\int_{t_1}^{t_2} (f_t^n, (u_t^n)^+)dt + \|(u_{t_2}^n)^+\|_2^2,$$

where $\varepsilon \leq t_1 \leq t_2 \leq T$ for $\varepsilon > 0$. The convergence of all terms, which do not depend on f, follows by the same arguments as the proof of Lemma 3.1.8. Since u is bounded, it is easy to see that u^n is uniformly bounded. Then we have

$$\begin{split} \lim_{n \to \infty} |\int_{t_1}^{t_2} (f_s^n, (u_s^n)^+) ds - \int_{t_1}^{t_2} (f_s, u_s^+) ds| \\ \leq & M \lim_{n \to \infty} \int_{t_1}^{t_2} \|f_s^n - f_s\|_1 ds + \lim_{n \to \infty} \int_{t_1}^{t_2} (f_s, (u_s^n)^+ - u_s^+) ds \\ = & 0. \end{split}$$

Finally, we obtain that

$$||u_{t_1}^+||_2^2 \le 2 \int_{t_1}^{t_2} (f_t, u_t^+) dt + ||u_{t_2}^+||_2^2$$

where $\varepsilon \leq t_1 \leq t_2 \leq T$ for $\varepsilon > 0$. Letting $\varepsilon \to 0$ the results follows.

The next proposition is a modification of [BPS05, Proposition 2.9]. It represents a version of the maximum principle.

Proposition 3.1.10 Let $u \in \hat{F}$ be bounded and $f \in L^1(dt \times d\mu), f \ge 0$, be such that the weak relation (3.1.7) is satisfied with test functions in bC_T and some function $\phi \ge 0, \phi \in L^2 \cap L^\infty$. Then $u \ge 0$ and it is represented by the following relation:

$$u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds,$$

where we use P_t is a C_0 -semigroup on $L^1(E;\mu)$ to make $P_{s-t}f_s$ meaningful.

Proof Let $(f^n)_{n \in \mathbb{N}}$ be a sequence of bounded functions on $[0, T] \times E$ such that

$$0 \le f^n \le f^{n+1} \le f, \qquad \lim_{n \to \infty} f^n = f$$

Since f^n is bounded, we have $f^n \in L^1([0,T]; L^2)$. Next we define

$$u_t^n = P_{T-t}\phi + \int_t^T P_{s-t}f_s^n ds$$

100

Then $u^n \in \hat{F}$ is the unique generalized solution for the data (ϕ, f^n) by Proposition 3.1.7. Clearly $0 \leq u^n \leq u^{n+1}$ for $n \in \mathbb{N}$. Define $y := u^n - u$ and $\tilde{f} := f^n - f$. Then $\tilde{f} \leq 0$ and y satisfies the weak relation (3.1.7) for the data $(0, \tilde{f})$. Therefore by Lemma 3.1.9, we have for $t_1 \in [0, T]$

$$||y_{t_1}^+||_2^2 \le 2 \int_{t_1}^T (\tilde{f}_s, y_s^+) ds \le 0.$$

We conclude that $||y_{t_1}^+||_2^2 = 0$. Therefore, $u \ge u^n \ge 0$ for $n \in \mathbb{N}$. Set $v := \lim_{n \to \infty} u^n$. By relation (3.1.5) for u^n and f^n we have

$$\|u_t^n\|_2^2 + 2\int_t^T \mathcal{E}^A(u_s^n) ds \le 2\int_t^T (f_s^n, u_s^n) ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|u_s^n\|_2 ds,$$

which implies that

$$\|u_t^n\|_2^2 + 2\int_t^T \mathcal{E}^A(u_s^n) ds \le 2M \int_t^T \int |f_s^n| d\mu ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|u_s^n\|_2^2 ds$$

By Gronwall's lemma we have $\sup_n \sup_{t \in [0,T]} \|u_t^n\|_2^2 \le \text{const.}$ We obtain that $\lim_{n \to \infty} \|u_t^n - v_t\|_2^2 = 0$ and

$$\lim_{n \to \infty} \left| \int_t^T \int (f_s^n u_s^n - f_s v_s) d\mu ds \right| = 0.$$

By [MR92, Lemma 2.12] we have

$$\int_{t}^{T} \mathcal{E}^{A}(v_{s}) ds \leq \int_{t}^{T} \liminf_{n \to \infty} \mathcal{E}^{A}(u_{s}^{n}) ds \leq \liminf_{n \to \infty} \int_{t}^{T} \mathcal{E}^{A}(u_{s}^{n}) ds$$

Finally, we get for $t \in [0, T]$

$$\begin{split} \|v_t\|_2^2 + 2\int_t^T \mathcal{E}^A(v_s) ds &\leq \lim_{n \to \infty} \|u_t^n\|_2^2 + 2\liminf_{n \to \infty} \int_t^T \mathcal{E}^A(u_s^n) ds \\ &\leq \lim_{n \to \infty} (2\int_t^T (f_s^n, u_s^n) ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|u_s^n\|_2 ds) \\ &= 2\int_t^T (f_s, v_s) ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|v_s\|_2 ds. \end{split}$$

Since the right side of this inequality is finite and $t \mapsto v_t$ is L^2 -continuous, it follows that $v \in \hat{F}$.

Now we show that v satisfies the weak relation (3.1.7) for the data (ϕ, f) . As

 $\varphi^n(t) := \|u_t^n - v_t\|_2$ is continuous and decreasing to 0, we conclude by Dini's theorem

$$\lim_{n \to \infty} \sup_{t \in [0,T]} \|u_t^n - v_t\|_2 = 0,$$

and therefore

$$\lim_{n \to \infty} \int_0^T \|u_t^n - v_t\|_2^2 = 0.$$

Furthermore, there exists $K \in \mathbb{R}_+$ and a subsequence $(n_k)_{k \in \mathbb{N}}$ such that

$$\left|\int_{0}^{T} \mathcal{E}^{A}(u_{s}^{n_{k}}) ds\right| \leq K \qquad \forall k \in \mathbb{N}.$$

In particular,

$$\int_0^T \int |D_{A^{1/2}} u_s^{n_k}|_H^2 d\mu ds \le K \qquad \forall k \in \mathbb{N}.$$

We obtain

$$\lim_{k \to \infty} \int_0^T \mathcal{E}^A(u_s^{n_k}, \varphi_s) ds = \int_0^T \mathcal{E}^A(v_s, \varphi_s) ds$$

and

$$\lim_{k\to\infty}\int_0^T\int \langle A^{1/2}b, D_{A^{1/2}}u_s^{n_k}\rangle_H\varphi_s d\mu ds = \int_0^T\int \langle A^{1/2}b, D_{A^{1/2}}v_s\rangle_H\varphi_s d\mu ds,$$

which implies (3.1.7) for v associated to (ϕ, f) . Clearly u - v satisfies (3.1.7) with data (0,0) for $\varphi \in b\mathcal{C}_T$. By Proposition 3.1.7 we have u - v = 0. Since

$$v_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds,$$

the assertion follows.

Corollary 3.1.11 Let $u \in \hat{F}$ be bounded and $f \in L^1(dt \times d\mu)$ be such that the weak relation (3.1.7) is satisfied with test functions in $b\mathcal{C}_T$ and some function $\phi \in L^2 \cap L^\infty$. Assume there exists $g \in bL^1(dt \times d\mu)$ such that $f \leq g$. Then u has the following representation:

$$u_t = P_{T-t}\phi + \int_t^T P_{s-t}f_s ds.$$

Proof Define $f^n := (f \lor (-n)) \land g, n \in \mathbb{N}$. Then $(f^n)_{n \in \mathbb{N}}$ is a sequence of bounded functions such that $f^n \downarrow f$ and $f^n \leq g$ then by the same arguments as the proof of Proposition 3.1.10, the assertion follows.

3.1. Preliminaries

The following proposition is a modification of [BPS05, Proposition 2.10]. It is essential for the analytical treatment of the non-linear equation (1.4) which is done in the next section.

Proposition 3.1.12 Let $u = (u^1, ..., u^l)$ be a vector valued function where each component is a generalized solution of the linear equation (3.1.3) associated to data (f^i, ϕ^i) , which are assumed to be bounded and to satisfy the condition in Proposition 3.1.5 (ii) for i = 1, ..., l. Let ϕ, f denote the vectors $\phi = (\phi^1, ..., \phi^l), f = (f^1, ..., f^l)$ and $D_{A^{1/2}u}$ denote the matrix whose rows consist of $D_{A^{1/2}u^i}$. Then the following relations hold μ -almost everywhere:

(3.1.16)
$$|u_t|^2 + 2\int_t^T P_{s-t}(|D_{A^{1/2}}u_s|_H^2)ds = P_{T-t}|\phi|^2 + 2\int_t^T P_{s-t}\langle u_s, f_s\rangle ds$$

and

$$(3.1.17) |u_t| \le P_{T-t} |\phi| + \int_t^T P_{s-t} \langle \hat{u}_s, f_s \rangle ds$$

Here we write $\hat{x} = x/|x|$, for $x \in \mathbb{R}^l$, $x \neq 0$ and $\hat{x} = 0$, if x = 0.

Proof By Proposition 3.1.5 (ii) we have $u \in b\mathcal{C}_T$.

First we assume l = 1. If we check that u^2 satisfies (3.1.7) with data $(2uf - 2|D_{A^{1/2}}u|_{H}^{2}, \phi^2)$ for $\varphi \in bC_T$, then (3.1.16) will follow by Corollary 3.1.11. We have the following relations:

$$\int_{0}^{T} (u_{t}^{2}, \partial_{t}\varphi_{t})dt = 2 \int_{0}^{T} (u_{t}, \partial_{t}(u_{t}\varphi_{t}))dt + (u_{0}^{2}, \varphi_{0}) - (u_{T}^{2}, \varphi_{T}),$$
$$\mathcal{E}^{A}(u_{t}^{2}, \varphi_{t}) = 2\mathcal{E}^{A}(u_{t}, u_{t}\varphi_{t}) - 2(|D_{A^{1/2}}u_{t}|_{H}^{2}, \varphi_{t}),$$

and

$$\int \langle A^{1/2}b, D_{A^{1/2}}(u_t^2) \rangle_H \varphi_t d\mu = 2 \int \langle A^{1/2}b, D_{A^{1/2}}u_t \rangle_H u_t \varphi_t d\mu.$$

Since u is a generalized solution of (3.1.3), we have

$$\int_0^T (u_t, \partial_t(u_t\varphi_t))dt - (u_T, u_T\varphi_T) + (u_0, u_0\varphi_0) - \int_0^T (f_t, u_t\varphi_t)dt$$
$$= -\int_0^T \mathcal{E}^A(u_t, u_t\varphi_t)dt - \int_0^T \langle A^{1/2}b, D_{A^{1/2}}u_t \rangle_H u_t\varphi_t d\mu dt.$$

By the above relation, we obtain

$$\begin{aligned} &(3.1.18) \\ &\int_0^T (u_t^2, \partial_t \varphi_t) dt + (u_0^2, \varphi_0) - (u_T^2, \varphi_T) + \int_0^T (\mathcal{E}^A(u_t^2, \varphi_t) + \langle A^{1/2}b, D_{A^{1/2}}(u_t^2) \rangle_H \varphi_t d\mu) dt \\ &= 2 \int_0^T (f_t u_t, \varphi_t) dt - \int_0^T 2(|D_{A^{1/2}} u_t|_H^2, \varphi_t) dt. \end{aligned}$$

Hence, by Corollary 3.1.11, (3.1.16) holds in the case l = 1. To deduce this relation in the case l > 1 it suffices to add the relations corresponding to the components $|u_t^i|^2, i = 1, ..., l$. For (3.1.17), let us define for $\varepsilon > 0$, $h_{\varepsilon}(t) := \sqrt{t + \varepsilon} - \sqrt{\varepsilon}$ for $t \ge 0$. We have the following relations by integration by parts:

$$\begin{aligned} \mathcal{E}^{A}(h_{\varepsilon}(|u|^{2}),\varphi) &= \mathcal{E}^{A}(|u|^{2},h_{\varepsilon}'(|u|^{2})\varphi) - (h_{\varepsilon}''(|u|^{2})|D_{A^{1/2}}(|u|^{2})|_{H}^{2},\varphi), \\ \int_{0}^{T}(h_{\varepsilon}(|u_{t}|^{2}),\partial_{t}\varphi_{t})dt &= \int_{0}^{T}(|u_{t}|^{2},\partial_{t}(\varphi_{t}h_{\varepsilon}'(|u_{t}|^{2})))dt - (|u_{T}|^{2},\varphi_{T}h_{\varepsilon}'(|u_{T}|^{2})) \\ &+ (|u_{0}|^{2},\varphi_{0}h_{\varepsilon}'(|u_{0}|^{2})) + (h_{\varepsilon}(|u_{T}|^{2}),\varphi_{T}) - (h_{\varepsilon}(|u_{0}|^{2}),\varphi_{0}). \end{aligned}$$

If we choose $\varphi h_{\varepsilon}'(|u|^2)$ as a test function in (3.1.18), we have

$$\begin{split} &\int_{0}^{T} (|u_{t}|^{2}, \partial_{t}(\varphi_{t}h_{\varepsilon}'(|u_{t}|^{2})))dt + (|u_{0}|^{2}, \varphi_{0}h_{\varepsilon}'(|u_{0}|^{2})) - (|u_{T}|^{2}, \varphi_{T}h_{\varepsilon}'(|u_{T}|^{2})) \\ &+ \int_{0}^{T} (\mathcal{E}^{A}(|u_{t}|^{2}, \varphi_{t}h_{\varepsilon}'(|u|_{t}^{2})) + \int \langle A^{1/2}b, D_{A^{1/2}}(|u_{t}|^{2}) \rangle_{H}\varphi_{t}h_{\varepsilon}'(|u_{t}|^{2})d\mu)dt \\ &= 2\int_{0}^{T} (\langle f_{t}, u_{t} \rangle, \varphi_{t}h_{\varepsilon}'(|u_{t}|^{2}))dt - \int_{0}^{T} 2(|D_{A^{1/2}}u_{t}|_{H}^{2}, \varphi_{t}h_{\varepsilon}'(|u_{t}|^{2}))dt. \end{split}$$

By the above relations we have

$$\begin{split} &\int_{0}^{T} (h_{\varepsilon}(|u_{t}|^{2}),\partial_{t}\varphi_{t})dt - (h_{\varepsilon}(|u_{T}|^{2}),\varphi_{T}) + (h_{\varepsilon}(|u_{0}|^{2}),\varphi_{0}) \\ &+ \int_{0}^{T} (\mathcal{E}^{A}(h_{\varepsilon}(|u_{t}|^{2}),\varphi_{t}) + \int \langle A^{1/2}b, D_{A^{1/2}}(h_{\varepsilon}(|u_{t}|^{2})) \rangle_{H}\varphi_{t}d\mu)dt \\ &= -\int_{0}^{T} (h_{\varepsilon}''(|u_{t}|^{2})|D_{A^{1/2}}(|u_{t}|^{2})|_{H}^{2},\varphi_{t})dt + 2\int_{0}^{T} (\langle f_{t}, u_{t} \rangle h_{\varepsilon}'(|u_{t}|^{2}),\varphi_{t})dt \\ &- \int_{0}^{T} 2(h_{\varepsilon}'(|u_{t}|^{2})|D_{A^{1/2}}u_{t}|_{H}^{2},\varphi_{t})dt. \end{split}$$

As

$$|D_{A^{1/2}}(|u|^2)|_H^2 = 4|\sum_i u^i D_{A^{1/2}} u^i|_H^2 \le 4(\sum_i |u^i| \cdot |D_{A^{1/2}} u^i|_H)^2 \le 4(\sum_i (u^i)^2)(\sum_i |D_{A^{1/2}} u^i|_H^2),$$

3.1. Preliminaries

we deduce

$$\begin{split} & 2\langle f, u \rangle h_{\varepsilon}'(|u|^2) - 2h_{\varepsilon}'(|u|^2)|D_{A^{1/2}}u|_{H}^2 - h_{\varepsilon}''(|u|^2)|D_{A^{1/2}}(|u|^2)|_{H}^2 \\ & = \frac{\langle f, u \rangle - |D_{A^{1/2}}u|_{H}^2}{(|u|^2 + \varepsilon)^{\frac{1}{2}}} + \frac{|\sum_{i} u^i D_{A^{1/2}}u|_{H}^2}{(|u|^2 + \varepsilon)^{\frac{3}{2}}} \\ & \leq \frac{\langle f, u \rangle}{(|u|^2 + \varepsilon)^{\frac{1}{2}}} - \frac{\varepsilon |D_{A^{1/2}}u|_{H}^2 + |u|^2 |D_{A^{1/2}}u|_{H}^2 - \sum_{i} (u^i)^2 \sum_{i} \langle D_{A^{1/2}}u^i, D_{A^{1/2}}u^i \rangle_{H}}{(|u|^2 + \varepsilon)^{\frac{3}{2}}} \\ & \leq \frac{\langle f, u \rangle}{(|u|^2 + \varepsilon)^{\frac{1}{2}}}. \end{split}$$

By Proposition 3.1.10 we deduce that

$$h_{\varepsilon}(|u_t|^2) \le P_{T-t}h_{\varepsilon}(|\phi|^2) + \int_t^T P_{s-t}\frac{\langle f_s, u_s \rangle}{(|u_s|^2 + \varepsilon)^{\frac{1}{2}}} ds$$

Letting $\varepsilon \to 0$, the assertion follows.

The next corollary is a version of the above proposition for general data. Here we use P_t is a C_0 -semigroup on L^1 .

Corollary 3.1.13 Let $u = (u^1, ..., u^l)$ be a vector valued function where each component is a generalized solution of the linear equation (3.1.3) associated to data $f^i \in L^1([0,T]; L^2), \phi^i \in L^2$ for i = 1, ..., l. Let ϕ, f denote the vectors $\phi = (\phi^1, ..., \phi^l), f = (f^1, ..., f^l)$ and $D_{A^{1/2}}u$ denote the matrix whose rows consist of $D_{A^{1/2}}u^i$. Then the following relations hold μ -almost everywhere

(3.1.19)
$$|u_t|^2 + 2\int_t^T P_{s-t}(|D_{A^{1/2}}u_s|_H^2)ds = P_{T-t}|\phi|^2 + 2\int_t^T P_{s-t}\langle u_s, f_s\rangle ds,$$

and

$$(3.1.20) |u_t| \le P_{T-t} |\phi| + \int_t^T P_{s-t} \langle \hat{u}_s, f_s \rangle ds.$$

Proof Analogously to the proof of Proposition 3.1.12 it is enough to verify (3.1.19) for l = 1. For $\phi \in L^2$, $f \in L^1([0,T], L^2)$, take ϕ_n , f_n as in Proposition 3.1.7. Then we have

- (a). $u_{n,t} := P_{T-t}\phi_n + \int_t^T P_{s-t}f_{n,s}ds$ is a generalized solution,
- (b). $\lim_{n\to\infty} \int_t^T \|f_{n,s} f_s\|_2 ds = 0,$
- (c). $\lim_{n \to \infty} \|\phi_n \phi\|_2 = 0$,
- (d). $\lim_{n \to \infty} ||u_n u||_T = 0.$

By Proposition 3.1.12 we have

$$(3.1.21) \quad |u_{n,t}|^2 + 2\int_t^T P_{s-t}(|D_{A^{1/2}}u_{n,s}|_H^2)ds = P_{T-t}|\phi_n|^2 + 2\int_t^T P_{s-t}\langle u_{n,s}, f_{n,s}\rangle ds$$

By (b) and (d) we obtain

$$\begin{split} &\|\int_{t}^{T} P_{s-t}((u_{n,s}, f_{n,s}) - (u_{s}, f_{s}))ds\|_{1} \\ \leq &C\int_{t}^{T}(\|u_{n,s}\|_{2}\|f_{n,s} - f_{s}\|_{2} + \|f_{s}\|_{2}\|u_{n,s} - u_{s}\|_{2})ds \\ \leq &C(\sup_{s \in [0,T]} \|u_{n,s}\|_{2} \int_{t}^{T} \|f_{n,s} - f_{s}\|_{2}ds + \sup_{s \in [0,T]} \|u_{n,s} - u_{s}\| \int_{t}^{T} \|f_{s}\|_{2}ds) \\ \rightarrow &0, \text{ as } n \to \infty. \end{split}$$

Here we used P_t is a C_0 -semigroup on $L^1(E;\mu)$. By (d) we obtain

$$\begin{split} &\int_{t}^{T} \||D_{A^{1/2}}u_{n,s}|_{H}^{2} - |D_{A^{1/2}}u_{s}|_{H}^{2}\|_{1}ds \\ \leq &((\int_{t}^{T} \||D_{A^{1/2}}u_{n,s}|_{H}\|_{2}^{2}ds)^{\frac{1}{2}} + (\int_{t}^{T} \||D_{A^{1/2}}u_{s}|_{H}\|_{2}^{2}ds)^{\frac{1}{2}})(\int_{t}^{T} \||D_{A^{1/2}}u_{n,s} - D_{A^{1/2}}u_{s}|_{H}\|_{2}^{2}ds)^{\frac{1}{2}} \\ = &((\int_{t}^{T} \mathcal{E}^{A}(u_{n,s})ds)^{\frac{1}{2}} + (\int_{t}^{T} \mathcal{E}^{A}(u_{s})ds)^{\frac{1}{2}})(\int_{t}^{T} \mathcal{E}^{A}(u_{n,s} - u_{s})ds)^{\frac{1}{2}} \\ \to 0, \text{ as } n \to \infty, \end{split}$$

and obtain

$$\lim_{n \to \infty} \int_t^T P_{s-t}(|D_{A^{1/2}}u_{n,s}|_H^2) ds = \int_t^T P_{s-t}|D_{A^{1/2}}u_s|_H^2 ds.$$

Passing to the limit in equation (3.1.21) we obtain (3.1.19). (3.1.20) follows by the same method.

Lemma 3.1.14 If $f, g \in L^1([0,T]; L^2)$ and $\phi \in L^2$, then the following relations hold μ -a.e.:

(3.1.22)
$$\int_{t}^{T} P_{s-t}(f_{s}P_{T-s}\phi)ds \leq \frac{1}{2}P_{T-t}\phi^{2} + \int_{t}^{T} \int_{s}^{T} P_{s-t}(f_{s}P_{r-s}f_{r})drds$$

Proof Define

$$h_t = P_{T-t}\phi, \qquad v_t = \int_t^T P_{s-t}f_s ds$$

By (3.1.19) we deduce

$$h_t^2 + 2\int_t^T P_{s-t} |D_{A^{1/2}}h_s|_H^2 ds = P_{T-t}\phi^2,$$
$$v_t^2 + 2\int_t^T P_{s-t} |D_{A^{1/2}}v_s|_H^2 ds = 2\int_t^T P_{s-t} (f_s \int_s^T P_{r-s}f_r dr) ds,$$

and

$$h_t v_t + 2 \int_t^T P_{s-t} \langle D_{A^{1/2}} h_s, D_{A^{1/2}} v_s \rangle_H ds = \int_t^T P_{s-t} (f_s P_{T-s} \phi) ds.$$

So, we have

$$\int_{t}^{T} P_{s-t}(f_{s}P_{T-s}\phi)ds = h_{t}v_{t} + 2\int_{t}^{T} P_{s-t}\langle D_{A^{1/2}}h_{s}, D_{A^{1/2}}v_{s}\rangle_{H}ds$$

$$\leq \frac{1}{2}(h_{t}^{2} + v_{t}^{2}) + \int_{t}^{T} P_{s-t}(|D_{A^{1/2}}h_{s}|_{H}^{2} + |D_{A^{1/2}}v_{s}|_{H}^{2})ds$$

$$= \frac{1}{2}P_{T-t}\phi^{2} + \int_{t}^{T} \int_{s}^{T} P_{s-t}(f_{s}P_{r-s}f_{r})drds.$$

3.2 The Non-linear Equation

In the case of non-linear equations, we are going to treat systems of equations, with the unknown functions and their first-order derivatives mixed in the non-linear term of the equation. The non-linear term is a given measurable function $f : [0, T] \times E \times \mathbb{R}^l \times H^l \to \mathbb{R}^l$, $l \in \mathbb{N}$. We are going to treat the following system of equations.

(3.2.1)
$$(\partial_t + L)u + f(\cdot, \cdot, u, D_{A^{1/2}}u) = 0, \qquad u_T = \phi.$$

The function ϕ is assumed to be in $L^2(E, d\mu; \mathbb{R}^l)$.

Definition 3.2.1 [Generalized solution of the nonlinear equation] A generalized solution of equation (3.2.1) is a system $u = (u^1, u^2, ..., u^l)$ of l elements in \hat{F} , which has the property that each function $f^i(\cdot, \cdot, u, D_{A^{1/2}}u)$ belongs to $L^1([0, T]; L^2(\mu))$ and such that there is a sequence $\{u_n\}$ which consists of strong solutions to (3.2.1) with data (ϕ_n, f_n) such that

$$||u_n - u||_T \to 0, ||\phi_n - \phi||_2 \to 0,$$

and

$$\lim_{n \to \infty} f_n(\cdot, \cdot, u_n, D_{A^{1/2}}u_n) = f(\cdot, \cdot, u, D_{A^{1/2}}u) \text{ in } L^1([0, T]; L^2(\mu))$$

Definition 3.2.2 [*Mild solution*] A mild solution of equation (3.2.1) is a system $u = (u^1, u^2, ..., u^l)$ of l elements in \hat{F} , which has the property that each function $f^i(\cdot, \cdot, u, D_{A^{1/2}}u)$ belongs to $L^1([0, T]; L^2(\mu))$ and such that for every $i \in \{1, ..., l\}$, the following equation holds

(3.2.2)
$$u^{i}(t,x) = P_{T-t}\phi^{i}(x) + \int_{t}^{T} P_{s-t}f^{i}(s,\cdot,u_{s},D_{A^{1/2}}u_{s})(x)ds, \mu - a.e.$$

Lemma 3.2.3 u is a generalized solution of the nonlinear equation (3.2.1) if and only if it is a mild solution of equation (3.2.1).

Proof The assertion follows by Proposition 3.1.7.

We will use the following notation:

$$\begin{split} |u|_{H} &:= \sum |u^{i}|_{H}, u \in L^{2}(E; H^{l}, d\mu), \\ \|\phi\|_{2}^{2} &:= \sum_{i=1}^{l} \|\phi^{i}\|_{2}^{2}, \phi \in L^{2}(E, d\mu; \mathbb{R}^{l}), \\ \mathcal{E}(u, v) &:= \sum_{i=1}^{l} \mathcal{E}(u^{i}, v^{i}), \ \mathcal{E}^{A}(u, v) := \sum_{i=1}^{l} \mathcal{E}^{A}(u^{i}, v^{i}), u, v \in F^{l}, \\ \|u\|_{T}^{2} &:= \sup_{t \leq T} \|u_{t}\|_{2}^{2} + \int_{0}^{T} \mathcal{E}^{A}(u_{t}) dt, u \in \hat{F}^{l}. \end{split}$$

3.2.1 The Case of Lipschitz Conditions

In this subsection we consider a measurable function $f:[0,T]\times E\times \mathbb{R}^l\times H^l\to \mathbb{R}^l$ such that

$$(3.2.3) |f(t, x, y, z) - f(t, x, y', z')| \le C(|y - y'| + |z - z'|_H),$$

with t, x, y, y', z, z' arbitrary and C a constant independent of t, x. We set $f^0(t, x) := f(t, x, 0, 0)$.

Proposition 3.2.4 Suppose that conditions (A1)-(A3) hold and f satisfies condition (3.2.3), $f^0 \in L^2([0,T] \times E, dt \times d\mu; \mathbb{R}^l)$ and $\phi \in L^2(E; \mathbb{R}^l)$. Then the equation

(3.2.1) admits a unique solution $u \in \hat{F}^l$ and it satisfies the following estimate

$$||u||_T^2 \le e^{T(1+2C+C^2+2\alpha)} (||\phi||_2^2 + ||f^0||_{L^2([0,T]\times E)}^2).$$

Proof If $u \in \hat{F}^l$, then by relation (3.2.3) we have

$$|f(\cdot, \cdot, u, D_{A^{1/2}}u)| \le |f(\cdot, \cdot, u, D_{A^{1/2}}u) - f(\cdot, \cdot, 0, 0)| + |f(\cdot, \cdot, 0, 0)|$$
$$\le C(|u| + |D_{A^{1/2}}u|_H) + |f^0|.$$

As $f^0 \in L^2([0,T] \times E, dt \times d\mu; \mathbb{R}^l)$ and $|D_{A^{1/2}}u|_H$ is an element of $L^2([0,T] \times E)$, we get $f(\cdot, \cdot, u, D_{A^{1/2}}u) \in L^2([0,T] \times E; \mathbb{R}^l)$.

Now we define the operator $A: \hat{F}^l \to \hat{F}^l$ by

$$(Au)^{i}(t,x) = P_{T-t}\phi^{i}(x) + \int_{t}^{T} P_{s-t}f^{i}(s,\cdot,u_{s},D_{A^{1/2}}u_{s})(x)ds, \qquad i = 1,...,l.$$

Then Proposition 3.1.7 implies that $Au \in \hat{F}^l$. In the following we write $f_{u,s}^i := f^i(s, \cdot, u_s, D_{A^{1/2}}u_s)$. Since $(Au)_t^i - (Av)_t^i = \int_t^T P_{s-t}(f_{u,s}^i - f_{v,s}^i)ds$ is the mild solution with data $(f_u^i - f_v^i, 0)$, by the same argument as the proof of Proposition 3.1.7 we have

$$\begin{split} \| \int_{t}^{T} P_{s-t}(f_{u,s}^{i} - f_{v,s}^{i}) ds \|_{[t,T]}^{2} &\leq M_{T}(\int_{t}^{T} \| f_{u,s} - f_{v,s} \|_{2} ds)^{2} \\ &\leq M_{T}(T-t) \int_{t}^{T} \| f_{u,s} - f_{v,s} \|_{2}^{2} ds \\ &\leq M_{T}(T-t) \int_{t}^{T} (\| u_{s} - v_{s} \|_{2}^{2} + \| \| D_{A^{1/2}} u_{s} - D_{A^{1/2}} v_{s} \|_{H} \|_{2}^{2}) ds \\ &\leq M_{T}(T-t) \| u - v \|_{[t,T]}^{2}, \end{split}$$

where M_T may change from line to line. Here

$$||u||_{[T_a,T_b]} := (\sup_{t \in [T_a,T_b]} ||u_t||_2^2 + \int_{T_a}^{T_b} \mathcal{E}^A(u_t) dt)^{\frac{1}{2}},$$

where $0 \le T_a \le T_b \le T$. Fix T_1 sufficiently small such that $\gamma := M_T(T - T_1) < 1$. Then we have :

$$||Au - Av||_{[T_1,T]}^2 < \gamma ||u - v||_{[T_1,T]}^2$$

Then there exists a unique $u_1 \in \hat{F}_{[T_1,T]}$ such that $Au_1 = u_1$ where $\hat{F}_{[T_a,T_b]} := C([T_a,T_b];L^2) \cap L^2((T_a,T_b);F)$ for $T_a \in [0,T]$ and $T_b \in [T_a,T]$.

By the same method as above, we define the operator $A^1: \hat{F}^l \to \hat{F}^l$ by

$$(A^{1}u)^{i}(t,x) = P_{T_{1}-t}u_{1}^{i}(T_{1},x) + \int_{t}^{T_{1}} P_{s-t}f^{i}(s,\cdot,u_{s},D_{A^{1/2}}u_{s})(x)ds, \qquad i=1,...,l.$$

Then we have

$$||A^{1}u - A^{1}v||_{[t,T_{1}]}^{2} \le M_{T}(T_{1} - t)||u - v||_{[t,T_{1}]}^{2}.$$

Now we choose $T_2 < T_1$ such that $M_T(T_1 - T_2) < 1$. Then we have that there exists a unique $u_2 \in \hat{F}_{[T_2,T_1]}$ such that $A^1u_2 = u_2$. Define $u := u_1 \mathbf{1}_{[T_1,T]} + u_2 \mathbf{1}_{[T_2,T_1]}$. By a similar argument as in the proof of Proposition 2.2.4, u is a solution on $[T_2, T]$. Therefore, we construct a solution over the interval $[T_2, T]$. Clearly there exists $n \in \mathbb{N}$ such that $T < n(T - T_1)$. Hence, the construction is done after n steps.

In order to obtain the estimate in the statement, we write

$$\begin{split} &|\int_{t}^{T} (f_{u,s}, u_{s})ds| \\ &\leq \int_{t}^{T} |(f_{s}^{0}, u_{s})|ds + C \int_{t}^{T} \|u_{s}\|_{2}^{2}ds + C \int_{t}^{T} \||D_{A^{1/2}}u_{s}|_{H}\|_{2} \|u_{s}\|_{2}ds \\ &\leq \frac{1}{2} \int_{t}^{T} \|f_{s}^{0}\|_{2}^{2}ds + (\frac{1}{2} + C + \frac{1}{2}C^{2}) \int_{t}^{T} \|u_{s}\|_{2}^{2}ds + \frac{1}{2} \int_{t}^{T} \mathcal{E}^{A}(u_{s})ds. \end{split}$$

By relation (3.1.5) of Proposition 3.1.7 it follows that

$$\begin{aligned} \|u_t\|_2^2 &+ 2\int_t^T \mathcal{E}^A(u_s)ds \le 2\int_t^T (f_{u,s}, u_s)ds + \|\phi\|_2^2 + 2\alpha \int_t^T \|u_s\|_2^2 ds \\ &\le \|\phi\|_2^2 + \int_t^T \|f_s^0\|_2^2 ds + (1 + 2C + C^2 + 2\alpha) \int_t^T \|u_s\|_2^2 ds + \int_t^T \mathcal{E}^A(u_s)ds. \end{aligned}$$

Now by Gronwall's lemma the desired estimate follows.

[Uniqueness] Let u_1 and u_2 be two solutions of equation (3.2.1). By using (3.1.5) for the difference $u_1 - u_2$ we get

$$\begin{split} \|u_{1,t} - u_{2,t}\|_{2}^{2} &+ 2\int_{t}^{T} \mathcal{E}^{A}(u_{1,s} - u_{2,s})ds \\ \leq & 2\int_{t}^{T} (f(s, \cdot, u_{1,s}, D_{A^{1/2}}u_{1,s}) - f(s, \cdot, u_{2,s}, D_{A^{1/2}}u_{2,s}), u_{1,s} - u_{2,s})ds + 2\alpha \int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds \\ \leq & 2\int_{t}^{T} C(|D_{A^{1/2}}u_{1,s} - D_{A^{1/2}}u_{2,s}|, |u_{1,s} - u_{2,s}|)ds + (2\alpha + C)\int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds \\ \leq & (2\alpha + C^{2} + C)\int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds + \int_{t}^{T} \mathcal{E}^{A}(u_{1,s} - u_{2,s})ds. \end{split}$$

By Gronwall's lemma it follows that

$$||u_{1,t} - u_{2,t}||_2^2 = 0,$$

hence $u_1 = u_2$.

3.2.2 The Case of Monotonicity Conditions

Let $f : [0,T] \times E \times \mathbb{R}^l \times H^l \to \mathbb{R}^l$ be a measurable function and $\phi \in L^2(E,\mu;\mathbb{R}^l)$ be the final condition of (3.2.1). In this subsection we impose the following conditions:

(H1) [Lipschitz condition in z] There exists a fixed constant C > 0 such that for t, x, y, z, z' arbitrary

$$|f(t, x, y, z) - f(t, x, y, z')| \le C|z - z'|_{H}.$$

(H2) [Monotonicity condition in y] For x, y, y', z arbitrary, there exists a function $\mu \in L^1([0, T]; \mathbb{R})$ such that

$$\langle y-y', f(t,x,y,z) - f(t,x,y',z) \rangle \leq \mu_t |y-y'|^2$$

We set $\alpha_t := \int_0^t \mu_s ds$.

(H3) [Continuity condition in y] For t, x and z fixed, the map

$$\mathbb{R}^l \ni y \mapsto f(t, x, y, z)$$

is continuous.

We need the following notation

$$f^{0}(t,x) := f(t,x,0,0), \qquad f'(t,x,y) := f(t,x,y,0) - f(t,x,0,0),$$
$$f^{\prime,r}(t,x) := \sup_{|y| \le r} |f'(t,x,y)|.$$

- (H4) For each $r > 0, f'^{,r} \in L^1([0,T]; L^2).$
- (H5) $\|\phi\|_{\infty} < \infty, \|f^0\|_{\infty} < \infty.$

As $\mu(E) < \infty$ we have $|\phi| \in L^2$, $|f^0| \in L^2([0,T]; L^2)$. The conditions (H1), (H4), and (H5) imply that if $u \in \hat{F}$ is bounded, then $|f(u, D_{A^{1/2}}u)| \in L^1([0,T]; L^2)$. Under the above conditions, even if E is equal to a Hilbert space, it seems impossible to apply general monotonicity methods to the map $\mathcal{V} \ni u \mapsto f(t, \cdot, u(\cdot), D_{A^{1/2}}u) \in \mathcal{V}'$

because of lack of a suitable reflexive Banach space \mathcal{V} such that $\mathcal{V} \subset \mathcal{H} \subset \mathcal{V}'$. Therefore, also here we proceed developing a hands-on approach to prove existence and uniqueness of solutions for equation (3.2.1) as done in [BPS05], [S09] and in particular, Chapter 2. Then by the same arguments as the proof of Lemma 2.2.5, the following lemma follows:

Lemma 3.2.5 In (H2) without loss of generality we assume that $\mu_t \equiv 0$.

Lemma 3.2.6 Suppose that conditions (A1)-(A3), (H1) and the following weaker form of condition (H2) (with $\mu_t \equiv 0$) hold,

$$(H2')\langle y, f'(t, x, y)\rangle \le 0,$$

for all t, x, y. If u is a solution of (3.2.1), then there exists a constant K which depends on C, T, α such that

$$||u||_T^2 \le K(||\phi||_2^2 + \int_0^T ||f_t^0||_2^2 dt).$$

Proof Since u is a generalized solution of (3.2.1), we have by Proposition 3.1.7

$$\|u_t\|_2^2 + 2\int_t^T \mathcal{E}^A(u_s)ds \le 2\int_t^T (f_s, u_s)ds + \|u_T\|_2^2 + 2\alpha\int_t^T \|u_s\|_2^2ds.$$

Conditions (H1) and (H2') yield

$$\begin{split} \langle f_s(u_s, D_{A^{1/2}}u_s), u_s \rangle = & \langle f_s(u_s, D_{A^{1/2}}u_s) - f_s(u_s, 0) + f'_s(u_s) + f^0_s, u_s \rangle \\ \leq & |f_s(u_s, D_{A^{1/2}}u_s) - f_s(u_s, 0)| |u_s| + \langle f'_s(u_s), u_s \rangle + |f^0_s| |u_s| \\ \leq & (C|D_{A^{1/2}}u_s|_H + |f^0_s|) |u_s|. \end{split}$$

Hence, it follows that

$$\begin{aligned} \|u_t\|_2^2 + 2\int_t^T \mathcal{E}^A(u_s)ds \\ \leq & 2\int_t^T \int (C|D_{A^{1/2}}u_s|_H + |f_s^0|)|u_s|d\mu ds + \|u_T\|_2^2 + 2\alpha \int_t^T \|u_s\|_2^2 ds \\ \leq & \int_t^T \mathcal{E}^A(u_s)ds + (C^2 + 1 + 2\alpha) \int_t^T \|u_s\|_2^2 ds + \int_t^T \|f_s^0\|_2^2 ds + \|u_T\|_2^2. \end{aligned}$$

Then Gronwall's lemma yields

$$||u||_T^2 \le K(||\phi||_2^2 + \int_0^T ||f_t^0||_2^2 dt).$$

Lemma 3.2.7 Assume that the conditions (A1)-(A3), (H1) and (H2') hold. If u is a generalized solution of (3.2.1), then there exists a constant K, which depends on C, α and T such that

(3.2.4)
$$||u||_{\infty} \le K(||\phi||_{\infty} + ||f^0||_{\infty}).$$

Proof By Corollary 3.1.13, we have

$$(3.2.5) |u_t|^2 + 2\int_t^T P_{s-t}(|D_{A^{1/2}}u_s|_H^2)ds = P_{T-t}|\phi|^2 + 2\int_t^T P_{s-t}\langle u_s, f_s(u_s, D_{A^{1/2}}u_s)\rangle ds.$$

Follow the same arguments as the proof of Lemma 3.2.6 we deduce

 $\langle f_s(u_s, D_{A^{1/2}}u_s), u_s \rangle \le (C|D_{A^{1/2}}u_s|_H + |f_s^0|)|u_s|.$

By Corollary 3.1.13 (3.1.20) we get

$$|u_s| \le P_{T-s}|\phi| + \int_s^T P_{r-s}(C|D_{A^{1/2}}u_r|_H + |f_r^0|)dr.$$

Then we have

$$\begin{split} &\int_{t}^{T} P_{s-t} \langle f_{s}(u_{s}, D_{A^{1/2}}u_{s}), u_{s} \rangle ds \\ &\leq \int_{t}^{T} P_{s-t} [(P_{T-s}|\phi| + \int_{s}^{T} P_{r-s}(C|D_{A^{1/2}}u_{r}|_{H} + |f_{r}^{0}|)dr)(C|D_{A^{1/2}}u_{s}|_{H} + |f_{s}^{0}|)] ds. \end{split}$$

So by (3.2.5) and Lemma 3.1.14 we have

$$\begin{split} &|u_t|^2 + 2\int_t^T P_{s-t}(|D_{A^{1/2}}u_s|^2)ds \\ \leq &P_{T-t}|\phi|^2 + 2(\int_t^T P_{s-t}[(P_{T-s}|\phi| + \int_s^T P_{r-s}(C|D_{A^{1/2}}u_r| + |f_r^0|)dr)(C|D_{A^{1/2}}u_s| + |f_s^0|)]ds) \\ \leq &3P_{T-t}|\phi|^2 + 2C^2\int_t^T \int_s^T P_{s-t}(|D_{A^{1/2}}u_s|P_{r-s}|D_{A^{1/2}}u_r|)drds + 2\int_t^T \int_s^T P_{s-t}(|f_s^0|P_{r-s}|f_r^0|)drds \\ &+ 2\int_t^T \int_s^T P_{s-t}[P_{r-s}(C|D_{A^{1/2}}u_r| + |f_r^0|)(C|D_{A^{1/2}}u_s| + |f_s^0|)]drds. \end{split}$$

Furthermore,

$$\begin{split} &\int_{t}^{T}\int_{s}^{T}P_{s-t}[P_{r-s}(C|D_{A^{1/2}}u_{r}|+|f_{r}^{0}|)(C|D_{A^{1/2}}u_{s}|+|f_{s}^{0}|)]drds \\ \leq &\frac{1}{2}\int_{t}^{T}\int_{s}^{T}[P_{s-t}(C|D_{A^{1/2}}u_{s}|+|f_{s}^{0}|)^{2}]+P_{s-t}[(P_{r-s}(C|D_{A^{1/2}}u_{r}|+|f_{r}^{0}|))^{2}]drds \\ \leq &\int_{t}^{T}\int_{s}^{T}C^{2}P_{s-t}|D_{A^{1/2}}u_{s}|^{2}+P_{s-t}|f_{s}^{0}|^{2}+\frac{1}{2}P_{r-t}(C|D_{A^{1/2}}u_{r}|+|f_{r}^{0}|)^{2}drds \\ \leq &2C^{2}(T-t)\int_{t}^{T}P_{s-t}|D_{A^{1/2}}u_{s}|^{2}ds+2(T-t)\int_{t}^{T}P_{s-t}|f_{s}^{0}|^{2}ds. \end{split}$$

By Schwartz's inequality one has

$$\begin{split} &\int_{t}^{T}\int_{s}^{T}P_{s-t}(|D_{A^{1/2}}u_{s}|_{H}P_{r-s}|D_{A^{1/2}}u_{r}|_{H})drds \\ &\leq \int_{t}^{T}\int_{s}^{T}\frac{1}{2}(P_{s-t}|D_{A^{1/2}}u_{s}|_{H}^{2})drds + \int_{t}^{T}\int_{s}^{T}\frac{1}{2}(P_{r-t}|D_{A^{1/2}}u_{r}|_{H}^{2})drds \\ &\leq (T-t)\int_{t}^{T}P_{s-t}|D_{A^{1/2}}u_{s}|_{H}^{2}ds. \end{split}$$

Hence we conclude

$$\begin{split} &|u_t|^2 + 2\int_t^T P_{s-t}(|D_{A^{1/2}}u_s|_H^2)ds\\ \leq &2P_{T-t}|\phi|^2 + 6C^2(T-t)\int_t^T P_{s-t}|D_{A^{1/2}}u_s|_H^2ds + 6(T-t)\int_t^T \int_s^T P_{s-t}(|f_s^0|^2)drds. \end{split}$$

Hence, we deduce by iteration the estimate over the interval [0, T]. We obtain from the above estimate:

$$|u_t|^2 \leq \sup_{t \in [0,T]} \sup_{x \in E} \tilde{K}(P_{T-t}|\phi|^2 + (T-t) \int_t^T P_{s-t}|f_s^0|^2 ds)$$

$$\leq \sup_{t \in [0,T]} \tilde{K}(\|\phi^2\|_{\infty} + T^2\|f^0\|_{\infty}^2)$$

$$\leq K^2(\|\phi\|_{\infty}^2 + \|f^0\|_{\infty}^2),$$

which implies (3.2.4).

The proof here is different from the finite dimensional case, i.e. the proof of Theorem 2.2.8, since a unit ball in H is not compact. And it is inspired from the probabilistic approach to prove the existence of the solution of the BSDE of [BDHPS03].

Theorem 3.2.8 Suppose the conditions (A1)-(A3), (H1)-(H5) hold. Then there exists a unique generalized solution of equation (3.2.1), and it satisfies the following estimates with constants K_1 and K_2 independent of u, ϕ, f

$$||u||_T^2 \le K_1(||\phi||_2^2 + \int_0^T ||f_t^0||_2^2 dt),$$

and

$$||u||_{\infty} \le K_2(||\phi||_{\infty} + ||f^0||_{\infty}).$$

Proof [Uniqueness] Let u_1 and u_2 be two solutions of equation (3.2.1). By using (3.1.5) for the difference $u_1 - u_2$ we get

$$\begin{split} \|u_{1,t} - u_{2,t}\|_{2}^{2} &+ 2\int_{t}^{T} \mathcal{E}^{A}(u_{1,s} - u_{2,s})ds \\ &\leq 2\int_{t}^{T} (f(s, \cdot, u_{1,s}, D_{A^{1/2}}u_{1,s}) - f(s, \cdot, u_{2,s}, D_{A^{1/2}}u_{2,s}), u_{1,s} - u_{2,s})ds + 2\alpha \int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds \\ &\leq 2\int_{t}^{T} C(|D_{A^{1/2}}u_{1,s} - D_{A^{1/2}}u_{2,s}|_{H}, |u_{1,s} - u_{2,s}|)ds + 2\alpha \int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds \\ &\leq (C^{2} + 2\alpha)\int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds + \int_{t}^{T} \mathcal{E}^{A}(u_{1,s} - u_{2,s})ds. \end{split}$$

By Gronwall's lemma it follows that

$$||u_{1,t} - u_{2,t}||_2^2 = 0,$$

hence $u_1 = u_2$.

[Existence] The existence will be proved in two steps.

Step 1. Suppose f is bounded. We define

$$M := \sup |f(t, x, y, z)|.$$

We need the following proposition.

Proposition 3.2.9 If f satisfies the condition in **Step 1**, then for $v \in \hat{F}^l$, there exists a unique generalized solution $u \in \hat{F}^l$ for the equation

$$(\partial_t + L)u + f(\cdot, \cdot, u, D_{A^{1/2}}v) = 0, \qquad u_T = \phi.$$

Following the same arguments as in Lemma 3.2.5, we assume that $2C^2 + 2\alpha + \mu_t \leq 0$.

For each $v \in \hat{F}^l$, we define Av = u where u is the unique generalized solution

obtained by Proposition 3.2.9. Let $v_1, v_2 \in \hat{F}^l$. By applying (3.1.5) to the difference $u_1 - u_2$ we obtain

$$\begin{split} \|u_{1,t} - u_{2,t}\|_{2}^{2} &+ 2\int_{t}^{T} \mathcal{E}^{A}(u_{1,s} - u_{2,s})ds \\ \leq & 2\int_{t}^{T} (f(s, \cdot, u_{1,s}, D_{A^{1/2}}v_{1,s}) - f(s, \cdot, u_{2,s}, D_{A^{1/2}}v_{2,s}), u_{1,s} - u_{2,s})ds + 2\alpha \int_{t}^{T} \|u_{1,s} - u_{2,s}\|_{2}^{2}ds \\ \leq & 2\int_{t}^{T} C(|D_{A^{1/2}}v_{1,s} - D_{A^{1/2}}v_{2,s}|_{H}, |u_{1,s} - u_{2,s}|)ds + \int_{t}^{T} (2\alpha + \mu_{s}) \|u_{1,s} - u_{2,s}\|_{2}^{2}ds \\ \leq & \int_{t}^{T} (2C^{2} + 2\alpha + \mu_{s}) \|u_{1,s} - u_{2,s}\|_{2}^{2}ds + \frac{1}{2}\int_{t}^{T} \mathcal{E}^{A}(v_{1,s} - v_{2,s})ds \\ \leq & \frac{1}{2}\int_{t}^{T} \mathcal{E}^{A}(v_{1,s} - v_{2,s})ds. \end{split}$$

Consequently we have $||Av_1 - Av_2||_T \leq \frac{1}{2} ||v_1 - v_2||_T$. Then the fixed point u of A is the solution for (3.2.1).

Proof of Proposition 3.2.9 We write $f(t, x, y) = f(t, x, y, D_{A^{1/2}}v)$ and uniqueness follows as above. Now we prove the existence of the solution.

We regularize f with respect to the variable y by convolution:

$$f_n(t, x, y, z) = n^l \int_{\mathbb{R}^l} f(t, x, y') \varphi(n(y - y')) dy'$$

where φ is a smooth nonnegative function with support contained in the ball $\{|y| \leq 1\}$ such that $\int \varphi = 1$. Then $f = \lim_{n \to \infty} f_n$ and for each n, $\partial_{y_i} f_n$ are uniformly bounded. Then each f_n satisfies a Lipschitz condition with respect to both y and z. Thus by Proposition 3.2.4 each f_n determines a solution $u_n \in \hat{F}^l$ of (3.2.1) with data (ϕ, f_n) . By the same arguments as in [S09, Theorem 4.19], we have that each f_n satisfies conditions (H1) and (H2') with C = 0 and $\mu = 0$. Since

$$|f_n(t, x, 0, 0)| \le n^l \int_{|y'| \le \frac{1}{n}} |f(t, x, y')| |\varphi(n(-y'))| dy'$$

$$\le M,$$

one deduces from Lemma 3.2.7 that $||u_n||_{\infty} \leq K$ and $||u_n||_T \leq K_T$.

Since the convolution operators approximate the identity uniformly on compact sets, we get for fixed t, x,

$$\lim_{n \to \infty} d'_{n,K}(t,x) := \sup_{|y| \le K} |f(t,x,y) - f_n(t,x,y)| = 0.$$

Next we will show that $(u_n)_{n \in \mathbb{N}}$ is a $\|\cdot\|_T$ -Cauchy sequence. By (3.1.5) for the difference $u_l - u_n$, we have

$$\begin{split} \|u_{l,t} - u_{n,t}\|_{2}^{2} &+ 2\int_{t}^{T} \mathcal{E}^{A}(u_{l,s} - u_{n,s})ds \\ \leq & 2\int_{t}^{T} (f_{l}(s, \cdot, u_{l,s}) - f_{n}(s, \cdot, u_{n,s}), u_{l,s} - u_{n,s})ds + 2\alpha \int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds \\ \leq & 2\int_{t}^{T} (|f_{l}(s, \cdot, u_{l,s}) - f(s, \cdot, u_{l,s})|, |u_{l,s} - u_{n,s}|)ds \\ & + 2\int_{t}^{T} (|f_{n}(s, \cdot, u_{n,s}) - f(s, \cdot, u_{n,s})|, |u_{l,s} - u_{n,s}|)ds \\ & + 2\int_{t}^{T} (f(s, \cdot, u_{l,s}) - f(s, \cdot, u_{n,s}), u_{l,s} - u_{n,s})ds + 2\alpha \int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds \\ \leq & 2\int_{t}^{T} (d'_{l,K}(s, \cdot) + d'_{n,K}(s, \cdot), |u_{l,s} - u_{n,s}|)ds + 2\alpha \int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds \\ \leq & \int_{t}^{T} \|d'_{l,K}(s, \cdot)\|_{2}^{2}ds + \int_{t}^{T} \|d'_{n,K}(s, \cdot)\|_{2}^{2}ds + (2 + 2\alpha) \int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds, \end{split}$$

and that $\lim_{n\to\infty} \int_t^T \|d'_{n,r}(s,\cdot)\|_2^2 ds = 0$. Thus, for l, n large enough, we get for an arbitrary $\varepsilon > 0$

$$\|u_{l,t} - u_{n,t}\|_{2}^{2} + \int_{t}^{T} \mathcal{E}^{A}(u_{l,s} - u_{n,s}) ds \leq \varepsilon + \tilde{K} \int_{t}^{T} \|u_{l,t} - u_{n,t}\|_{2}^{2} ds,$$

where \tilde{K} depends on C, M, μ, α . It is easy to see that Gronwall's lemma then implies that $(u_n)_{n \in \mathbb{N}}$ is a Cauchy-sequence in \hat{F} . Define $u := \lim_{n \to \infty} u_n$ and take a subsequence $(n_k)_{k \in \mathbb{N}}$ such that $u_{n_k} \to u$ a.e. We have

$$f(\cdot, \cdot, u_{n_k}) \to f(\cdot, \cdot, u)$$
 in $L^2(dt \times d\mu)$.

Since $||u_{n_k} - u||_T \to 0$, we obtain

$$|||D_{A^{1/2}}u - D_{A^{1/2}}u_{n_k}|_H||_{L^2(dt \times d\mu)} \to 0.$$

We conclude that

$$\begin{split} &\lim_{k \to \infty} \|f_{n_k}(u_{n_k}) - f(u)\|_{L^2(dt \times d\mu)} \\ &\leq \lim_{k \to \infty} \|f_{n_k}(u_{n_k}) - f(u_{n_k})\|_{L^2(dt \times d\mu)} + \lim_{k \to \infty} \|f(u_{n_k}) - f(u)\|_{L^2(dt \times d\mu)} \\ &\leq \lim_{k \to \infty} \|d'_{n_k,r}\|_{L^2(dt \times d\mu)} + \lim_{k \to \infty} \|f(u_{n_k}) - f(u)\|_{L^2(dt \times d\mu)} \\ &= 0. \end{split}$$

By passing to the limit in the mild equation associated to u_{n_k} with data (ϕ, f_{n_k}) , it follows that u is the solution associated to $(\phi, f(u, D_{A^{1/2}}v))$.

Step 2. Now we consider the general case. Let r be a positive real number such that

$$r \ge 1 + K(\|\phi\|_{\infty} + \|f^0\|_{\infty}),$$

where K is the constant appearing in Lemma 3.2.7 (3.2.4). Let θ_r be a smooth function such that $0 \leq \theta_r \leq 1, \theta_r(y) = 1$ for $|y| \leq r$ and $\theta_r(y) = 0$ if $|y| \geq r+1$. For each $n \in \mathbb{N}$, we set $q_n(z) := z \frac{n}{|z|_H \vee n}$ and

$$h_n(t, x, y, z) := \theta_r(y)(f(t, x, y, q_n(z)) - f_t^0) \frac{n}{f'^{,r+1} \vee n} + f_t^0.$$

We have

$$\begin{aligned} |h_n(t,x,y,z)| &\leq |f(t,x,y,q_n(z)) - f(t,x,y,0) + f(t,x,y,0) - f_t^0| \mathbf{1}_{\{|y| \leq r+1\}} \frac{n}{f',r+1} \vee n}{f',r+1} + f_t^0 \\ &\leq C|q_n(z)|_H + \frac{nf',r+1}{f',r+1} \vee n} + f_t^0 \\ &\leq (1+C)n + f_t^0. \end{aligned}$$

We easily show that h_n satisfies (H1) and (H3). So, we only need to prove (H2). For $y, y' \in \mathbb{R}^l$, if |y| > r+1, |y'| > r+1, the inequality is trivially satisfied and thus we concentrate on the case $|y'| \leq r+1$. We have

$$\langle y - y', h_n(t, x, y, z) - h_n(t, x, y', z) \rangle = \theta_r(y) \frac{n}{f'^{,r+1} \vee n} \langle y - y', f(t, x, y, q_n(z)) - f(t, x, y', q_n(z)) \rangle + \frac{n}{f'^{,r+1} \vee n} (\theta_r(y) - \theta_r(y')) \langle y - y', f(t, x, y', q_n(z)) - f_t^0 \rangle.$$

The first term of the right hand side of the previous equality is negative. For the second term, since θ_r is C(r)-Lipschitz, we obtain

$$\begin{aligned} (\theta_r(y) - \theta_r(y')) \langle y - y', f(t, x, y', q_n(z)) - f_t^0 \rangle &\leq C(r) |y - y'|^2 |f(t, x, y', q_n(z)) - f_t^0| \\ &\leq C(r) (Cn + f'_{r+1}(t)) |y - y'|^2, \end{aligned}$$

and thus

$$\frac{n}{f'^{,r+1} \vee n} (\theta_r(y) - \theta_r(y')) \langle y - y', f(t, x, y', q_n(z)) - f_t^0 \rangle \le C(r)(C+1)n|y - y'|^2.$$

Then each h_n satisfies the assumptions in **Step 1**, and denote u_n is the generalized

solution of (3.2.1) with data (h_n, ϕ) . We have

$$\langle y, h'_n(t, x, y) \rangle = \langle y, h_n(t, x, y, 0) - h_n(t, x, 0, 0) \rangle = \langle y, f(t, x, y, 0) - f_t^0 \rangle \frac{n\theta_r(y)}{f'^{r+1} \vee n} \le 0.$$

Hence, Lemma 3.2.7 implies that $||u_n||_{\infty} \leq r-1$, $||u_n||_T \leq K_T$. So, u_n is a solution with data (f_n, ϕ) , where $f_n(t, x, y, z) = (f(t, x, y, q_n(z)) - f_t^0) \frac{n}{f', r+1 \vee n} + f_t^0$. For this function (H2) is satisfied with $\mu_t = 0$. Conditions (H1) and (H2) yield

$$|(f_l(u_l, D_{A^{1/2}}u_l) - f_n(u_n, D_{A^{1/2}}u_n), u_l - u_n)| \le C(|D_{A^{1/2}}u_l - D_{A^{1/2}}u_n|_H, |u_l - u_n|) + |(f_l(u_n, D_{A^{1/2}}u_n) - f_n(u_n, D_{A^{1/2}}u_n), u_l - u_n)|.$$

For $n \leq l$, we have

$$|f_l(u_n, D_{A^{1/2}}u_n) - f_n(u_n, D_{A^{1/2}}u_n)| \le 2C|D_{A^{1/2}}u_n|_H \mathbb{1}_{\{|D_{A^{1/2}}u_n|_H \ge n\}} + 2C|D_{A^{1/2}}u_n|_H \mathbb{1}_{\{f', r+1 > n\}} + 2f'^{,r+1}\mathbb{1}_{\{f', r+1 > n\}}.$$

Then we have

$$\begin{split} \|u_{l,t} - u_{n,t}\|_{2}^{2} &+ 2\int_{t}^{T} \mathcal{E}^{A}(u_{l,s} - u_{n,s})ds \\ &\leq 2\int_{t}^{T}(f_{l}(u_{l,s}, D_{A^{1/2}}u_{l,s}) - f_{n}(u_{n,s}, D_{A^{1/2}}u_{n,s}), u_{l,s} - u_{n,s})ds + 2\alpha\int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds \\ &\leq 2\int_{t}^{T} C(|D_{A^{1/2}}u_{l} - D_{A^{1/2}}u_{n}|_{H}, |u_{l} - u_{n}|)ds + 2\int_{t}^{T} (2C|D_{A^{1/2}}u_{n}|_{H}1_{\{|D_{A^{1/2}}u_{n}|_{H} \ge n\}}, |u_{l} - u_{n}|)ds \\ &+ 2\int_{t}^{T} (2C|D_{A^{1/2}}u_{n}|_{H}1_{\{f', r+1 > n\}}, |u_{l} - u_{n}|)ds + 2\int_{t}^{T} (2f', r+1_{\{f', r+1 > n\}}, |u_{l} - u_{n}|)ds \\ &+ 2\alpha\int_{t}^{T} \|u_{l,s} - u_{n,s}\|_{2}^{2}ds \\ &\leq (C^{2} + 2\alpha)\int_{t}^{T} \|u_{l} - u_{n}\|_{2}^{2}ds + \int_{t}^{T} \mathcal{E}^{A}(u_{l} - u_{n})ds \\ &+ 8C(r-1)\int_{t}^{T} \int |D_{A^{1/2}}u_{n}|_{H}1_{\{|D_{A^{1/2}}u_{n}|_{H} \ge n\}}d\mu ds \\ &+ 8C(r-1)\int_{t}^{T} |D_{A^{1/2}}u_{n}|_{H}1_{\{f', r+1 > n\}}d\mu ds + 8C(r-1)\int_{t}^{T} \int f', r+1_{\{f', r+1 > n\}}d\mu ds. \end{split}$$

Since $||u_n||_T^2 \leq K_T$, we have $\int_0^T ||D_{A^{1/2}}u_n|_H||_2^2 ds \leq K_T$. Hence,

$$n^{2} \int_{t}^{T} \|1_{\{|D_{A^{1/2}}u_{n}|_{H} \ge n\}}\|_{2}^{2} ds \le \int_{t}^{T} \||D_{A^{1/2}}u_{n}1_{\{|D_{A^{1/2}}u_{n}|_{H} \ge n\}}|_{H}\|_{2}^{2} ds \le K_{T}.$$

 As

$$\lim_{n \to \infty} \int_t^T \int_{\{f', r > n\}} f', r d\mu ds = 0,$$

and

$$\int_{t}^{T} \int_{\{f',r>n\}} |D_{A^{1/2}}u_{n}|_{H} d\mu dt \leq ||1_{\{f',r>n\}}||_{L^{2}(dt\times d\mu)} ||D_{A^{1/2}}u_{n}|_{H}||_{L^{2}(dt\times d\mu)} \to 0,$$

for n big enough we obtain

$$\|u_{l,t} - u_{n,t}\|_{2}^{2} + \int_{t}^{T} \mathcal{E}^{A}(u_{l,s} - u_{n,s}) ds \le (C^{2} + 2\alpha) \int_{t}^{T} \|u_{l} - u_{n}\|_{2}^{2} ds + \varepsilon.$$

By Gronwall's lemma it is easy to see that $(u_n)_{n\in\mathbb{N}}$ is a Cauchy sequence in \hat{F}^l . Hence, $u := \lim_{n\to\infty} u_n$ is well defined. We find a subsequence such that $(u_{n_k}, D_{A^{1/2}}u_{n_k}) \to (u, D_{A^{1/2}}u)$ a.e.

$$f(u_{n_k}, D_{A^{1/2}}u) \to f(u, D_{A^{1/2}}u),$$

and conclude that

$$\begin{split} &|f_{n_k}(u_{n_k}, D_{A^{1/2}}u_{n_k}) - f(u, D_{A^{1/2}}u)| \\ \leq & 1_{\{f', r \leq n_k\}} |f(u, D_{A^{1/2}}u) - f(u_{n_k}, q_{n_k}(D_{A^{1/2}}u_{n_k}))| \\ &+ 1_{\{f', r > n_k\}} [|f(u, D_{A^{1/2}}u) - f^0| + |f(u, D_{A^{1/2}}u) - f(u_{n_k}, q_{n_k}(D_{A^{1/2}}u_{n_k}))|] \\ \leq & |f(u, D_{A^{1/2}}u) - f(u_{n_k}, q_{n_k}(D_{A^{1/2}}u_{n_k}))| + 1_{\{f', r > n_k\}} |f(u, D_{A^{1/2}}u) - f^0| \\ \leq & |f(u_{n_k}, D_{A^{1/2}}u) - f(u_{n_k}, q_{n_k}(D_{A^{1/2}}u_{n_k}))| + |f(u_{n_k}, D_{A^{1/2}}u) - f(u, D_{A^{1/2}}u)| \\ &+ 1_{\{f', r > n_k\}} |f(u, D_{A^{1/2}}u) - f^0| \\ \rightarrow 0 \text{ a.e..} \end{split}$$

Since

$$\begin{aligned} &|f_{n_k}(u_{n_k}, D_{A^{1/2}}u_{n_k}) - f(u, D_{A^{1/2}}u)| \\ \leq &|f(u, 0) - f(u, D_{A^{1/2}}u)| + |f_{n_k}(u_{n_k}, D_{A^{1/2}}u_{n_k}) - f_{n_k}(u_{n_k}, 0)| \\ &+ |f_{n_k}(u_{n_k}, 0) - f^0| + |f^0 - f(u, 0)| \\ \leq &C(|D_{A^{1/2}}u|_H + |D_{A^{1/2}}u_{n_k}|_H) + 2f^{',r}, \end{aligned}$$

we obtain

$$f_{n_k}(u_{n_k}, D_{A^{1/2}}u_{n_k}) \to f(u, D_{A^{1/2}}u)$$

in $L^1([0,T], L^2)$. We conclude that u is a generalized solution of (3.2.1) associated to the data (ϕ, f) .

3.3 Martingale representation for the processes

Some of the basic results on backward equations rely on the following well-known representation theorem (see, e.g., [FT02]). The Wiener process in a Hilbert space has the *martingale representation property*: any martingale with respect to the filtration generated by the Wiener process can be expressed as an Itô integral against the Wiener process. Now we extend this martingale representation theorem for the process associated with the operator L.

3.3.1 Representation under P^x

In order to obtain the results for the probabilistic part, we need that \mathcal{E} is a quasiregular generalized Dirichlet form (Definition 1.3) in the sense of Remark 3.1.1 (iii) with $\hat{c} \equiv 0$. There is a Markov process $X = (\Omega, \mathcal{F}_{\infty}, \mathcal{F}_t, X_t, P^x)$ which is properly associated in the resolvent sense with \mathcal{E} , i.e. $R_{\alpha}f := E^x \int_0^{\infty} e^{-\alpha t} f(X_t) dt$ is \mathcal{E} -quasicontinuous *m*-version of the resolvent G_{α} of \mathcal{E} for $\alpha > 0$ and $f \in \mathcal{B}_b(E) \cap L^2(E;\mu)$. The coform $\hat{\mathcal{E}}$ introduced in Section 1.1 is a generalized Dirichlet form with the associated resolvent $(\hat{G}_{\alpha})_{\alpha>0}$ and there exists an μ -tight special standard process properly associated in the resolvent sense with $\hat{\mathcal{E}}$. We always assume that $(\mathcal{F}_t)_{t\geq 0}$ is the (universally completed) natural filtration of X_t . From now on, we obtain all the results under the above assumptions.

As mentioned in Remark 3.1.1 (vii), such a process can be constructed by quasiregularity ([St2, IV. 1. Definition 1.7]) and a structural condition ([St2, IV. 2. D3] on the domain \mathcal{F} of the generalized Dirichlet form.

We use the spaces $\mathcal{M}, \dot{\mathcal{M}}$ which are introduced in Section 1.2. Define for $k \in E$,

$$\mathcal{E}_k(u,v) := \int \frac{\partial u}{\partial k} \frac{\partial v}{\partial k} d\mu, \qquad u,v \in \mathcal{F}C_b^{\infty}.$$

 $k \in E$ is called μ -admissible if $(\mathcal{E}_k, \mathcal{F}C_b^{\infty})$ is closable on $L^2(E; \mu)$. We consider the following conditions:

(A4) There exist constants $c, C_1 > 0$ such that $cId_H \leq A(z) \leq C_1Id_H$ for all $z \in E$. There exists a countable dense subset $\{e_k\}$ of E', which is an orthonormal basis of H, consisting of μ -admissible elements in E, and $u_k(\cdot) :=_{E'} \langle e_k, \cdot \rangle_E \in \mathcal{F}$.

(A4') There exists a countable dense subset $\{e_k\}$ of E', which is an orthonormal basis of H, consisting of μ -admissible elements in E, and $u_k(\cdot) =_{E'} \langle e_k, \cdot \rangle_E \in \mathcal{F}$. Furthermore, $A(z)e_k = \lambda_k(z)e_k$ for some non-negative Borel measurable functions λ_k .

Remark 3.3.1 Condition (A4) can be replaced by condition (A4') and all results below can be proved by the same arguments. For simplicity, we only give the proof under condition (A4).

By the existence of $\{e_k\}$, (A1) follows from [MR92, Proposition 3.8]. Set

$$\mathcal{F}C_b^{\infty}(\{e_k\}) := \{ f(_{E'}\langle e_1, \cdot \rangle_E, \dots, _{E'} \langle e_m, \cdot \rangle_E) | m \in \mathbb{N}, f \in C_b^{\infty}(\mathbb{R}^m) \}.$$

(A5) The process X associated with \mathcal{E} above is a continuous conservative Hunt process in the state space $E \cup \{\partial\}$, $\alpha \hat{G}_{\alpha}$ is sub-Markovian and strongly continuous on \mathcal{V} , and $\hat{\mathcal{E}}$ is quasi-regular. Furthermore, $\mathcal{F}C_b^{\infty}(\{e_k\}) \subset \mathcal{F}$ and for $u \in \mathcal{F}$, there exists a sequence $\{u_n\} \subset \mathcal{F}C_b^{\infty}(\{e_k\})$ such that $\mathcal{E}(u_n - u) \to 0, n \to \infty$.

If \mathcal{E} satisfies (A2) and (A4), the bilinear form can be written as

$$\mathcal{E}(u,v) := \int \langle A(z)\nabla u(z), \nabla v(z) \rangle_H d\mu(z) + \int \langle A(z)b(z), \nabla u(z) \rangle_H v(z)d\mu(z), u \in F, v \in bF.$$

Again we set $D_{A^{1/2}}u := A^{1/2} \nabla u$.

For an initial distribution $\mu \in \mathcal{P}(E)$ (where $\mathcal{P}(E)$ denotes all the probabilities on E,) we will prove that the *Fukushima reprensentation property* mentioned in [QY10] holds for X, i.e. there is an algebra $K(E) \subset \mathcal{B}_b(E)$ which generates the Borel σ -algebra $\mathcal{B}(E)$ and is invariant under R^{α} for $\alpha > 0$, and there are countable continuous martingales $M^i, i \in \mathbb{N}$, over $(\Omega, \mathcal{F}^{\mu}, \mathcal{F}^{\mu}_t, P^{\mu})$ such that for any potential $u = R^{\alpha} f$ where $\alpha > 0$ and $f \in K(E)$, the martingale part $M^{[u]}$ of the semimartingale $u(X_t) - u(X_0)$ has a martingale representation in terms of M^i , that is, there are predictable processes $F_i, i \in \mathbb{N}$ on $(\Omega, \mathcal{F}^{\mu}, \mathcal{F}^{\mu}_t)$ such that

$$M_t^{[u]} = \sum_{j=1}^{\infty} \int_0^t F_s^j dM_s^j \qquad P^{\mu} - a.e..$$

By Theorem 1.7, if \hat{G}_{α} is sub-Markovian and strongly continuous on \mathcal{V} , the Fukushima's decomposition holds for $u \in \mathcal{F}$. In this case we set $M^k := M^{[u_k]}$, with $u_k(\cdot) := \langle e_k, \cdot \rangle_H$. These martingales are called *coordinate martingales*.

Let us first calculate the energy measure related to $\langle M^{[u]} \rangle, u \in \mathcal{F}C_b^{\infty}$. By [Tr2,

formula (23)], for bounded $g \in L^2(E,\mu)$, we have

$$\begin{split} &\int \hat{G}_{\gamma}gd\mu_{\langle M^{[u]}\rangle} \\ &= \lim_{\alpha \to \infty} \alpha (U^{\alpha+\gamma}_{\langle M^{[u]}\rangle} 1, \hat{G}_{\gamma}g) \\ &= \lim_{\alpha \to \infty} \lim_{t \to \infty} E_{\hat{G}_{\gamma}g\cdot\mu} (\alpha e^{-(\gamma+\alpha)t} \langle M^{[u]}\rangle_t) + \lim_{\alpha \to \infty} E_{\hat{G}_{\gamma}g\cdot\mu} (\int_0^{\infty} \langle M^{[u]}\rangle_t \alpha(\gamma+\alpha) e^{-(\gamma+\alpha)t} dt) \\ &= \lim_{\alpha \to \infty} \lim_{t \to \infty} \alpha \langle \mu_{\langle M^{[u]}\rangle}, e^{-(\gamma+\alpha)t} \int_0^t \hat{P}_s \hat{G}_{\gamma}g ds \rangle \\ &+ \lim_{\alpha \to \infty} \alpha(\gamma+\alpha) (\int_0^{\infty} e^{-(\gamma+\alpha)t} E_{\hat{G}_{\gamma}g\cdot\mu} ((u(X_t)-u(X_0)-N_t^{[u]})^2) dt) \\ &= \lim_{\alpha \to \infty} \alpha(\gamma+\alpha) (\int_0^{\infty} e^{-(\gamma+\alpha)t} E_{\hat{G}_{\gamma}g\cdot\mu} ((u(X_t)-u(X_0))^2) dt) \\ &= \lim_{\alpha \to \infty} 2\alpha(u-\alpha G_{\alpha}u, u\hat{G}_{\gamma}g) - \alpha(u^2, \hat{G}_{\gamma}g - \alpha \hat{G}_{\alpha}\hat{G}_{\gamma}g) \\ &= 2(-Lu, u\hat{G}_{\gamma}g) - (-Lu^2, \hat{G}_{\gamma}g) \\ &= 2\mathcal{E}^A(u, u\hat{G}_{\gamma}g) - \mathcal{E}^A(u^2, \hat{G}_{\gamma}g) + 2\int \langle Ab, \nabla u \rangle_H u\hat{G}_{\gamma}g\mu(dx) \\ &- \int \langle Ab, \nabla(u^2) \rangle_H \hat{G}_{\gamma}g\mu(dx) \\ &= 2\mathcal{E}^A(u, u\hat{G}_{\gamma}g) - \mathcal{E}^A(u^2, \hat{G}_{\gamma}g) \\ &= 2\int \langle A \nabla u, \nabla (u\hat{G}_{\gamma}g) \rangle_H d\mu - \int \langle A \nabla (u^2), \nabla (\hat{G}_{\gamma}g) \rangle_H d\mu \\ &= 2\int \langle A \nabla u, \nabla u \rangle_H \hat{G}_{\gamma}gd\mu. \end{split}$$

Then by [Tr2, Theorem 2.5], we have

$$\mu_{\langle M^{[u]}\rangle} = 2\langle A\nabla u, \nabla u \rangle_H \cdot d\mu.$$

By [Tr2, Proposition 2.19], for $u \in \mathcal{F}C_b^{\infty}$ and $u = f(E' \langle e_1, \cdot \rangle_E, \dots, E' \langle e_m, \cdot \rangle_E)$, we have

$$M_t^{[u]} = \sum_{i=1}^n \int_0^t \langle \nabla u(X_s), e_i \rangle_H dM_s^i.$$

Then by the same arguments as in [F90, Theorem 3.1] (see also proof of Theorem 2.3.6 and Lemma 3.3.2), we have that under P^x for quasi every point x (where the

exceptional set depends on u, v, and every $u, v \in \mathcal{F}$,

(3.3.1)
$$M_t^{[u]} = \sum_{i=1}^{\infty} \int_0^t \langle \nabla u(X_s), e_i \rangle_H dM_s^i,$$

where $\sum_{i=1}^{\infty} \int_{0}^{t} \langle \nabla u(X_s), e_i \rangle_H dM_s^i = \lim_{n \to \infty} \sum_{i=1}^{n} \int_{0}^{t} \langle \nabla u(X_s), e_i \rangle_H dM_s^i$ in $(\dot{\mathcal{M}}, e)$ and we have

(3.3.2)
$$\langle M^{[u]}, M^{[v]} \rangle_t = 2 \int_0^t \langle A(X_s) \nabla u(X_s), \nabla v(X_s) \rangle_H ds.$$

In particular,

(3.3.3)
$$\langle M^i, M^j \rangle_t = 2 \int_0^t a_{ij}(X_s) ds,$$

where $a_{ij}(z) := \langle A(z)e_i, e_j \rangle_H$.

Lemma 3.3.2 Suppose (A4)(A5) hold. For $u \in \mathcal{F}$ and V_t is a continuous adapted process, with $|V_t| \leq M, \forall t, \omega$, we have for q.e. $x \in E$,

(3.3.4)
$$\int_0^t V_s dM_s^{[u]} = \sum_{i=1}^\infty \int_0^t V_s \langle \nabla u(X_s), e_i \rangle_H dM_s^i \qquad P^x - a.s..$$

If (A4'), (A5) hold, then for some $\psi \in \tilde{\nabla} u$, we have for q.e. $x \in E$,

$$\int_0^t V_s dM_s^{[u]} = \sum_{i=1}^\infty \int_0^t V_s \langle \psi(X_s), e_i \rangle_H dM_s^i \qquad P^x - a.s..$$

Proof By [Tr2, Remark 2.2], for $\nu \in \hat{S}_{00}$ and $B^n := \sum_{i=1}^n \int_0^t V_s \langle \nabla u(X_s), e_i \rangle_H dM_s^i$ we have

$$E^{\nu}(B^{n+m} - B^n)^2 = E^{\nu} \left(\sum_{i=n+1}^{n+m} \int_0^t V_s \langle \nabla u(X_s), e_i \rangle_H dM_s^i \right)^2$$
$$= E^{\nu} \left(\sum_{i,j=n+1}^{n+m} \int_0^t V_s^2 a_{ij}(X_s) \langle \nabla u(X_s), e_i \rangle_H \langle \nabla u(X_s), e_j \rangle_H ds \right)$$
$$\leq C_1 M^2 E^{\nu} \left(\sum_{i=n+1}^{n+m} \int_0^t \langle \nabla u(X_s), e_i \rangle_H^2 ds \right)$$
$$\leq C_1 M^2 e^t |\hat{U}_1 \nu|_{\infty} \sup_t \frac{1}{t} E^{\mu} \left(\sum_{i=n+1}^{n+m} \int_0^t \langle \nabla u(X_s), e_i \rangle_H^2 ds \right)$$

$$=C_1 M^2 e^t |\hat{U}_1 \nu|_{\infty} \sum_{i=n+1}^{n+m} \int \langle \nabla u(z), e_i \rangle_H^2 \mu(dz) \to 0, \text{ as } n, m \to \infty,$$

where $\hat{S}_{00}, \hat{U}_1 \nu$ are introduced in Section 1.2. Then we define $\sum_{i=1}^{\infty} \int_0^t V_s \langle \nabla u(X_s), e_i \rangle_H dM_s^i := \lim_{n \to \infty} B^n$ in $(\dot{\mathcal{M}}, e)$. Furthermore, we have

$$E^{\nu} \left(\int_{0}^{t} V_{s} dM_{s}^{[u]} - \sum_{i=1}^{n} \int_{0}^{t} V_{s} \langle \nabla u(X_{s}), e_{i} \rangle_{H} dM_{s}^{i} \right)^{2}$$

$$\leq E^{\nu} \int_{0}^{t} \sum_{i,j=n+1}^{\infty} V_{s}^{2} a_{ij}(X_{s}) \langle \nabla u(X_{s}), e_{i} \rangle_{H} \langle \nabla u(X_{s}), e_{j} \rangle_{H} ds$$

$$\leq M^{2} C_{1} e^{t} |\hat{U}_{1}\nu|_{\infty} \sup_{t} \frac{1}{t} E^{\mu} \int_{0}^{t} \sum_{i=n+1}^{\infty} \langle \nabla u(X_{s}), e_{i} \rangle_{H}^{2} ds$$

$$\leq M^{2} C_{1} e^{t} |\hat{U}_{1}\nu|_{\infty} \int_{0}^{t} \sum_{i=n+1}^{\infty} \langle \nabla u(z), e_{i} \rangle_{H}^{2} \mu(dz) \to 0, \text{ as } n \to \infty.$$

So, we have

$$\int_0^t V_s dM_s^{[u]} = \sum_{i=1}^\infty \int_0^t V_s \langle \nabla u(X_s), e_i \rangle_H dM_s^i \qquad P^\nu - a.s..$$

Then by [Tr2, Theorem 2.5] and Theorem 1.4, the assertions follow.

Moreover, by a modification of the proof of [QY10, Theorem 3.1], we have the martingale representation theorem for X which is similar to [BPS05].

Theorem 3.3.3 Suppose that (A4) or (A4') and (A5) hold. Then there exists some exceptional set \mathcal{N} such that the following representation result holds: For every bounded \mathcal{F}_{∞} -measurable random variable ξ , there exists predictable processes $\phi : [0, \infty) \times \Omega \to H$, such that for each probability measure ν , supported by $E \setminus \mathcal{N}$, one has

$$\xi = E^{\nu}(\xi|\mathcal{F}_0) + \sum_{i=0}^{\infty} \int_0^\infty \langle \phi_s, e_i \rangle_H dM_s^i \qquad P^{\nu} - a.e.,$$

where $M^i = M^{[u_i]}$ with $u_i :=_{E'} \langle e_i, \cdot \rangle_E, i \in \mathbb{N}$ are the coordinate martingales, and

$$E^{\nu} \int_0^{\infty} \langle A(X_s) \phi_s, \phi_s \rangle_H ds \leq \frac{1}{2} E^{\nu} \xi^2.$$

If another predictable process ϕ' satisfies the same relations under a certain measure P^{ν} , then one has $A^{1/2}(X_t)\phi'_t = A^{1/2}(X_t)\phi_t$, $dt \times dP^{\nu} - a.e.$

Proof Suppose that \mathcal{N} is some fixed exceptional set. By \mathcal{K} we denote the class of bounded random variables for which the statement holds outside this set. We claim

that if $(\xi_n) \subset \mathcal{K}$ is a uniformly bounded increasing sequence and $\xi = \lim_{n \to \infty} \xi_n$, then $\xi \in \mathcal{K}$. Indeed, we have $E^x |\xi_n - \xi|^2 \to 0$. Let ϕ^n denote the process which represents ξ_n . Then

$$E^{x} \int_{0}^{\infty} |\phi_{s}^{n} - \phi_{s}^{p}|_{H}^{2} ds \leq \frac{1}{c} E^{x} \int_{0}^{\infty} \langle A(X_{s})(\phi_{s}^{n} - \phi_{s}^{p}), \phi_{s}^{n} - \phi_{s}^{p} \rangle_{H} ds \leq \frac{1}{2c} E^{x} |\xi_{p} - \xi_{n}|^{2}.$$

Now we want to pass to the limit with ϕ^n pointwise, so that the limit become predictable. For each l = 0, 1, ... set

$$n_l(x) := \inf\{n | E^x (\xi - \xi_n)^2 < \frac{1}{2^l}\}$$
$$\bar{\xi}_l := \xi_{n_l(X_0)}.$$

Then one has $\bar{\xi}_l = \xi_{n_l(x)}$ on the set $\{X_0 = x\}$, and $E^x(\xi - \bar{\xi}_l)^2 < \frac{1}{2^l}$ for any $x \in \mathcal{N}^c$. The process which represents $\bar{\xi}_l$ is simply obtained by the formula $\bar{\phi}^l = \phi^{n_l(X_0)}$. Then define $\phi_s = \lim_{l \to \infty} \bar{\phi}_s^l$ in H. By the same arguments as the proof of Lemma 3.3.1, we obtain

$$E^{x} (\sum_{i=1}^{\infty} \int_{0}^{\infty} \langle \phi_{s} - \bar{\phi}_{s}^{l}, e_{i} \rangle_{H} dM_{s}^{i})^{2} = \lim_{k \to \infty} E^{x} (\sum_{i=1}^{k} \int_{0}^{\infty} \langle \phi_{s} - \bar{\phi}_{s}^{l}, e_{i} \rangle_{H} dM_{s}^{i})^{2}$$
$$= \lim_{k \to \infty} E^{x} (\sum_{i,j=1}^{k} \int_{0}^{\infty} a_{ij}(X_{s}) \langle \phi_{s} - \bar{\phi}_{s}^{l}, e_{i} \rangle_{H} \langle \phi_{s} - \bar{\phi}_{s}^{l}, e_{j} \rangle_{H} ds)$$
$$\leq C_{1} E^{x} \int_{0}^{\infty} |\phi_{s} - \bar{\phi}_{s}^{l}|_{H}^{2} ds \to 0, \text{ as } l \to \infty.$$

Therefore, we have $\xi \in \mathcal{K}$.

Let $K(E) \subset \mathcal{B}_b(E)$ be a countable set which is closed under multiplication, generates the Borel σ -algebra $\mathcal{B}(E)$ and $R^{\alpha}(K(E)) \subset K(E)$ for $\alpha \in \mathbb{Q}^+$. Such K(E) can be constructed as follows. We choose a countable set $N_0 \subset b\mathcal{B}(E)$ which generates the Borel σ -algebra $\mathcal{B}(E)$. Since E as a separable Banach space is strongly Lindelöf, such a set N_0 can easily be constructed (see [MR92, Section 3.3]). For $l \geq 1$ we define $N_{l+1} = \{g_1 \cdot \ldots \cdot g_k, U^{\alpha}g_1 \cdot g_2 \cdot \ldots \cdot g_k, g_i \in N_l, k \in \mathbb{N} \cup \{0\}, \alpha \in \mathbb{Q}^+\}$ and $K(E) := \bigcup_{l=0}^{\infty} N_l$ (c.f. [FOT94, Lemma 7.1.1]).

Let \mathcal{C}_0 be all $\xi = \xi_1 \cdots \xi_n$ for some $n \in \mathbb{N}$, $\xi_j = \int_0^\infty e^{-\alpha_j t} f_j(X_t) dt$, where $\alpha_j \in \mathbb{Q}^+$, $f_j \in K(E), j = 1, ..., n$. Since the results in Lemma 2.3.7 also hold in this case, we see that the universal completion of the σ -algebra generated by \mathcal{C}_0 is \mathcal{F}_∞ . By the first claim, a monotone class argument reduces the proof to the representation of a random variable in \mathcal{C}_0 .

Let $\xi \in \mathcal{C}_0$. By Markov property of the process (see e.g. [QY10, Theorem 3.1]),

we have

$$N_t = E^x(\xi|\mathcal{F}_t) = \sum_m Z_t^m$$

where the sum is finite, and for each $m, Z^m = Z_t$ has the following form

$$Z_t = V_t u(X_t)$$

(the superscript *m* will be dropped if no confusion may arise), where $V_t = \prod_{i=1}^{k'} \int_0^t e^{-\beta_i s} g_i(X_s) ds$ and $u(x) = U^{\beta_1 + \ldots + \beta_k} (h_1(U^{\beta_2 + \ldots + \beta_k} h_2 \ldots (U^{\beta_k} h_k) \ldots))$ for $\beta_i \in \mathbb{Q}^+, g_i, h_i \in K(E)$. Obviously, $u \in K(E)$. Hence, by Fukushima's decomposition and Fukushima's representation property we have

$$(3.3.5) \ u(X_t) - u(X_0) = M_t^{[u]} + A_t^{[u]} = \sum_{j=1}^{\infty} \int_0^t \langle \nabla u(X_s), e_j \rangle_H dM_s^j + A_t^{[u]} \qquad P^x - a.s.$$

Then by Itô's formula and Lemma 3.3.2, we have

$$Z_t = Z_0 + \int_0^t u(X_s) dV_s + \int_0^t V_s dA_s^{[u]} + \int_0^t V_s dM_s^{[u]}$$

= $Z_0 + \int_0^t u(X_s) dV_s + \int_0^t V_s dA_s^{[u]} + \sum_{i=1}^\infty \int_0^t V_s \langle \nabla u(X_s), e_i \rangle_H dM_s^i.$

Hence,

$$N_t = \sum_{i=1}^{\infty} \int_0^t V_s \langle \nabla u(X_s), e_i \rangle_H dM_s^i \qquad P^x - a.s..$$

Define $\phi_s = V_s \nabla u(X_s)$, then the representation holds for $\xi \in C_0$. As (3.3.4) and (3.3.5) hold for every x outside of an exceptional set of null capacity, the exceptional set \mathcal{N} in the statement will be the union of all these exceptional sets corresponding to $u \in K(E)$ and the exceptional sets related to V in Lemma 3.3.2.

If in the preceding theorem, ξ is nonnegative, we drop the boundedness assumption.

Corollary 3.3.4 Suppose that (A4) or (A4') and (A5) hold. Let \mathcal{N} be the set obtained in Theorem 3.3.3. Then for any \mathcal{F}_{∞} -measurable nonnegative random variable $\xi \geq 0$ there exists a predictable process $\phi : [0, \infty) \times \Omega \to H$ such that

$$\xi = E^x(\xi|\mathcal{F}_0) + \sum_{i=0}^{\infty} \int_0^\infty \langle \phi_s, e_i \rangle_H dM_s^i \qquad P^x - a.e.,$$

where $M^i, i \in \mathbb{N}$, are as in Theorem 3.3.3, and

$$E^x \int_0^\infty \langle A(X_s)\phi_s, \phi_s \rangle_H ds \le \frac{1}{2} E^x \xi^2,$$

for each point $x \in \mathcal{N}^c$ such that $E^x \xi < \infty$.

If another predictable process ϕ' satisfies the same relations under a certain measure P^x , then one has $A^{1/2}(X_t)\phi'_t = A^{1/2}(X_t)\phi_t, dt \times dP^x - a.e.$

3.3.2 Representation under P^{μ}

As usual we set $\int_0^t \psi_s dM_s = \sum_{i=0}^\infty \int_0^t \langle \psi_s, e_i \rangle_H dM_s^i$.

Lemma 3.3.5 Suppose that (A1)-(A3), (A5) and (A4) or (A4') hold. If $u \in \mathcal{D}(L)$, $\psi \in \tilde{\nabla}u$, then

$$u(X_t) - u(X_0) = \int_0^t \psi(X_s) dM_s + \int_0^t Lu(X_s) ds \qquad P^{\mu} - a.s..$$

Proof Corollary 3.3.4 and (3.3.1) imply the assertion.

Then by the same arguments as the proof of Lemma 2.3.6 we extend this representation to time dependent functions u(t, x).

Lemma 3.3.6 Suppose that (A1)-(A3), (A5) and (A4) or (A4') hold. Let $u : [0,T] \times E \to \mathbb{R}$ be such that

- (i) $\forall s, u_s \in \mathcal{D}(L)$ and $s \to Lu_s$ is continuous in L^2 .
- (ii) $u \in C^1([0,T]; L^2)$.

Then clearly $u \in \mathcal{C}_T$, and, moreover, for any $\psi \in \tilde{\nabla} u$ and any s, t > 0 such that s + t < T, the following relation holds P^{μ} -a.s.

$$u(s+t, X_t) - u(s, X_0) = \int_0^t \psi_{s+r}(X_r) dM_r + \int_0^t (\partial_s + L) u_{s+r}(X_r) dr.$$

Theorem 3.3.7 Suppose that (A1)-(A3), (A5) and (A4) or (A4') hold. Let $f \in L^1([0,T]; L^2)$ and $\phi \in L^2(E)$ and define

$$u_t := P_{T-t}\phi + \int_t^T P_{s-t}f_s ds$$

Then for each $\psi \in \tilde{\nabla} u$ and for each $s \in [0, T]$, the following relation holds P^{μ} -a.s.

$$u(s+t, X_t) - u(s, X_0) = \int_0^t \psi(s+r, X_r) dM_r - \int_0^t f(s+r, X_r) dr.$$

Furthermore, if u is a generalized solution of PDE (3.2.1), for each $t \in [s, T]$ the following BSDE holds P^{μ} -a.s.

$$u(t, X_{t-s}) = \phi(X_{T-s}) + \int_{t}^{T} f(r, X_{r-s}, u(r, X_{r-s}), A^{1/2} \nabla u(r, X_{r-s})) dr$$
$$- \int_{t-s}^{T-s} \psi(s+r, X_r) dM_r.$$

Proof First assume that ϕ and f satisfy the conditions in Proposition 3.1.5 (ii). Then we have that u satisfies the conditions in Lemma 3.3.6 and by Lemma 3.3.6, the assertion follows. For the general case we choose u^n associated with (f^n, ϕ^n) as in Proposition 3.1.7. Then $||u^n - u||_T \to 0$ as $n \to \infty$. For u^n we have

(3.3.6)
$$u^{n}(s+t,X_{t}) - u^{n}(s,X_{0}) = \int_{0}^{t} \nabla u^{n}_{s+r}(X_{r}) dM_{r} - \int_{0}^{t} f^{n}(s+r,X_{r}) dr.$$

As

$$E^{\mu} | \int_{0}^{t} (\nabla u_{s+r}^{n}(X_{r}) - \nabla u_{s+r}^{p}(X_{r})) dM_{r}|^{2} \\ \leq E^{\mu} \int_{0}^{t} \langle A(X_{r})(\nabla u_{s+r}^{n}(X_{r}) - \nabla u_{s+r}^{p}(X_{r})), \nabla u_{s+r}^{n}(X_{r}) - \nabla u_{s+r}^{p}(X_{r}) \rangle_{H} dr \\ \leq \int_{0}^{t} \mathcal{E}^{A}(u_{s+r}^{n} - u_{s+r}^{p}) dr,$$

letting $n \to \infty$ in (3.3.6), we obtain the assertions.

3.4 BSDE's and Weak Solutions

The set \mathcal{N} obtained in Theorem 3.3.3 will be fixed throughout this section. By Theorem 3.3.3, we solve the BSDE under all measures P^x , $x \in \mathcal{N}^c$, at the same time. We will treat systems of l equations, $l \in \mathbb{N}$, associated to \mathbb{R}^l -valued functions $f : [0,T] \times \Omega \times \mathbb{R}^l \times H^l \mapsto \mathbb{R}^l$, assumed to be predictable. This means that we consider the map $(s, \omega) \mapsto f(s, \omega, \cdot, \cdot)$ as a process which is predictable with respect to the canonical filtration of our process (\mathcal{F}_t) .

Lemma 3.4.1 Suppose that (A4) or (A4') and (A5) hold. Let ξ be an \mathcal{F}_{T} -

measurable random variable and $f : [0, T] \times \Omega \mapsto \mathbb{R}$ an $(\mathcal{F}_t)_{t \geq 0}$ -predictable process. Let D be the set of all points $x \in \mathcal{N}^c$ for which the following integrability condition holds

$$E^{x}(|\xi| + \int_{0}^{T} |f(s,\omega)|ds)^{2} < \infty.$$

Then there exists a pair $(Y_t, Z_t)_{0 \le t \le T}$ of predictable processes $Y : [0, T) \times \Omega \mapsto \mathbb{R}, Z : [0, T) \times \Omega \mapsto H$, such that under all measures $P^x, x \in D$, they have the following properties:

(i) Y is continuous;

(ii) Z satisfies the integrability condition

$$\int_0^T |A^{1/2}(X_t)Z_t|_H^2 dt < \infty, \qquad P^x - a.s.;$$

(iii) the local martingale obtained integrating Z against the coordinate martingales, i.e. $\int_0^t Z_s dM_s$, is a uniformly integrable martingale; (iv)

$$Y_t = \xi + \int_t^T f(s,\omega)ds - \int_t^T Z_s dM_s, \qquad P^x - a.s., 0 \le t \le T$$

If another pair (Y'_t, Z'_t) of predictable processes satisfies the above conditions (i),(ii),(iii),(iv), under a certain measure P^{ν} with the initial distribution ν supported by D, then one has $Y_{\cdot} = Y'_{\cdot}, P^{\nu} - a.s.$ and $A^{1/2}(X_t)Z_t = A^{1/2}(X_t)Z'_t, dt \times P^{\nu} - a.s.$

Proof The representations of the positive and negative parts of the random variable $\xi + \int_0^T f_s ds$ give us a predictable process Z such that

$$\xi + \int_0^T f_s ds = E^{X_0}(\xi + \int_0^T f_s ds) + \int_0^T Z_s dM_s.$$

Then we obtain the desired process Y by the formula

$$Y_t = E^{X_0}(\xi + \int_0^T f_s ds) + \int_0^T Z_s dM_s - \int_0^t f_s ds.$$

Definition 3.4.2 Let ξ be an \mathbb{R}^l -valued, \mathcal{F}_T -measurable, random variable and $f: [0,T] \times \Omega \times \mathbb{R}^l \times H^l \mapsto \mathbb{R}^l$ a measurable \mathbb{R}^l -valued function such that $(s,\omega) \mapsto f(s,\omega,\cdot,\cdot)$ is a predictable process. Let p > 1 and ν be a probability measure supported by \mathcal{N}^c such that $E^{\nu}|\xi|^p < \infty$. We say that a pair $(Y_t, Z_t)_{0 \leq t \leq T}$ of predictable processes $Y: [0,T) \times \Omega \mapsto \mathbb{R}^l$, $Z: [0,T) \times \Omega \mapsto H^l$ is a solution of the BSDE in $L^p(P^{\nu})$ with data (ξ, f) provided Y is continuous under P^{ν} and it satisfies both the

integrability conditions

$$\int_0^T |f(t, \cdot, Y_t, A^{1/2}(X_t)Z_t)| dt < \infty, \qquad P^{\nu} - a.s.,$$

and

$$E^{\nu} \left(\int_0^T |A^{1/2}(X_t)Z_t|_H^2 dt\right)^{p/2} < \infty,$$

and the following equation holds

(3.4.1)

$$Y_t = \xi + \int_t^T f(s, \omega, Y_s, A^{1/2}(X_s)Z_s)ds - \int_t^T Z_s dM_s, \qquad P^{\nu} - a.s., 0 \le t \le T.$$

Let $f : [0,T] \times \Omega \times \mathbb{R}^l \times H^l \mapsto \mathbb{R}^l$ be a measurable \mathbb{R}^l -valued function such that $(s,\omega) \mapsto f(s,\omega,\cdot,\cdot)$ is predictable and it satisfies the following conditions:

(\Omega1) [Lipschitz condition in z] There exists a constant C>0 such that for all t,ω,y,z,z'

$$|f(t,\omega,y,z) - f(t,\omega,y,z')| \le C|z - z'|_{H^{1}}$$

($\Omega 2$) [Monotonicity condition in y] For ω, y, y', z arbitrary, there exists a function $\mu_t \in L^1([0,T],\mathbb{R})$ such that

$$\langle y-y', f(t,\omega,y,z)-f(t,\omega,y',z)\rangle \leq \mu_t |y-y'|^2,$$

and $\alpha_t := \int_0^t \mu_s ds$.

(Ω 3) [Continuity condition in y] For t, ω and z fixed, the map

$$y \mapsto f(t, \omega, y, z)$$

is continuous.

We use the following notations:

$$f^{0}(t,\omega) := f(t,\omega,0,0), \qquad f'(t,\omega,y) := f(t,\omega,y,0) - f(t,\omega,0,0),$$
$$f'^{,r}(t,\omega) := \sup_{|y| \le r} |f'(t,\omega,y)|.$$

Let ξ be an \mathbb{R}^l -valued, \mathcal{F}_T -measurable, random variable and, for each p > 0 let A_p denote the set of all points $x \in \mathcal{N}^c$ such that

(3.4.2)
$$E^x \int_0^T f_t'^{,r} dt < \infty, \qquad \forall r \ge 0,$$

and

$$E^{x}(|\xi|^{p} + (\int_{0}^{T} |f^{0}(s,\omega)|ds)^{p}) < \infty.$$

Let A_{∞} denote the set of points $x \in \mathcal{N}^c$ for which (3.4.2) holds and with the property that $|\xi|, |f^0| \in L^{\infty}(P^x)$.

The method to prove the following proposition is standard, and it is included in the Appendix.

Proposition 3.4.3 Assume that (A4) or (A4') and (A5) holds. Under conditions $(\Omega 1), (\Omega 2), (\Omega 3)$ there exists a pair $(Y_t, Z_t)_{0 \le t \le T}$ of predictable processes $Y : [0, T) \times \Omega \mapsto \mathbb{R}^l, Z : [0, T) \times \Omega \mapsto H^l$ that forms a solution of the BSDE (3.4.1) in $L^p(P^x)$ with data (ξ, f) for each point $x \in A_p$. Moreover, the following estimate holds with some constant K that depends only on c, C, μ and T,

$$E^{x}(\sup_{t\in[0,T]}|Y_{t}|^{p} + (\int_{0}^{T}|A^{1/2}(X_{t})Z_{t}|^{2}_{H}dt)^{p/2}) \leq KE^{x}(|\xi|^{p} + (\int_{0}^{T}|f^{0}(s,\omega)|ds)^{p}), \qquad x \in A_{p}.$$

If $x \in A_{\infty}$, then $\sup_{t \in [0,T]} |Y_t| \in L^{\infty}(P^x)$.

If (Y'_t, Z'_t) is another solution in $L^p(P^x)$ for some point $x \in A_p$, then one has $Y_t = Y'_t$ and $A^{1/2}(X_t)Z_t = A^{1/2}(X_t)Z'_t$, $dt \times P^x - a.s.$.

We shall now look at the connection between the solutions of BSDE's introduced in this section and the PDE's studied in Section 3.2. In order to do this we have to consider BSDE's over time intervals [s, T], with $0 \le s \le T$ as done in Section 2.4.

Let us give a formal definition for the natural notion of solution over a time interval [s, T]. Let ξ be an \mathcal{F}_{T-s} -measurable, \mathbb{R}^l -valued, random variable and $f : [s, T] \times \Omega \times \mathbb{R}^l \times H^l \to \mathbb{R}^l$ an \mathbb{R}^l -valued, measurable map such that $(f(s + l, \omega, \cdot, \cdot))_{l \in [0, T-s]}$ is predictable with respect to $(\mathcal{F}_l)_{l \in [0, T-s]}$. Let ν be a probability measure supported by \mathcal{N}^c such that $E^{\nu} |\xi|^p < \infty$. We say a pair $(Y_t, Z_t)_{s \leq t \leq T}$ of processes $Y : [s, T] \times \Omega \to \mathbb{R}^l, Z : [s, T] \times \Omega \to H^l$ is a solution in $L^p(P^{\nu})$ of the BSDE over the interval [s, T] with data (ξ, f) , provided they have the property that reindexed as $(Y_{s+l}, Z_{s+l})_{l \in [0, T-s]}$ these processes are $(\mathcal{F}_l)_{l \in [0, T-s]}$ -predictable, Y is continuous and together they satisfy the integrability conditions

$$\int_{s}^{T} |f(t, \cdot, Y_{t}, A^{1/2}(X_{t-s})Z_{t})| dt < \infty, \qquad P^{\nu} - a.s.,$$

and

$$E^{\nu} \left(\int_{s}^{T} |A^{1/2}(X_{t-s})Z_{t}|_{H}^{2} dt \right)^{p/2} < \infty,$$

and the following equation under P^{ν} holds

(3.4.3)
$$Y_t = \xi + \int_t^T f(r, Y_r, A^{1/2}(X_{r-s})Z_r) dr - \int_{t-s}^{T-s} Z_{s+l} dM_l, \quad s \le t \le T.$$

The next result gives the probabilistic interpretation of Theorem 3.2.8. Let us assume that $f: [0,T] \times E \times \mathbb{R}^l \times H^l \to \mathbb{R}^l$ is the measurable function appearing in the basic equation (3.2.1).

Let $\phi: E \to \mathbb{R}^l$ be measurable and for each p > 1, let A_p denote the set of all points $(s, x) \in [0, T) \times \mathcal{N}^c$ such that

$$E^x \int_s^T f'^{,r}(t, X_{t-s}) dt < \infty, \qquad \forall r \ge 0,$$

and

$$E^{x}(|\phi|^{p}(X_{T-s}) + (\int_{s}^{T} |f^{0}(t, X_{t-s})|ds)^{p}) < \infty.$$

Set $D := \bigcup_{p>1} A_p, A_{p,s} := \{x \in \mathcal{N}^c, (s, x) \in A_p\}$, and $A_s := \bigcup_{p>1} A_{p,s}, s \in [0, T)$.

By the same arguments as Theorem 2.4.4, we obtain the following results.

Theorem 3.4.4 Assume that (A4) or (A4') and (A5) holds and that the function f satisfies conditions (H1),(H2),(H3). Then there exist nearly Borel measurable functions $(u, \psi), u : D \to \mathbb{R}^l, \psi : D \to H^l$, such that, for each $s \in [0, T)$ and each $x \in A_{p,s}$, the pair $(u(t, X_{t-s}), \psi(t, X_{t-s}))_{s \le t \le T}$ solves BSDE (3.4.3) in $L^p(P^x)$ with data $(\phi(X_{T-s}), f(t, X_{t-s}, y, z))$ over the interval [s, T].

In particular, the functions u, ψ satisfy the following estimate, for $(s, x) \in A_p$,

$$E^{x}(\sup_{t\in[s,T]}|u(t,X_{t-s})|^{p} + (\int_{s}^{T}|A^{1/2}\psi(t,X_{t-s})|^{2}dt)^{p/2})$$

$$\leq KE^{x}(|\phi(X_{T-s})|^{p} + (\int_{s}^{T}|f^{0}(t,X_{t-s})|dt)^{p}).$$

Moreover, if (A1)-(A3) hold and f, ϕ satisfy conditions (H4) and (H5), then the complement of $A_{2,s}$ is μ -negligible (i.e. $\mu(A_{2,s}^c) = 0$) for each $s \in [0, T)$, the class of $u1_{A_2}$ is an element of \hat{F}^l which is a generalized solution of (3.2.1), ψ represents a version of ∇u and the following relation holds for each $(s, x) \in D$ and $1 \leq i \leq l$, (3.4.4)

$$u^{i}(s,x) = E^{x}(\phi^{i}(X_{T-s})) + \int_{s}^{T} E^{x} f^{i}(t, X_{t-s}, u(t, X_{t-s}), A^{1/2}(X_{t-s})\psi(t, X_{t-s})) dt.$$

3.5 Examples

In this section, we give some examples satisfying our assumptions (A1)-(A5).

Example 3.5.1 (Ornstein-Uhlenbeck semigroup) Given two separable Hilbert spaces H and U, consider the stochastic differential equation

(3.5.1)
$$dX(t) = A_1 X(t) dt + B dW(t), \qquad X(0) = x \in H,$$

where $A_1: D(A_1) \subset H \to H$ is the infinitesimal generator of a strongly continuous semigroup e^{tA_1} , $B: U \to H$ is a bounded linear operator, and W is a cylindrical Wiener process in U. Assume

- (i) $||e^{tA_1}|| \leq M e^{\omega t}$ for $\omega < 0$, $M \geq 0$, and all $t \geq 0$.
- (ii) For any t > 0 the linear operator Q_t , defined as

$$Q_t x = \int_0^t e^{sA_1} C e^{sA_1^*} x ds, \ x \in H, t \ge 0,$$

where $C = BB^*$, is of trace class.

(iii) $Ce^{tA_1^*} = e^{tA_1}C.$

 μ denotes the Gaussian measure in H with mean 0 and covariance operator $Q_\infty.$ Then the bilinear form

$$\mathcal{E}(u,u) := \frac{1}{2} \int_{H} |C^{1/2} \nabla u|^2 d\mu, \ u \in \mathcal{F}C_b^{\infty},$$

is closable. The closure of $\mathcal{F}C_b^{\infty}$ with respect to \mathcal{E}_1 is denoted by F. (\mathcal{E}, F) is a generalized Dirichlet form in the sense of Remark 3.1.1 (iii) with $(E_1, \mathcal{B}(E_1), m) = (H, \mathcal{B}(H), \mu)$, $(\mathcal{A}, \mathcal{V}) = (\mathcal{E}, F)$ and $\Lambda = 0$. In particular, it is a symmetric Dirichlet form associated with the O-U process given by (3.5.1) and satisfies conditions (A1)-(A5) (see [D04, ChapterII]).

Example 3.5.2 Let H be a real separable Hilbert space (with scalar product $\langle \cdot, \cdot \rangle$ and norm denoted by $|\cdot|$) and μ a finite positive measure on H. We denote its Borel σ -algebra by $\mathcal{B}(H)$. For $\rho \in L^1_+(H,\mu)$ we consider the following bilinear form

$$\mathcal{E}^{\rho}(u,v) = \frac{1}{2} \int_{H} \langle \nabla u, \nabla v \rangle \rho(z) \mu(dz), u, v \in \mathcal{F}C_{b}^{\infty},$$

where $L^1_+(H,\mu)$ denotes the set of all non-negative elements in $L^1(H,\mu)$. There are many examples for ρ such that \mathcal{E}^{ρ} is closable. For example if $\rho d\mu$ is a "Log-Concave" measure in the sense of [ASZ09], and more examples can be found in [MR92]. The closure of $\mathcal{F}C_b^{\infty}$ with respect to \mathcal{E}_1 is denoted by F. (\mathcal{E}, F) is a generalized Dirichlet form in the sense of Remark 3.1.1 (iii) with $(E_1, \mathcal{B}(E_1), m) = (H, \mathcal{B}(H), \mu), (\mathcal{A}, \mathcal{V}) =$ (\mathcal{E}, F) and $\Lambda = 0$. In particular, it is a symmetric Dirichlet form and satisfies (A1)-(A5). Assume that: $A_1 : D(A_1) \subset H \to H$ is a linear self-adjoint operator on H such that $\langle A_1 x, x \rangle \geq \delta |x|^2 \ \forall x \in D(A_1)$ for some $\delta > 0$ and A_1^{-1} is of trace class. μ will denote the Gaussian measure in H with mean 0 and covariance operator

$$Q := \frac{1}{2} A_1^{-1}$$

We are concerned with the following two cases.

1. Choose $\rho = \frac{e^{-2U(x)}}{\int_H e^{-2U(y)} dy}$ for a Borel map $U : H \to (-\infty, +\infty]$ with $\int_H e^{-2U(y)} dy \in (0,\infty)$. Under some regular condition for U, the process associated with \mathcal{E}^{ρ} is the solution of the following SPDE

$$dX(t) = (A_1X(t) + \nabla U(X(t))dt + dW(t), \qquad X(0) = x \in H.$$

2. $\rho = 1_{\{|x|_H \leq 1\}}$. This case has been studied in [ASZ09], [RZZ] and it is associated with a reflected O-U process ([RZZ]). The Kolomogorov equation associated with \mathcal{E} has been studied in [BDT09] and the solution corresponds to the Kolomogorov equation with Neumann boundary condition.

Example 3.5.3 Consider the same situation in Example 3.5.1 and assume that, in addition we are given a nonlinear function $F: H \to H$ such that there exists K > 0, $|F(x) - F(y)|_H \leq K|x - y|, x, y \in H$ and $\langle F(x) - F(y), x - y \rangle \leq 0, x, y \in H$. A_1 is an operator which satisfies the condition in Example 3.5.2 and $A_1^{-1+\delta}$ is trace-class for some $\delta \in (0, \frac{1}{2})$. We are concerned with the stochastic differential equation

(3.5.2)
$$dX(t) = (A_1X(t) + F(X(t))dt + BdW(t), \qquad X(0) = x \in H.$$

The Kolomogrov operator associated with (3.5.2) is given by

$$K_0\varphi = \frac{1}{2}Tr[CD^2\varphi] + \langle x, A_1^*D\varphi \rangle + \langle F(x), D\varphi \rangle, \varphi \in \mathcal{E}_{A_1}(H),$$

where $\mathcal{E}_{A_1}(H) :=$ linear span $\{\varphi_h(x) = e^{i\langle h, x \rangle} : h \in D(A_1^*)\}$. Assume the semigroup e^{tA_1} is analytic. Then by [DZ02, Theorem 11.2.21] there exists a unique invariant measure ν for K_0 i.e.

$$\int K_0 \varphi d\nu = 0, \text{ for all } \varphi \in \mathcal{E}_{A_1}(H),$$

and ν is absolutely continuous with respect to μ from Example 3.5.1 and for $\rho = \frac{d\nu}{d\mu}$

we have that $\rho \in W^{1,2}(H,\mu)$ and $D \log \rho \in W^{1,2}(H,\nu;H)$.

Then by [Tr2, Section 4.2], we know that the bilinear form on $L^2(H;\nu)$ associated with K_0 is a generalized Dirichlet form in the sense of Remark 3.1.1 (iii) with $(E_1, \mathcal{B}(E_1), m) = (H, \mathcal{B}(H), \nu), (\mathcal{A}, \mathcal{V}) = (0, L^2(H, \nu))$ and $\Lambda = K_0$. It satisfies conditions (A1)-(A5). There are even more general conditions on F and A_1 which can be found in [DRW09, Theorem 5.2] such that conditions (A1)-(A5) hold.

The following example is given in [Tr2, Section 4.2].

Example 3.5.4 Assume that E is a separable real Hilbert space with inner product $\|\cdot\|_{E}^{1/2}$ and $H \subset E$ densely by a Hilbert-Schmidt map. Let $B : E \to E$ be a Borel measurable vector field satisfying the following conditions:

(B.1) $\lim_{\|z\|_E \to \infty} \langle B(z), z \rangle = -\infty.$

(B.2) $_{E'}\langle l, B \rangle_E : E \to \mathbb{R}$ is weakly continuous for all $l \in E'$.

(B.3) There exist $C_1, C_2, d \in (0, \infty)$, such that $||B(z)||_E \le C_1 + C_2 ||z||_E^d$.

Then by [BRS00, Theorem 5.2] there exists a probability measure μ on $(E, \mathcal{B}(E))$ such that $_{E'}\langle l, B \rangle_E \in L^2(E; \mu)$ for all $l \in E'$ and such that

$$\int \frac{1}{2} \Delta_H u + \frac{1}{2} {}_{E'} \langle \nabla u, B \rangle_E d\mu = 0 \text{ for all } u \in \mathcal{F}C_b^{\infty},$$

where Δ_H is the Gross-Laplacian, i.e., $\Delta_H u = \sum_{i,j=1}^m \frac{\partial f}{\partial x_i \partial x_j} (l_1(z), ..., l_m(z)) \langle l_i, l_j \rangle_H$ for $u = f(l_1, ..., l_m) \in \mathcal{F}C_b^{\infty}$. Assume $B(z) = -z + v(z), v : E \to H$. For the bilinear form associated with $Lu = \frac{1}{2}\Delta_H u + \frac{1}{2}_{E'} \langle \nabla u, B \rangle_E, u \in \mathcal{F}C_b^{\infty}$ on $L^2(E, \mu)$ is a generalized Dirichlet form in the sense of Remark 3.1.1 (iii) with $(E_1, \mathcal{B}(E_1), m) =$ $(H, \mathcal{B}(H), \nu), (\mathcal{A}, \mathcal{V}) = (0, L^2(H, \nu))$ and $\Lambda = L$. It satisfies conditions (A1)-(A5).

3.6 A control problem

In this section, we assume conditions (A1)-(A5) and $||A^{1/2}(\cdot)||_{L^{\infty}(H)}$, $||A^{-1/2}(\cdot)||_{L^{\infty}(H)} \in L^{\infty}(E,\mu)$ and consider the case l = 1.

Proposition 3.6.1 Let (β, γ) be an (\mathbb{R}, H) -valued predictable process, $\varphi \in \bigcap_{x \in \mathcal{N}^c} M_x^2(\mathbb{R})$ and $\xi \in \bigcap_{x \in \mathcal{N}^c} L^2(P^x)$. β is assumed to be bounded from above, $|\gamma|_H$ is bounded and $\xi \in \mathcal{F}_T$. Then the linear BSDE

$$-dY_t = [\varphi_t + Y_t\beta_t + \langle A^{1/2}(X_t)Z_t, \gamma_t \rangle_H]dt - Z_t dM_t, \qquad Y_T = \xi,$$

has a solution (Y, Z) in $M^2_x(\mathbb{R}) \times M^2_x(H), \forall x \in \mathcal{N}^c$, where $M^2_x(\mathbb{R})$ and $M^2_x(H)$ are

defined in the appendix. Moreover, Y_t is given by the closed formula

$$Y_t = E^x [\xi \Gamma_T^t + \int_t^T \Gamma_s^t \varphi_s ds | \mathcal{F}_t] P^x - a.s., \forall x \in \mathcal{N}^c,$$

where Γ_s^t is the adjoint process defined for $s \ge t$ by the forward linear SDE

$$d\Gamma_s^t = \Gamma_s^t [\beta_s ds + A^{-1/2}(X_s)\gamma_s . dM_s], \qquad \Gamma_t^t = 1.$$

In particular, if ξ and φ are nonnegative, the process Y is nonnegative. If, in addition, $Y_0 = 0$, then, for any t, $Y_t = 0$ a.s., $\xi = 0$ a.s., and $\varphi_t = 0 dP^x \otimes dt$ a.s..

Proof By the same arguments as in [BPS02, Lemma 7.1] the assertion follows. \Box

Theorem 3.6.2 (Comparison Theorem). Let $x \in A_2$. Let (f^1, ξ^1) and (f^2, ξ^2) be two standard parameters of BSDE (3.4.1), where f^1, f^2 satisfy conditions $(\Omega 1) - (\Omega 3)$ and

$$f^{i}(s,\omega,y,z) := h_{i}(s,\omega) + \langle A^{1/2}(X_{s})c_{i}(s,\omega), z \rangle_{H}, \text{ for } i = 1, 2,$$

where $(h_i, c_i), i = 1, 2$ are bounded (\mathbb{R}, H) -valued predictable processes. Let (Y^1, Z^1) and (Y^2, Z^2) be the associated square-integrable solutions. We suppose that

(a) $\xi^1 \ge \xi^2 P^x$ -a.s. (b) $\delta_2 f_t = f^1(t, Y_t^2, A^{1/2}(X_t)Z_t^2) - f^2(t, Y_t^2, A^{1/2}(X_t)Z_t^2) \ge 0, dP^x \otimes dt$ -a.s.

Then we have that P^x -almost surely: $Y_t^1 \ge Y_t^2$, for all $t \ge 0$.

Proof The pair $(\delta Y, \delta Z)$ is the solution of the following linear BSDE:

$$-d\delta Y_t = [\langle \Delta_z f(t), A^{1/2}(X_t) \delta Z_t \rangle_H + \delta_2 f_t] dt - \delta Z_t dM_t, \qquad \delta Y_T = \xi^1 - \xi^2,$$
$$\langle \Delta_z f(t), e_i \rangle_H := \langle A^{1/2}(X_s) c_1, e_i \rangle_H,$$

if $\langle A^{1/2}(X_t)(Z_t^1-Z_t^2), e_i \rangle_H$ is not equal to 0, and $\langle \Delta_z f(t), e_i \rangle_H := 0$, otherwise. Then by the same arguments as in [BPS02, Theorem 7.2] and using Proposition 3.6.1, the assertion follows.

Now we consider a control problem associated to the Markov process X. An admissible control is a process $\theta(t, \omega)$ which is progressively measurable with respect to the filtration $(\mathcal{F}_t)_{t\geq 0}$ and which takes values in a compact subset K of some metric space. We denote by Θ the class of admissible controls.

A bounded measurable function $c: [0,T] \times E \times K \to H$ is given and we suppose that it is continuous with respect of the last variable. For a given admissible control θ we define $N_t^{\theta} = \int_0^t c_s(X_s, \theta_s) dM_s, \Gamma_t^{\theta} = \exp(N_t^{\theta} - \frac{1}{2} \langle N^{\theta} \rangle_t)$, and $P^{\theta,x} = \Gamma^{\theta} P^x$. The payoff function of the control problem is defined as

$$J^{\theta}(x) = E^{\theta,x}[\phi(X_T) + \int_0^T h(s, X_s, \theta_s)ds],$$

where ϕ and h are bounded measurable functions and h is continuous in θ . One wants to minimize the payoff function, that is to calculate the value function

$$J^*(x) = \inf_{\theta \in \Theta} J^{\theta}(x),$$

and to find an optimal control θ^* , that is an admissible control such that $J^*(x) = J^{\theta^*}(x)$.

In what follows we restrict our analysis to points $x \in \mathcal{N}^c$. We calculate $J^{\theta}(x)$ by solving the BSDE

(3.6.1)
$$Y_t^{\theta} = \phi(X_T) + \int_t^T g_s(X_s, A^{1/2}(X_s)Z_s^{\theta}, \theta_s) ds - \int_t^T Z_s^{\theta} dM_s$$

where $g:[0,T] \times E \times H \times K \to \mathbb{R}$ is the Hamiltonian defined by

$$g(s, x, z, \theta) = h(s, x, \theta) + 2\langle A^{1/2}(x)c(s, x, \theta), z \rangle_{H^1}$$

Then by Itô's formula and the same arguments as in [BPS02, Section 7] we have

$$J^{\theta}(x) = Y_0^{\theta, x}$$

where $Y_0^{\theta,x}$ is the initial value of the solution of the preceding equation (3.6.1) under P^x .

In order to calculate the value function and to produce the optimal control we have to solve the following BSDE

(3.6.2)
$$Y_t^* = \phi(X_T) + \int_t^T g^*(s, X_s, A^{1/2}(X_s)Z_s^*) ds - \int_t^T Z_s^* dM_s,$$

where $g^*(s, x, z) = \inf_{\theta \in K} g(s, x, z, \theta)$.

It is easy to check that $z \to g^*(s, x, z)$ is Lipschitz continuous, so that there exists a unique solution (Y^*, Z^*) of equation (3.6.2). Then we know the initial value of the solution is a constant: $Y_0^{*,x} = E^x Y_0^*$.

Since g is continuous as a function of θ and K is a compact set, the infimum is attained at a point θ^* and one may choose a measurable function $(s, x, z) \rightarrow \theta^*(s, x, z)$ which realizes the infimum. We construct the optimal control in the following way: $\theta_s^* := \theta^*(s, \omega) = \theta^*(s, X_s(\omega), Z_s^*(\omega)).$ **Corollary 3.6.3** Under the above hypotheses $J^*(x) = Y_0^{*,x}$ and θ^* is an optimal control.

Proof Using Theorem 3.6.2 and by the same arguments as in [BPS02, Corollary 7.3], the assertion follows. \Box

We now may interpret the solution of the Hamilton Jacobi Bellman equation as the value function of the above control problem. The HJB equation is

$$(\partial_t + L)u + g^*(t, x, D_{A^{1/2}}u) = 0, u(T, x) = \phi.$$

We have proved that this equation admits a unique solution in the sense of mild equations and it satisfies $u(0, x) = Y_0^{*,x} = J^*(x)$.

3.7 Appendix

For $p \geq 1$, let $M_x^p(\mathbb{R}^l)$ denote the set of all (equivalent classes of) predictable processes $\{\phi_t\}_{t\in[0,T]}$ with values in \mathbb{R}^l such that

$$\|\phi\|_{M^p_x} = (E^x[(\int_0^T |\phi_r|^2 dr)^{p/2}])^{1/p} < \infty.$$

Let $M_x^p(H^l)$ denote the set of all (equivalent classes of) predictable processes $\{\phi_t\}_{t\in[0,T]}$ with values in H^l such that

$$\|\phi\|_{M^p_x} = \left(E^x\left[\left(\int_0^T |\phi_r|_H^2 dr\right)^{p/2}\right]\right)^{1/p} < \infty.$$

 $\mathcal{S}^p_x(\mathbb{R}^l)$ denotes the set of all \mathbb{R}^l -valued, adapted and càdlàg processes $\{\phi_t\}_{t\in[0,T]}$ such that

$$\|\phi\|_{\mathcal{S}^p_x(\mathbb{R}^l)} = E^x [\sup_t |X_t|^p]^{1/p} < \infty.$$

Let ν be a measure supported by \mathcal{N}^c .

Lemma A.1 Let $\{K_t\}_{t\in[0,T]}$ and $\{H_t\}_{t\in[0,T]}$ be two progressively measurable processes with values in \mathbb{R}^l and H^l respectively, such that P^{ν} -a.s.

$$\int_0^T (|K_t| + |A^{1/2}(X_t)H_t|_H^2) dt < \infty.$$

We consider the \mathbb{R}^l -valued semimartingale $\{Y_t\}_{t\in[0,T]}$ defined by

$$Y_t = Y_0 + \int_0^t K_s ds + \int_0^t H_s dM_s.$$

Then, for any $p \ge 1$, we have

$$\begin{split} |Y_t|^p - \mathbf{1}_{p=1} L_t = &|Y_0|^p + p \int_0^t |Y_s|^{p-1} \langle \hat{Y}_s, K_s \rangle ds + p \int_0^t |Y_s|^{p-1} \langle \hat{Y}_s, H_s. dM_s \rangle \\ &+ \frac{p}{2} \int_0^t |Y_s|^{p-2} \mathbf{1}_{Y_s \neq 0} \{ (2-p)(|A^{1/2}(X_s)H_s|_H^2 \\ &- \sum_{j,k=1}^l \langle A(X_s)H_s^j, H_s^k \rangle_H Y_s^j Y_s^k / |Y_s|^2) + (p-1)|A^{1/2}(X_s)H_s|_H^2 \} ds, \end{split}$$

where $\{L_t\}_{t\in[0,T]}$ is a continuous, increasing process with $L_0 = 0$, which increases only on the boundary of the random set $\{t \in [0,T], Y_t = 0\}$.

Proof We consider the function $u_{\varepsilon}(x) = (|x|^2 + \varepsilon^2)^{1/2}$. We have

$$\nabla u_{\varepsilon}^{p}(x) = p u_{\varepsilon}^{p-2}(x) x, \qquad D^{2} u_{\varepsilon}^{p}(x) = p u_{\varepsilon}^{p-2}(x) I + p(p-2) u_{\varepsilon}^{p-4}(x) (x \otimes x).$$

Then Itô's formula leads to the equality,

$$\begin{split} u_{\varepsilon}^{p}(Y_{t}) = & u_{\varepsilon}^{p}(Y_{0}) + p \int_{0}^{t} u_{\varepsilon}^{p-2}(Y_{s}) \langle Y_{s}, K_{s} \rangle ds + p \int_{0}^{t} u_{\varepsilon}^{p-2}(Y_{s}) \langle Y_{s}, H_{s}.dM_{s} \rangle \\ & + \frac{1}{2} \sum_{j,k} \int_{0}^{t} D_{jk}^{2} u_{\varepsilon}^{p}(Y_{s}) \langle A(X_{s})H_{s}^{j}, H_{s}^{k} \rangle_{H} ds. \end{split}$$

It remains to pass to the limit when $\varepsilon \to 0$ in this identity. We have

$$\int_0^t u_{\varepsilon}^{p-2}(Y_s) \langle Y_s, K_s \rangle ds \to \int_0^t |Y_s|^{p-1} \langle \hat{Y}_s, K_s \rangle ds.$$

and uniformly on [0, T] in P^{ν} -probability,

$$\int_0^t u_{\varepsilon}^{p-2}(Y_s) \langle Y_s, H_s. dM_s \rangle \to \int_0^t |Y_s|^{p-1} \langle \hat{Y}_s, H_s. dM_s \rangle.$$

This convergence of stochastic integrals follows from the following convergence

$$\int_0^T |Y_r|^2 \mathbb{1}_{Y_r \neq 0} |A^{1/2}(X_s) H_r|_H^2 (|Y_r|^{p-2} - u_{\varepsilon}^{p-2}(Y_r))^2 dr \to 0,$$

which is clear from the dominated convergence theorem.

We have

$$\begin{split} &\sum_{j,k} D_{jk}^2 u_{\varepsilon}^p(Y_s) \langle A(X_s) H_s^j, H_s^k \rangle_H \\ = &p(2-p) (|Y_s| u_{\varepsilon}^{-1}(Y_s))^{4-p} |Y_s|^{p-2} 1_{Y_s \neq 0} (|A^{1/2}(X_s) H_s|_H^2 \\ &- \sum_{j,k=1}^l \langle A(X_s) H_s^j, H_s^k \rangle_H Y_s^j Y_s^k / |Y_s|^2) \\ &+ p(p-1) (|Y_s| u_{\varepsilon}^{-1}(Y_s))^{4-p} |Y_s|^{p-2} 1_{Y_s \neq 0} |A^{1/2}(X_s) H_s|_H^2 + C_s^{\varepsilon}(p), \end{split}$$

where $C_s^{\varepsilon}(p) = p\varepsilon^2 |A^{1/2}(X_s)H_s|_H^2 u_{\varepsilon}^{p-4}(Y_s)$. Furthermore, one has

$$|A^{1/2}(X_s)H_s|_H^2 \ge \sum_{j,k=1}^l \langle A(X_s)H_s^j, H_s^k \rangle_H Y_s^j Y_s^k / |Y_s|^2.$$

Moreover,

$$\frac{|Y_s|}{u_{\varepsilon}(Y_s)} \uparrow 1_{Y_s \neq 0},$$

as $\varepsilon \to 0$. Hence by monotone convergence, as $\varepsilon \to 0$,

$$\int_{0}^{t} (|Y_{s}|u_{\varepsilon}^{-1}(Y_{s}))^{4-p}|Y_{s}|^{p-2} \mathbb{1}_{Y_{s}\neq0} \{(2-p)(|A^{1/2}(X_{s})H_{s}|_{H}^{2} - \sum_{j,k=1}^{l} \langle A(X_{s})H_{s}^{j}, H_{s}^{k} \rangle_{H} Y_{s}^{j} Y_{s}^{k} / |Y_{s}|^{2}) + (p-1)|A^{1/2}(X_{s})H_{s}|_{H}^{2} \} ds$$

converges to

$$\begin{split} &\int_{0}^{t}|Y_{s}|^{p-2}1_{Y_{s}\neq0}\{(2-p)(|A^{1/2}(X_{s})H_{s}|_{H}^{2}-\sum_{j,k=1}^{l}\langle A(X_{s})H_{s}^{j},H_{s}^{k}\rangle_{H}Y_{s}^{j}Y_{s}^{k}/|Y_{s}|^{2}) \\ &+(p-1)|A^{1/2}(X_{s})H_{s}|_{H}^{2}\}ds, \end{split}$$

 P^{ν} -a.s. for all $0 \leq t \leq T$. By the same arguments as in [BDHPS03, Lemma 2.2], we have $\{L_t^{\varepsilon}(p) := \int_0^t C_s^{\varepsilon}(p) ds\}_{t \in [0,T]}$ converges, as $\varepsilon \to 0$, to a continuous increasing process $\{L_t(p)\}_{t \in [0,T]}$, and $L_t(p) \equiv 0$ for $p \neq 1$. Furthermore, $L_t(1)$ increases only on the boundary of the random set $\{t \in [0,T], Y_t = 0\}$. Now the assertions follows. \Box

Corollary A.2 If (Y, Z) is a solution of the BSDE, $p \ge 1$, $c(p) = p[(p-1) \land 1]/2$ and $0 \le t \le u \le T$, then

$$|Y_t|^p + c(p) \int_t^u |Y_s|^{p-2} \mathbb{1}_{Y_s \neq 0} |A^{1/2}(X_s) Z_s|_H^2 ds$$

$$\leq |Y_u|^p + p \int_t^u |Y_s|^{p-1} \langle \hat{Y}_s, f(s, Y_s, A^{1/2}(X_s)Z_s) \rangle ds$$
$$- p \int_t^u |Y_s|^{p-1} \langle \hat{Y}_s, Z_s. dM_s \rangle.$$

Now we state some estimates concerning solutions to the BSDE. In the following we assume that p > 1 and make use of the following assumption

(A):
$$\forall (t, y, z) \in [0, T] \times \mathbb{R}^l \otimes H^l, \qquad \langle \hat{y}, f(t, y, z) \rangle \le f_t + \mu |y| + C|z|, \ P^\nu - a.s.,$$

where $\mu \in \mathbb{R}, C \ge 0$ and $\{f_t\}_{t \in [0,T]}$ is a non-negative progressively measurable process. Let us set $F := \int_0^T f_r dr$.

Lemma A.3 Suppose assumption (A) holds and that for some p > 0, F^p is integrable. Let (Y, Z) be a solution to the BSDE. If $Y \in \mathcal{S}^p_{\nu}$, then $Z \in \mathcal{M}^p_{\nu}$ and there exists a constant C_p depending only on p such that for every $a \ge \mu + C^2$,

$$E^{\nu}[(\int_{0}^{T} e^{2at} |A^{1/2}(X_{r})Z_{r}|_{H}^{2} dr)^{p/2}] \leq C_{p} E^{\nu}[\sup_{t} e^{apt} |Y_{t}|^{p} + (\int_{0}^{T} e^{ar} f_{r} dr)^{p}].$$

Proof We note that (Y, Z) solves the BSDE with data (ξ, f) iff

$$(\bar{Y}_t, \bar{Z}_t) := (e^{at}Y_t, e^{at}Z_t)$$

solves the BSDE with data $(e^{aT}\xi, f')$, where

$$f'(t, y, z) := e^{at} f(t, e^{-at}y, e^{-at}z) - ay.$$

We restrict ourselves to the case that a = 0 and $\mu + C^2 \leq 0$. For each integer $n \geq 1$, let us introduce the stopping time

$$\tau_n := \inf\{t \in [0,T], \int_0^t |A^{1/2}(X_r)Z_r|_H^2 dr \ge n\} \wedge T.$$

By Itô's formula we get

$$|Y_0|^2 + \int_0^{\tau_n} |A^{1/2}(X_r)Z_r|_H^2 dr = |Y_{\tau_n}|^2 + 2\int_0^{\tau_n} \langle Y_r, f(r, Y_r, Z_r) \rangle dr - 2\int_0^{\tau_n} \langle Y_r, Z_r.dM_r \rangle.$$

By (A) and since $\mu + C^2 \leq 0$, we have

$$2\langle y, f(r, y, z) \rangle \le 2|y|f_r + 2\mu|y|^2 + 2C^2|y|^2 + |z|^2/2 \le 2|y|f_r + |z|^2/2.$$

Hence we deduce that

$$\frac{1}{2} \int_0^{\tau_n} |A^{1/2}(X_r)Z_r|_H^2 dr \le Y_*^2 + 2Y_* \int_0^T f_r dr + 2|\int_0^{\tau_n} \langle Y_r, Z_r.dM_r \rangle|,$$

where Y_* denotes $\sup_{t \in [0,T]} |Y_t|$, and thus

$$\left(\int_{0}^{\tau_{n}} |A^{1/2}(X_{r})Z_{r}|_{H}^{2} dr\right)^{p/2} \leq c_{p}(Y_{*}^{p} + (\int_{0}^{T} f_{r} dr)^{p} + |\int_{0}^{\tau_{n}} \langle Y_{r}, Z_{r}.dM_{r}\rangle|^{p/2}).$$

By the BDG inequality, we get

$$c_{p}E^{\nu}[|\int_{0}^{\tau_{n}} \langle Y_{r}, Z_{r}.dM_{r} \rangle|^{p/2}] \leq d_{p}E^{\nu}[(\int_{0}^{\tau_{n}} |Y_{r}|^{2}|A^{1/2}(X_{r})Z_{r}|_{H}^{2}dr)^{p/4}]$$

$$\leq d_{p}E^{\nu}[|Y_{*}|^{p/2}(\int_{0}^{\tau_{n}} |A^{1/2}(X_{r})Z_{r}|_{H}^{2}dr)^{p/4}]$$

$$\leq \frac{d_{p}^{2}}{2}E^{\nu}[Y_{*}^{p}] + \frac{1}{2}E^{\nu}[(\int_{0}^{\tau_{n}} |A^{1/2}(X_{r})Z_{r}|_{H}^{2}dr)^{p/2}].$$

Furthermore, we have

$$E^{\nu}[(\int_0^{\tau_n} |A^{1/2}(X_r)Z_r|_H^2 dr)^{p/2}] \le C_p E^{\nu}[Y_*^p + (\int_0^T f_r dr)^p].$$

Letting $n \to \infty$ we obtain

$$E^{\nu}[(\int_{0}^{T} |A^{1/2}(X_{r})Z_{r}|_{H}^{2} dr)^{p/2}] \leq C_{p}E^{\nu}[Y_{*}^{p} + (\int_{0}^{T} f_{r} dr)^{p}].$$

Proposition A.4 Suppose that assumption (A) holds and that for some p > 1, F belongs to L^p . Let (Y, Z) be a solution to the BSDE where Y belongs to S^p_{ν} . Then there exists a constant C_p , depending only on p, such that for every $a \ge \mu + C^2/[1 \land (p-1)]$,

$$E^{\nu}[\sup_{t} e^{apt}|Y_{t}|^{p} + (\int_{0}^{T} e^{2at}|A^{1/2}(X_{r})Z_{r}|^{2}_{H}dr)^{p/2}] \leq C_{p}E^{\nu}[e^{apT}|\xi|^{p} + (\int_{0}^{T} e^{ar}f_{r}dr)^{p}].$$

Proof We restrict ourselves the proof to the case a = 0 and $\mu + C^2/[1 \wedge (p-1)] \leq 0$; By Corollary A.2 and $\langle \hat{y}, f(r, y, z) \rangle \leq f_r + \mu |y| + C|z|$, we have

$$|Y_t|^p + c(p) \int_t^T |Y_r|^{p-2} \mathbf{1}_{Y_s \neq 0} |A^{1/2}(X_r) Z_r|_H^2 dr$$

$$\leq |\xi|^{p} + p \int_{t}^{T} |Y_{r}|^{p-1} f_{r} + \mu |Y_{r}|^{p} + C|Y_{r}|^{p-1} |A^{1/2}(X_{r})Z_{r}|_{H} dr - p \int_{t}^{T} |Y_{r}|^{p-1} \langle \hat{Y}_{r}, Z_{r}.dM_{r} \rangle$$

Now by the same arguments as in [BDHPS03, Proposition 3.2] the assertion follows. \Box

Proof of Proposition 3.4.3 We consider P^x for $x \in A_p$.

We note that (Y, Z) solves the BSDE with data (ξ, f) iff

$$(\bar{Y}_t, \bar{Z}_t) := (e^{\alpha_t} Y_t, e^{\alpha_t} Z_t)$$

solves the BSDE with data $(e^{\alpha_T}\xi, f')$, where

$$f'(t, y, z) := e^{\alpha_t} f(t, e^{-\alpha_t} y, e^{-\alpha_t} z) - \mu_t y.$$

We replace ($\Omega 2$) by the condition that for t, ω, y, y', z arbitrary,

$$\langle y - y', f(t, \omega, y, z) - f(t, \omega, y', z) \rangle \le 0.$$

Step 1. We assume that f is Lipschitz continuous with respect to both y and z, and that f, ξ are bounded. We define a mapping Φ from $B_x^2 := M_x^2(\mathbb{R}^l) \times M_x^2(H^l)$ into itself as follows. Given $(U, V) \in B_x^2$, $\Phi(U, V) := (Y, Z)$, where (Y, Z) is the solution of the BSDE (3.4.1) associated with the data $(\xi, f(U, A^{1/2}(X)V))$ by Lemma 3.4.1. As

$$\langle \int_0^t \phi_s . dM_s \rangle_t = \int_0^t \langle A(X_s) \phi_s, \phi_s \rangle_H ds,$$

by Itô's formula and the BDG inequality, we have

$$E^x[\sup_{t\in[0,T]}|Y_t|^2]<\infty.$$

Let $(U,V), (U',V') \in B_x^2, (Y,Z) = \Phi(U,V), (Y',Z') = \Phi(U',V'), (\overline{U},\overline{V}) = (U - U', V - V'), (\overline{Y}, \overline{Z}) = (Y - Y', Z - Z')$. It follows from Itô's formula that for each $\gamma \in \mathbb{R}$,

$$\begin{split} & e^{\gamma t} E^{x} |\bar{Y}_{t}|^{2} + E^{x} \int_{t}^{T} e^{\gamma s} (\gamma |\bar{Y}_{s}|^{2} + |A^{1/2}(X_{s})\bar{Z}_{s}|_{H}^{2}) ds \\ & \leq 2K E^{x} \int_{t}^{T} e^{\gamma s} |\bar{Y}_{s}| (|\bar{U}_{s}| + |A^{1/2}(X_{s})\bar{V}_{s}|_{H}) ds \\ & \leq 4K^{2} E^{x} \int_{t}^{T} e^{\gamma s} |\bar{Y}_{s}|^{2} + \frac{1}{2} E^{x} \int_{t}^{T} e^{\gamma s} (|\bar{U}_{s}|^{2} + |A^{1/2}(X_{s})\bar{V}_{s}|_{H}^{2}) ds, \end{split}$$

where K is the Lipschitz constant of f. We choose $\gamma = 1 + 4K^2$. Then

$$E^{x} \int_{0}^{T} e^{\gamma s} (|\bar{Y}_{s}|^{2} + |A^{1/2}(X_{s})\bar{Z}_{s}|_{H}^{2}) ds \leq \frac{1}{2} E^{x} \int_{0}^{T} e^{\gamma s} (|\bar{U}_{s}|^{2} + |A^{1/2}(X_{s})\bar{V}_{s}|_{H}^{2}) ds,$$

from which it follows that Φ is a strict contraction on B_x^2 equipped with the norm:

$$|||(Y,Z)|||_{\gamma}^{x} = (E^{x} \int_{0}^{T} e^{\gamma t} (|Y_{t}|^{2} + |A^{1/2}(X_{t})Z_{t}|_{H}^{2}) dt)^{1/2}.$$

We define a sequence (Y^n, Z^n) by $(Y^{n+1}, Z^{n+1}) = \Phi(Y^n, Z^n)$. For $\gamma = 1 + 4K^2$, we have

$$E^{x} \int_{0}^{T} e^{\gamma s} (|Y_{s}^{n} - Y_{s}^{n+1}|^{2} + |A^{1/2}(X_{s})(Z_{s}^{n} - Z_{s}^{n+1})|_{H}^{2}) ds$$

$$\leq \frac{1}{2^{n}} E^{x} \int_{0}^{T} e^{\gamma s} (|Y_{s}^{0} - Y_{s}^{1}|^{2} + |A^{1/2}(X_{s})(Z_{s}^{0} - Z_{s}^{1})|_{H}^{2}) ds.$$

Then for any $x \in A^2$ we have the a.e. pointwise convergence of (Y_s^n, Z_s^n) under P^x . We denote the limit by (Y_s, Z_s) . Then it is the fixed point of Φ under the norm $|||(Y, Z)|||_{\gamma}^x$. So we have (Y_s, Z_s) is the solution of the BSDE.

Step 2. We assume f, ξ are bounded.

We need the following proposition.

Proposition A.5 Suppose f, ξ are bounded. Given $V_t \in \bigcap_x M_x^2(H^l)$, there exists a unique pair of predictable processes $(Y_t, Z_t) \in M_x^2 \times M_x^2(H^l), \forall x \in \mathcal{N}^c$, satisfying for all $P^x, x \in \mathcal{N}^c$

$$Y_t = \xi + \int_t^T f(s, Y_s, V_s) ds - \int_t^T Z_s dM_s, \qquad 0 \le t \le T.$$

Using Proposition A.5, we construct a mapping Φ from B_x^2 into itself as follows. For any $(U, V) \in B_x^2$, $(Y, Z) = \Phi(U, V)$ is the solution of the BSDE

$$Y_t = \xi + \int_t^T f(s, Y_s, V_s) ds - \int_t^T Z_s dM_s, \qquad 0 \le t \le T.$$

Then as in **Step 1**, we have

$$e^{\gamma t} E^{x} |\bar{Y}_{t}|^{2} + E^{x} \int_{t}^{T} e^{\gamma s} (\gamma |\bar{Y}_{s}|^{2} + |A^{1/2}(X_{s})\bar{Z}_{s}|_{H}^{2}) ds$$

$$= 2E^{x} \int_{t}^{T} e^{\gamma s} \langle \bar{Y}_{s}, f(Y_{s}, A^{1/2}(X_{s})V_{s}) - f(Y'_{s}, A^{1/2}(X_{s})V'_{s}) \rangle ds$$

$$\leq 2KE^{x} \int_{t}^{T} e^{\gamma s} |\bar{Y}_{s}| \times |A^{1/2}(X_{s})\bar{V}_{s}|_{H} ds$$

$$\leq E^{x} \int_{t}^{T} e^{\gamma s} (2K^{2}|\bar{Y}_{s}|^{2} + \frac{1}{2}|A^{1/2}(X_{s})\bar{V}_{s}|^{2}_{H}) ds.$$

Then by the same argument as in Step 1, the assertion of Proposition 3.4.3 follows, if f, ξ are bounded.

Proof of Proposition A.5 We shall write f(s, y) for $f(s, y, V_s)$ and $|\xi|^2 + \sup_t |f(t, 0)|^2 \le C$ a.s.. We define

$$f^n(t,y) := (\rho_n * f(t,\cdot))(y),$$

where $\rho_n : \mathbb{R}^l \mapsto \mathbb{R}^+, n \in \mathbb{N}$ is a sequence of smooth functions with compact support which approximates the Dirac measure at 0, satisfying $\int \rho_n(z) dz = 1$. Then f^n is locally Lipschitz in y, uniformly with respect to s and ω .

Define for each $m \in \mathbb{N}$

$$f^{n,m}(t,y) := f^n(t, \frac{\inf(m, |y|)}{|y|}y).$$

Then $f^{n,m}$ is globally Lipschitz and bounded, uniformly w.r.t. (t, ω) . As in Step 1, we have a unique pair $(Y_t^{n,m}, Z_t^{n,m}) \in M_x^2(\mathbb{R}^l) \times M_x^2(H^l)$ such that

$$Y_t^{n,m} = \xi + \int_t^T f^{n,m}(s, Y_s^{n,m}) ds - \int_t^T Z_s^{n,m} dM_s, \qquad 0 \le t \le T.$$

By Itô's formula, we have

$$|Y_t^{n,m}|^2 \le e^T C, \qquad 0 \le t \le T.$$

Consequently, for $m^2 > e^T C$, $(Y_t^{n,m}, Z_t^{n,m})$ does not depend on m. Therefore, we denote it by (Y_t^n, Z_t^n) . $(Y^l - Y^k, Z^l - Z^k)$ is the solution of BSDE associated with $(f^l(t, y + Y_t^k) - f^k(Y_t^k), 0)$. Hence by Proposition A.4, we have

$$E^{x}(\sup_{0 \le t \le T} |Y_{t}^{k} - Y_{t}^{l}|^{2}) + E^{x}(\int_{0}^{T} |A^{1/2}(X_{t})(Z_{t}^{k} - Z_{t}^{l})|_{H}^{2}dt) \le KE^{x}[\int_{0}^{T} |f^{k}(t, Y_{t}^{k}) - f^{l}(t, Y_{t}^{k})|^{2}dt]$$

For fixed (t, ω) we have,

$$\sup_{k>l} \int_0^T |f^k(t, Y_t^k) - f^l(t, Y_t^k)|^2 dt \to 0, \text{ as } l \to \infty$$

Then we obtain

$$\sup_{k>l} E^x \int_0^T |f^k(t, Y_t^k) - f^l(t, Y_t^k)|^2 dt \le E^x \sup_{k>l} \int_0^T |f^k(t, Y_t^k) - f^l(t, Y_t^k)|^2 dt \to 0, \text{ as } l \to \infty$$

Therefore, we obtain a sequence of representable variables that converges rapidly enough under all measures $P^x, x \in \mathcal{N}^c$. For each $l = 0, 1, \dots$ set

$$n_{l}(x) = \inf\{n > n_{l-1}(x); \sup_{k \ge n} E^{x} [\int_{0}^{T} |f^{k}(t, Y_{t}^{k}) - f^{n}(t, Y_{t}^{k})|^{2} dt] < \frac{1}{2^{l}} \}$$
$$\bar{Y}^{l} = Y^{n_{l}(X_{0})}, \bar{Z}^{l} = Z^{n_{l}(X_{0})}.$$

With this sequence one may pass to the limit and define $Z_s := \limsup_{l \to \infty} \overline{Z}_s^l$ and the assertion follows..

Now we continue the proof of Proposition 3.4.3.

Step 3. Now we assume that ξ and $\sup_t |f_t^0|$ are bounded random variables. Let r be a positive real number such that

$$\sqrt{e^{(1+C^2)T}(\|\xi\|_{\infty} + T\|f^0\|_{\infty})} < r.$$

Let θ_r be a smooth function such that $0 \leq \theta_r \leq 1, \theta_r(y) = 1$ for $|y| \leq r$ and $\theta_r(y) = 0$, if $|y| \geq r+1$. For $n \in \mathbb{N}$, we set $q_n(z) := z \frac{n}{|z|_H \vee n}$ and

$$h^{n}(t, y, z) := \theta_{r}(y)(f(t, y, q_{n}(z)) - f_{t}^{0})\frac{n}{f'^{,r+1} \vee n} + f_{t}^{0}.$$

By [BDHPS03, Theorem 4.2], we have that each h_n satisfies ($\Omega 2$) with a positive constant and (ξ, h^n) are bounded. The BSDE associated to (ξ, h^n) has a unique solution (Y^n, Z^n) in the space $S_x^2 \times \mathcal{M}_x^2(H^l)$. By the same arguments as in [Pa99, Proposition 2.4] we have $||Y^n||_{\infty} < r$. By Proposition A.4, $||Z^n||_{\mathcal{M}_x^2(H^l)} \leq r'$. Hence (Y^n, Z^n) is a solution to the BSDE associated to (ξ, f_n) where

$$f^{n}(t, y, z) := (f(t, y, q_{n}(z)) - f_{t}^{0}) \frac{n}{f'^{,r+1} \vee n} + f_{t}^{0}.$$

Since $(Y^{n+i} - Y^n, Z^{n+i} - Z^n)$ is the solution of BSDE associated with $(f^{n+i}(t, y + Y_t^n, z + Z_t^n) - f^n(Y_t^n, Z_t^n), 0)$ and

$$\begin{aligned} &\langle y, f^{n+i}(t, y+Y_t^n, z+Z_t^n) - f^n(t, Y_t^n, Z_t^n) \rangle \\ &= \langle y, f^{n+i}(t, y+Y_t^n, z+Z_t^n) - f^{n+i}(t, Y_t^n, Z_t^n) \rangle + \langle y, f^{n+i}(t, Y_t^n, Z_t^n) - f^n(t, Y_t^n, Z_t^n) \rangle \\ &\leq C |y||z| + |y||f^{n+i}(t, Y_t^n, Z_t^n) - f^n(t, Y_t^n, Z_t^n)|, \end{aligned}$$

by Proposition A.4 we have

$$E^{x}(\sup_{0 \le t \le T} |Y_{t}^{n+i} - Y_{t}^{n}|^{2}) + E^{x}(\int_{0}^{T} |A^{1/2}(X_{t})(Z_{t}^{n+i} - Z_{t}^{n})|_{H}^{2}dt)$$

$$\leq KE^{x}[\int_{0}^{T} |f^{n+i}(t, Y_{t}^{n}, Z_{t}^{n}) - f^{n}(t, Y_{t}^{n}, Z_{t}^{n})|^{2}dt].$$

Since $||Y^n||_{\infty} \leq r$, we have

$$|f^{n+i}(t, Y^n_t, Z^n_t) - f^n(t, Y^n_t, Z^n_t)| \le 2C|Z^n_t|_H \mathbf{1}_{|Z^n_t|_H > n} + 2C|Z^n_t|_H \mathbf{1}_{f', r+1 > n} + 2f', r+1(t)\mathbf{1}_{f', r+1 > n},$$

and the above formula converges to 0, uniformly in i as $n \to \infty$. Follow the same arguments as in the proof of Proposition A.5, the assertion follows in this case.

Step 4. Consider the general case. For each $n \in \mathbb{N}$, let us define

$$\xi^n := q_n(\xi), \qquad f^n(t, y, z) := f(t, y, z) - f_t^0 + q_n(f_t^0).$$

For each pair (ξ^n, f^n) , the BSDE has a unique solution (Y^n, Z^n) in L^2 by Step 3. By Proposition A.4, we have

$$E^{x}(\sup_{0 \le t \le T} |Y_{t}^{n+i} - Y_{t}^{n}|^{p}) + E^{x}(\int_{0}^{T} |A^{1/2}(X_{t})(Z_{t}^{n+i} - Z_{t}^{n})|_{H}^{2} dt)^{p/2}$$

$$\leq K_{1}E^{x}[|\xi^{n+i} - \xi^{n}|^{p} + (\int_{0}^{T} |q_{n+i}(f_{t}^{0}) - q_{n}(f_{t}^{0})|dt)^{p}].$$

The right hand side of the last inequality clearly tends to 0, as $n \to \infty$, uniformly in *i* and the assertion follows.

Chapter 4

Stochastic quasi-geostrophic equation

In this chapter, we study the 2D stochastic quasi-geostrophic equation in \mathbb{T}^2 for general parameter $\alpha \in (0, 1)$ and multiplicative noise. We prove it is uniquely ergodic provided the noise is non-degenerate for $\alpha > \frac{2}{3}$. In this case, the convergence to the (unique) invariant measure is exponentially fast. In the general case, we prove the existence of Markov selections. In Section 4.1, we introduce some notations and preliminaries for quasi-geostrophic equation. In Section 4.2, we prove the existence of Markov selections for the solution of the stochastic quasi-geostrophic equation with $\alpha \in (0, 1)$. In Section 4.3, we prove the Markov semigroup associated with the solution of the stochastic quasi-geostrophic equation is strong Feller and irreducible if the noise is non-degenerate. Furthermore, it is strongly mixing. In Section 4.4 we prove that the convergence to the (unique) invariant measure is exponentially fast. In Section 4.5, we prove the above results if the noise is mildly degenerate. The main results of this chapter have already been submitted for publication, see [RZZ12].

4.1 Notations and Preliminaries

We consider the usual abstract form of equations (1.7)-(1.9). In the following, we will restrict ourselves to flows which have zero average on the torus, i.e.

$$\int_{\mathbb{T}^2} \theta d\xi = 0.$$

Thus (1.9) can be restated as

$$u = \left(-\frac{\partial \psi}{\partial \xi_2}, \frac{\partial \psi}{\partial \xi_1}\right) \text{ and } (-\triangle)^{1/2} \psi = -\theta.$$

Set $H = \{f \in L^2(\mathbb{T}^2) : \int_{\mathbb{T}^2} f d\xi = 0\}$ and let $|\cdot|$ and $\langle ., . \rangle$ denote the norm and inner product in H respectively. On the periodic domain \mathbb{T}^2 , $\{\sin(k\xi)|k \in \mathbb{Z}^2_+\} \cup$ $\{\cos(k\xi)|k \in \mathbb{Z}^2_-\}$ form an eigenbasis of $-\Delta$. Here $\mathbb{Z}^2_+ = \{(k_1, k_2) \in \mathbb{Z}^2|k_2 > 0\} \cup \{(k_1, 0) \in \mathbb{Z}^2|k_1 > 0\}, \mathbb{Z}^2_- = \{(k_1, k_2) \in \mathbb{Z}^2| - k \in \mathbb{Z}^2_+\}, x \in \mathbb{T}^2$, and the corresponding eigenvalues are $|k|^2$. Define

$$||f||_{H^s}^2 = \sum_k |k|^{2s} \langle f, e_k \rangle^2$$

and let H^s denote the Sobolev space of all f for which $||f||_{H^s}$ is finite. Set $\Lambda = (-\Delta)^{1/2}$. Then

$$||f||_{H^s} = |\Lambda^s f|.$$

By the singular integral theory of Calderón and Zygmund (cf. [St70, Chapter 3]), for any $p \in (1, \infty)$, there is a constant C = C(p), such that

(4.1.1)
$$||u||_{L^p} \le C(p) ||\theta||_{L^p}.$$

Fix $\alpha \in (0,1)$ and define the linear operator $A: D(A) = H^{2\alpha}(\mathbb{T}^2) \subset H \to H$ as $Au := \kappa (-\Delta)^{\alpha} u$. The operator A is positive definite and selfadjoint with the same eigenbasis as that of $-\Delta$ mentioned above. Denote the eigenvalues of A by $0 < \lambda_1 \leq \lambda_2 \leq \cdots$, and renumber the above eigenbasis correspondingly as e_1, e_2, \ldots We also set $||u|| := |A^{1/2}u|$, then $||\theta||^2 \geq \lambda_1 |\theta|^2$.

First we recall the following important product estimates (cf. [Re95, Lemma A.4]):

Lemma 4.1.1 Suppose that s > 0 and $p \in (1, \infty)$. If $f, g \in S$, the Schwartz class, then

(4.1.2)
$$\|\Lambda^{s}(fg)\|_{L^{p}} \leq C(\|f\|_{L^{p_{1}}}\|g\|_{H^{s,p_{2}}} + \|g\|_{L^{p_{3}}}\|f\|_{H^{s,p_{4}}}),$$

with $p_i \in (1, \infty), i = 1, ..., 4$ such that

$$\frac{1}{p} = \frac{1}{p_1} + \frac{1}{p_2} = \frac{1}{p_3} + \frac{1}{p_4}.$$

We shall use as well the following useful Sobolev inequality (cf [St70, Chapter

V]):

Lemma 4.1.2 Suppose that $q > 1, p \in [q, \infty)$ and

$$\frac{1}{p} + \frac{\sigma}{2} = \frac{1}{q}.$$

Suppose that $\Lambda^{\sigma} f \in L^q$, then $f \in L^p$ and there is a constant $C \ge 0$ such that

$$\|f\|_{L^p} \le C \|\Lambda^{\sigma} f\|_{L^q}.$$

We consider the abstract stochastic evolution equation in place of Eqs (1.7)-(1.9),

(4.1.3)
$$\begin{cases} d\theta(t) + A\theta(t)dt + u(t) \cdot \nabla \theta(t)dt = G(\theta(t))dW(t), \\ \theta(0) = \theta_0, \end{cases}$$

where u satisfies (1.9) and W(t) is a cylindrical Wiener process in a separable Hilbert space K defined on a probability space (Ω, \mathcal{F}, P) . Here G is a mapping from H^{α} to $L_2(K, H)$, where $L_2(K, H)$ denote all the Hilbert-Schmidt operator from K to H.

Consider the following conditions:

(G.1) (i) $|G(\theta)|^2_{L_2(K,H)} \leq \lambda_0 |\theta|^2 + \rho, \theta \in H^{\alpha}$, for some positive real numbers λ_0 and ρ .

(ii) If $y, y_n \in H^{\alpha}$ such that $y_n \to y$ in H, then $\lim_{n\to\infty} \|G(y_n)^*(v) - G(y)^*(v)\|_K = 0$ for all $v \in C^{\infty}(\mathbb{T}^2)$.

Remark 4.1.3 Note that, because divu = 0 for regular functions θ and v, we have

$$\langle u(s) \cdot \nabla(\theta(s) + \psi), \theta(s) + \psi \rangle = 0,$$

 \mathbf{SO}

$$\langle u(s) \cdot \nabla \theta(s), \psi \rangle = -\langle u(s) \cdot \nabla \psi, \theta(s) \rangle$$

4.2 Markov selections in the general case

In this section, we will use [GRZ09, Theorem 4.7] to get an almost sure Markov family $(P_x)_{x \in L^2}$ for Eq. (4.1.3). Here we use the same notations as [GRZ09]. Below we choose

$$H = \mathbb{Y} = L^2(\mathbb{T}^2)$$

and

$$\mathbb{X} = (H^{2+2\alpha})^*, \qquad \mathbb{X}^* = H^{2+2\alpha}.$$

Then X is a Hilbert space and $X^* \subset Y$ compactly. Let $\mathcal{E} = \{e_i, i \in \mathbb{N}\}$ be the orthonormal basis of H introduced in Section 4.1. We define the operator \mathcal{A} as follows: for $\theta \in C^{\infty}(\mathbb{T}^2)$

$$\mathcal{A}(\theta) := -\kappa (-\Delta)^{\alpha} \theta - u \cdot \nabla \theta,$$

where u satisfies (1.9). Then by Lemma 4.2.3 below, \mathcal{A} can be extended to an operator $\mathcal{A}: H \to \mathbb{X}$. For θ not in H define $\mathcal{A}(\theta) := \infty$.

 Set

$$\Omega := C([0,\infty); \mathbb{X}),$$

and let \mathcal{B} denote the σ -field of Borel sets of Ω and let $\mathcal{P}(\Omega)$ denote the set of all probability measures on (Ω, \mathcal{B}) . Define the canonical process $\xi : \Omega \to \mathbb{X}$ as

$$\xi_t(\omega) = \omega(t).$$

For each t, $\mathcal{B}_t = \sigma(\xi_s : 0 \leq s \leq t)$. Given $P \in \mathcal{P}(\Omega)$ and t > 0, let $P(\cdot | \mathcal{B}_t)(\omega)$ denote a regular conditional probability distribution of P given \mathcal{B}_t . In particular, $P(\cdot | \mathcal{B}_t)(\omega) \in \mathcal{P}(\Omega)$ for every $\omega \in \Omega$ and for any bounded \mathcal{B} -measurable function fon Ω

$$E^{P}[f|\mathcal{B}_{t}] = \int_{\Omega} f(y)P(dy|\mathcal{B}_{t}), \quad P-a.s.,$$

and there exists a *P*-null set $N \in \mathcal{B}_t$ such that for every ω not in *N*

$$P(\cdot|\mathcal{B}_t)(\omega)|_{\mathcal{B}_t} = \delta_{\omega}(= \text{ Dirac measure at } \omega),$$

hence

$$P(\{y: y(s) = \omega(s), s \in [0, t]\} | \mathcal{B}_t)(\omega) = 1$$

In particular, we can consider $P(\cdot|\mathcal{B}_t)(\omega)$ as a measure on $(\Omega^t, \mathcal{B}^t)$, i.e.,

$$P(\cdot|\mathcal{B}_t)(\omega) \in \mathcal{P}(\Omega^t),$$

where $\Omega^t := C([t,\infty); \mathbb{X})$ and $\mathcal{B}^t := \sigma(\xi_s : s \ge t)$.

We say $P \in \mathcal{P}(\Omega)$ is concentrated on the paths with values in H, if there exists $A \in \mathcal{B}$ with P(A) = 1 such that $A \subset \{\omega \in \Omega : \xi_t(\omega) \in H, \forall t \ge 0\}$. The set of such measures is denoted by $\mathcal{P}_H(\Omega)$. The shift operator $\Phi_t : \Omega \to \Omega^t$ is defined by

$$\Phi_t(\omega)(s) = \omega(s-t), \ s \ge t.$$

Following [GRZ09, Definitions 2.5], we introduce the following notions.

Definition 4.2.1 A family $(P_x)_{x \in H}$ of probability measures in $\mathcal{P}_H(\Omega)$, is called an *almost sure Markov family* if for any $A \in \mathcal{B}$, $x \mapsto P_x(A)$ is $\mathcal{B}(H)/\mathcal{B}([0,1])$ measurable, and for each $x \in H$ there exists a Lebesgue null set $T_{P_x} \subset (0, \infty)$ such that for all t not in T_{P_x} and P_x -almost all $\omega \in \Omega$

$$P_x(\cdot|\mathcal{B}_t)(\omega) = P_{\omega(t)} \circ \Phi_t^{-1}.$$

We now introduce the following notion of a martingale solution to Eq. (4.1.3) and write $\xi(t)$ instead of ξ_t .

Definition 4.2.2 Let $x_0 \in H$. A probability measure $P \in \mathcal{P}(\Omega)$ is called a martingale solution of Eq. (4.1.3) with initial value x_0 , if:

(M1) $P(\xi(0) = x_0) = 1$ and for any $n \in \mathbb{N}$

$$P\{\xi \in \Omega : \int_0^n \|\mathcal{A}(\xi(s))\|_{\mathbb{X}} ds + \int_0^n \|G(\xi(s))\|_{L_2(K;H)}^2 ds < +\infty\} = 1;$$

(M2) for every $l \in \mathcal{E}$, the process

$$M_l(t,\xi) :=_{\mathbb{X}} \langle \xi(t), l \rangle_{\mathbb{X}^*} - \int_0^t {}_{\mathbb{X}} \langle \mathcal{A}(\xi(s)), l \rangle_{\mathbb{X}^*} ds$$

is a continuous square-integrable \mathcal{F}_t -martingale under P, whose quadratic variation process is given by

$$\langle M_l \rangle(t,\xi) := \int_0^t \|G^*(\xi(s))(l)\|_K^2 ds,$$

where the asterisk denotes the adjoint operator of $G(\xi(s))$;

(M3) for any $p \in \mathbb{N}$, there exist a continuous positive real function $t \mapsto C_{t,p}$ (only depending on p and \mathcal{A}, G), a lower semi-continuous positive real functional $\mathcal{N}_p : \mathbb{Y} \to [0, \infty]$, and a Lebesgue null set $T_P \subset (0, \infty)$ such that for all $0 \leq s \in [0, \infty) \setminus T_P$ and for all $t \geq s$

$$E^{P}[\sup_{r\in[s,t]}|\xi(r)|^{2p} + \int_{s}^{t} \mathcal{N}_{p}(\xi(r))dr|\mathcal{B}_{s}] \leq C_{t-s}(|\xi(s)|^{2p} + 1).$$

First, we prove the following lemma.

Lemma 4.2.3 For any $\theta_1, \theta_2 \in C^{\infty}(\mathbb{T}^2)$,

$$\|(-\Delta)^{\alpha}\theta_1 - (-\Delta)^{\alpha}\theta_2\|_{\mathbb{X}} \le C_1|\theta_1 - \theta_2|,$$

$$\|u_1 \cdot \nabla \theta_1 - u_2 \cdot \nabla \theta_2\|_{\mathbb{X}} \le C_2(|\theta_1| + |\theta_2|)|\theta_1 - \theta_2|$$

for constants C_1, C_2 . In particular, the operator $\mathcal{A} : C^{\infty}(\mathbb{T}^2) \to \mathbb{X}$ extends to an operator $\mathcal{A} : H \to \mathbb{X}$ by continuity.

Proof We only prove the second assertion, the first can be proved analogously. By the Sobolev embedding theorem we have

$$\begin{split} & \|u_1 \cdot \nabla \theta_1 - u_2 \cdot \nabla \theta_2\|_{\mathbb{X}} \\ &= \sup_{w \in C^{\infty}(\mathbb{T}^2): \|w\|_{H^{2+2\alpha} \leq 1}} |\langle u_1 \cdot \nabla \theta_1 - u_2 \cdot \nabla \theta_2, w\rangle| \\ &= \sup_{w \in C^{\infty}(\mathbb{T}^2): \|w\|_{H^{2+2\alpha} \leq 1}} |\langle u_1 \cdot \nabla w, \theta_1 \rangle - \langle u_2 \cdot \nabla w, \theta_2 \rangle| \\ &= \sup_{w \in C^{\infty}(\mathbb{T}^2): \|w\|_{H^{2+2\alpha} \leq 1}} |\langle (u_1 - u_2) \cdot \nabla w, \theta_1 \rangle + \langle u_2 \cdot \nabla w, \theta_1 - \theta_2 \rangle| \\ &\leq C[\sup_{w \in C^{\infty}(\mathbb{T}^2): \|w\|_{H^{2+2\alpha} \leq 1}} \|\nabla w\|_{C(\mathbb{T}^2)}] (|u_1 - u_2| \cdot |\theta_1| + |\theta_1 - \theta_2| \cdot |u_2|) \\ &\leq C(|\theta_1| + |\theta_2|) |\theta_1 - \theta_2|. \end{split}$$

In the last inequality we use (4.1.1) and the constant C changes from line to line. \Box

In order to use [GRZ09, Theorem 4.7], we define the functional \mathcal{N}_1 on \mathbb{Y} as follows:

$$\mathcal{N}_1(\theta) := \begin{cases} |\Lambda^{\alpha} \theta|^2, & \text{if } \theta \in H^{\alpha}, \\ +\infty, & \text{otherwise}. \end{cases}$$

It is obvious that $\mathcal{N}_1 \in \mathfrak{U}^2$, defined in [GRZ09, Section 4]. We recall that a lower semicontinuous function $\mathcal{N} : \mathbb{Y} \to [0, \infty]$ belongs to \mathfrak{U}^2 if $\mathcal{N}(x) = 0$ implies x = 0, $\mathcal{N}(cy) \leq c^2 \mathcal{N}(y), \forall c \geq 0, y \in \mathbb{Y}$ and $\{y \in \mathbb{Y} : \mathcal{N}(y) \leq 1\}$ is relatively compact in \mathbb{Y} .

Theorem 4.2.4 Let $\alpha \in (0, 1)$ and assume G satisfies (G.1). Then for each $x_0 \in H$, there exists a martingale solution $P \in \mathcal{P}(\Omega)$ starting from x_0 to Eq. (4.1.3) in the sense of Definition 4.2.2.

Proof We only need to check (C1)-(C3) in [GRZ09, Section 4] for the above \mathcal{A} and G. For the reader's convenience, we give them as follows:

(C1)(Demi-Continuity) For any $x \in \mathbb{X}^*$, if y_n strongly converges to y in \mathbb{Y} , then

$$\lim_{n \to \infty} \mathbb{X} \langle \mathcal{A}(y_n), x \rangle_{\mathbb{X}^*} =_{\mathbb{X}} \langle \mathcal{A}(y), x \rangle_{\mathbb{X}^*},$$

and

$$\lim_{n \to \infty} \|G^*(y_n)(x) - G^*(y)(x)\|_K = 0.$$

(C2)(Coercivity Condition) There exist $\lambda_1 \geq 0$ and $N_1 \in \mathfrak{U}^2$ such that for all $x \in \mathbb{X}^*$

$$\mathbb{X}\langle \mathcal{A}(x), x \rangle_{\mathbb{X}^*} \leq -\mathcal{N}_1(x) + \lambda_1(1+|x|^2).$$

(C3)(Growth Condition) There exist $\lambda_2, \lambda_3, \lambda_4 > 0$ and $\gamma' \ge \gamma > 1$ such that for all $x \in \mathbb{Y}$

$$\|A(x)\|_{\mathbb{X}}^{\gamma} \le \lambda_2 \mathcal{N}_1(x) + \lambda_3 (1 + |x|^{\gamma'}),$$

$$\|G(x)\|_{L_2(K;H)}^2 \le \lambda_4 (1 + |x|^2),$$

where \mathcal{N}_1 is as in (C2).

(C1) holds since Lemma 4.2.3 implies demi-continuity of \mathcal{A} and G.

(C2) follows, because noting that for $\theta \in \mathbb{X}^*$

$$\langle u\cdot\nabla\theta,\theta\rangle=0,$$

we have

$$\langle \mathcal{A}(\theta), \theta \rangle = -\mathcal{N}_1(\theta).$$

Also (C3) is clear since by Lemma 4.2.3

$$\|\mathcal{A}(\theta)\|_{\mathbb{X}} \le C|\theta|^2$$

and

$$||G(\theta)||_{L_2(K;H)} \le C(|\theta|+1).$$

The set of all such martingale solutions with initial value x_0 is denoted by $\mathcal{C}(x_0)$. Using [GRZ09, Theorem 4.7], we now obtain the following:

Theorem 4.2.5 Let $\alpha \in (0, 1)$. Assume G satisfies (G.1). Then there exists an almost sure Markov family $(P_{x_0})_{x_0 \in H}$ for Eq. (4.1.3) and $P_{x_0} \in \mathcal{C}(x_0)$ for each $x_0 \in H$.

4.3 Ergodicity for $\alpha > \frac{2}{3}$

In this section, we assume that $\alpha > \frac{2}{3}$, K = H, and that G satisfies:

Assumption 4.3.1 There are an isomorphism Q_0 of H and a number $s \ge 1$ such

that $G = A^{-\frac{s+\alpha}{2\alpha}} Q_0^{1/2}$, and furthermore, G satisfies

$$\int (\sum_j |Ge_j|^2)^{p/2} d\xi \le C,$$

for some fixed $p \in ((\alpha - \frac{1}{2})^{-1}, \infty)$, (which is e.g. always the case if $Q_0 = I$).

For $x := \theta_0 \in L^p$, let P_x denote the law of the corresponding solution θ to (4.1.3). Then by [RZZ12, Theorems 5.4 and 5.5] the measures $P_x, x \in L^p$, form a Markov process. Let $(P_t)_{t\geq 0}$ be the associated transition semi-group on $\mathcal{B}_b(H)$, defined as

(4.3.1)
$$P_t(\varphi)(x) := E_x[\varphi(\xi_t)], \qquad x \in L^p, \varphi \in \mathcal{B}_b(H),$$

where E_x denotes expectation under P_x .

4.3.1 The strong Feller property for $\alpha > \frac{2}{3}$

In this subsection we prove that its transition semigroup has the strong Feller property under appropriate conditions.

Remark 4.3.2 (i) Since in our case $\alpha < 1$, the linear part $(-\Delta)^{\alpha}$ in (1.7) is less regularizing. As $G = A^{-\frac{s+\alpha}{2\alpha}} Q_0^{1/2}$, we get the trajectories z of the associated O-U process to be in $C([0, \infty), H^{s+2\alpha-1-\varepsilon})$ for every $\varepsilon > 0$ (c.f. [DZ92, Theorem 5.16], [DO06, Proposition 3.1]). However, in order to prove the weak-strong uniqueness principle (see (4.3.2) and Theorem 4.3.4 below) and the strong Feller property of the semigroup associated with the solution of the cutoff equation (see Proposition 4.3.5 below), we need $z \in C([0, \infty), H^{s+1-\alpha+\sigma_1})$ for some $\sigma_1 > 0$. Therefore, we need $s + 2\alpha - 1 > s + 1 - \alpha$, i.e. $\alpha > \frac{2}{3}$. The situation of the 3D-Navier-Stokes equation is different. While in our case the needed regularity of z is higher than the regularity of our solution space $C((0, \infty), H^s)$ for the cutoff equation (4.3.2), for the 3-D Navier-Stokes equation the needed regularity of z is the same as for the solution of the cutoff equation.

(ii) Since $\alpha < 1$, we can't use the same type of estimate as in [FR08] (c.f. [FR08, Lemma D.2]) to obtain our results. We use Lemma 4.1.1 and choose suitable parameters (s, σ_1, σ_2) such that the approach in [FR08] can be modified to apply here (see (4.3.6)-(4.3.10), (4.3.13) and so on).

(iii) It seems difficult to use the Kolmogorov equation method as in [DD03], [DO06] or a coupling approach as in [O07] in our situation. In fact, to get a uniform H^s -norm estimate for the solutions of the Galerkin approximations of the equation (1.1) for some s > 0, the regularity, needed for the trajectories of the associated Ornstein-Uhlenbeck (O-U) process z is higher than H^s , which is entirely different from the situation of the 3-D Navier-Stokes equation. According to the method in [DD03], DO06] and [O07], we should use the solutions' $H^{s+\alpha}$ -norm to control the $H^{s+\alpha}$ -norm of the derivative of the solutions as required for the Bismut-Elworthy-Li formula. In particular, the associated O-U process z should be also in $H^{s+\alpha}$. However, under Assumption 4.3.1 for the noise, our O-U process z is only in $L^2([0,T], H^{s+2\alpha-1})$. As a result, for their method to apply here, we need even $\alpha \geq 1$.

Fix $s \ge 1$ as in Assumption 4.3.1 and set $\mathcal{W} := H^s$ and $|x|_{\mathcal{W}} := ||x||_{H^s}$.

Now we state the main result of this section.

Theorem 4.3.3 Under Assumption 4.3.1, $(P_t)_{t\geq 0}$ is \mathcal{W} -strong Feller, i.e. for every t > 0 and $\psi \in \mathcal{B}_b(H)$, $P_t \psi \in C_b(\mathcal{W})$.

We shall use [FR08, Theorem 5.4], which is an abstract result to prove the strong Feller property. In order to use [FR08, Theorem 5.4], we follow the idea of [FR08, Theorem 5.11] to construct $P_x^{(R)}$. We introduce an equation which differs from the original one by a cut-off only, so that with large probability they have the same trajectories on a small random time interval (see (4.3.3) below). We consider the equation

(4.3.2)
$$d\theta(t) + A\theta(t)dt + \chi_R(|\theta|_{\mathcal{W}}^2)u(t) \cdot \nabla\theta(t)dt = GdW(t),$$

where $\chi_R : \mathbb{R} \to [0, 1]$ is of class C^{∞} such that $\chi_R(|\theta|) = 1$ if $|\theta| \leq R$, $\chi_R(|\theta|) = 0$ if $|\theta| > R + 1$ and with its first derivative bounded by 1. Then, if we can prove the following Theorem 4.3.4 and Proposition 4.3.5, Theorem 4.3.3 follows.

Theorem 4.3.4 (Weak-strong uniqueness) Suppose Assumption 4.3.1 holds. Then for every $x \in \mathcal{W}$, Eq. (4.3.2) has a unique martingale solution $P_x^{(R)}$, with

$$P_x^{(R)}[C([0,\infty);\mathcal{W})] = 1.$$

Let $\tau_R: \Omega \to [0,\infty]$ be defined as

$$\tau_R(\omega) = \inf\{t \ge 0 : |\omega(t)|_{\mathcal{W}}^2 \ge R\},\$$

and $\tau_R(\omega) = \infty$ if this set is empty. If $x \in \mathcal{W}$ and $|x|_{\mathcal{W}}^2 < R$, then

(4.3.3)
$$\lim_{\varepsilon \to 0} P_{x+h}^{(R)}[\tau_R \ge \varepsilon] = 1, \text{ uniformly in } h \in \mathcal{W}, |h|_{\mathcal{W}} < 1.$$

Moreover,

(4.3.4)
$$E^{P_x^{(R)}}[\varphi(\xi_t)1_{[\tau_R \ge t]}] = E^{P_x}[\varphi(\xi_t)1_{[\tau_R \ge t]}]$$

for every $t \ge 0$ and $\varphi \in \mathcal{B}_b(H)$, where P_x is the martingale solution of (4.1.3).

Proof Let z denote the solution to

$$dz(t) + Az(t)dt = GdW(t),$$

with initial data z(0) = 0 and let $v_x^{(R)}$ be the solution to the auxiliary problem

(4.3.5)
$$\frac{dv^{(R)}(t)}{dt} + Av^{(R)}(t) + u^{(R)}(t) \cdot \nabla(v^{(R)}(t) + z(t))\chi_R(|v^{(R)} + z|_{\mathcal{W}}^2) = 0,$$

with $v^{(R)}(0) = x$. Here $u^{(R)}(t) = u_{v^{(R)}}(t) + u_z(t)$, $u_{v^{(R)}}$ and u_z satisfy (1.9) with θ replaced by $v^{(R)}$ and z, respectively. Moreover, define $\theta^{(R)} := v^{(R)} + z$, which is a martingale solution to equation (4.3.2). We denote its law on Ω by $P_x^{(R)}$. By Assumption 4.3.1 the trajectories of the noise belong to

$$\Omega^* := \bigcap_{\beta \in (0,\frac{1}{2}), \kappa \in [0, \frac{s+\alpha}{2\alpha} - \frac{1}{2\alpha})} C^{\beta}([0,\infty); D(A^{\kappa})),$$

with probability one. Hence, the analyticity of the semigroup generated by A implies that for each $\omega \in \Omega^*$, $z(\omega) \in C([0,\infty), D(\Lambda^{s+2\alpha-1-\varepsilon}))$ for every $\varepsilon > 0$.

Now, for $\omega \in \Omega^*$ we prove that Eq. (4.3.5) with $z(\omega)$ replacing z has a unique global weak solution in the space $C([0,\infty); \mathcal{W})$. First, we obtain the following a-priori estimate for suitable $\sigma_1, \sigma_2 > 0$ with $\sigma_2 \leq s, \sigma_2 + \sigma_1 = 1, s + \sigma_1 - \alpha + 1 < s + 2\alpha - 1 < s + \alpha$, where we used that $\alpha > \frac{2}{3}$ since $0 < \sigma_1 < 3\alpha - 2$:

$$\frac{1}{2} \frac{d}{dt} |\Lambda^{s} v^{(R)}|^{2} + \kappa |\Lambda^{s+\alpha} v^{(R)}|^{2} \\
\leq C \chi_{R}(|\theta^{(R)}|_{W}^{2}) |\Lambda^{s-\alpha+1} R(u^{(R)} \theta^{(R)})| \cdot |\Lambda^{s+\alpha} v^{(R)}| \\
\leq C \chi_{R}(|\theta^{(R)}|_{W}^{2}) |\Lambda^{s-\alpha+1+\sigma_{1}} \theta^{(R)}| |\Lambda^{\sigma_{2}} \theta^{(R)}| \cdot |\Lambda^{s+\alpha} v^{(R)}| \\
\leq C \chi_{R}(|\theta^{(R)}|_{W}^{2}) (|\Lambda^{s-\alpha+1+\sigma_{1}} v^{(R)}| + |\Lambda^{s-\alpha+1+\sigma_{1}} z|) \cdot |\Lambda^{s+\alpha} v^{(R)}| \\
\leq C \chi_{R}(|\theta^{(R)}|_{W}^{2}) (C |\Lambda^{s} v^{(R)}|^{1-r} |\Lambda^{s+\alpha} v^{(R)}|^{r} + |\Lambda^{s-\alpha+1+\sigma_{1}} z|) \cdot |\Lambda^{s+\alpha} v^{(R)}| \\
\leq C \chi_{R}(|\theta^{(R)}|_{W}^{2}) (|\Lambda^{s} v^{(R)}|^{2} + |\Lambda^{s-\alpha+1+\sigma_{1}} z|^{2}) + \frac{\kappa}{2} |\Lambda^{s+\alpha} v^{(R)}|^{2} \\
\leq C \chi_{R}(|\theta^{(R)}|_{W}^{2}) (C(R) + |\Lambda^{s-\alpha+1+\sigma_{1}} z|^{2}) + \frac{\kappa}{2} |\Lambda^{s+\alpha} v^{(R)}|^{2},$$

where $r := \frac{1-\alpha+\sigma_1}{\alpha}$. Here in the second inequality we used Lemmas 4.1.1 and 4.1.2,

and in the fourth inequality we used the Gagliardo-Nirenberg inequality and in the fifth inequality we used Young's inequality. Let P_n be the orthogonal projection in H onto the space spanned by $e_1, \ldots e_n$. Consider the ordinary differential equation

$$\frac{dv_n(t)}{dt} + Av_n(t) + P_n(u_n(t) \cdot \nabla(v_n(t) + z(t))\chi_R(|v_n + z|_{\mathcal{W}}^2)) = 0,$$

with initial condition

$$v_n(0) = P_n v_0.$$

Here u_n satisfies (1.9) with θ replaced by $v_n + z$. Denote the solution of the following approximate equation by v_n . We obtain that the sequence v_n is bounded in $L^{\infty}(0,T;H)$ and in $L^2(0,T;H^{\alpha})$. It is obvious that there exists an element $v^{(R)} \in L^{\infty}(0,T;H) \cap L^2(0,T;H^{\alpha})$ and a sub-sequence v'_m such that

$$v'_m \to v^{(R)}$$
 in $L^2(0,T;H^{s+\alpha})$ weakly, and in $L^{\infty}(0,T;H^s)$ weak-star, as $m' \to \infty$.

In order to prove the strong convergence in $L^2(0, T; H^s)$, we need to use [FG95, Theorem 2.1]. So we just need to prove that $||v_n||_{W^{\gamma,2}(0,T,H^{-3})}$ is bounded for some $1/2 < \gamma < 1$, which can be obtained by estimated each term of the approximate equation. Then by compact embedding, we have $v'_m \to v^{(R)}$ in $L^2(0,T;H^s) \cap C([0,T];H^{-\beta})$ strongly for some $\beta > 3$. Note that v_n also satisfies

$$\langle v_n(t),\psi\rangle + \int_0^t \langle A^{1/2}v_n(s),A^{1/2}\psi\rangle ds - \int_0^t \chi_R(|v_n+z|_{\mathcal{W}}^2) \langle u_n(s)\cdot\nabla\psi,v_n(s)+z(s)\rangle ds = \langle P_nv_0,\psi\rangle,$$

for all $t \in [0, T]$ and all $\psi \in C^1(\mathbb{T}^2)$. Then taking the limit in above equation, we obtain that (4.3.5) has a weak solution in $L^{\infty}([0, T], \mathcal{W})$.

$$\begin{split} & [\text{Continuity}] \text{ For each } \omega \in \Omega^*, \, \sigma_1 \text{ and } \sigma_2 \text{ as above, since } s - \alpha + 1 + \sigma_1 < s + 2\alpha - 1, \\ & \text{we have } z \in C([0,\infty); D(\Lambda^{s-\alpha+1+\sigma_1})). \text{ For } s > 3 - 3\alpha, \, s_0 = s - \alpha, \text{ multiplying the} \\ & \text{equations } (4.3.5) \text{ by } \frac{d}{dt}\Lambda^{2s_0}v^{(R)}, \text{ we obtain} \\ & (4.3.7) \\ & \frac{\kappa}{2}\frac{d}{dt}|\Lambda^{s_0+\alpha}v^{(R)}|^2 + |\Lambda^{s_0}\dot{v}^{(R)}|^2 \leq C\chi_R(|\theta^{(R)}|_W^2)|\Lambda^{s_0+1}R(u^{(R)}\theta^{(R)})| \cdot |\Lambda^{s_0}\dot{v}^{(R)}| \\ & \leq C\chi_R(|\theta^{(R)}|_W^2)|\Lambda^{s_0+1+\sigma_1}\theta^{(R)}||\Lambda^{\sigma_2}\theta^{(R)}| \cdot |\Lambda^{s_0}\dot{v}^{(R)}| \\ & \leq C\chi_R(|\theta^{(R)}|_W^2)(|\Lambda^{s+\alpha}v^{(R)}|^2 + |\Lambda^{s_0+\alpha}v^{(R)}|^2 + |\Lambda^{s_0+1+\sigma_1}z|^2) \\ & \quad + \frac{1}{2}|\Lambda^{s_0}\dot{v}^{(R)}|^2. \end{split}$$

Here in the second inequality we used Lemmas 4.1.1 and 4.1.2, and in the third inequality we used the Gagliardo-Nirenberg inequality and Young's inequality.

As $\int_0^T |\Lambda^{s+\alpha} v^{(R)}(t_1)|^2 dt_1$ can be dominated by the same arguments as (4.3.6),

we get an a-priori estimate for the time derivative $\frac{d}{dt}v^{(R)}$ in $L^2(0,T;H^{s_0})$. Then by [Te84], we obtain $v^{(R)} \in C([0,T], \mathcal{W})$.

[Uniqueness] Let θ_1, θ_2 be two solutions of Eq. (4.3.5) in $C([0, \infty); \mathcal{W})$ and set $w := \theta_1 - \theta_2$ and $u_w := u_1 - u_2$. Then by a similar argument as in the proof of [RZZ12, Theorem 5.1], we have for small $\varepsilon_0 > 0$

$$\begin{aligned} \frac{1}{2} \frac{d}{dt} |\Lambda^{s_0} w|^2 + \kappa |\Lambda^{s_0 + \alpha} w|^2 &= -\left(\chi_R(|\theta_1|_{\mathcal{W}}^2) - \chi_R(|\theta_2|_{\mathcal{W}}^2)\right) \langle \Lambda^{s_0 + \varepsilon_0 - \alpha}(u_1 \cdot \nabla \theta_1), \Lambda^{s_0 + \alpha - \varepsilon_0} w \rangle \\ &- \chi_R(|\theta_2|_{\mathcal{W}}^2) \langle \Lambda^{s_0 - \alpha}(u_1 \cdot \nabla w + u_w \cdot \nabla \theta_2), \Lambda^{s_0 + \alpha} w \rangle \\ &= I + II + III. \end{aligned}$$

As

$$|\chi_R(|\theta_1|_{\mathcal{W}}^2) - \chi_R(|\theta_2|_{\mathcal{W}}^2)| \le C(R)|w|_{\mathcal{W}}[1_{[0,R+1]}(|\theta_1|_{\mathcal{W}}^2) + 1_{[0,R+1]}(|\theta_2|_{\mathcal{W}}^2)]$$

we have for σ_1, σ_2 as above,

$$\begin{aligned} (4.3.8) \\ I \leq & C[1_{[0,R+1]}(|\theta_1|_{\mathcal{W}}^2) + 1_{[0,R+1]}(|\theta_2|_{\mathcal{W}}^2)]|w|_{\mathcal{W}} \cdot |\Lambda^{s_0 - \alpha + \varepsilon_0 + 1 + \sigma_1}\theta_1||\Lambda^{\sigma_2}\theta_1| \cdot |\Lambda^{s_0 + \alpha - \varepsilon_0}w| \\ \leq & C(R, |\theta_1|_{\mathcal{W}}, |\theta_2|_{\mathcal{W}})|w|_{\mathcal{W}}|\Lambda^{s_0 + \alpha - \varepsilon_0}w| \\ \leq & C(R, |\theta_1|_{\mathcal{W}}, |\theta_2|_{\mathcal{W}})|\Lambda^{s_0}w|^2 + \frac{\kappa}{4}|w|_{\mathcal{W}}^2, \end{aligned}$$

where $s_0 + \alpha = s$. Here in the first inequality we used Lemmas 4.1.1 and 4.1.2, and in the third inequality we used the Gagliardo-Nirenberg inequality and Young's inequality. In a similar way, we obtain

$$II \leq C(R, |\theta_1|_{\mathcal{W}})|\Lambda^{s_0}w|^2 + \frac{\kappa}{4}|w|_{\mathcal{W}}^2,$$

and

$$III \le C(R, |\theta_2|_{\mathcal{W}})|\Lambda^{s_0}w|^2 + \frac{\kappa}{4}|w|_{\mathcal{W}}^2.$$

Then we obtain

$$\frac{1}{2}\frac{d}{dt}|\Lambda^{s_0}w|^2 + \kappa|\Lambda^{s_0+\alpha}w|^2 \le C(R, \sup_{t\in[0,T]}|\theta_1(t)|_{\mathcal{W}}, \sup_{t\in[0,T]}|\theta_2(t)|_{\mathcal{W}})|\Lambda^{s_0}w|^2 + \frac{3\kappa}{4}|w|_{\mathcal{W}}^2.$$

By Gronwall's lemma we have $|\Lambda^{s_0}w| = 0$, which implies w = 0.

So Eq. (4.3.5) has a unique global weak solution in the space $C([0, \infty); \mathcal{W})$.

Next, we prove (4.3.3). In order to do so, it is sufficient to show that $P_x^{(R)}[\tau_R < \varepsilon] \le C(\varepsilon, R)$ with $C(\varepsilon, R) \downarrow 0$ as $\varepsilon \downarrow 0$, for all $x \in \mathcal{W}$, with $|x|_{\mathcal{W}}^2 \le \frac{R}{8}$. So, fix $\varepsilon > 0$ small enough, let $\Theta_{\varepsilon,R} := \sup_{t \in [0,\varepsilon]} |\Lambda^{s-\alpha+1+\sigma_1}z(t)|$ and assume that $\Theta_{\varepsilon,R}^2 \le \frac{R}{8}$. Setting $\varphi(t) := |v^{(R)}|_{\mathcal{W}}^2 + \Theta_{\varepsilon,R}^2$, by (4.3.6) we get $\dot{\varphi} \le C(R)$. This implies, together with the bounds on x and $\Theta_{\varepsilon,R}$, that

$$|\theta^{(R)}(t)|_{\mathcal{W}}^2 \le 2(|v^{(R)}(t)|_{\mathcal{W}}^2 + |z(t)|_{\mathcal{W}}^2) \le R,$$

for ε small enough. In particular, since this holds for all $t \leq \varepsilon$, it follows that $\tau_R \geq \varepsilon$. Hence

$$P_x^{(R)}[\tau_R < \varepsilon] \le P_x^{(R)}[\sup_{t \in [0,\varepsilon]} |\Lambda^{s+1+\sigma_1-\alpha} z(t)|^2 > \frac{R}{8}].$$

Letting $\varepsilon \downarrow 0$, we have $P_x^{(R)}[\tau_R < \varepsilon] \rightarrow 0$, and the claim is proved, since the probability above is independent of x.

Finally, the same arguments as in the proof of [RZZ12 Theorem 5.1] imply that

$$\theta_x(t \wedge \tau_R(\theta_x^{(R)})) = \theta_x^{(R)}(t \wedge \tau_R(\theta_x^{(R)})) \quad \forall t, P_x - a.s..$$

Moreover, since θ is *H*-valued weakly continuous, we obtain $\tau_R(\theta_x^{(R)}) = \tau_R(\theta)$. \Box

In order to apply [FR08, Theorem 5.4], we now only need the following result.

Proposition 4.3.5 For every R > 0, the transition semi-group $(P_t^{(R)})_{t \ge 0}$ associated to Eq. (4.3.2) is \mathcal{W} -strong Feller.

Proof We shall provide formal estimates, that can, however, be made rigorous through Galerkin approximations. Let $(\Sigma, \mathcal{F}, (\mathcal{F}_t)_{t\geq 0}, \mathbb{P})$ be a filtered probability space, $(W_t)_{t\geq 0}$ a cylindrical Wiener process on H and, for every $x \in \mathcal{W}$, let $\theta_x^{(R)}$ be the solution to Eq. (4.3.2). By the Bismut, Elworthy and Li formula,

$$D_y(P_t^{(R)}\psi)(x) = \frac{1}{t} E^{\mathbb{P}}[\psi(\theta_x^{(R)}(t)) \int_0^t \langle G^{-1}D_y \theta_x^{(R)}(s), dW(s) \rangle],$$

where $D_y(P_t^{(R)}\psi)$ denotes $\langle D(P_t^{(R)}\psi), y \rangle$ for $y \in H$, and thus, for $\|\psi\|_{\infty} \leq 1$, by the B-D-G inequality

$$|(P_t^{(R)}\psi)(x_0+h) - (P_t^{(R)}\psi)(x_0)| \le \frac{C}{t} \sup_{\eta \in [0,1]} E^{\mathbb{P}}[(\int_0^t |G^{-1}D_h \theta_{x_0+\eta h}^{(R)}(s)|^2 ds)^{1/2}].$$

The proposition is proved once we prove that the right-hand side of the above inequality converges to 0 as $|h|_{\mathcal{W}} \to 0$.

Fix $x \in \mathcal{W}$, $y \in H$ and write $\theta = \theta_x^{(R)}$, $D\theta = D_y\theta$, $Du = D_yu$. The term $D\theta$ solves the following equation

$$\frac{d}{dt}D\theta + \kappa\Lambda^{2\alpha}(D\theta) = -[\chi_R(|\theta|_{\mathcal{W}}^2)[Du \cdot \nabla\theta + u \cdot \nabla D\theta] + 2\chi_R'(|\theta|_{\mathcal{W}}^2)\langle\theta, D\theta\rangle_{\mathcal{W}}u \cdot \nabla\theta].$$

Multiplying the above equation with $\Lambda^{2s}D\theta$ and taking the inner product in L^2 , we have

$$\frac{1}{2}\frac{d}{dt}|\Lambda^{s}D\theta|^{2} + \kappa|\Lambda^{s+\alpha}(D\theta)|^{2}$$

= $-\langle [\chi_{R}(|\theta|_{\mathcal{W}}^{2})[Du \cdot \nabla\theta + u \cdot \nabla D\theta] + 2\chi_{R}'(|\theta|_{\mathcal{W}}^{2})\langle\theta, D\theta\rangle_{\mathcal{W}}u \cdot \nabla\theta], \Lambda^{2s}D\theta\rangle$

For the first term on the left hand side, we have for $|\theta|_{\mathcal{W}}^2 \leq R$ (4.3.9)

$$\begin{split} &|\langle Du \cdot \nabla \theta, \Lambda^{2s} D\theta \rangle| \\ &= |\langle \Lambda^{s-\alpha} (Du \cdot \nabla \theta), \Lambda^{s+\alpha} D\theta \rangle| \\ &\leq C|\Lambda^{s-\alpha+1+\sigma_1} \theta| \cdot |\Lambda^{\sigma_2} D\theta| \cdot |\Lambda^{s+\alpha} D\theta| + C|\Lambda^{s-\alpha+1+\sigma_1} D\theta| \cdot |\Lambda^{\sigma_2} \theta| \cdot |\Lambda^{s+\alpha} D\theta| \\ &\leq \varepsilon |\Lambda^{s+\alpha} D\theta|^2 + C(C(R) + |\Lambda^{s+\alpha} v|^2 + |\Lambda^{s-\alpha+1+\sigma_1} z|^2) |\Lambda^s D\theta|^2, \end{split}$$

for σ_1, σ_2 as above, where we used Lemmas 4.1.1, 4.1.2 in the first inequality as well as the Gagliardo-Nirenberg inequality and Young's inequality in the second inequality.

The second term can be estimated similarly. For the third term, by Lemmas 4.1.1, 4.1.2 we have

$$(4.3.10) \qquad \begin{aligned} |\langle u \cdot \nabla \theta, \Lambda^{2s} D \theta \rangle| &= |\langle \Lambda^{s-\alpha} (u \cdot \nabla \theta), \Lambda^{s+\alpha} D \theta \rangle| \\ &\leq C |\Lambda^{s-\alpha+1+\sigma_1} \theta| |\Lambda^{\sigma_2} \theta| \cdot |\Lambda^{s+\alpha} D \theta| \\ &\leq C (|\Lambda^{s+\alpha} v| + |\Lambda^{s-\alpha+1+\sigma_1} z|) |\Lambda^s \theta| |\Lambda^{s+\alpha} D \theta|. \end{aligned}$$

Then we obtain

$$\frac{1}{2}\frac{d}{dt}|\Lambda^s D\theta|^2 + \kappa|\Lambda^{s+\alpha}(D\theta)|^2 \leq \frac{\kappa}{2}|\Lambda^{s+\alpha}(D\theta)|^2 + C(C(R) + |\Lambda^{s+\alpha}v|^2 + |\Lambda^{s-\alpha+1+\sigma_1}z|^2)|\Lambda^s D\theta|^2$$

From Gronwall's inequality we finally obtain

$$\int_0^t |\Lambda^{s+\alpha}(D\theta(l))|^2 dl \le \exp(C \int_0^t (C(R) + |\Lambda^{s+\alpha}v|^2 + |\Lambda^{s-\alpha+1+\sigma_1}z|^2 dl))|\Lambda^s h|^2.$$

By (4.3.6) we obtain

$$E\int_{0}^{t} |\Lambda^{s+\alpha}(D\theta(l))|^{2} dl \leq \sum_{n=1}^{\infty} \exp(Ct(C(R)+cn^{2}))P(\sup_{(0,t)} |\Lambda^{s-\alpha+1+\sigma_{1}}z| > n)|\Lambda^{s}h|^{2}.$$

Because of Assumption 4.3.1 and since z is a Gaussian process, one deduces that

there exist $\eta, C > 0$ such that

$$P[\sup_{l \in [0,t]} |\Lambda^{s-\alpha+1+\sigma_1} z(l)|^2 > R_0] \le C e^{-\eta \frac{R_0^2}{t}},$$

(see e.g. [FR07, Proposition 15]). Then for $t_0^2 \leq \frac{\eta}{cC}$, we obtain

$$E\int_0^{t_0} |\Lambda^{s+\alpha}(D\theta(s))|^2 ds \le c(t_0, R)|\Lambda^s h|^2,$$

which, as $G = Q_0^{-1/2} \Lambda^{s+\alpha}$, implies the assertion for t_0 . For general t, by the semigroup property the assertion follows easily.

4.3.2 A support theorem for $\alpha > 2/3$

A Borel probability measure μ on H is fully supported on \mathcal{W} if $\mu(U) > 0$ for every non-empty open set $U \subset \mathcal{W}$. Set $\mathcal{W}_1 := D(\Lambda^{s-\alpha+1+\sigma_1})$, where σ_1 is the same as in the proof of Theorem 4.3.4 and we will use it below.

Lemma 4.3.6 (Approximate controllability) Let R > 0, T > 0. Let $x \in W$ and $y \in W$, with $Ay \in W_1$, such that

$$|x|_{\mathcal{W}}^2 \le \frac{R}{2} \qquad |y|_{\mathcal{W}}^2 \le \frac{R}{2}.$$

Then there exist (a control function) $\omega \in \operatorname{Lip}([0,T]; \mathcal{W}_1)$ and

$$\theta \in C([0,T]; \mathcal{W}) \cap L^2([0,T]; D(\Lambda^{s+\alpha})),$$

such that θ solves the equation

(4.3.11)
$$\theta(t) - x + \int_0^t A\theta(r) + \chi_R(|\theta|_W^2)u(r) \cdot \nabla\theta(r)dr = \omega(t) \quad dt - a.e.t \in [0, T],$$

with $\theta(0) = x$ and $\theta(T) = y$, and

(4.3.12)
$$\sup_{t\in[0,T]}|\theta(t)|_{\mathcal{W}}^2 \le R.$$

Proof First consider $\omega = 0$. Then by an inequality similar to (4.3.6), we get

$$\frac{d}{dt}|\theta|_{\mathcal{W}}^2 + \kappa |\Lambda^{\alpha}\theta|_{\mathcal{W}}^2 \le C(R)|\theta|_{\mathcal{W}}^2.$$

Hence by Gronwall's lemma $\theta(t) \in D(\Lambda^{s+\alpha})$ for almost every $t \in [0,T]$ and, by

solving again the equation with one of these regular points as initial condition, by Lemma 4.1.1 we have

$$\frac{d}{dt}|\Lambda^{\alpha+s}\theta|^2 + \kappa|\Lambda^{2\alpha+s}\theta|_{\mathcal{W}}^2 \le C|\Lambda^{2\alpha+s}\theta||\Lambda^{s+1+\sigma}\theta|||\theta||_{L^p} \le C(R)|\Lambda^{s+\alpha}\theta|^2 + \frac{\kappa}{2}|\Lambda^{s+2\alpha}\theta|^2,$$

where $\sigma = \frac{2}{p} < 2\alpha - 1$ and where we used the L_p -estimate in the same way as in the proof of [Re95, Theorem 3.3]. Then we find a small $T_* \in (0, \frac{T}{2})$ such that $|\theta(t)|^2_{\mathcal{W}} \leq R$ and $A\theta(T_*) \in \mathcal{W}_1$ for all $t \leq T_*$. Define θ to be the solution above for $t \in [0, T_*]$ and extended by linear interpolation between y and $\theta(T_*)$ in $[T_*, T]$. Then obviously (4.3.12) follows.

Next, if we set

$$\eta := \partial_t \theta + A\theta + \chi_R(|\theta|^2_{\mathcal{W}})u \cdot \nabla \theta, \quad T_* \le t \le T,$$

 $\omega := 0$ for $t \leq T_*$ and $\omega(t) = \int_{T_*}^t \eta_s ds$ for $t \in [T_*, T]$, we also have (4.3.11). It remains to prove that $\eta \in L^{\infty}(0,T; \mathcal{W}_1)$. For the first two terms of η this is obvious. For the non-linear term we have that

$$|u \cdot \nabla \theta|_{\mathcal{W}_1} \le C |\Lambda^{2\alpha} \theta|_{\mathcal{W}_1}^2,$$

).

for any $\theta \in D(\Lambda^{s+\sigma_1+1+\alpha})$

Let $l \in (0, \frac{1}{2})$ and p > 1 such that $l - \frac{1}{p} > 0$. Under this assumption we see that for every $\alpha_1 < \frac{s+\alpha-1}{2\alpha}$ the map

$$\omega \mapsto z(\cdot, \omega) : W^{l,p}([0,T]; D(A^{\alpha_1})) \to C([0,T]; D(A^{\alpha_1+l-\frac{1}{p}-\varepsilon}))$$

is continuous, for all $\varepsilon > 0$, where z is the solution to the Stokes problem

$$z(t) + \int_0^t Az(s)ds = \omega(t).$$

In particular, it is possible to find $\alpha_1 \in (0, \frac{s+\alpha-1}{2\alpha})$, s and p such that the above map is continuous from $W^{l,p}([0,T]; D(A^{\alpha_1}))$ to $C([0,T]; D(\Lambda^{s-\alpha+1+\sigma_1}))$.

Lemma 4.3.7 (Continuity with respect to the control functions) Let l, p and α_1 be chosen as above, and let $\omega_n \to \omega$ in $W^{l,p}([0,T]; D(A^{\alpha_1}))$. Let θ be the solution to equation (4.3.11) corresponding to ω and some initial condition x, and let

$$\tau = \inf\{t \ge 0 : |\theta(t)|_{\mathcal{W}}^2 \ge R\}$$

where as usual we set $\inf \emptyset = \infty$. For each $n \in \mathbb{N}$, define similarly θ_n and τ_n

corresponding to ω_n with the same initial condition x. If $\tau > T$, then $\tau_n > T$ for n large enough and

$$\theta_n \to \theta$$
 in $C([0,T]; \mathcal{W})$.

Proof Set $v_n := \theta_n - z_n$ for each $n \in \mathbb{N}$, and $v := \theta - z$, where z_n, z are the solutions to the Stokes problem corresponding to ω_n, ω respectively. Since $\omega_n \to \omega$ in $W^{l,p}([0,T]; D(A^{\alpha_1}))$, we can find a common lower bound for $(\tau_n)_{n \in \mathbb{N}}$ and τ . For every time smaller than this lower bound t_0 , by (4.3.6) we have

$$\sup_{(0,t_0)} |\Lambda^s \theta_n|^2 \le R, \qquad \sup_{(0,t_0)} |\Lambda^s \theta|^2 \le R, \qquad \sup_{(0,t_0)} |\Lambda^{s-\alpha+1+\sigma_1} z_n| \le C(R),$$

and

$$\sup_{(0,t_0)} |\Lambda^{s-\alpha+1+\sigma_1} z| \le C(R), \qquad \int_0^{t_0} |\Lambda^{s+\alpha} v_n(l)|^2 dl \le C(R), \qquad \int_0^{t_0} |\Lambda^{s+\alpha} v(l)|^2 dl \le C(R),$$

where C(R) is a constant depending only on R. Moreover, we obtain for $t \leq t_0$

$$\frac{d}{dt}|v-v_{n}|_{\mathcal{W}}^{2}+2\kappa|\Lambda^{\alpha}(v_{n}-v)|_{\mathcal{W}}^{2}$$

$$=\langle u_{n}\cdot\nabla\theta_{n},\Lambda^{2s}(v-v_{n})\rangle-\langle u\cdot\nabla\theta,\Lambda^{2s}(v-v_{n})\rangle$$

$$=[\langle (u_{v_{n}}-u_{v})\cdot\nabla\theta_{n},\Lambda^{2s}(v-v_{n})\rangle+\langle u\cdot\nabla(v_{n}-v),\Lambda^{2s}(v-v_{n})\rangle$$

$$+\langle (u_{z_{n}}-u_{z})\cdot\nabla\theta_{n},\Lambda^{2s}(v-v_{n})\rangle+\langle u\cdot\nabla(z_{n}-z),\Lambda^{2s}(v-v_{n})\rangle].$$

For the first term on the right hand side, by using Lemmas 4.1.1, 4.1.2 we have

$$(4.3.13) \begin{aligned} |\langle (v_n - v) \cdot \nabla \theta_n, \Lambda^{2s}(v - v_n) \rangle| \\ \leq C |\Lambda^{s+\alpha}(v - v_n)| |\Lambda^{s-\alpha+1+\sigma_1}(v - v_n)| |\Lambda^{\sigma_2}\theta_n| \\ + C |\Lambda^{s+\alpha}(v - v_n)| |\Lambda^{s-\alpha+1+\sigma_1}\theta_n| |\Lambda^{\sigma_2}(v - v_n)| \\ \leq \frac{\kappa}{4} |\Lambda^{s+\alpha}(v - v_n)|^2 + (C(R) + |\Lambda^s v_n|^2 + |\Lambda^{s+\alpha}v_n|^2) |\Lambda^s(v - v_n)|^2 \\ + c |\Lambda^{s-\alpha+1+\sigma_1}z_n|^2 |\Lambda^s(v - v_n)|^2. \end{aligned}$$

The other term can be estimated similarly. Then we obtain

$$\frac{d}{dt}|v - v_n|_{\mathcal{W}}^2 + 2\kappa|\Lambda^{\alpha}(v_n - v)|_{\mathcal{W}}^2$$

$$\leq \kappa|\Lambda^{\alpha}(v_n - v)|_{\mathcal{W}}^2 + C(C(R) + |\Lambda^{\alpha}v_n|_{\mathcal{W}}^2 + |\Lambda^{\alpha}v|_{\mathcal{W}}^2)(|v - v_n|_{\mathcal{W}}^2 + |\Lambda^{s - \alpha + 1 + \sigma_1}(z - z_n)|^2).$$

Here σ_1, σ_2 are as above. Then by Gronwall's lemma

$$|v-v_n|_{\mathcal{W}}^2 \leq \Theta_n \exp(C\int_0^t (C(R)+|\Lambda^{\alpha}v_n|_{\mathcal{W}}^2+|\Lambda^{\alpha}v|_{\mathcal{W}}^2)dl)\int_0^t (C(R)+|\Lambda^{\alpha}v_n|_{\mathcal{W}}^2+|\Lambda^{\alpha}v|_{\mathcal{W}}^2)dl,$$

where $\Theta_n = \sup_{[0,T]} |\Lambda^{s-\alpha+1+\sigma_1}(z-z_n)|$. We conclude $\theta_n \to \theta$ in $C([0,T]; \mathcal{W})$. Now, since $\tau > T$, if $S = \sup_{t \in [0,T]} |\Lambda^s \theta(t)|^2$, then S < R and we find $\delta > 0$ (depending only on R and S) and $n_0 \in \mathbb{N}$ such that $\Theta_n < \delta$ and $|v_n - v|_{\mathcal{W}}^2 < \delta$ for all $n \ge n_0$, and so

$$|\theta_n(t)|_{\mathcal{W}} \le |v_n(t) - v(t)|_{\mathcal{W}} + \Theta_n + |\theta(t)|_{\mathcal{W}} \le 2\sqrt{\delta} + \sqrt{S} \le \sqrt{R - \delta}.$$

Then $\tau_n > T$ for all $n \ge n_0$.

Theorem 4.3.8 Suppose Assumption 4.3.1 holds and for $x \in H$ let P_x be the distribution of the solution of (4.1.3) with initial value $\theta_0 = x$. Then for every $x \in W$ and every T > 0, the image measure of P_x at time T is fully supported on W.

Proof Fix $x \in \mathcal{W}$ and T > 0. We need to show that for every $y \in \mathcal{W}$ and $\varepsilon > 0$, $P_x[|\theta_T - y|_{\mathcal{W}} < \varepsilon] > 0$. Let $\bar{y} \in \mathcal{W} \cap D(A)$ such that $A\bar{y} \in \mathcal{W}_1$ and $|y - \bar{y}|_{\mathcal{W}} < \frac{\varepsilon}{2}$. Choose R > 0 such that $3|x|_{\mathcal{W}}^2 < R$ and $3|y|_{\mathcal{W}}^2 < R$. Then by Theorem 4.3.4,

$$P_x[|\theta_T - y|_{\mathcal{W}} < \varepsilon] \ge P_x[|\theta_T - \bar{y}|_{\mathcal{W}} < \frac{\varepsilon}{2}] \ge P_x[|\theta_T - \bar{y}|_{\mathcal{W}} < \frac{\varepsilon}{2}, \tau_R > T]$$
$$= P_x^{(R)}[|\theta_T - \bar{y}|_{\mathcal{W}} < \frac{\varepsilon}{2}, \tau_R > T].$$

By Lemma 4.3.6, there is a control $\bar{\omega} \in W^{l,p}([0,T]; D(A^{\alpha_1}))$, with l, p and α_1 chosen as in Lemma 4.3.7, such that the solution $\bar{\theta}$ to the control problem (4.3.11) corresponding to $\bar{\omega}$ satisfies $\bar{\theta}(0) = x, \bar{\theta}(T) = \bar{y}$ and $|\bar{\theta}(t)|_{\mathcal{W}}^2 \leq \frac{2}{3}R$. By Lemma 4.3.7, there exists $\delta > 0$ such that for all $\omega \in W^{l,p}([0,T]; D(A^{\alpha_1}))$ with $|\omega - \bar{\omega}|_{W^{l,p}([0,T]; D(A^{\alpha_1}))} < \delta$, we have

$$|\theta(T,\omega) - \bar{y}|_{\mathcal{W}} < \frac{\varepsilon}{2} \text{ and } \sup_{t \in [0,T]} |\theta(t,\omega)|_{\mathcal{W}}^2 < R,$$

where $\theta(\cdot, \omega)$ is the solution to the control problem (4.3.11) corresponding to ω and starting at x. Hence

$$P_x^{(R)}[|\theta_T - \bar{y}|_{\mathcal{W}} < \frac{\varepsilon}{2}, \tau_R > T] \ge P_x^{(R)}[|\eta - \bar{\omega}|_{W^{l,p}([0,T];D(A^{\alpha_1}))} < \delta],$$

where $\eta_t = \theta_t - x + \int_0^t (A\theta_s + \chi_R(|\theta_s|_W^2) u \cdot \nabla \theta_s) ds$, hence $\theta_T = \theta(T, \eta)$, and the right hand side of the inequality above is strictly positive since by Assumption 4.3.1 η is a Brownian motion in $D(A^{\alpha_1})$.

4.3.3 Existence of invariant measures for $\alpha > \frac{2}{3}$

In this subsection, we prove the existence of invariant measures. Let θ_n denote the solution of the usual Galerkin approximation

$$\begin{cases} d\theta_n(t) + A\theta_n(t)dt + P_n(u_n(t) \cdot \nabla \theta_n(t))dt = P_n G(\theta_n(t))dW(t), \\ \theta_n(0) = P_n x. \end{cases}$$

Lemma 4.3.9 Let $\alpha > \frac{2}{3}$. If $x \in H^1, n \in \mathbb{N}, t > 0$, then there exist $\delta_1 > 0$ and $\gamma_0 > 0$ such that

$$E[\int_0^t |A^{\delta_1} \theta_n|_{\mathcal{W}}^{2\gamma_0} dr] \le C(1+t)(|x|^2+1),$$

where C is independent of x and R.

Proof We apply Itô's formula to the function $(1 + |\Lambda^{\delta}\theta|^2)^{-p}$ for $\delta > 2 - 2\alpha$ and get

$$\begin{aligned} &\frac{1}{(1+|\Lambda^{\delta}\theta|^2)^p} - \frac{1}{(1+|\Lambda^{\delta}x|^2)^p} \\ =& 2p \int_0^t \frac{|\Lambda^{\delta+\alpha}\theta|^2}{(1+|\Lambda^{\delta}\theta|^2)^{p+1}} dr + 2p \int_0^t \frac{\langle\Lambda^{\delta-\alpha}(u\cdot\nabla\theta),\Lambda^{\delta+\alpha}\theta\rangle}{(1+|\Lambda^{\delta}\theta|^2)^{p+1}} dr \\ &- 2p \int_0^t \frac{\langle\Lambda^{\delta}\theta,\Lambda^{\delta}GdW_r\rangle}{(1+|\Lambda^{\delta}\theta|^2)^{p+1}} - p \int_0^t \frac{\mathrm{Tr}[\mathrm{GG}^*\Lambda^{2\delta}]}{(1+|\Lambda^{\delta}\theta|^2)^{p+1}} dr \\ &+ 2p(p+1) \int_0^t \frac{|\Lambda^{\delta}G\theta|^2}{(1+|\Lambda^{\delta}\theta|^2)^{p+1}} dr, \end{aligned}$$

where for simplicity we write $\theta = \theta_n$. Choosing σ'_1, σ'_2 with $\sigma'_2 \leq \delta, \sigma'_2 + \sigma'_1 = 1, \delta + \sigma'_1 - \alpha + 1 < \delta + \alpha$ the non-linear part is estimated as follows:

$$\begin{aligned} |\langle \Lambda^{\delta-\alpha}(u \cdot \nabla \theta), \Lambda^{\delta+\alpha} \theta \rangle| &\leq C |\Lambda^{\delta-\alpha+1+\sigma_1'} \theta| \cdot |\Lambda^{\sigma_2'} \theta| |\Lambda^{\delta+\alpha} \theta| \\ &\leq C |\Lambda^{\delta} \theta|^m + |\Lambda^{\delta+\alpha} \theta|^2, \end{aligned}$$

with $m = \frac{2(3\alpha - 1 - \sigma_1')}{2\alpha - 1 - \sigma_1'}$.

Then for p big enough we obtain

$$E\int_0^t \frac{|\Lambda^{\delta+\alpha}\theta_n|^2}{(1+|\Lambda^{\delta}\theta_n|^2)^{p+1}}dr \leq C(1+t).$$

Since by Young's inequality

$$|\Lambda^{\delta+\alpha}\theta_n|^{2\gamma_p} \le c[\frac{|\Lambda^{\delta+\alpha}\theta_n|^2}{(1+|\Lambda^{\delta}\theta_n|^2)^{p+1}} + 1 + |\Lambda^{\delta}\theta_n|^2],$$

for $\delta \leq \alpha$ we obtain

(4.3.14)
$$E[\int_0^t |\Lambda^{\delta+\alpha}\theta_n|^{2\gamma_p} dr] \le C(1+t)(|x|^2+1).$$

If $\delta > \alpha$, we already know that some power of $|\Lambda^{\delta}\theta_n|$ is integrable with respect to $dt \otimes P$. Then one proceeds as in the previous case to obtain (4.3.14). We choose $\delta + \alpha > s$ and obtain the assertions.

Theorem 4.3.10 Let $\alpha > \frac{2}{3}$ and suppose Assumption 4.3.1 holds. Then there exists a unique invariant measure ν on \mathcal{W} for the transition semigroup $(P_t)_{t\geq 0}$. Moreover:

(i) The invariant measure ν is ergodic.

(ii) The transition semigroup $(P_t)_{t\geq 0}$ is \mathcal{W} -strong Feller, irreducible, and therefore strongly mixing. Furthermore, $P_t(x, dy), t > 0, x \in \mathcal{W}$, are mutually equivalent.

(iii) There are $\delta_1 > 0$ and $\gamma_0 > 0$ such that

$$\int |A^{\delta_1}x|_{\mathcal{W}}^{2\gamma_0} d\nu < \infty.$$

Proof Choose $x_0 \in H^1$ and define

$$\mu_t = \frac{1}{t} \int_0^t P_r^* \delta_{x_0} dr.$$

Since

$$\int |A^{\delta_1}x|_{\mathcal{W}}^{2\gamma_0}\mu_t(dx) = \frac{1}{t}E_{x_0}[\int_0^t |A^{\delta_1}\theta|_{\mathcal{W}}^{2\gamma_0}dr],$$

by Lemma 4.3.9 we obtain

$$\int |A^{\delta_1} x|_{\mathcal{W}}^{2\gamma_0} \mu_t(dx) \le C.$$

This implies that μ_t is tight on \mathcal{W} . The strong Feller property of P_t follows from Theorem 4.3.3. Hence, a limit point of μ_t is an invariant measure for $(P_t)_{t\geq 0}$. Therefore, by Doob's theorem, the strongly mixing property is a consequence of the irreducibility.

Remark 4.3.11 If we don't assume that G satisfies $\int (\sum_j |G(e_j)|^2)^{p/2} d\xi \leq C$, for some fixed $p \in ((\alpha - \frac{1}{2})^{-1}, \infty)$, the solution of equation (4.1.3) may be not unique. Then we can also prove the above results for each Markov selection $P_x, x \in \mathcal{W}$, corresponding to (4.1.3) and the respective semigroup $(P_t)_{t>0}$ by similar arguments as [R08].

Remark 4.3.12 (i) (Mildly degenerate noise) We can also consider the ergodicity of the equation driven by a mildly degenerate noise as in [EH01]. For this we have to use an extension of the Bismut-Elworthy-Li formula. We have the same problem as explained in Remark 4.3.2. So, we can just get the result for $\alpha > 2/3$.

(ii) (Degenerate noise) There are many papers considering 2D Navier-Stokes equation driven by degenerate noise. Contrary to the 2D Navier-Stokes equation, no Foias-Prodi type estimate is available for the quasi-geostrophic equation. It seems impossible to use a coupling approach as in [KS02], [BKL02], [M02] to prove ergodicity in the case where equation (4.1.3) is driven by a degenerate noise. It also seems difficult to use the method in [HM06] to prove ergodicity.

4.4 Exponential convergence for $\alpha > \frac{2}{3}$

First we introduce the same approximation as in the proof of [RZZ12, Theorem 4.3]: We pick a smooth $\phi \ge 0$, with $\operatorname{supp} \phi \subset [1, 2], \int_0^\infty \phi = 1$, and for $\delta > 0$ let

$$U_{\delta}[heta](t) := \int_0^\infty \phi(au) (k_{\delta} * R^{\perp} heta) (t - \delta au) d au,$$

where k_{δ} is the periodic Poisson Kernel in \mathbb{T}^2 given by $\hat{k}_{\delta}(\zeta) = e^{-\delta|\zeta|}, \zeta \in \mathbb{Z}^2$, and we set $\theta(t) = 0, t < 0$. We take a zero sequence δ_n and consider the equation:

$$d\theta_n(t) + A\theta_n(t)dt + u_n(t) \cdot \nabla \theta_n(t)dt = k_{\delta_n} * G(\theta)dW(t),$$

with initial data $\theta_n(0) = \theta_0$ and $u_n = U_{\delta_n}[\theta_n]$, where $k_{\delta} * G(\theta)$ means for $y \in K$, $k_{\delta} * G(\theta)(y) = k_{\delta} * (G(\theta)(y))$. For a fixed *n*, this is a linear equation in θ_n on each subinterval $[t_k, t_{k+1}]$ with $t_k = k\delta_n$, since u_n is determined by the values of θ_n on the two previous subintervals.

As will be seen below, we shall need uniform L^p -estimates, and a crucial ingredient to prove them is Krylov's L^p -Itô formula. In order to obtain a uniform estimate, the L^p -estimate known from the deterministic case (see e.g. [Re95]) is not strong enough for our purpose. Therefore, we need the following result, which is an improved version of the "positivity lemma" from [Re95, Lemma 3.2].

Lemma 4.4.1 (Improved positivity Lemma) For $\alpha \in (0, 1)$, and $\theta \in L^p$ with $\Lambda^{2\alpha}\theta \in L^p$, for some 2 ,

$$\int |\theta|^{p-2} \theta(\Lambda^{2\alpha} - \frac{2\lambda_1}{p})\theta \ge 0.$$

Proof Denote the semigroup with respect to $-\Lambda^{2\alpha} + \frac{2\lambda_1}{p}$ and $-\Lambda^{2\alpha}$ in L^2 by P_t^0 and P_t^1 , respectively. Then we have $P_t^0 f = e^{2t\lambda_1/p}P_t^1 f$. Since

$$\|P_t^1 f\|_{L^2} \le e^{-\lambda_1 t} \|f\|_{L^2},$$

and

$$\|P_t^1 f\|_{L^{\infty}} \le \|f\|_{L^{\infty}},$$

by the interpolation theorem, we have

$$||P_t^1 f||_{L^p} \le e^{-2\lambda_1 t/p} ||f||_{L^p}.$$

Thus,

$$||P_t^0 f||_{L^p} \le ||f||_{L^p}$$

Then we have

$$\frac{d}{dt} \|P_t^0 \theta\|_{L^p}^p = \int |P_t^0 \theta|^{p-2} (P_t^0 \theta) (P_t^0 (-\Lambda^{2\alpha} + \frac{2\lambda_1}{p})\theta) dx \le 0$$

Letting $t \to 0$, we obtain our result.

Proposition 4.4.2 Let $\alpha > \frac{1}{2}$. Let θ denote the solution of equation (4.1.3). Then for 2

$$E\|\theta(t)\|_{L^{p}}^{p} \leq \|x\|_{L^{p}}^{p}e^{-\lambda_{1}t} + \frac{C}{\lambda_{1}}(1 - e^{-\lambda_{1}t}).$$

Proof Using [Kr10, Lemma 5.1] for θ_n , we obtain

$$\begin{split} \|\theta(t)\|_{L^{p}}^{p} &= \|\theta(s)\|_{L^{p}}^{p} + \int_{s}^{t} [-p \int_{\mathbb{T}^{2}} |\theta(l)|^{p-2} \theta(l) (\Lambda^{2\alpha} \theta(l) + u(l) \cdot \nabla \theta(l)) d\xi dl \\ &+ \frac{1}{2} p(p-1) \int_{\mathbb{T}^{2}} |\theta(l)|^{p-2} (\sum_{j} |k_{\delta_{n}} * G(e_{j})|^{2}) d\xi] dl \\ &+ p \int_{s}^{t} \int_{\mathbb{T}^{2}} |\theta(l)|^{p-2} \theta(l) k_{\delta_{n}} * Gd\xi dW(l) \\ &\leq \|\theta(s)\|_{L^{p}}^{p} - 2\lambda_{1} \int_{s}^{t} \int_{\mathbb{T}^{2}} |\theta(l)|^{p} d\xi dl \\ &+ \int_{s}^{t} \frac{1}{2} p(p-1) \int_{\mathbb{T}^{2}} |\theta(l)|^{p-2} (\sum_{j} |k_{\delta_{n}} * G(e_{j})|^{2}) d\xi dl \\ &+ p \int_{s}^{t} \int_{\mathbb{T}^{2}} |\theta(l)|^{p-2} \theta(l) k_{\delta_{n}} * Gd\xi dW(l) \end{split}$$

4.4. Exponential convergence for $\alpha > \frac{2}{3}$

$$\leq \|\theta(s)\|_{L^p}^p - 2\lambda_1 \int_s^t \int_{\mathbb{T}^2} |\theta(l)|^p d\xi dl + \int_s^t (\varepsilon \int_{\mathbb{T}^2} |\theta(l)|^p d\xi + C(\varepsilon) \int (\sum_j |k_{\delta_n} * G(e_j)|^2)^{p/2} d\xi) dl + p \int_s^t \int_{\mathbb{T}^2} |\theta(l)|^{p-2} \theta(l) k_{\delta_n} * Gd\xi dW(l),$$

where we used Lemma 4.4.1 to get the first inequality and our assumption on G to get the last inequality. Here for simplicity we write $\theta(t) = \theta_n(t, x)$. Taking expectation we obtain

$$E\|\theta_n(t)\|_{L^p}^p \le E\|\theta_n(s)\|_{L^p}^p - E\lambda_1 \int_s^t \int_{\mathbb{T}^2} |\theta_n(t)|^p d\xi dt + C(\varepsilon, p)(t-s).$$

Then by Gronwall's lemma we have

$$E \|\theta_n(t)\|_{L^p}^p \le \|\theta(0)\|_{L^p}^p e^{-\lambda_1 t} + \frac{C}{\lambda_1} (1 - e^{-\lambda_1 t}).$$

Then taking the limit $n \to \infty$ in the above inequality we deduce

$$E\|\theta(t)\|_{L^{p}}^{p} \leq \|x\|_{L^{p}}^{p}e^{-\lambda_{1}t} + \frac{C}{\lambda_{1}}(1 - e^{-\lambda_{1}t}).$$

Lemma 4.4.3 Let $\alpha > \frac{2}{3}$ and suppose Assumption 4.1.3 is satisfied with $s > 3 - 2\alpha$. Let θ denote the solution of (4.1.3) and take p as in Assumption 4.1.3. Then for every $R_0 \ge 1$, there exist values $T_1 = T_1(R_0)$ and $K_1 = K_1(R_0)$ such that if $|\theta_0| \le R_0$, $\sup_{t \in [0,T_1]} \|\theta\|_{L^p}^p \le R_0$, and $\sup_{t \in [0,T_1]} |\Lambda^{s-\alpha+1+\sigma_1+\delta}z(t)|^2 \le R_0$ for some $0 < \delta < 3\alpha - 2 - \sigma_1$, then $|\Lambda^{s+\delta}\theta(T_1)|^2 \le K_1$.

Proof By Itô's formula, we obtain that there exists $K_0 = K_0(R_0) > 0$ and for *P*-a.s. $\omega, \exists t_0(\omega) > 0$ such that

$$|\Lambda^{\alpha}\theta(t_0)|^2 \le K_0.$$

For any r > 0, by Lemmas 4.1.1, 4.1.2 we have the following a-priori estimate for $N = \frac{\alpha}{\alpha - \frac{1}{2} - \frac{1}{p}}$ and $\sigma = \frac{2}{p}$,

$$(4.4.1) \qquad \frac{d}{dt} |\Lambda^r v|^2 + |\Lambda^{r+\alpha} v|^2 \le |\langle u \cdot \nabla \theta, \Lambda^{2r} v \rangle|$$
$$\le C |\Lambda^{r+\alpha} v| \cdot |\Lambda^{r-\alpha+1+\sigma} \theta| \cdot ||\theta||_{L^p}$$
$$\le \frac{1}{4} |\Lambda^{r+\alpha} v|^2 + C ||\theta||_{L^p}^N |\Lambda^r v|^2 + C |\Lambda^{r-\alpha+1+\sigma} z|^2 \cdot ||\theta||_{L^q}^2.$$

We choose the approximation θ_n as at the beginning of this subsection with initial time t = 0 replaced by the initial time $t = t_0(\omega)$ and $\theta_n(t_0) = \theta(t_0)$. Set $z_n =$

 $\int_{t_0}^t e^{-(t-s)A} k_{\delta_n} * GdW(s).$ Then we have the following L^p -norm estimate for $v_n := \theta_n - z_n$,

$$\frac{d}{dt} \|v_n\|_{L^p}^p \le pC \|\nabla z_n\|_{\infty} (\|v_n\|_{L^p}^p + \|z_n\|_{L^p} \|v_n\|_{L^p}^{p-1}).$$

Thus we have

$$\frac{d}{dt} \|v_n\|_{L^p} \le C \|\nabla z_n\|_{\infty} (\|v_n\|_{L^p} + \|z_n\|_{L^p}).$$

Then by Gronwall's lemma and since $s > 3 - 2\alpha$, we obtain the desired uniform L^p -norm estimates for v_n . Moreover, by (4.4.1) and Gronwall's lemma we obtain the uniform H^r -norm estimates for v_n . By a similar argument as in the proof of [RZZ12, Theorem 3.4] we have that v_n converges to some process \tilde{v} in $L^2([t_0, T], H)$ such that $\tilde{v} + z$ is the solution of (4.1.3) in $[t_0, T]$. Then by the uniqueness proof in [RZZ12, Theorem 5.1] we have $\tilde{v} = v$ in $[t_0, \infty)$, which implies that $v \in L^{\infty}([t_0, \infty), H^r) \cap L^2_{loc}([t_0, \infty), H^{r+\alpha})$ *P*-a.s. ω . Therefore, (4.4.1) also holds for v with $t \in [t_0, \infty)$.

Then by (4.4.1) for $r = \alpha$, we obtain that there exist $K_1 = K_1(R_0) > 0$ and $t_1 = t_1(\omega) > t_0(\omega)$ such that $|\Lambda^{2\alpha}v(t_1)| \leq K_1$. Using (4.4.1) for $r = 2\alpha$ we obtain that there exists $T_0 = T_0(R_0)$ such that $|\Lambda^{2\alpha}v(T_0)| \leq K_1$. Then we proceed analogously and obtain that there exists $T_1 = T_1(R_0)$ such that $|\Lambda^{s+\delta}v(T_1)| \leq K_1$ for some $0 < \delta < 3\alpha - 2 - \sigma_1$.

Lemma 4.4.4 Let $\alpha > \frac{2}{3}$ and suppose Assumption 4.1.3 holds with $s > 3 - 2\alpha$. Then for each $R \ge 1$ there are $T_1 > 0$ and a compact subset $K \subset \mathcal{W}$ such that

$$\inf_{\|x\|_{L^p} \le R} P_{T_1}(x, K) > 0,$$

for p as in Assumption 4.1.3.

Proof Define $K := \{x : |\Lambda^{s+\delta}x|^2 \leq K_1(R_0)\}$, where $K_1(R_0), \delta$ comes from the previous lemma. By Lemma 4.4.3, for $R \leq R_0$ we have

$$\inf_{\|x\|_{L^{p}} \leq R} P_{T_{1}}(x, K) \geq \inf_{\|x\|_{L^{p}} \leq R} (1 - P_{x}[\sup_{t \in [0, T_{1}]} |\Lambda^{s - \alpha + 1 + \sigma_{1} + \delta} z(t)|^{2} > R_{0}] - P_{x}[\sup_{t \in [0, T_{1}]} \|\theta\|_{L^{p}}^{p} > R_{0}]),$$

where we used Lemma 4.4.3 in the last step. Under Assumption 4.1.3, since z is a Gaussian process, one deduces that there exist $\eta, C > 0$ such that

$$P_x[\sup_{t\in[0,T_1]} |\Lambda^{s-\alpha+1+\sigma_1+\delta} z(t)|^2 > R_0] \le C e^{-\eta \frac{R_0}{T_1}},$$

(see e.g. [FR06, Proposition 15]). Also by [RZZ12, Theorem 4.3], we obtain

$$\sup_{\|x\|_{L^{p}} \le R} P_{x}[\sup_{t \in [0,T_{1}]} \|\theta\|_{L^{p}}^{p} > R_{0}] \le \sup_{\|x\|_{L^{p}} \le R} \frac{E_{x}[\sup_{t \in [0,T_{1}]} \|\theta\|_{L^{p}}^{p}]}{R_{0}} \le \frac{C(R)}{R_{0}}.$$

Choosing R_0 big enough, we prove the assertion.

The exponential convergence now follows from Lemma 4.4.4 and an abstract result of [GM05, Theorem 3.1]. For p in Assumption 4.1.3, let $V : L^p \to \mathbb{R}$ be a measurable function and define $\|\phi\|_V := \sup_{x \in L^p} \frac{|\phi(x)|}{V(x)}$ and $\|\nu\|_V := \sup_{\|\phi\|_V \leq 1} \langle \nu, \phi \rangle$ for a signed measure ν .

Theorem 4.4.5 Let $\alpha > \frac{2}{3}$. Suppose that Assumption 4.1.3 holds with $s > 3-2\alpha$ and let $V(x) := 1 + ||x||_{L^p}^p$ for p as in Assumption 4.1.3. Then there exist $C_{\exp} > 0$ and a > 0 such that

$$\|P_t^*\delta_{x_0} - \mu\|_{TV} \le \|P_t^*\delta_{x_0} - \mu\|_V \le C_{\exp}(1 + \|x_0\|_{L^p}^p)e^{-at},$$

for all t > 0 and $x_0 \in L^p$, where $\|\cdot\|_{TV}$ is the total variation distance on measures.

Proof By similar arguments as in the proof of Lemma 4.4.3 we obtain $P_t(x, W) = 1$ for $x \in L^p$. By [GM05, Theorem 3.1], we need to verify the following four conditions,

- 1. the measures $(P_t(x, \cdot))_{t>0, x\in L^p}$ are equivalent,
- 2. $x \to P_t(x, \Gamma)$ is continuous in \mathcal{W} for all t > 0 and all Borel sets $\Gamma \subset H$,
- 3. for each $R \ge 1$ there exist $T_1 > 0$ and a compact subset $K \subset \mathcal{W}$ such that

$$\inf_{\|x\|_{L^p} \le R} P_{T_1}(x, K) > 0,$$

4. there exist k, b, c > 0 such that for all $t \ge 0$,

$$E^{P_x}[\|\theta(t)\|_{L^p}^p] \le k \|x\|_{L^p}^p e^{-bt} + c.$$

Condition 1 can be verified by [GM05, Lemma 3.2] and since $P_t(x, W) = 1$ for $x \in L^p$. The other conditions can be verified by Theorem 4.3.2, Lemma 4.4.4 and Proposition 4.4.2.

Remark 4.4.6 (i) For $\alpha > \frac{3}{4}$ we can get a better result following a similar argument as in [R08]. Namely, there exist $C_{exp} > 0$ and a > 0 such that

$$||P_t^*\delta_{x_0} - \mu||_{TV} \le ||P_t^*\delta_{x_0} - \mu||_V \le C_{\exp}(1 + |x_0|^2)e^{-at},$$

for all t > 0 and $x_0 \in H$. Here P_t could be every Markov selection associated to the solution of equation (4.1.3).

(ii) The reason why $\alpha > \frac{3}{4}$ is needed, is as follows: As in Theorem 4.3.3, we can prove P_t is H^s -strong Feller with $s > 3 - 3\alpha$. And for a solution θ of equation (4.1.3) starting from $x \in H$, we can prove that it will enter H^{α} only under our condition on the noise. If the process θ enters H^s , we can prove that it satisfies the above four conditions. Hence, to obtain exponential convergence for every $x \in H$, we need the process starting from $x \in H$ to enter H^s . Hence we need $3 - 3\alpha < s < \alpha$, i.e. $\alpha > \frac{3}{4}$.

4.5 Ergodicity for $\alpha > 3/4$ driven by mildly degenerate noises

In this section, we assume that $\alpha > \frac{3}{4}$, W(t) is a cylinder Wiener process on Hand G is additive. We recall here that on the periodic domain \mathbb{T}^2 , $\{\cos(kx)|k \in \mathbb{Z}^2_+\} \cup \{\sin(kx)|k \in \mathbb{Z}^2_-\}$ form an eigenbasis of $-\Delta$, which we denote by e_k in this section. Here $\mathbb{Z}^2_+ = \{(k_1, k_2) \in \mathbb{Z}^2 | k_2 > 0\} \cup \{(k_1, 0) \in \mathbb{Z}^2 | k_1 > 0\}, \mathbb{Z}^2_- = \{(k_1, k_2) \in \mathbb{Z}^2 | -k \in \mathbb{Z}^2_+\}, x \in \mathbb{T}^2$ and the corresponding eigenvalues are $|k|^2$. Define $\mathbb{Z}^2_* = \mathbb{Z}^2 \setminus \{(0, 0)\}$. Moreover, denote for any $N > 0, Z_L(N) = [-N, N]^2 \setminus (0, 0)$ and $Z_H(N) = \mathbb{Z}^2_* \setminus Z_L(N)$.

4.5.1 The strong Feller property for $\alpha > 3/4$

Given $N \ge 1$, let $\pi_N : H \to H$ be the projection onto the subspace of H generated by all modes k such that $|k|_{\infty} := \max |k_i| \le N$. Assume that $\alpha > \frac{3}{4}$ and G satisfies:

Assumption 4.5.1 The operator $G : H \to H$ is linear bounded and there are $\gamma > 1$ and an integer $N_0 \ge 1$ such that

- [A1] (diagonality) G is diagonal on the basis $(e_k)_{k \in \mathbb{Z}^2_*}$,
- [A2] (finite degeneracy) $\pi_{N_0}G = 0$ and $\ker((Id \pi_{N_0})G) = \{0\},\$
- [A3] (regularity) $(Id \pi_{N_0})A^{\frac{\gamma+\alpha}{2\alpha}}G$ is bounded invertible on $(Id \pi_{N_0})H$.

Under Assumption 4.5.1, (G.1) is satisfied obviously and $QdW = \sum_{k \in Z_H(N_0)} e_k g_k dw_k$, where $(w_k)_{k \in Z_H(N_0)}$ is a sequence of independent 2D Brownian motions. Let $(P_x)_{x \in H}$ be any a.s. Markov process obtained in Theorem 4.2.5, and $(P_t)_{t \ge 0}$ be the associated transition semi-group on $\mathcal{B}_b(H)$, defined as

(4.5.1)
$$P_t(\varphi)(x) = E^x[\varphi(\xi_t)], \qquad x \in H, \varphi \in \mathcal{B}_b(H).$$

In this subsection we prove it has the strong Feller property under Assumption 4.5.1.

Choose s such that $\gamma < \gamma - \alpha + 1 < s < s - \alpha + 1 < \gamma + 2\alpha - 1$, and set $\mathcal{W} = D(\Lambda^s)$ and $|x|_{\mathcal{W}} = |\Lambda^s x|$. By this choice, we know that $\alpha > \frac{3}{4}$ is required.

Now we state the main result of this section.

Theorem 4.5.2 Under Assumption 4.5.1. Then $(P_t)_{t\geq 0}$ is \mathcal{W} -strong Feller, i.e. for every t > 0 and $\psi \in \mathcal{B}_b(H), P_t \psi \in C_b(\mathcal{W})$.

In order to prove Theorem 4.5.2, we follow the approach of [EH01], [RX10] to construct P_x^{ρ} . We introduce an equation which differs from the original one by a cut-off only, so that with large probability they have the same trajectories on a small deterministic time interval. We consider the equation

(4.5.2)
$$d\theta^{\rho}(t) + A\theta^{\rho}(t)dt + \chi(\frac{|\theta^{\rho}|_{\mathcal{W}}}{3\rho})u^{\rho}(t) \cdot \nabla\theta^{\rho}(t)dt = G(\theta^{\rho})dW(t),$$

where

$$G(\theta^{\rho}) = G + (1 - \chi(\frac{|\theta^{\rho}|_{\mathcal{W}}}{\rho}))\bar{G},$$

and \bar{G} is non-degenerate operator on $\pi_{N_0}H$, and $\chi : \mathbb{R} \to [0,1]$ of class C^{∞} such that $\chi(r) = 1$ if $r \leq 1$, $\chi(r) = 0$ if $r \geq 2$ and with derivative bounded by 1. We could choose \bar{G} is diagonal on the basis (e_k) i.e. $\bar{G}e_k = g_k e_k$, for $k \in Z_L(N_0)$. By the same arguments as in Theorem 4.3.4, we obtain the following results, where we used $s - \alpha + 1 < \gamma + 2\alpha - 1$.

Theorem 4.5.3 (Weak-strong uniqueness) Under Assumption 4.5.1. For every $x \in \mathcal{W}$, Eq. (4.5.2) has a unique martingale solution P_x^{ρ} , with

$$P_x^{\rho}[C([0,\infty);\mathcal{W})] = 1.$$

Let $\tau_{\rho}: \Omega \to [0, \infty]$ be defined as

$$\tau_{\rho}(\omega) = \inf\{t \ge 0 : |\omega(t)|_{\mathcal{W}} \ge \rho\},\$$

and $\tau_{\rho}(\omega) = \infty$ if this set is empty. If $x \in \mathcal{W}$ and $|x|_{\mathcal{W}} < \rho$, then

(4.5.3)
$$\lim_{\varepsilon \to 0} P^{\rho}_{x+h}[\tau_{\rho} \ge \varepsilon] = 1, \text{ uniformly in } h \in \mathcal{W}, |h|_{\mathcal{W}} < 1.$$

Moreover, on $[0, \tau_{\rho}]$, the probability measure P_x^{ρ} coincides with any martingale solution P_x of the equation (4.1.3), namely

(4.5.4)
$$E^{P_x^{\nu}}[\varphi(\xi_t)1_{[\tau_{\rho} \ge t]}] = E^{P_x}[\varphi(\xi_t)1_{[\tau_{\rho} \ge t]}],$$

for every $t \geq 0$ and $\varphi \in \mathcal{B}_b(H)$.

4.5.2 Strong-Feller property of cutoff dynamics

In order to apply [FR08, Theorem 5.4], we need to prove the following result.

Theorem 4.5.4 There is $\rho_0 > 0$ such that for $\rho \ge \rho_0$, the transition semi-group $(P_t^{\rho})_{t\ge 0}$ associated to Eq. (4.5.2) is \mathcal{W} -strong Feller.

Fix $N \ge N_0$ (whose value will be suitably chosen later). In this and the following subsection we shall denote with the superscript L the quantities projected onto the modes smaller than N and with the superscript H those projected onto the modes larger than N. We divide the equation (4.5.2) into the low and high frequency parts (dropping the ρ in θ^{ρ} for simplicity),

(4.5.5)
$$\begin{cases} d\theta^L + A\theta^L dt + \chi(\frac{|\theta|_W}{3\rho})B_L(\theta,\theta)dt = G_L(\theta)dW_t^L, \\ d\theta^H + A\theta^H dt + \chi(\frac{|\theta|_W}{3\rho})B_H(\theta,\theta)dt = G_H dW_t^H, \end{cases}$$

where $\theta^L = \pi_N \theta$, $\theta^H = (Id - \pi_N)u$, $W^L = \pi_N W$, $W^H = (Id - \pi_N)W$, $B_L = \pi_N B$, $B_H = (Id - \pi_N)B$ for $B(\theta, \theta) = u \cdot \nabla \theta$, $G_L(\theta) = G(\theta)\pi_N$ and $G_H = G(\theta)(Id - \pi_N)$.

With the above separation for the dynamics, it is natural to define the Frechet derivatives for their low and high frequency parts. We will always use the notations D_L and D_H to denote the derivatives with respect to H^L (resp. H^H) of a differentiable function defined on H. For instance, for any stochastic process X(t, x) on Hwith X(0, x) = x, $D_H X^L(t, x) : H^H \to H^L$ is defined by

$$D_H X^L(t,x)h = D_h X^L(t,x) = \lim_{\varepsilon \to 0} \frac{1}{\varepsilon} [X^L(t,x+\varepsilon h) - X^L(t,x)], h \in H^H.$$

Let $C_b^k(\mathcal{W})$ be the set of functions on \mathcal{W} with bounded 0-th,...,k-th order derivatives. Similarly, for $\psi \in C_b^1(\mathcal{W})$, $D_L\psi(x)$ and $D_H\psi(x)$ can be defined.

To prove Theorem 4.5.4, we need to approximate (4.5.5) by the following more regular dynamics

(4.5.6)
$$\begin{cases} d\theta^{\delta,\rho} + A\theta^{\delta,\rho}dt + e^{-A_H\delta}\chi(\frac{|\theta^{\delta,\rho}|_{\mathcal{W}}}{3\rho})B(\theta^{\delta,\rho},\theta^{\delta,\rho})dt = G(\theta^{\delta,\rho})dW_t,\\ \theta^{\delta,\rho}(0) = x, \end{cases}$$

where $\delta > 0$ and $A_H = (Id - \pi_N)A$ (the existence and uniqueness of weak solution to equation (4.5.6) is standard).

Define two maps $\Phi_t(\cdot)$ and $\Phi_t^{\delta}(\cdot)$ from H to H by

$$\Phi_t(x) := \theta^{\rho}(t) \text{ and } \Phi_t^{\delta}(x) := \theta^{\delta,\rho}(t),$$

where $\theta^{\rho}(t), \theta^{\delta,\rho}(t)$ are the solutions to (4.5.5) and (4.5.6) respectively.

The key points for the proofs of Proposition 4.5.6 are the following two inequalities and Lemma 4.5.5. For given $\beta \geq 1$, there exist constants $C_1 > 0, C_2 > 0$ such that for every $\theta_1, \theta_2 \in D(\Lambda^{\beta+1+\sigma_1}), \sigma_1 + \sigma_2 = 1$,

(4.5.7)
$$|\Lambda^{\beta} B(\theta_1, \theta_2)| \le C_1(|\Lambda^{\beta+1+\sigma_1} \theta_1| |\Lambda^{\sigma_2} \theta_2| + |\Lambda^{\beta+1+\sigma_1} \theta_1| |\Lambda^{\sigma_2} \theta_2|).$$

(4.5.8)
$$|\Lambda^{\beta} e^{-At} B(\theta_1, \theta_2)|_H \le \frac{C_2}{t^{1-\frac{\varepsilon}{2\alpha}}} |\Lambda^{\beta} \theta_1|_H |\Lambda^{\beta} \theta_2|_H,$$

for some $0 < \varepsilon < 2\alpha - 1$. (4.5.7) and (4.5.8) can be obtained by Lemma 4.1.1.

Lemma 4.5.5 (c.f. [DZ92]) Let $G : H \to H$ be a linear bounded operator such that $\Lambda^{\gamma+\alpha}G$ is also bounded. Then for any $\varepsilon_1 < \frac{1}{2}(\gamma + 2\alpha - 1 - s)$ and $\beta_1 < \frac{1}{2}(\gamma + 2\alpha - 1 - s) - \varepsilon_1$, there exists $C(\varepsilon, \beta, p, T) > 0$ such that

$$E[\sup_{0 \le t \le T} |\Lambda_1^{\varepsilon} \int_0^t e^{-A(t-l)} G dW_l|_{\mathcal{W}}^p] \le C(\varepsilon_1, \beta_1, p, T) T^{\beta_1 p}.$$

By (4.5.7), (4.5.8) and Lemma 4.5.5 we obtain the following estimates.

Proposition 4.5.6 For every T > 0 and $p \ge 2$, there exist some $C_i = C_i(p, \rho, \gamma, \alpha) > 0, i = 1, 2$ such that

(4.5.9)
$$E[\sup_{0 \le t \le T} |\Phi_t - \Phi_t^{\delta}|_{\mathcal{W}}^p] \le C_1 e^{C_1 T} |e^{-A\delta} - Id|_{\mathcal{L}(\mathcal{W})}^p,$$

(4.5.10)
$$E[\sup_{0 \le t \le T} |D\Phi_t - D\Phi_t^{\delta}|_{\mathcal{L}(W)}^p] \le C_2 e^{C_2 T} |e^{-A\delta} - Id|_{\mathcal{L}(W)}^p.$$

For any $\psi \in C_b^1(\mathcal{W}), h \in \mathcal{W} \text{ and } t > 0$,

(4.5.11)
$$\lim_{\delta \to 0+} |D_h E[\psi(\Phi_t^{\delta})] - D_h E[\psi(\Phi_t)]| = 0$$

Proof Denote $\Psi_t = \Phi_t - \Phi_t^{\delta}$, we have

$$\begin{split} \Psi_t &= \int_0^t e^{-A(t-r)} [B(\Phi_r, \Phi_r) \chi(\frac{|\Phi_r|_{\mathcal{W}}}{3\rho}) - e^{-A\delta} B(\Phi_r^{\delta}, \Phi_r^{\delta}) \chi(\frac{|\Phi_r^{\delta}|_{\mathcal{W}}}{3\rho})] dr \\ &+ \int_0^t e^{-A(t-r)} [G(\Phi_r) - G(\Phi_r^{\rho})] dW_r \\ &:= \int_0^t I_1 dr + \int_0^t I_2 dW_r. \end{split}$$

By (4.5.7) and (4.5.8) we obtain

$$|I_1|_{\mathcal{W}} \le C_1(t-r)^{-1+\frac{\varepsilon}{2\alpha}} |Id-e^{-A\delta}|_{\mathcal{L}(\mathcal{W})} + C_2(t-r)^{-1+\frac{\varepsilon}{2\alpha}} |\Psi_r|_{\mathcal{W}},$$

for ε in (4.5.8). By Lemma 4.5.5 we obtain

$$E[\sup_{0 \le t \le T} |\int_0^t I_2 dW_r|^p] \le C_3 T^{\beta_1 p} E[\sup_{0 \le t \le T} |\Psi_t|^p_{\mathcal{W}}],$$

with $p \ge 2$ and β_1 in Lemma 4.5.5. By the above two estimates and induction argument (4.5.9) follows. (4.5.10) can be obtained by the same method and (4.5.11) follows by (4.5.9) and (4.5.10).

Moreover, we obtain the following estimates by using (4.5.7), (4.5.8) and Lemma 4.5.5. We choose $0 < \varepsilon_0 < \frac{1}{2}(\gamma + 2\alpha - 1 - s), 0 < \varepsilon_1 < \frac{2\alpha - 1}{2\alpha}$ and define $\widetilde{\mathcal{W}} := D(\Lambda^{s + \varepsilon_0})$ and $|x|_{\widetilde{\mathcal{W}}} = |\Lambda^{s + \varepsilon_0} x|$.

Lemma 4.5.7 For any $T > 0, p \ge 2$ and $\delta \ge 0$, there exist some $C_i = C_i(p, \rho, \gamma, \alpha) > 0, i = 1, ..., 7$ such that

(4.5.12)
$$E[\sup_{0 \le t \le T} |\Phi_t^{\delta}|_{\mathcal{W}}^p] \le C_1 e^{C_1 T} |x|_{\mathcal{W}}^p,$$

(4.5.13)
$$E[\sup_{0 \le t \le T} |\Phi_t^{\delta}|_{\widetilde{\mathcal{W}}}^p] \le C_2 e^{C_2 T} |x|_{\widetilde{\mathcal{W}}}^p,$$

(4.5.14)
$$E[\sup_{0 \le t \le T} |t^{\frac{\varepsilon_0}{2\alpha}} \Phi_t^{\delta}|_{\widetilde{\mathcal{W}}}^p] \le C_3 e^{C_3 T} |x|_{\mathcal{W}}^p,$$

(4.5.15)
$$E[\sup_{0 \le t \le T} |D_h \Phi_t^{\delta}|_{\mathcal{W}}^p] \le C_4 e^{C_4 T} |h|_{\mathcal{W}}^p, \qquad h \in \mathcal{W},$$

(4.5.16)
$$E[\int_0^t |\Lambda^{\alpha+\gamma} D_h \Phi_\tau^\delta|^2 d\tau] \le C_5 e^{C_5 t} |h|_{\mathcal{W}}^2, \qquad h \in \mathcal{W},$$

(4.5.17)
$$E[\sup_{0 \le t \le T} |D_{h^L} \Phi_t^{\delta, H}|_{\mathcal{W}}^p] \le (T^{p\varepsilon_1} \lor T^{\varepsilon_0 p}) C_6 e^{C_6 T} |h^L|_{\mathcal{W}}^p, \qquad h^L \in \mathcal{W}^L,$$

(4.5.18)
$$E[\sup_{0 \le t \le T} |D_{h^H} \Phi_t^{\delta, L}|_{\mathcal{W}}^p] \le (T^{\varepsilon_1 p} \lor T^{\varepsilon_0 p}) C_7 e^{C_7 T} |h^H|_{\mathcal{W}}^p, \qquad h \in \mathcal{W}^H.$$

Proof (4.5.12)-(4.5.14) can be proved by a similar argument as the proof of (4.5.9), so we omit them here. For (4.5.15), we have that for every $h \in \mathcal{W}$, $D_h \Phi_t$ satisfies the following equation

$$\begin{aligned} D_h \Phi_t = & e^{-At} h + \int_0^t e^{-A(t-r)} (B(D_h \Phi_r, \Phi_r) + B(\Phi_r, D_h \Phi_r)) \chi(\frac{|\Phi_r|_{\mathcal{W}}}{3\rho}) \\ &+ e^{-A(t-r)} B(D_h \Phi_r, \Phi_r) \chi'(\frac{|\Phi_r|_{\mathcal{W}}}{3\rho}) \frac{1}{3\rho} \frac{\langle D_h \Phi_r, \Phi_r \rangle_{\mathcal{W}}}{|\Phi_r|_{\mathcal{W}}} dr \\ &- \int_0^t e^{-A(t-r)} \chi'(\frac{|\Phi_r|_{\mathcal{W}}}{\rho}) \frac{1}{\rho} \frac{\langle D_h \Phi_r, \Phi_r \rangle_{\mathcal{W}}}{|\Phi_r|_{\mathcal{W}}} G_L dW_s^L, \end{aligned}$$

By (4.5.7), (4.5.8) and Lemma 4.5.5, (4.5.15) follows. Similarly we obtain (4.5.17) and (4.5.18).

Let us prove (4.5.16). By Itô's formula, we have for $\sigma_1 + \sigma_2 = 1, 0 < \sigma_1 < (s - \gamma + \alpha - 1) \land (2\alpha - 1)$,

$$\begin{split} E|\Lambda^{\gamma}D_{h}\Phi_{l}|^{2} + 2\int_{0}^{t}E|\Lambda^{\gamma+\alpha}D_{h}\Phi_{l}|^{2}dl \\ \leq |\Lambda^{\gamma}h|^{2} + C\int_{0}^{t}E|\Lambda^{s}D_{h}\Phi_{l}|^{2}dl + C\int_{0}^{t}E[|\Lambda^{\gamma+\alpha}D_{h}\Phi_{l}||\Lambda^{\gamma-\alpha}B(\Phi_{l},D_{h}\Phi_{l})|\chi(\frac{|\Phi_{l}|_{\mathcal{W}}}{3\rho})]dl \\ \leq |\Lambda^{\gamma}h|^{2} + C\int_{0}^{t}E|\Lambda^{s}D_{h}\Phi_{l}|^{2}dl + C\int_{0}^{t}E[|\Lambda^{\gamma+\alpha}D_{h}\Phi_{l}||\Lambda^{\gamma-\alpha+1+\sigma_{1}}\Phi_{l}||\Lambda^{\sigma_{2}}D_{h}\Phi_{l}| \\ + |\Lambda^{\gamma+\alpha}D_{h}\Phi_{l}||\Lambda^{\gamma-\alpha+1+\sigma_{1}}D_{h}\Phi_{l}||\Lambda^{\sigma_{2}}\Phi_{l}|)|\chi(\frac{|\Phi_{l}|_{\mathcal{W}}}{3\rho})|]dl. \end{split}$$

Then by $\gamma - \alpha + 1 < s, \gamma - \alpha + 1 + \sigma_1 < \gamma + \alpha$, we obtain

$$E|\Lambda^{\gamma}D_h\Phi_t|^2 + 2\int_0^t E|\Lambda^{\gamma+\alpha}D_h\Phi_l|^2 dl \le |\Lambda^{\gamma}h|^2 + C\int_0^t E|\Lambda^{\gamma}D_h\Phi_l|^2 dl,$$

which implies (4.5.16) by Gronwall's lemma.

Lemma 4.5.8 There exists some constant p > 1 such that for every $x \in \widetilde{\mathcal{W}}$, $h \in \mathcal{W}^L$, $\psi \in C_b^1(H)$

$$|E[(D_L\psi)(\Phi_t^{\delta}(x))D_h\Phi_t^{\delta,L}(x)]| \leq \frac{Ce^{Ct}(1+|x|_{\widetilde{\mathcal{W}}})^p}{t^p} \|\psi\|_{\infty} |h|_{\mathcal{W}}.$$

The proof of Lemma 4.5.8 will be given in the next section. Now we could prove Theorem 4.5.4 by using Lemmas 4.5.7, 4.5.8. The proof is a modification of the proof of [RX10, Theorem 3.1].

Proof of Theorem 4.5.4 Set $S_t \psi(x) = E[\psi(\Phi_t^{\delta})]$ for any $\psi \in C_b^2(\mathcal{W})$, we prove the theorem in the following two steps.

Step 1. Estimate $DS_t\psi(x)$ for all $x \in \widetilde{\mathcal{W}}$. By (4.5.16) we know $y_t^H = G_H^{-1}D_{h^H}\Phi_t^{\delta,H} \in H^H dt \times dP$ -a.s.. Hence we could proceed as in the proof of Proposition 5.2 of [EH01] to get

$$D_{h^{H}}S_{t}\psi(x) = \frac{2}{t}E[\psi(\Phi_{t}^{\delta})\int_{\frac{t}{4}}^{\frac{3t}{4}} \langle y_{s}^{H}, dW_{s}^{H} \rangle_{H}] + \frac{2}{t}E[\int_{\frac{t}{4}}^{\frac{3t}{4}} D_{L}S_{t-s}\psi(\Phi_{s}^{\delta})D_{h^{H}}\Phi_{s}^{\delta,L}ds]$$

Hence, by B-D-G inequality and (4.5.16), we obtain

$$\begin{split} |D_{h^{H}}S_{t}\psi(x)| &\leq \frac{2}{t} \|\psi\|_{\infty} (\int_{\frac{t}{4}}^{\frac{3t}{4}} E|y_{r}^{H}|^{2}dr)^{1/2} + \frac{2}{t} \int_{\frac{t}{4}}^{\frac{3t}{4}} E[|D_{L}S_{t-r}\psi(\Phi_{s}^{\delta})|_{\mathcal{W}'}|D_{h^{H}}\Phi_{r}^{\delta,L}|_{\mathcal{W}}]dr \\ &\leq \frac{2}{t}ce^{ct}\|\psi\|_{\infty}|h^{H}|_{\mathcal{W}} + \frac{2}{t} \int_{\frac{t}{4}}^{\frac{3t}{4}} E[|D_{L}S_{t-r}\psi(\Phi_{s}^{\delta})|_{\mathcal{W}'}|D_{h^{H}}\Phi_{r}^{\delta,L}|_{\mathcal{W}}]dr. \end{split}$$

For the low frequency part, according to Lemma 4.5.8, we obtain

$$\begin{aligned} |D_{h^{L}}S_{t}\psi(x)| &= |E[D_{L}S_{\frac{t}{2}}\psi(\Phi_{t/2}^{\delta})D_{h^{L}}\Phi_{t/2}^{\delta,L}]| + |E[D_{H}S_{\frac{t}{2}}\psi(\Phi_{t/2}^{\delta})D_{h^{L}}\Phi_{t/2}^{\delta,H}]| \\ &\leq C_{2}e^{C_{2}t}(1+|x|_{\widetilde{W}})^{p}t^{-p}\|\psi\|_{\infty}|h^{L}|_{\mathcal{W}} + E[|D_{H}S_{\frac{t}{2}}\psi(\Phi_{t/2}^{\delta})|_{\mathcal{W}'}|D_{h^{L}}\Phi_{t/2}^{\delta,H}|_{\mathcal{W}}], \end{aligned}$$

where p > 1 is the constant in Lemma 4.5.8.

By this we obtain for $0 < t \leq T$ and T sufficiently small, (see e.g. [RX10, Theorem 3.1])

$$|DS_t\psi(x)|_{\mathcal{W}'} \le \frac{C(1+|x|_{\widetilde{\mathcal{W}}}^p)^p}{t^p} \|\psi\|_{\infty},$$

with $C = C(T, \rho, \gamma, \alpha)$.

Step 2. Strong Feller property of P_t^{ρ} . For any $h \in \mathcal{W}$ and $0 < t \leq T$, we have

$$\begin{aligned} |D_h S_{2t} \psi(x)|^2 &= |E[DS_t \psi(\Phi_t^{\delta}) D_h \Phi_t^{\delta}]|^2 \le E[|DS_t \psi(\Phi_t^{\delta})|_{\mathcal{W}'}^2] E[|D_h \Phi_t^{\delta}|_{\mathcal{W}}^2] \\ &\le \frac{C}{t^{2p}} \|\psi\|_{\infty}^2 E[(1+|\Phi_t^{\delta}|_{\widetilde{\mathcal{W}}})^{2p}] |h|_{\mathcal{W}}^2 \le \frac{C}{t^{2p+\varepsilon_0 p/\alpha}} \|\psi\|_{\infty}^2 (1+|x|_{\mathcal{W}})^{2p} |h|_{\mathcal{W}}^2, \end{aligned}$$

where $C = C(T, \rho, \alpha, \gamma)$. Let $\delta \to 0$, we have

$$|D_h P_{2t} \psi(x)| \le \frac{C}{t^{p+\varepsilon_0 p/(2\alpha)}} \|\psi\|_{\infty} (1+|x|_{\mathcal{W}})^p |h|_{\mathcal{W}}, 0 < t \le T.$$

This implies that $(P_t^{\rho})_{t \in (0,T]}$ is strong Feller. The extension of the strong Feler property to arbitrary T > 0 is standard.

4.5.3 Malliavin calculus

Proof of Lemma 4.5.8

In this subsection, we will only study the equation (4.5.6), following the idea of [N85], [EH01]. We will simply write $\Phi_t = \Phi_t^{\delta}$ throughout this subsection since all the estimates obtained are independent of δ .

Given $v \in L^2_{loc}(\mathbb{R}^+, H)$, the Malliavin derivative of Φ_t in direction v, denoted by $\mathcal{D}_v \Phi_t$, is defined by

$$\mathcal{D}_v \Phi_t = \lim_{\varepsilon \to 0} \frac{\Phi_t(W + \varepsilon V, x) - \Phi_t(W, x)}{\varepsilon},$$

where $V(t) = \int_0^t v(s) ds$. The direction v can be random and is adapted to the filtration generated by W. The Malliavin derivatives on the low and high frequency parts, denoted by $\mathcal{D}_v \Phi_t^L$ and $\mathcal{D}_v \Phi_t^H$, can be defined in a similar way. $\mathcal{D}_v \Phi_t^L$ and $\mathcal{D}_v \Phi_t^H$ satisfy the following two SPDEs respectively:

$$(4.5.19) \qquad d\mathcal{D}_{v}\Phi^{L} + [A\mathcal{D}_{v}\Phi^{L} + \tilde{B}_{L}(\mathcal{D}_{v}\Phi^{L}, \Phi)\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}) + \tilde{B}_{L}(\mathcal{D}_{v}\Phi^{H}, \Phi)\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}) + B_{L}(\Phi, \Phi)D_{L}(\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}))\mathcal{D}_{v}\Phi^{L} + B_{L}(\Phi, \Phi)D_{H}(\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}))\mathcal{D}_{v}\Phi^{H}]dt = [D_{L}G_{L}(\Phi)\mathcal{D}_{v}\Phi^{L} + D_{H}G_{L}(\Phi)\mathcal{D}_{v}\Phi^{H}]dW_{t}^{L} + G_{L}(\Phi)v^{L}dt,$$

$$(4.5.20)$$

$$d\mathcal{D}_{v}\Phi^{H} + [A\mathcal{D}_{v}\Phi^{H} + e^{-A_{H}\delta}\tilde{B}_{H}(\mathcal{D}_{v}\Phi^{L}, \Phi)\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}) + e^{-A_{H}\delta}\tilde{B}_{H}(\mathcal{D}_{v}\Phi^{H}, \Phi)\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho})$$

$$+ e^{-A_{H}\delta}B_{H}(\Phi, \Phi)D_{L}(\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}))\mathcal{D}_{v}\Phi^{L} + e^{-A_{H}\delta}B_{H}(\Phi, \Phi)D_{H}(\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}))\mathcal{D}_{v}\Phi^{H}]dt$$

$$= G_{H}v^{H}dt,$$

with $\mathcal{D}_v \Phi_0^L = 0$ and $\mathcal{D}_v \Phi_0^H = 0$, where $\tilde{B}(u, v) = B(u, v) + B(v, u)$. Moreover, we define a flow between s and t by $J_{s,t}(x), s \leq t$, which satisfies the following equation: for all $h \in H^L$ (4.5.21)

$$dJ_{s,t}h + [AJ_{s,t}h + \tilde{B}_L(J_{s,t}h, \Phi)\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}) + B_L(\Phi, \Phi)D_L(\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}))J_{s,t}h]dt$$
$$= D_L G_L(\Phi)J_{s,t}hdW_t^L,$$

with $J_{s,s}(x) = Id \in \mathcal{L}(H^L, H^L)$. The inverse $J_{s,t}^{-1}(x)$ satisfies (4.5.22) $dJ_{s,t}^{-1}h + J_{s,t}^{-1}[Ah + \tilde{B}_L(h, \Phi)\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}) + B_L(\Phi, \Phi)D_L(\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}))h - \text{Tr}((D_LG_L(\Phi_t))^2)h]dt$ $= -J_{s,t}^{-1}D_LG_L(\Phi)hdW_t^L.$

Simply writing $J_t = J_{0,t}$, clearly $J_{s,t} = J_t J_s^{-1}$.

We follow the ideas in Section 6.1 of [EH01] to develop a Malliavin calculus for (4.5.6). One of the key points for this approach is to find an adapted process $v \in L^2_{loc}(\mathbb{R}^+, H)$ so that (4.5.23)

$$G_H v^H(t) = e^{-A_H \delta} \tilde{B}_H(\mathcal{D}_v \Phi^L, \Phi) \chi(\frac{|\Phi|_W}{3\rho}) + e^{-A_H \delta} B_H(\Phi, \Phi) D_L(\chi(\frac{|\Phi|_W}{3\rho})) \mathcal{D}_v \Phi^L,$$

which implies that $\mathcal{D}_v \Phi_t^H = 0$ for all t > 0.

Proposition 4.5.9 There exists $v \in L^2_{loc}(\mathbb{R}^+, H)$ satisfying (4.5.23), and

$$\mathcal{D}_v \Phi_t^L = J_t \int_0^t J_s^{-1} G_L(\Phi_s) v^L(s) ds \text{ and } \mathcal{D}_v \Phi_t^H = 0.$$

Proof We first claim that

$$(4.5.24) e^{-A_H\delta}\tilde{B}_H(\mathcal{D}_v\Phi^L,\Phi)\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}) + e^{-A_H\delta}B_H(\Phi,\Phi)D_L(\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}))\mathcal{D}_v\Phi^L \in (D(\Lambda^{\gamma+\alpha}))^H.$$

Since $\mathcal{D}_v \Phi_t^L$ is finite dimensional, $\mathcal{D}_v \Phi_t^L \in \mathcal{W}$. The two terms on the left hand of (4.5.24) can all be bounded in the same way, for instance

$$\begin{split} |\Lambda^{\gamma+\alpha}e^{-A_H\delta}\tilde{B}_H(\mathcal{D}_v\Phi^L,\Phi)\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho})| = |\Lambda^{\gamma+\alpha}e^{-A_H\delta}\tilde{B}_H(\mathcal{D}_v\Phi^L,\Phi)\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho})| \\ \leq C_1\delta^{\frac{-\alpha-\gamma}{2\alpha}}|\mathcal{D}_v\Phi_t^L|_{\mathcal{W}}|\Phi_t|_{\mathcal{W}}, \end{split}$$

and (4.5.24) follows immediately. Hence by Assumption [A3] for G, there exists at least one $v^H \in L^2_{loc}(\mathbb{R}^+, H^H)$ satisfying (4.5.23). Thus equation (4.5.20) is a homogeneous linear equation and has a unique solution

$$\mathcal{D}_v \Phi_t^H = 0,$$

for all t > 0. Then equation (4.5.19) now reads as

$$d\mathcal{D}_v \Phi^L + [A\mathcal{D}_v \Phi^L + \tilde{B}_L(\mathcal{D}_v \Phi^L, \Phi)\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}) + B_L(\Phi, \Phi)D_L(\chi(\frac{|\Phi|_{\mathcal{W}}}{3\rho}))\mathcal{D}_v \Phi^L]dt$$

= $D_L G_L(\Phi)\mathcal{D}_v \Phi^L dW_t^L + G_L(\Phi)v^L dt,$

with $\mathcal{D}_v \Phi_0^L = 0$. Applying the Itô's formula to the product $J^{-1} \mathcal{D}_v \Phi^L$ we see immediately that

(4.5.25)
$$\mathcal{D}_v \Phi_t^L = J_t \int_0^t J_s^{-1} G_L(\Phi_s) v^L(s) ds.$$

Let $N \ge N_0$ be the integer fixed and M is the dimension of $\pi_N H$. Consider $v_1, ..., v_L \in L^2_{\text{loc}}(\mathbb{R}^+; H)$ with each of them satisfying Proposition 4.5.9. Set

$$v = [v_1, \dots, v_M],$$

we have

$$\mathcal{D}_v \Phi_t^H = 0, \qquad \mathcal{D}_v \Phi_t^L = J_t \int_0^t J_s^{-1} G_L(\Phi_s) v^L(s) ds$$

Choose $v^L(s) = (J_s^{-1}G_L(\Phi_s))^*$ and define the Malliavin matrix

$$\mathcal{M}_{t} = \int_{0}^{t} J_{s}^{-1} G_{L}(\Phi_{s}) (J_{s}^{-1} G_{L}(\Phi_{s}))^{*} ds.$$

The following two lemmas are crucial for the proof of Lemma 4.5.8. The first one can be proved by a similar argument as Lemma 4.5.7 and [RX10, Lemma 4.2], so we omit it here.

Lemma 4.5.10 For any T > 0 and $p \ge 2$, there exist some $C_i = C_i(p, \rho, \gamma, \alpha) > 0$ 0(i = 1, ..., 4) such that

(4.5.26)
$$E(\sup_{0 \le t \le T} |J_t(x)h^L|_{\mathcal{W}}^p) \le C_1 e^{C_1 T} |h^L|_{\mathcal{W}}^p$$

(4.5.27)
$$E(\sup_{0 \le t \le T} |J_t^{-1}(x)h^L|_{\mathcal{W}}^p) \le C_2 e^{C_2 T} |h^L|_{\mathcal{W}}^p,$$

(4.5.28)
$$E(\sup_{0 \le t \le T} |J_t^{-1}(x)h^L - h^L|_{\mathcal{W}}^p) \le T^{p/2}C_3 e^{C_3 T} |h^L|_{\mathcal{W}}^p,$$

(4.5.29)
$$E(\sup_{0 \le t \le T} |\Phi_t(x) - e^{-At}x|_{\mathcal{W}}^p) \le (T^{\varepsilon_0 p} \lor T^{p\varepsilon_1})C_4 e^{C_4 T},$$

Suppose that v_1, v_2 satisfy Proposition 4.5.9 and $p \ge 2$, then

(4.5.30)
$$E(\sup_{0 \le t \le T} |\mathcal{D}_{v_1} \Phi_t^L(x)|_{\mathcal{W}}^p) \le C_5 e^{C_5 T} E[\int_0^T |v_1^L(s)|_{\mathcal{W}}^p ds],$$

(4.5.31)

$$E(\sup_{0 \le t \le T} |\mathcal{D}_{v_1 v_2}^2 \Phi_t^L(x)|_{\mathcal{W}}^p) \le C_6 e^{C_6 T} (E[\int_0^T |v_1^L(s)|_{\mathcal{W}}^{2p} ds])^{1/2} (E[\int_0^T |v_2^L(s)|_{\mathcal{W}}^{2p} ds])^{1/2},$$

(4.5.32)
$$E(\sup_{0 \le t \le T} |\mathcal{D}_{v_1} D_h \Phi_t^L(x)|_{\mathcal{W}}^p) \le C_7 e^{C_7 T} |h|_{\mathcal{W}}^p (E[\int_0^T |v_1^L(s)|_{\mathcal{W}}^{2p} ds])^{1/2},$$

with $h \in \mathcal{W}$ and $C_i = C_i(p, \rho, \gamma, \alpha) > 0, i = 5, 6, 7.$

Lemma 4.5.11 Suppose that Φ_t is the solution to equation (4.5.6) with initial data $x \in \widetilde{\mathcal{W}}$. Then $\mathcal{M}_t \in \mathcal{L}(\mathcal{W}^L, \mathcal{W}^L)$ is invertible almost surely. Denote $\lambda_{\min}(t)$ the smallest eigenvalue of \mathcal{M}_t . Then there exists some q > 1 such that for every p > 0 there is some $C = C(p, \rho, \alpha, \gamma)$ such that

(4.5.33)
$$P[|1/\lambda_{\min}(t)| \ge 1/\varepsilon^q] \le \frac{C\varepsilon^{\varepsilon_0 p}(1+|x|_{\widetilde{\mathcal{W}}})^p}{t^p}.$$

Now we are ready to prove Lemma 4.5.8, which is a modification from [EH01, Proposition 7.12].

Proof of Lemma 4.5.8 Let us consider for i, k = 1...M,

$$\psi_{ik}(\Phi_t) = \psi(\Phi_t) \sum_{j=1}^{M} [(\mathcal{D}_v \Phi_t^L)^{-1}]_{ij} [D_L \Phi_t^L]_{jk}.$$

For $h \in \mathcal{W}^L$, we obtain

$$D_L \psi_{ik}(\Phi_t) \mathcal{D}_v \Phi_t^L h = D_L \psi(\Phi_t) \mathcal{D}_v \Phi_t^L h \sum_{j=1}^M [(\mathcal{D}_v \Phi_t^L)^{-1}]_{ij} [D_L \Phi_t^L]_{jk}$$
$$+ \psi(\Phi_t) \sum_{j=1}^M \mathcal{D}_{vh} \{ [(\mathcal{D}_v \Phi_t^L)^{-1}]_{ij} [D_L \Phi_t^L]_{jk} \}.$$

Set $h = h_i$, where h_i is the standard orthonormal basis of \mathbb{R}^M and sum over *i*, we

obtain

$$E[D_L \psi(\Phi_t) \mathcal{D}_{h_k} \Phi_t^L h] = E[\sum_{i=1}^M \mathcal{D}_{vh_i} \psi_{ik}(\Phi_t)] - E[\psi(\Phi_t) \sum_{i,j=1}^M \mathcal{D}_{vh_i} \{ [(\mathcal{D}_v \Phi_t^L)^{-1}]_{ij} [D_L \Phi_t^L]_{jk} \}].$$

For the first term on the right hand of the above equality, by Bismut integration by parts formula (see e.g. [EH01, Proposition 6.1]) and $\mathcal{D}_v \Phi_t^L = J_t \mathcal{M}_t$,

$$|E[\sum_{i=1}^{M} D_L \psi_{ik}(\Phi_t) \mathcal{D}_{vh_i} \Phi_t^L]|$$

$$\leq \sum_{i,j=1}^{M} |E[\psi(\Phi_t)[J_t^{-1} \mathcal{M}_t^{-1}]_{ij}[D_L \Phi_t^L]_{jk} \int_0^t \langle v^L h_i, dW_r \rangle]|$$

$$\leq ||\psi||_{\infty} \sum_{i,j=1}^{M} E[\frac{1}{\lambda_{\min}} |J_t^{-1} h_j| [D_{h_k} \Phi_t^L]_{jk} |\int_0^t \langle v^L h_i, dW_r \rangle|].$$

Then by B-D-G inequality, (4.5.33), (4.5.27), we obtain

$$|E[\sum_{i=1}^{M} \mathcal{D}_{vh_i} \psi_{ik}(\Phi_t)]| \le ||\psi||_{\infty} C e^{Ct} (1+|x|_{\widetilde{\mathcal{W}}})^p t^{-p}.$$

The other term can be estimated similarly, the assertion follows.

4.5.4 Hörmander's systems

Let us consider the SPDE for θ^L in Stratanovich form as (4.5.34)

$$d\theta^L + A\theta^L dt + \chi(\frac{|\theta|_{\mathcal{W}}}{3\rho})B_L(\theta,\theta)dt - \frac{1}{2}\sum_{k\in Z_L(N_0)} Dg_k(\theta)e_k \cdot g_k(\theta)e_k dt = \sum_{k\in Z_L(N)} g_k(u)\circ dw_k(t)e_k \cdot g_k(\theta)e_k \cdot g_k(\theta)e_k dt = \sum_{k\in Z_L(N)} g_k(u)\circ dw_k(t)e_k \cdot g_k(\theta)e_k \cdot g_k(\theta)e_k \cdot g_k(\theta)e_k dt = \sum_{k\in Z_L(N)} g_k(u)\circ dw_k(t)e_k \cdot g_k(\theta)e_k \cdot g_k(\theta)e_k$$

where $g_k(\theta) = (1 - \chi(\frac{|\theta|_{\mathcal{W}}}{\rho}))g_k$ for $k \in Z_L(N_0)$ and $g_k(\theta) = g_k$ for $k \in Z_L(N) \setminus Z_L(N_0)$. For any $x \in \mathcal{W}$, it is clear that if $k \in Z_L(N_0)$

$$Dg_k(x)e_k \cdot g_k(x)e_k = -\frac{1}{\rho}\chi'(\frac{|x|_{\mathcal{W}}}{\rho})(1-\chi(\frac{|x|_{\mathcal{W}}}{\rho}))g_k^2\frac{\langle x, e_k\rangle_{\mathcal{W}}}{|x|_{\mathcal{W}}}e_k.$$

For any two Banach spaces E_1, E_2 , we denote the set of all C^{∞} functions $E_1 \to E_2$, which are polynomially bounded together with all their derivatives by $P(E_1, E_2)$. If

 $K \in P(H, H^L)$ and $X \in P(H, H)$, define $[X, K]_L$ by

$$[X,K]_L(x) = DK(x)X(x) - D_L X^L(x)K(x), x \in H.$$

For instance, $[A, K]_L \in P(D(A), H^L)$ with $[A, K]_L(x) = DK(x)Ax - A_LK(x)$. Define

$$X^{0}(x) = Ax + \chi(\frac{|x|_{\mathcal{W}}}{\rho})e^{-\delta A_{H}}B(x,x) + \sum_{k \in Z_{L}(N_{0})} \frac{1}{2\rho}\chi'(\frac{|x|_{\mathcal{W}}}{\rho})(1 - \chi(\frac{|x|_{\mathcal{W}}}{\rho}))g_{k}^{2}\frac{\langle x, e_{k}\rangle_{\mathcal{W}}}{|x|_{\mathcal{W}}}e_{k}.$$

Definition 4.5.12 The Hörmander's system **K** for equation (4.5.34) is defined as follows: given any $y \in \mathcal{W}$, define

$$\begin{aligned} \mathbf{K}_{0}(y) &= \{g_{k}(y)e_{k} : k \in Z_{L}(N)\}, \\ \mathbf{K}_{1}(y) &= \{[X^{0}(y), g_{k}(y)e_{k}]_{L} : k \in Z_{L}(N)\} \\ \mathbf{K}_{2}(y) &= \{[g_{k}(y)e_{k}, K(y)]_{L} : K \in \mathbf{K}_{1}(y), k \in Z_{L}(N)\}, \end{aligned}$$

and $\mathbf{K}(y) = \mathbf{K}_0(y) \cup \mathbf{K}_1(y) \cup \mathbf{K}_2(y)$.

Proposition 4.5.13 There exist $\bar{\rho} > 0$ and $\bar{N} \ge N_0$ (which depend only on N_0 and G) such that if $\rho \ge \bar{\rho}$ and $N \ge \bar{N}$, then the following property holds : for every $x \in \mathcal{W}$ and $h \in H^L$ there exist $\sigma > 0$ and R > 0 such that

$$\inf_{\delta > 0} \sup_{K \in \mathbf{K}} \inf_{|y-x|_{\mathcal{W}} \le R} |\langle K(y), h \rangle_{\mathcal{W}}| \ge \sigma |h|_{\mathcal{W}}.$$

Proof The basic idea of the proof follows from [EH01, Theorem 7.8] and [RX10, Proposition 4.5]. It is sufficient to show that there is a (finite) set $\tilde{\mathbf{K}} \subset \mathbf{K}(y)$ such that span $\tilde{\mathbf{K}} = H^L$. We choose $R \leq \frac{1}{4}\rho$.

Case 1: $|x|_{\mathcal{W}} \ge R + 2\rho$. In this case $g_k(y) = g_k$, we can take $\mathbf{\tilde{K}} = \mathbf{K}_0$.

Case 2: $|x|_{\mathcal{W}} \leq \rho - R$. In this case $X^0(y) = Ay + e^{-\delta A_H}B(y, y)$. Since

$$\langle B(e^{il\cdot x}, e^{im\cdot x}), e^{ik\cdot x}) \rangle = \begin{cases} -\frac{1}{|l|}(l^{\perp} \cdot m), & \text{if } l+m=k, \\ 0, & \text{if } l+m \neq k, \end{cases}$$

we could calculate $B(e_l, e_m)$ easily. For instance, for $l, m, l - m \in \mathbb{Z}^2_+$, we have

$$B(e_l, e_m) = -\frac{1}{2|l|} (l^{\perp} \cdot m) e_{l+m} + \frac{1}{2|l|} (l^{\perp} \cdot m) e_{l-m}$$

For $l, -m, l+m \in \mathbb{Z}^2_+$, we have

$$B(e_l, e_m) = -\frac{1}{2|l|} (l^{\perp} \cdot m) e_{-l-m} + \frac{1}{2|l|} (l^{\perp} \cdot m) e_{m-l}.$$

We have for $l, m \in Z_L(N) \setminus Z_L(N_0), l + m = k \in Z_L(N_0),$

$$[g_l e_l, [B(y, y), g_m e_m]_L]_L = -\pi_N B(g_l e_l, g_m e_m) - \pi_N B(g_m e_m, g_l e_l).$$

Hence, we choose $N \geq N_0$ large enough so that for each $k \in Z_L(N_0)$ there are $l, m \in Z_L(N) \setminus Z_L(N_0)$ such that l and m are linearly independent and k = l + m (or k = l - m). Then the vectors $[g_l e_l, [B(y, y), g_m e_m]_L]_L, g_l e_l$, where l, m run over $Z_L(N) \setminus Z_L(N_0)$ span H^L . Then we can take $\tilde{\mathbf{K}} = \mathbf{K}_0 \cup \mathbf{K}_2$.

Case 3: $\rho - R \leq |x|_{\mathcal{W}} \leq 2\rho + R$. Write $X^0(y) = X^{01}(y) + X^{02}(y)$ where $X^{01}(y) = Ay + e^{-\delta A_H} B(y, y)$ and $X^{02}(y) = \sum_{k \in Z_L(N_0)} \frac{1}{2\rho} \chi'(\frac{|x|_{\mathcal{W}}}{\rho}) (1 - \chi(\frac{|x|_{\mathcal{W}}}{\rho})) g_k^2 \frac{\langle x, e_k \rangle_{\mathcal{W}}}{|x|_{\mathcal{W}}} e_k$. By Case 2, we obtain $[g_l e_l, [X^{01}(y), g_m e_m]_L]_L, g_l e_l$ span the whole H^L . And it is easy to see $|[g_l e_l, [X^{01}(y), g_m e_m]_L]_L| \leq \frac{c}{\rho^3}$. So, for ρ large enough, $[g_l e_l, [X^0(y), g_m e_m]_L]_L$ span H^L . Take $\tilde{\mathbf{K}} = \mathbf{K}_0 \cup \mathbf{K}_2$.

4.5.5 Proof of Lemma 4.5.11

We follow the idea of the proof of [N85, Theorem 4.2] by using Proposition 4.5.13 and the following Norris's Lemma ([N85, Lemma 4.1]).

Lemma 4.5.14 (Norris' Lemma). Let $a, y \in \mathbb{R}$. Let β_t be a real-valued predictable process and γ_t and u_t be adapted *H*-valued processes. Let

$$a_t = a + \int_0^t \beta_s ds + \int_0^t \langle \gamma_s, dW_s \rangle, \qquad Y_t = y + \int_0^t a_s ds + \int_0^t \langle u_s dW_s \rangle,$$

Suppose that $T < t_0$ is a bounded stopping time such that for some constant $C < \infty$:

$$|\beta_t|, |\gamma_t|, |a_t|, |u_t| \leq C$$
 for all $t \leq T$.

Then for any r > 8 and $\nu > \frac{r-8}{9}$ there is $C = C(T, q, \nu)$ such that

$$P[\int_0^T Y_t^2 dt < \varepsilon^r, \int_0^T (|a_t|^2 + |u_t|^2) dt \ge \varepsilon] < Ce^{-\frac{1}{\varepsilon^\nu}}.$$

Proof of Lemma 4.5.11 Denote $\mathcal{S}^L = \{\eta \in \mathcal{W}^L; |\eta|_{\mathcal{W}^L} = 1\}$. As for $\eta \in \mathcal{S}^L$,

$$\langle \mathcal{M}_t \eta, \eta \rangle_{\mathcal{W}} = \sum_{k \in Z_L(N)} \frac{1}{|k|^{2s}} \int_0^t |\langle J_s^{-1}(g_k(\Phi_s)e_k), \eta \rangle_{\mathcal{W}}|^2 ds,$$

(4.5.33) is equivalent to (4.5.35)

$$P[\inf_{\eta\in\mathcal{S}^L}\sum_{k\in\mathbb{Z}_L(N)}\frac{1}{|k|^{2s}}\int_0^t |\langle J_s^{-1}(g_k(\Phi_s)e_k),\eta\rangle_{\mathcal{W}}|^2ds\leq\varepsilon^q]\leq\frac{C\varepsilon^{\frac{\varepsilon_0p}{2\alpha}}(1+|x|_{\widetilde{\mathcal{W}}})^p}{t^p},$$

for all p > 0, where $g_k(\theta) = (1 - \chi(\frac{|\theta|_{\mathcal{W}}}{\rho}))g_k$ for $k \in Z_L(N_0)$ and $g_k(\theta) = g_k$ for $k \in Z_L(N) \setminus Z_L(N_0)$.

Define a stopping time τ by

$$\tau = \inf\{s > 0 : |\Phi_s(x) - x|_{\mathcal{W}} > R, |J_s^{-1} - Id|_{\mathcal{L}(\mathcal{W})} > c\},\$$

where R > 0 is the same as in Proposition 4.5.13 and c > 0 is sufficiently small. By (4.5.29) and the easy fact $|e^{-At}x - x|_{\mathcal{W}} \leq Ct^{\frac{\varepsilon_0}{2\alpha}}|x|_{\widetilde{\mathcal{W}}}$, we have for any $p \geq 2$

$$E[|\sup_{0 \le t \le T} |\Phi_t - x|_{\mathcal{W}}^p] \le E[\sup_{0 \le t \le T} |e^{-At}x - x|_{\mathcal{W}}^p + \sup_{0 \le t \le T} |\Phi_t(x) - e^{-At}x|_{\mathcal{W}}^p]$$
$$\le C_1(1 + |x|_{\widetilde{\mathcal{W}}})^p (T^{\frac{p\varepsilon_0}{2\alpha}} \lor T^{p\varepsilon_1}).$$

Combining the above inequality and (4.5.28), we have for $\varepsilon_0 < \varepsilon_1$

(4.5.36)
$$P(\tau \le \varepsilon) = C_1 \varepsilon^{\frac{p \varepsilon_0}{2\alpha}} (1 + |x|_{\widetilde{\mathcal{W}}})^p,$$

for all p > 0.

By the same arguments as [N85, p127], (4.5.35) holds as long as for any $\eta \in \mathcal{S}^L$, we have some neighborhood $\mathcal{N}(\eta)$ of η and some $k \in Z_L(N)$ so that

$$(4.5.37) \qquad \sup_{\eta' \in \mathcal{N}(\eta)} P\left[\int_0^{t \wedge \tau} |\langle J_s^{-1}(g_k(\Phi_s)e_k), \eta \rangle_{\mathcal{W}}|^2 ds \le \varepsilon^q\right] \le \frac{C\varepsilon^{\frac{\varepsilon_0 p}{2\alpha}}(1+|x|_{\widetilde{\mathcal{W}}})^p}{t^p}.$$

According to Definition 4.5.12 and Proposition 4.5.13, for any $\eta \in S^L$, there exists a $K \in \mathbf{K}$ and a neighborhood \mathcal{N} of η in S^L such that

$$\inf_{|y-x|_{\mathcal{W}} \leq R} \inf_{|V-Id|_{\mathcal{L}(\mathcal{W})}} \inf_{\eta \in \mathcal{N}} |\langle VK(y), \eta \rangle_{\mathcal{W}}| \geq \frac{\sigma}{2}$$

By this and (4.5.36) we deduce that for any $\eta \in \mathcal{S}^L$, we have some neighborhood

$$\mathcal{N}(\eta) \text{ of } \eta$$

$$(4.5.38)$$

$$\sup_{\eta' \in \mathcal{N}(\eta)} P[\int_0^{t \wedge \tau} |\langle J_s^{-1} K(\Phi_s), \eta \rangle_{\mathcal{W}}|^2 ds \le \varepsilon^q] \le P[\tau \wedge t < 2\varepsilon/\sigma] \le \frac{C\varepsilon^{\frac{\varepsilon_0 p}{2\alpha}} (1+|x|_{\widetilde{\mathcal{W}}})^p}{t^p}.$$

Now we prove that (4.5.38) implies (4.5.37). Without loss of generality, assume that $K \in \mathbf{K}_2$, so there exists some $g_k e_k$ and $g_l e_l$ such that

$$K_0(y) := g_k(y)e_k, K_1(y) := [X^0(y), g_k(y)e_k], K = K_2 := [g_l(y)e_l, K_1(y)]$$

Take $Y(t) = \langle J_t^{-1} K_1(\Phi_t), \eta \rangle$, $a(t) = \langle J_t^{-1} [X_0, K_1](\Phi_t), \eta \rangle$ and $u^i(t) = \langle J_t^{-1} [g_i e_i, K_1](\Phi_t), \eta \rangle$. Applying Lemma 4.5.14, we obtain

$$P(\int_0^{t\wedge\tau} |\langle J_s^{-1}K_1(\Phi_s),\eta\rangle|^2 \le \varepsilon^r, \int_0^{t\wedge\tau} |\langle J_s^{-1}K_2(\Phi_s),\eta\rangle|^2 \ge \varepsilon) \le Ce^{-\frac{1}{\varepsilon^\nu}}.$$

Hence, by (4.5.38) we obtain

$$P[\int_0^{t\wedge\tau} |\langle J_s^{-1} K_1(\Phi_s), \eta \rangle_{\mathcal{W}}|^2 ds \le \varepsilon^r] \le \frac{C\varepsilon^{\frac{\varepsilon_0 p}{2\alpha}} (1+|x|_{\widetilde{\mathcal{W}}})^p}{t^p}$$

By a similar but simpler arguments we have (4.5.37).

4.5.6 Controllability and support

In this subsection we prove the support theorem for the solution of the equation (4.3.1) by the control theory.

Proposition 4.5.15 Suppose that Assumption 4.5.1 holds. Let $(P_x)_{x\in H}$ be the Markov solution of equaion (4.3.1). For every $x \in \mathcal{W}$ and T > 0, and every \mathcal{W} -open set $U \subset \mathcal{W}$, $P_x(\xi_T \in U) > 0$.

To prove Proposition 4.5.15, by the proof of Theorem 4.3.8, we only need to prove the following control problem.

Lemma 4.5.16 Given any $T > 0, x, y \in \mathcal{W}$ and $\varepsilon > 0$, there exist $\rho_0, \theta \in C([0,T]; \mathcal{W})$ and $\omega \in L^{\infty}([0,T]; H)$ such that θ solves the following equation,

(4.5.39)
$$\partial_t \theta + A\theta + B(\theta, \theta) = G\omega,$$

with $\theta(0) = x$ and $|\theta(T) - y| \leq \varepsilon$, and $\sup_{t \in [0,T]} |\theta(t)|_{\mathcal{W}} \leq \rho_0$.

Proof Let $z \in D(\Lambda^{3\alpha+s})$ such that $|y-z|_{\mathcal{W}} \leq \frac{\varepsilon}{2}$. It suffices to show that there exist

 θ, ω satisfying the conditions of lemma and $|u(T) - z|_{\mathcal{W}} \leq \frac{\varepsilon}{2}$. Decompose $\theta = \theta^H + \theta^L$, then equation (4.5.39) can be written as

(4.5.40)
$$d\theta^L + A\theta^L dt + B_L(\theta, \theta) dt = 0.$$

(4.5.41)
$$d\theta^H + A\theta^H dt + B_H(\theta, \theta) dt = G\omega.$$

We split [0, T] into the pieces $[0, T_1], [T_1, T_2], [T_2, T_3]$ and $[T_3, T]$ with T_1, T_2, T_3 to be chosen along the proof.

Step 1: regularization of the initial condition. Set $\omega = 0$ on $[0, T_1]$. By the same arguments as Lemma 4.3.6, we can find a time T_1 such that $\sup_{[0,T_1]} |\theta(t)|_{\mathcal{W}} \leq \rho_0$ and $\theta(T_1) \in D(\Lambda^{3\alpha+s})$.

Step 2: high modes led to zero. Choose a smooth function ψ on $[T_1, T_2]$ such that $0 \leq \psi \leq 1, \psi(T_1) = 1$ and $\psi(T_2) = 0$ and set $\theta^H(t) = \psi(t)\theta^H(T_1)$ for $t \in [T_1, T_2]$. As θ^L is finite dimensional, an estimate yields

$$\frac{d}{dt}|\theta^L|_{\mathcal{W}}^2 + |\Lambda^{\alpha}\theta^L|_{\mathcal{W}}^2 \le c(|\theta^L|_{\mathcal{W}}^2 + |\theta^H|_{\mathcal{W}}^2)^2,$$

and $|\theta(t)|_{\mathcal{W}}^2 \leq |\theta^L(t)|_{\mathcal{W}}^2 + |\theta^H(T_1)|_{\mathcal{W}}^2 \leq \rho_0$ for $T_1 \leq t \leq T_2 := \frac{T}{2} \wedge (T_1 + (4c|x|_{\mathcal{W}}^2)^{-1}).$ Plug θ^L in (4.5.41), take

$$\omega(t) = \psi'(t)G^{-1}\theta^{H}(T_{1}) + \psi(t)G^{-1}A\theta^{H}(T_{1}) + G^{-1}B_{H}(\theta(t), \theta(t)).$$

As $\theta(T_1) \in D(\Lambda^{3\alpha+s}), |G^{-1}A\theta^H(T_1)| < \infty$ and $|G^{-1}B_H(\theta(t), \theta(t))| \le c|A\theta(t)|_{\mathcal{W}}^2 \le C(|A\theta^H(T_1)|_{\mathcal{W}}^2 + |\theta^L(t)|_{\mathcal{W}}^2)$ for $t \in [T_1, T_2]$. Hence, $\omega \in L^{\infty}([T_1, T_2], H)$.

Step 3: low modes close to z. Let $\theta^L(t)$ be the linear interpolation between $\theta^L(T_2)$ and z^L for $t \in [T_2, T_3]$. Write $\theta(t) = \sum \theta_k(t)e_k$, then (4.5.40) is written as

(4.5.42)
$$\dot{\theta}_k + |k|^{2\alpha} \theta_k + B_k(\theta, \theta) = 0, k \in Z_L(N_0).$$

Let us choose a suitable θ^H to simplify the above $B_k(\theta, \theta)$. Consider the set $\{(l_k, m_k) : k \in Z_L(N_0)\}$ such that (a) $l_k + m_k = k$.

(b) $l_k \not\parallel m_k$ for all $k \in Z_L(N_0)$.

(c) For every $k \in Z_L(N_0), |l_k|, |m_k| \ge 2^{(2N_0+1)^2}$.

(d) If $k_1 \neq k_2$, then $|l_{k_1} \pm l_{k_2}|, |m_{k_1} \pm m_{k_2}|, |l_{k_1} \pm m_{k_2}|, |m_{k_1} \pm l_{k_2}| \ge 2^{(2N_0+1)^2}$. Define

$$\theta^H(t) = \sum_{k \in Z_L(N_0)} (\theta_{l_k}(t) e_{l_k} + \theta_{m_k} e_{m_k}),$$

with $\theta_{l_k}(t), \theta_{m_k}$ to be determined below. By (c) and (d), it is easy to see

$$B_k(\theta^L, \theta^H) = B_k(\theta^H, \theta^L) = 0,$$

$$B_k(e_{l_{k_1}}, e_{l_{k_2}}) = B_k(e_{m_{k_1}}, e_{m_{k_2}}) = B_k(e_{l_{k_1}}, e_{m_{k_2}}) = B_k(e_{m_{k_1}}, e_{l_{k_2}}) = 0.$$

Then (4.5.40) is simplified to the following equation

(4.5.43)
$$\dot{\theta}_k + |k|^{2\alpha} \theta_k + B_k(\theta^L, \theta^L) + \tilde{B}_k(\theta_{l_k} e_{l_k}, \theta_{m_k} e_{m_k}) = 0, k \in Z_L(N_0).$$

One can easily find a solution θ^H for equation (4.5.43) which is smooth in t and by construction θ is finite dimensional. Hence $\theta(t)$ is smooth in space and time for $t \in [T_2, T_3]$ and $\sup |\theta(t)|_{\mathcal{W}}$ can be bounded by $|\theta^L(T_2)|_{\mathcal{W}}, z^L$ and $T_3 - T_2$. We set $\omega = G^{-1}[\dot{\theta}^H + A\theta^H + B_H(\theta, \theta)]$ and $\omega \in L^{\infty}([T_2, T_3], H)$.

Step 4: high modes close to z. In the interval $[T_3, T]$ we choose θ^H as the linear interpolation between $\theta^H(T_3)$ and z^H . Let θ^L be the solution to (4.5.40) on $[T_3, T]$ with the choice of θ^H given above. Since $\theta(T_3) \in D(\Lambda^{3\alpha+s})$ and $\theta^L(T_3) = z^L$, we know $\sup_{T_3 \leq t \leq T} |\theta^L(t) - z^L|_W \leq \frac{\varepsilon}{2}$ if $T - T_3$ is small enough. Then as in Step 1, we can find $\omega \in L^{\infty}([T_3, T], H)$ solving (4.5.41). It is clear that $\sup_{T_3 \leq t \leq T} |\theta(t)|_W \leq c|z|_W + C|\theta(T_3)|_W$.

We also obtain the following ergodic properties by the same arguments as Theorems 4.3.12 and 4.4.3.

Theorem 4.5.17 Assume Assumption 4.5.1. There exists a Markov process $\theta(\cdot, \nu)$ on a probability space $(\Omega, \mathcal{F}, P_{\nu})$ which is a martingale stationary solution of the stochastic quasi-geostrophic equation (4.1.3). The law ν of $\theta(t, \nu)$ is the unique invariant measure on \mathcal{W} of the transition semigroup $(P_t)_{t>0}$. Moreover

(i) the invariant measure ν is ergodic,

(ii) the transition semigroup $(P_t)_{t\geq 0}$ is strong Feller, irreducible, and therefore strongly mixing.

(iii) there exist $C_{exp} > 0$ and a > 0 such that

$$\|P_t^*\delta_{x_0} - \mu\|_{TV} \le \|P_t^*\delta_{x_0} - \mu\|_V \le C_{\exp}(1 + |x_0|^2)e^{-at},$$

for all t > 0 and $x_0 \in H$.

Remark 4.5.18 We can also prove approximate controllability of the solution of the stochastic quasi-geostrophic equation for $\alpha > 1/2$ driven by finite dimensional noise. Since the proof is similar to [S06], we don't give all the details of the proof. For more details, we refer to [S06].

Bibliography

- [AR90] S. Albeverio, M. Röckner: Classical Dirichtet forms on topological vector spaces - closability and a Cameron-Martin formula. J. Funct. Anal. 88, (1990) 395-436
- [ASZ09] L. Ambrosio, G. Savaré and L. Zambotti, Existence and stability for Fokker-Planck equations with log-concave reference measure. *Probab. Theory Related Fields.* 145, (2009) 517-564
- [Ba10] V.Barbu, Nonlinear differential equations of monotone type in Banach spaces, Springer New York Dordrecht Heidelberg London, (2010)
- [BC08] P. Briand, F. Confortola, BSDEs with stochastic Lipschitz condition and quadratic PDEs in Hilbert spaces. *Stochastic Process. Appl.* 118 no. 5, (2008) 818-838.
- [BDHPS03] Ph.Briand, B. Delyon, Y.Hu, E.Pardoux and L.Stoica : L^p solutions of backward stochastic differential equations, Stochastic Process. Appl. 108, (2003)109-129
- [BDR09] V.I. Bogachev, G. Da Prato, M. Röckner, Fokker-Planck equations and maximal dissipativity for Kolmogorov operators with time dependent singular drifts in Hilbert spaces, J. Funct. Anal., V. 256 N 3, (2009) 1269-1298.
- [BDT09] V. Barbu, G. Da Prato and L. Tubaro, Kolmogorov equation associated to the stochastic reflection problem on a smooth convex set of a Hilbert spaces, *The Annals of Probability* 4, (2009) 1427-1458
- [BG68] R.M. Blumenthal, R.K. Getoor, Markov processes and potential theory, Academic Press, New York and London, (1968)
- [BKL02] J. Bricmont, A. Kupiainen, R. Lefevere : Exponential mixing for the 2D Stochastic NavierCStokes Dynamics. Commun. Math. Phys. 230(1), (2002) 87-132

- [BKR06] V.I. Bogachev, N.V. Krylov, M. Röckner, Elliptic equations for measures: regularity and global bounds of densities, J. Math. Pures Appl., V. 85 N 6, (2006) 743-757
- [BPS02] V.Bally, E.Pardoux and L.Stoica, Backward stochastic differential equations associated to a symmetric Markov process, *Research Report*, RR-4425, INRIA, (2002)
- [BPS05] V.Bally, E.Pardoux and L.Stoica, Backward stochastic differential equations associated to a symmetric Markov process, *Potential Analysis*, 22, (2005) 17-60
- [BR95] V.I. Bogachev, M. Rockner, Regularity of invariant measures on finite and infinite dimensional spaces and applications, J. Funct. Anal., V. 133 N 1, (1995) 168-223
- [BR01] V.I. Bogachev , M. Röckner Elliptic equations for measures on infinite dimensional spaces and applications. *Probab. Theory Relat. Fields.* V. 120, (2001) 445-496
- [BRS00] V.I. Bogachev, M. Röckner, W. Stannat, Uniqueness of invariant measures and maximal dissipativity of diffusion operators on L1, Infinite Dimensional Stochastic Analysis (Proceedings of the Colloquium, Amsterdam, 11-12 February, 1999), Ph. Clément, F. den Hollander, J. van Neerven and B. de Pagter eds., Royal Netherlands Academy of Arts and Sciences, Amsterdam, (2000) 39-54
- [CV06] L. Caffarelli, A. Vasseur, Drift diffusion equations with fractional diffusion and the quasi-geostrophic equation, Annals of Math., 171 No. 3, (2010) 1903-1930
- [CW99] P. Constantin, J. Wu, Behavior of solutions of 2D quasi-geostrophic equations, SIAM J. Math. Anal. 30, (1999) 937-948
- [D04] G. Da Prato: Kolmogorov equations for stochastic PDEs. Advanced Courses in Mathematics. CRM Barcelona. Birkhäuser Verlag, Basel, (2004)
- [DD03] G. Da Prato and A. Debussche, Ergodicity for the 3D stochastic Navier-Stokes equations, J. Math. Pures Appl. (9) 82 no. 8, (2003) 877-947.
- [DO06] A. Debussche, C. Odasso, Markov solutions for the 3D stochastic Navier-Stokes equations with state dependent noise, *J. evol. equ.* 6, (2006) 305-324

- [DR02] G. Da Prato, M. Röckner, Singular dissipative stochastic equations in Hilbert spaces, Probab. Theory Related Fields 124, (2002) 261-303
- [DRW09] G. Da Prato, M. Röckner and F.Y. Wang, Singular stochastic equations on Hilbert spaces: Harnack inequalities for their transition semigroups, *Journal* of Functional Analysis 257, (2009) 992-1017
- [DZ92] G. Da Prato, J. Zabczyk, Stochastic Equations in Infinite Dimensions. Cambridge University Press (1992)
- [DZ96] G. Da Prato, J. Zabczyk, Ergodicity for Infinite Dimensional Systems, London Mathematical Society Lecture Notes, n. 229, Cambridge University Press (1996)
- [DZ02] G. Da Prato, J. Zabczyk, Second Order Partial Differential Equations in Hilbert Spaces, London Mathematical Society, Lecture Notes, 293, Cambridge University Press, (2002)
- [EH01] J.-P. Eckmann, M. Hairer, Uniqueness of the invariant measure for a stochastic PDE driven by degenerate noise, *Comm. Math. Phys.* 219 no. 3, (2001) 523-565
- [F90] R.Z.Fan, Representation of martingale additive functionals on banach spaces, Acta Mathematicae applicatae sinica, Vol. 6 No.1, (1990) 74-80
- [FG95] F. Flandoli, D. Gatarek, Martingale and stationary solutions for stochastic Navier-Stokes equations, Probability Theory and Related Fields 102, (1995) 367-391
- [FH07] M.Fuhrman and Y.Hu, Backward Stochastic Differential Equations in Infinite Dimensions with Continuous Driver and Applications, Appl. Math. Optim. 56, (2007) 265-302
- [FOT94] M.Fukushima, Y. Oshima and M. Takeda, Dirichlet forms and symmetric Markov Processes, Walter de Gruyter, Berlin, (1994)
- [FR08] F. Flandoli, M. Romito, Markov selections for the 3D stochastic Navier-Stokes equations, *Probability Theory and Related Fields* 140, (2008) 407-458
- [FR07] F. Flandoli, M. Romito, Regularity of transition semigroups associated to a 3D stochastic Navier-Stokes equation, Stochastic Differential Equations: Theory and Application (P. Baxendale H. and S. Lototski V., eds.), Interdisciplinary Mathematical Sciences, vol. 2, World Scientific, Singapore, (2007)

- [FT02] M.Fuhrman and G.Tessitore, Nonlinear Kolmogorow equations in infinite dimensional spaces: The backward stochastic differential equations approach and applications to optimal control, *The Annals of Probability*, Vol. 30. No.3., (2002) 1397-1465
- [G96] F. Gozzi, Global regular solutions of second order HamiltonCJacobi equations in Hilbert spaces with locally Lipschitz nonlinearities. J. Math. Anal. Appl. 198, (1996) 399-443
- [GM05] B. Goldys and B. Maslowski, Exponential ergodicity for stochastic Burgers and 2D Navier-Stokes equations, J. Funct. Anal. 226 no. 1, (2005) 230-255
- [GRZ09] B. Goldys, M. Röckner and X.C. Zhang, Martingale solutions and Markov selections for stochastic partial differential equations, *Stochastic Processes and their Appliations* **119** (2009) 1725-1764
- [HM06] M. Hairer, J. C. Mattingly, Ergodicity of the 2D Navier-Stokes equations with degenerate stochastic forcing, Ann. of Math. (2), 164, (2006) 993-1032
- [Kr10] N. V. Krylov, Ito's formula for the L_p -norm of stochastic W_p^1 -valued processes Probab. Theory Relat. Fields 147, (2010) 583-605
- [KNV07] A. Kiselev, F. Nazarov, and A. Volberg. Global well-posedness for the critical 2D dissipative quasi-geostrophic equation. *Invent. math.* 167, (2007) 445-453
- [KS02] S. Kuksin, A. Shirikyan: Coupling approach to white-forced nonlinear PDEs. J. Math. Pures Appl. 1, (2002) 567-602
- [Ju04] N. Ju, Existence and Uniqueness of the Solution to the Dissipative 2D Quasi-Geostrophic Equations in the Sobolev Space, Communications in Mathematical Physics 251, (2004) 365-376
- [Ju05] N. Ju, On the two dimensional quasi-geostrophic equations. Indiana Univ. Math. J. 54 No. 3, (2005), 897-926
- [JY77] J. Jacod and M. Yor, Etude des solutions extrémales et représentation intégrale des solutions pour certains probl'emes de martingales, Z. W. 38, (1977) 83-125
- [L88] P. L. Lions, Viscosity solutions of fully nonlinear second-order equations and optimal stochastic control in infinite dimensions. I. The case of bounded stochastic evolutions. Acta Math. 161 (1988) 243-278

- [L89] P. L. Lions, Viscosity solutions of fully nonlinear second order equations and optimal stochastic control in infinite dimensions. II. Optimal control of Zakai's equation. Stochastic Partial Differential Equations and Applications II. Lecture Notes in Math. 1390 147-170. Springer, Berlin (1989)
- [L92] P. L. Lions, Viscosity solutions of fully nonlinear second-order equations and optimal stochastic control in infinite dimensions. III. Uniqueness of viscosity solutions for general second-order equations. J. Funct. Anal. 86, (1989) 1-18
- [L01] A.Lejay, BSDE driven by Dirichlet process and semi-linear parabalic PDE. Application to homogenization, Stochastic Processes and their Application, 97, 1,(2001) 1-39
- [M11] F. Masiero, Hamilton Jacobi Bellman equations in infinite dimensions with quadratic and superquadratic Hamiltonian, http://arxiv.org/abs/1007.1882v1
- [M02] J. Mattingly, Exponential convergence for the stochastically forced Navier-Stokes equations and other partially dissipative dynamics. Commun. Math. Phys. 230, (2002) 421-462
- [M67] P. A. Meyer, Processus de Markov, Lecture Notes in Mathematics 26, Springer-Verlag, 1967.
- [MOR95] Z.M.Ma, L. Overbeck, M. Röckner, Markov process associated with semi-Dirichlet forms, Osaka J. Math. 32, (1995) 97-119
- [MR92] Z.M.Ma, M. Röckner, Introduction to the theory of (Non-Symmetric) Dirichlet forms, Springer Verlag, Berlin Heidelberg New York, (1992)
- [N85] James Norris, Simplified Malliavin calculus, Seminaire de Probabilities, XX, 1984/85, Lecture Notes in Math., vol. 1204, Springer, Berlin, (1986) 101-130
- [O07] Odasso, C.: Exponential mixing for the 3D stochastic Navier-Stokes equations. Commun. Math. Phys. 270(1), (2007) 109-139
- [Pa99] E.Pardoux,:BSDE, weak convergence and homogenization of semilinear PDEs', in F.H. Clarke and R.J.Stern(eds), Nonlinear Analysis, Differential Equations and Control, Kluwer Academic Publishers, (1999) 505-549
- [PP90] E. Pardoux, S. G. Peng, Adapted solution of a backward stochastic dierential equation, System Control Lett. 14, No.1, (1990) 55-61
- [PP92] E. Pardoux, and S. Peng, Backward stochastic differential equations and quasilinear parabolic partial differential equations and their applications, in

B.L. Rozovskii and R.B. Sowers (eds), Stochastic Partial Differential Equations and Their Applications, Lecture Notes in Control Inform. Sci. 176, Springer, Berlin, (1992) 200-217

- [PR07] C. Prévôt and M. Röckner, A concise course on stochastic partial differential equations, Lecture Notes in Mathematics, vol. 1905, Springer, (2007)
- [QY10] Z.M. Qian and J.G. Ying, Martingale representations for diffusion processes and backward stochastic differential equations, arXiv:0910.4911v1
- [Re95] S. Resnick, Danamical Problems in Non-linear Advective Partial Differential Equations, PhD thesis, University of Chicago, Chicago (1995)
- [R08] M. Romito, Analysis of equilibrium states of Markov solutions to the Navier-Stokes equations J. Stat. Phys. 131, (2008) 415-444
- [RX10] Marco Romito and Lihu Xu, Ergodicity of the 3d stochastic navier-stokes equations driven by mildly degenerate noise, *Stochastic Processes and their Applications* vol. 121, 4, (2010) 673-700
- [RZZ] M. Röckner, R.-C. Zhu, X.-C. Zhu, The stochastic reflection problem on an infinite dimensional convex set and BV functions in a Gelfand triple, to appear in "Annals of Probability"
- [RZZ10] M. Röckner, R.-C. Zhu, X.-C. Zhu, BV functions in a Gelfand triple and the stochastic reflection problem on a convex set of a Hilbert space C. R. Math. Acad. Sci. Paris 348 no. 21-22, (2010) 1175-1178
- [RZZ12] M. Röckner, R.-C. Zhu, X.-C. Zhu, Sub- and supercritical stochastic quasigeostrophic equation, to preprint
- [Sh88] M. Sharpe, General Theory of Markov processes, Academic Press, Boston, 1988
- [S06] A. Shirikyan, Approximate controllability of three-dimensional Navier-Stokes equations, *Communications in Mathematical Physics* 266, 1,(2006) 123-151
- [S09] L. M. Sowa, Quasi-linear partial differential equations in Dirichlet spaces, Diploma degree thesis, Bielefeld, BiBoS Additional Preprint E09-04-320
- [St1] W. Stannat: (Nonsymmetric)Dirichlet operators on L¹: Existence, Uniqueness and associated Markov processes, Ann. Scuola Norm. Sup. Pisa, Serie IV, vol. XXVIII, 1,(1999) 99-140

- [St2] W. Stannat: The theory of generalized Dirichlet forms and its application in analysis and stochastics, Memoirs of the AMS, 678, (1999)
- [St70] E. Stein, Singular Integrals and Differentiability Properties of Functions, Princeton, NJ: Princeton University Press, (1970)
- [Te84] R. Temam, Navier-Stokes Equations, North-Holland, Amsterdam, (1984)
- [Tr1] G. Trutnau, Stochastic calculus of generalized Dirichlet forms and applications to stochastic differential equations in infinite dimensions, Osaka J.Math. 37, (2000) 315-343
- [Tr2] G. Trutnau, Stochastic calculus of generalized Dirichlet forms and applications, Doctor degree thesis, Bielefeld, SFB 343 Preprint-E00-004
- [Z] T.S. Zhang, A probabilistic approach to Dirichlet problems of semilinear elliptic PDEs with singular coefficients, to appear on *Annals of Probability*
- [ZR11] T.S. Zhang, Q.K. Ran, Backward SDEs and Sobolev solutions for semilinear parabolic PDEs with singular coefficients, *Infinite Dimensional Analysis*, *Quantum Probability and Related Topics*, 14, 3, (2011) 517-536
- [Zhu a] R.C. Zhu, BSDE and generalized Dirichlet forms: the finite dimensional case, submitted
- [Zhu b] R.C. Zhu, BSDE and generalized Dirichlet forms: the infinite dimensional case, submitted